

This PDF is available at http://nap.nationalacademies.org/29043











# The Science and Practice of Team Science (2025)

## **DETAILS**

268 pages | 6 x 9 | PAPERBACK ISBN 978-0-309-73441-7 | DOI 10.17226/29043

## **CONTRIBUTORS**

Committee on Research and Application in Team Science; Board on Behavioral, Cognitive, and Sensory Sciences; Board on Human-Systems Integration; Division of Behavioral and Social Sciences and Education; National Academies of Sciences, Engineering, and Medicine

# SUGGESTED CITATION

National Academies of Sciences, Engineering, and Medicine. 2025. The Science and Practice of Team Science. Washington, DC: The National Academies Press. https://doi.org/10.17226/29043.



FIND RELATED TITLES

Visit the National Academies Press at nap.edu and login or register to get:

- Access to free PDF downloads of thousands of publications
- 10% off the price of print publications
- Email or social media notifications of new titles related to your interests
- Special offers and discounts



All downloadable National Academies titles are free to be used for personal and/or non-commercial academic use. Users may also freely post links to our titles on this website; non-commercial academic users are encouraged to link to the version on this website rather than distribute a downloaded PDF to ensure that all users are accessing the latest authoritative version of the work. All other uses require written permission. (Request Permission)

This PDF is protected by copyright and owned by the National Academy of Sciences; unless otherwise indicated, the National Academy of Sciences retains copyright to all materials in this PDF with all rights reserved.



NATIONAL ACADEMIES PRESS

# The Science and Practice of Team Science

Committee on Research and Application in Team Science

Board on Behavioral, Cognitive, and Sensory Sciences

Board on Human-Systems Integration

Division of Behavioral and Social Sciences and Education

## NATIONAL ACADEMIES PRESS 500 Fifth Street, NW Washington, DC 20001

This activity was supported by contracts/grants between the National Academy of Sciences and the Keck Foundation (#9554) and the National Institutes of Health (HHSN263201800029I/75N98023F00012). Generous support of the work of the Board on Behavioral, Cognitive, and Sensory Sciences comes from the National Science Foundation (Award 2234961), the National Institute on Aging, the American Psychological Association, and the National Aeronautics and Space Administration. Any opinions, findings, conclusions, or recommendations expressed in this publication do not necessarily reflect the views of any organization or agency that provided support for the project.

International Standard Book Number-13: 978-0-309-73441-7 Digital Object Identifier: https://doi.org/10.17226/29043 Library of Congress Control Number: 2025943511

This publication is available from the National Academies Press, 500 Fifth Street, NW, Keck 360, Washington, DC 20001; (800) 624-6242; https://nap.nationalacademies.org.

The manufacturer's authorized representative in the European Union for product safety is Authorised Rep Compliance Ltd., Ground Floor, 71 Lower Baggot Street, Dublin Do2 P593 Ireland; www.arccompliance.com.

Copyright 2025 by the National Academy of Sciences. National Academies of Sciences, Engineering, and Medicine and National Academies Press and the graphical logos for each are all trademarks of the National Academy of Sciences. All rights reserved.

Printed in the United States of America.

Suggested citation: National Academies of Sciences, Engineering, and Medicine. 2025. *The Science and Practice of Team Science*. Washington, DC: National Academies Press. https://doi.org/10.17226/29043.

The **National Academy of Sciences** was established in 1863 by an Act of Congress, signed by President Lincoln, as a private, nongovernmental institution to advise the nation on issues related to science and technology. Members are elected by their peers for outstanding contributions to research. Dr. Marcia McNutt is president.

The **National Academy of Engineering** was established in 1964 under the charter of the National Academy of Sciences to bring the practices of engineering to advising the nation. Members are elected by their peers for extraordinary contributions to engineering. Dr. Tsu-Jae Liu is president.

The **National Academy of Medicine** (formerly the Institute of Medicine) was established in 1970 under the charter of the National Academy of Sciences to advise the nation on medical and health issues. Members are elected by their peers for distinguished contributions to medicine and health. Dr. Victor J. Dzau is president.

The three Academies work together as the **National Academies of Sciences**, **Engineering**, **and Medicine** to provide independent, objective analysis and advice to the nation and conduct other activities to solve complex problems and inform public policy decisions. The National Academies also encourage education and research, recognize outstanding contributions to knowledge, and increase public understanding in matters of science, engineering, and medicine.

Learn more about the National Academies of Sciences, Engineering, and Medicine at www.nationalacademies.org.

Consensus Study Reports published by the National Academies of Sciences, Engineering, and Medicine document the evidence-based consensus on the study's statement of task by an authoring committee of experts. Reports typically include findings, conclusions, and recommendations based on information gathered by the committee and the committee's deliberations. Each report has been subjected to a rigorous and independent peer-review process and it represents the position of the National Academies on the statement of task.

**Proceedings** published by the National Academies of Sciences, Engineering, and Medicine chronicle the presentations and discussions at a workshop, symposium, or other event convened by the National Academies. The statements and opinions contained in proceedings are those of the participants and are not endorsed by other participants, the planning committee, or the National Academies.

**Rapid Expert Consultations** published by the National Academies of Sciences, Engineering, and Medicine are authored by subject-matter experts on narrowly focused topics that can be supported by a body of evidence. The discussions contained in rapid expert consultations are considered those of the authors and do not contain policy recommendations. Rapid expert consultations are reviewed by the institution before release.

For information about other products and activities of the National Academies, please visit www.nationalacademies.org/about/whatwedo.

# COMMITTEE ON RESEARCH AND APPLICATION IN TEAM SCIENCE

DIANA BURLEY (Co-Chair), American University

MO WANG (Co-Chair), University of Florida

DOROTHY CARTER, Michigan State University

ERIN KAI-LING CHIOU, Arizona State University

DEBORAH DIAZGRANADOS, Virginia Commonwealth University

STEPHEN M. FIORE, University of Central Florida

MICHAEL C. FRANK, Stanford University

NICOLE B. MOTZER, Montana State University

MARK T. PETERS, MITRE Corporation

SUSAN J. SIMKINS, The Pennsylvania State University

BONNIE SWENOR, Johns Hopkins Medical Institutions (through July 2024)

KELLY D. TAYLOR, University of California, San Francisco

CORY VALENTE, Dow Chemical Company

CHRISTOPHER W. WIESE, Georgia Institute of Technology

# Study Staff

KELSEY N. SCHUCH, Study Director

SHARON BRITT, Program Coordinator

JACQUELINE COLE, Senior Program Assistant (through July 2024)

KATHERINE R. KANE, Research Associate (as of December 2024)

ASHTON RAY, Senior Program Assistant (through March 2025)

**EMANUEL ROBINSON**, Board Director, Board on Human-Systems Integration

DANIEL TALMAGE, Program Officer (through October 2024)

EMILY A. VARGAS, Program Officer (as of December 2024)

**DANIEL J. WEISS,** Board Director, Board on Behavioral, Cognitive, and Sensory Sciences

#### Consultants

JOE ALPER, Independent Science Writer HEATHER KREIDLER, Fact or Fiction LLC

# BOARD ON BEHAVIORAL, COGNITIVE, AND SENSORY SCIENCES

ELIZABETH A. PHELPS (*Chair*), Harvard University MIGUEL P. ECKSTEIN, University of California, Santa Barbara WILSON S. GEISLER, The University of Texas at Austin MICHELE J. GELFAND, Stanford University MARA MATHER, University of Southern California ULRICH MAYR, University of Oregon KATHERINE L. MILKMAN, University of Pennsylvania MARY C. MURPHY, Indiana University ELISSA L. NEWPORT, Georgetown University DON OPERARIO, Emory University DAVID E. POEPPEL, New York University KARL W. REID, Massachusetts Institute of Technology MO WANG, University of Florida DUANE WATSON, Vanderbilt University

**DANIEL J. WEISS, Director** 

## **BOARD ON HUMAN-SYSTEMS INTEGRATION**

FREDERICK L. OSWALD (Chair), Rice University TARA S. BEHREND, Michigan State University ANN M. BISANTZ, State University of New York at Buffalo LINDA NG BOYLE, New York University NANCY I. CURRIE-GREGG, Texas A&M University MARY CZERWINSKI, University of Washington ROLLIN J. FAIRBANKS III, MedStar Health QIWEI HE, Georgetown University JANAE N. LOCKETT-REYNOLDS, Department of Homeland Security **IULIE MARBLE**, University of Maryland NATHAN J. MCNEESE, Clemson University MICHAEL MULLER, International Business Machines Corporation **DENIZ ONES**, University of Minnesota MARK T. PETERS, MITRE Corporation CHRISTOPHER R. REID, The Boeing Company MICHELLE M. ROBERTSON, Northeastern University WENDY A. ROGERS, University of Illinois Urbana-Champaign EVA M. SAGE-GAVIN, Accenture LAWRENCE G. SHATTUCK, Naval Postgraduate School (Retired) HENG XU, University of Florida

EMANUEL ROBINSON, Director



# Reviewers

This consensus study report was reviewed in draft form by individuals chosen for their diverse perspectives and technical expertise. The purpose of this independent review is to provide candid and critical comments that will assist the National Academies of Sciences, Engineering, and Medicine in making each published report as sound as possible and to ensure that it meets the institutional standards for quality, objectivity, evidence, and responsiveness to the study charge. The review comments and draft manuscript remain confidential to protect the integrity of the deliberative process.

We thank the following individuals for their review of this report:

LESLIE DECHURCH, Northwestern University MICHAEL GOLDBERG, Columbia University POORNA KUSHALNAGAR, Gallaudet University KATE MITTENDORF, Vanderbilt University JUDITH OLSON, University of California–Irvine YVETTE E. PEARSON, University of Texas–Dallas EDUARDO SALAS, Rice University LE ZHOU, Texas A&M University

Although the reviewers listed above provided many constructive comments and suggestions, they were not asked to endorse the conclusions or recommendations of this report, nor did they see the final draft before its release. The review of this report was overseen by JULIE RYAN, Wyndrose Technical Group, and JONATHAN MORENO, University of Pennsylvania Health System. They were responsible for making certain that

x REVIEWERS

an independent examination of this report was carried out in accordance with the standards of the National Academies and that all review comments were carefully considered. Responsibility for the final content rests entirely with the authoring committee and the National Academies.

# Acknowledgments

The committee and staff thank the many individuals and groups who contributed to this report and made it possible. First, we acknowledge the generous support of the W. M. Keck Foundation and 17 institutes and centers at the National Institutes of Health (NIH). These include the Sexual & Gender Minority Research Office, National Institute of Neurological Disorders and Stroke (NINDS) BRAIN Initiative, NINDS, National Center for Advancing Translational Sciences, Environmental Influences on Child Health Outcomes Program, National Institute of Mental Health, National Cancer Institute, Chief Officer for Scientific Workforce Diversity, National Institute on Aging, Fogarty International Center, National Institute of General Medical Sciences, National Institute of Nursing Research, National Institute of Biomedical Imaging and Bioengineering, National Institute of Arthritis and Musculoskeletal and Skin Diseases, National Institute of Environmental Health Sciences, Office of Disease Prevention, and Office of Behavioral and Social Sciences Research. Generous support of the work of the Board on Behavioral, Cognitive, and Sensory Sciences comes from the National Science Foundation (Award 2234961), the National Institute on Aging, the American Psychological Association, and the National Aeronautics and Space Administration.

The committee also greatly benefited from presentations and discussions from external experts. These include Emily Ackerman, Harvard Medical School; Kathleen Bogart, Oregon State University; Jordan Clarence, Carelon Behavioral Health; Jeni Cross, Colorado State University; Alyssa Dolge, Office of Behavioral and Social Sciences Research, NIH; Kevin Doran, University of Colorado Boulder; Matthew Dye, Rochester Institute

xii

of Technology; Sindy Escobar Alvarez, Doris Duke Foundation; Leslie Fan Kohlenberg, Dow Chemical; Erin Fitzgerald, Applied Research Laboratory for Intelligence and Security; Sharon Glotzer, University of Michigan; Jeanne-Marie Guise, Beth Israel Deaconess Medical Center; C. K. Gunsalus, University of Illinois Urbana-Champaign; Kara Hall, National Cancer Institute; Holly Hapke, University of California, Irvine; Christine Hendren, Appalachian State University; Zeynab Jouzi, Cornell University; Pramod P. Khargonekar, University of California, Irvine; Poorna Kushalnagar, Gallaudet University; Songqi Liu, National Science Foundation; Michael O'Rourke, Michigan State University; Kim Patten, University of Arizona; Bryce Pilz, University of Michigan; Michael Potter, University of California, San Francisco; Maritza Salazar Campo, University of California, Irvine; Wendy Shaw, Pacific Northwest National Laboratory; Marissa Shuffler, Clemson University; Tori Simenec, University of Minnesota Twin Cities; Cynthia Spence, Spelman College; Erica Spotts, Office of Behavioral and Social Sciences Research, NIH; Daniel Stokols, University of California, Irvine; Barbara Walker, University of California, Santa Barbara; Melissa Walsh, Brown University; and Susan Wolf, University of Minnesota.

We thank the authors of the commissioned papers that accompany this report, which greatly helped in developing the report. These include Mahya Behesti, New York University; Megan Morris, University of Colorado School of Medicine; Kara Ayers, Cincinnati Children's Hospital; Sarika Gopalakrishnan, Envision Research Institute; Walter Wittich, University of Montreal; J. R. Rizzo, New York University; and Heng Xu, University of Florida.

We also thank Heather Kreidler and Joe Alper and their contributions as science writers, as well as the individuals who served as reviewers.

Finally, we thank our colleagues on the committee for their hard work, expertise, time, and dedication in serving on this consensus study.

# Contents

Preface xvi		
Acronyms and Abbreviations		
Sur	Summary	
1	Introduction STUDY BACKGROUND AND CHARGE TO THE COMMITTEE, 13 COMMITTEE APPROACH, 15 ORGANIZATION OF THE REPORT, 20 REFERENCES, 20	11
2	Review of Current Science of Team Science KEY TAKEAWAYS FROM ENHANCING THE EFFECTIVENESS OF TEAM SCIENCE, 23 IMPLEMENTING THE 2015 RECOMMENDATIONS: SUCCESSES AND CHALLENGES, 26 REFERENCES, 35	23
3	Best Practices EDUCATION, TRAINING, AND PROFESSIONAL DEVELOPMENT, 41 BEST PRACTICES FOR SUPPORTING SCIENCE TEAMS, 49 FUTURE RESEARCH QUESTIONS, 88	41

xiii

xivCONTENTS CONCLUSION AND RECOMMENDATION, 95 REFERENCES, 96 Institutional and External Supports for Team Science 115 INSTITUTIONAL INFRASTRUCTURE, 116 INSTITUTIONAL CULTURE AND POLICY, 128 FUNDERS, 137 SCIENTIFIC INCENTIVES, 140 CONCLUSIONS AND RECOMMENDATIONS, 144 REFERENCES, 145 159 **Evaluating Team Science** IMPORTANCE OF TEAM SCIENCE EVALUATION, 159 CHALLENGES OF TEAM SCIENCE EVALUATION, 161 A FRAMEWORK FOR GUIDING TEAM SCIENCE EVALUATION, 163 CONCLUSION AND RECOMMENDATION, 178 REFERENCES, 178 Forward-Looking Research Recommendations and Infrastructure Needs 187 BEST PRACTICES, 187 INSTITUTIONAL AND EXTERNAL SUPPORT, 189 EVALUATION, 190 IDENTIFICATION OF RESEARCH GAPS, 190 CONCLUSION AND RECOMMENDATION, 191 Appendix A Committee Biosketches 193 Appendix B **Team Science Background** 201 **High-Functioning Science Teams** 213 Appendix C Appendix D Sampling of Survey-Based Team Assessments 229 Appendix E Glossary 245

# Boxes and Tables

# **BOXES**

1-1	Statement of Task, 14
1-2	Interested Parties Relevant to Team Science, 16
1-3	Definitions of Collaborative Approaches, 19
2-1	Recommendations from Enhancing the Effectiveness of Team Science, 25
4-1 4-2	The Role of Open Science in Facilitating Team Science, 120 Using Open Products and Resource-Tracking to Align Team Science Incentives, 143
	TABLES
3-1 3-2	Research Questions for Understanding Effective Science Teams, 90 Summary of Best Practices for Science Teams, 94
D-1	Sampling of Survey-Based Team Assessments, 230



# Preface

This report highlights the growing importance of interdisciplinary collaboration, underscoring how modern scientific challenges often demand team-based approaches that integrate expertise from multiple fields. As contemporary research problems become more complex, no single discipline can address them fully, making team science critical for innovation. The National Academies of Sciences, Engineering, and Medicine's 2015 report, Enhancing the Effectiveness of Team Science, emphasized that over 90% of scientific and engineering publications were coauthored, illustrating the prominence of collaborative efforts. This new report builds on that foundation by addressing how recent developments—such as virtual collaboration and advancements in artificial intelligence—have reshaped the landscape of team science.

This report is particularly significant as it examines both the opportunities and challenges team science presents in today's rapidly evolving research environment. By investigating how to harness virtual and hybrid collaborations effectively, this report offers timely insights for best practices. It also addresses a critical need: preparing the scientific workforce with the skills, tools, and training it needs to navigate these collaborative environments. As such, this report provides valuable guidance for fostering effective, inclusive, and impactful team science, making it an essential resource for shaping the future of interdisciplinary research.

We wish to express our deep appreciation to the members of the committee for their diligent and dedicated contributions. Their expertise and knowledge were indispensable throughout our deliberations, and their efforts, which often required working nights and weekends, are particularly xviii PREFACE

notable given the incredibly challenging year it has been. We cannot thank them enough.

On behalf of the entire committee, we wish to thank the National Academies staff for their outstanding support and guidance. We are also deeply appreciative to Heather Kreidler for her writing and fact-checking, and the report benefited deeply from the editing skills of Joe Alper. Additionally, we want to express our sincere gratitude to everyone who contributed their time, expertise, and experiences to our committee. The presentations, resources, and insights contributed immensely to our deliberations. Finally, we wish to thank National Institutes of Health staff for their partnership and forthright participation throughout this process. We also thank the Keck Foundation for providing the funds for the committee to travel to in-person meetings.

Diana Burley and Mo Wang, *Co-Chairs* Committee on Research and Application in Team Science June 2025

# Acronyms and Abbreviations

ABCs attitudes, behaviors, and cognitive states

AI artificial intelligence

CBPR community-based participatory research

COMPETES Creating Opportunities to Meaningfully Promote

Excellence in Technology, Education, and Science

CRediT Contributor Roles Taxonomy
CTD component team distance

CTSA Clinical and Translational Science Award
CTSI Clinical and Translational Science Institute

DoE Department of Energy

EAGER Early-Concept Grants for Exploratory Research

ERC Engineering Research Center

FDA Food and Drug Administration

GPU graphics processing unit

HFE human factors or ergonomics

IGERT Integrative Graduate Education and Research Traineeship

IMOI input-mediator-output-input IRB institutional review board

xix

xx

#### ACRONYMS AND ABBREVIATIONS

MOU memorandum of understanding

NAKFI National Academies Keck Futures Initiative

NCATS National Center for Advancing Translational Sciences

NCI National Cancer Institute
NIH National Institutes of Health

NORDP National Organization of Research Development

**Professionals** 

NRC National Research Council NRT NSF Research Traineeship NSF National Science Foundation

OSF Open Science Framework

R&D research and development

SESYNC Socio-Environmental Synthesis Center STC Science and Technology Centers

STEM science, technology, engineering, and mathematics STEMM science, technology, engineering, mathematics, and

medicine

TIP Technology, Innovation and Partnerships (Directorate)

UN United Nations

# Summary

Modern scientific research frequently involves collaborative efforts, with teams tackling intricate problems that require integrating theories and methodologies from multiple scientific perspectives. Often, these science teams also reflect a wide range of demographic and geographic perspectives. Moreover, the challenges of managing large and complex teams, as well as growing interest in translating research findings for nonscientists, often means the teams include individuals who may not identify as scientists, such as administrators, funders, and community partners. Groundbreaking research, including the Human Genome Project, the discovery of the Higgs boson, and the invention of the internet, has been performed by science teams that vary in size, geographic distribution, and often disciplinary expertise.

A decade ago, the National Academies of Sciences, Engineering, and Medicine (National Academies) published *Enhancing the Effectiveness of Team Science*. That report substantially increased understanding of the importance of team science and highlighted the value of research on how team science is conducted and supported. However, shifts in the scientific and social landscape over the last decade—including technological developments, geopolitical tensions, and a global pandemic—have introduced new challenges for conducting team science.

Given those shifts and the growing importance of collaboration in modern scientific research, the National Academies Board on Behavioral, Cognitive, and Sensory Sciences and the Board on Human-Systems Integration, with support from the Keck Foundation and the National Institutes of Health, convened an expert committee to review and synthesize topics related to team science. The committee's statement of task covered five major topics:

- 1. Current team science practices, including in-person, virtual, and hybrid environments.
- Evidence of best practices in team science and how to enhance those efforts.
- 3. What is known about the best practices, effectiveness, benefits, and potential pitfalls of virtual and hybrid approaches for team science.
- 4. What evidence-based approaches and training have been designed to enhance the effectiveness of team science.
- 5. What types of outcomes, methods, and measures would be appropriate for evaluating team science training and performance.

It is important to note the term "team science" is often applied ambiguously. It sometimes refers to the narrow study of science teams, at other times refers to the broader study of teams across disciplines and contexts, and can also refer to the conduct of science by teams. While the committee drew on findings from the general study of teams, this report focuses on science team practices and the study of those practices, using the terms *team science* and the *science of team science* to refer to those.

To address its statement of task, the committee sought to understand developments in the practice and study of team science in the decade since the 2015 National Academies report. It also aimed to identify evidencebased best practices for assembling and managing science teams. The committee reviewed the scholarly literature on teams generally and research on teams in the context of science. The committee members gathered expert insight through public forums and drew on their own expertise and experiences in analyzing what they learned from these sources. The committee found empirical investigations of a range of teams; however, the body of literature focused specifically on science teams is limited. Furthermore, the committee notes that science teams range from a pair of individuals with a relatively narrow scope of inquiry to massive, multiparty collections of collaborators addressing the world's most intractable problems. Thus, the committee provides the conclusions and recommendations in this report with the clear acknowledgment that team size, complexity, degree of disciplinary integration, proximity of team members, permeability of team boundaries, and other characteristics are critical aspects to consider when assessing whether and how to implement a particular best practice.

SUMMARY 3

## CURRENT STATE OF RESEARCH ON TEAM SCIENCE

The committee found that though recognition of its importance has increased over the last decade, the science of team science is not yet well developed. Moreover, while some institutions and funding agencies have increased their investment in research and initiatives on team science, the field remains underfunded. As a result, there is a disconnect between theory, empirical evidence, and accepted best practices for implementing team science initiatives and in the training programs that prepare team members. In addition, new technologies, including virtual and hybrid collaboration technologies and artificial intelligence tools, have emerged over the last decade and enable new forms of collaboration. Further investigation is required to understand how these technologies can affect team science dynamics, performance, and individuals, as well as best practices for how to effectively integrate new technologies into team science.

Research in the field is also hampered by differences in the use of terms when discussing team science. This includes terms such as *cross-disciplinary*, *interdisciplinary*, *multidisciplinary*, and *transdisciplinary*, which have unique meanings in relation to the study of team science but may be used interchangeably in other disciplines. With the broad array of parties involved in team science, including scholars of team science, researchers, and funders, this inconsistency can interfere with understanding key concepts in the field, confuse communication, and potentially reduce the effectiveness of team science initiatives.

Much of the research informing team science practices derives from findings in organizational psychology and other social sciences—that is, from the general study of teams, not from studies that specifically focus on science teams. Although some of the findings from that body of research may apply, science teams have unique characteristics and needs that may limit the generalizability of those findings from the broader team literature. Research is needed to better understand these potential differences.

## BEST PRACTICES FOR CONDUCTING TEAM SCIENCE

Research on teams in general and on science teams specifically makes it clear that effective teams do not emerge naturally. Instead, the research points to several best practices or strategies that help science teams collaborate effectively to achieve their goals. These include practices that teams and their leaders can adopt, as well as strategies their supporting institutions can use to support strong science teams.

The committee identified best practices for science teams with two caveats. First, not all recommended practices will be suitable for every team. Science teams range from small groups that include members from

one or two disciplines studying a relatively narrow question to large, cross-disciplinary teams tackling broad, complex issues. Teams differ in their goals and the complexity of the tools and methods they share. Thus, deciding on which practices will be appropriate will require careful analysis of the team's characteristics and objectives. Second, research on best practices for team science is in an early stage. Much of the available relevant research has been conducted on teams in general, rather than on science teams specifically, so careful attention is needed as to the applicability of the research findings.

Despite these caveats, the committee identified best practices for developing the competencies science teams require. Research has demonstrated that coordination, information-sharing, and conflict management are key processes for effective teams. In addition, three psychological states—cohesion, shared mental models, and transactive memory systems—are linked closely to achieving team objectives.

Education, training, and professional development—through workshops, formal and informal courses, and mentoring—are necessary to build these competencies. The committee relied on an existing framework to temporally organize its thinking about the practices for which there is reasonable evidence of effectiveness (from the general literature on teams as well as case studies) that will likely translate to the science team context. This four-part framework includes development, conceptualization, implementation, and translation. The following conclusion highlights some best practices the committee identified, but it is not exhaustive.

Conclusion 3-1: The following strategies offer the potential to improve science team performance and outcomes, if adapted to specific contexts and circumstances.

- Development stage: careful team assembly that reflects a task analysis, consideration of the team composition, attention to the orientation of new members, and development of a shared language.
- Conceptualization stage: development of a team charter, deliberative team planning and project design, and attention to a shared mental model of the team's work.
- Implementation stage: systematic project management, regular team debriefs to identify what went well and what went poorly at each stage and to determine the best ways for the next project stage.
- Translation stage: working with community members and attention to external as well as internal validity, that is, understanding of the generalizability of the research findings.

SUMMARY 5

Recommendation 3-1: Research funders, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should provide resources enabling the study of team development, conceptualization, implementation, and translation.

# BEST PRACTICES FOR EXTERNAL SUPPORT OF TEAM SCIENCE

Effective science teams require supports of many kinds. In addition to those the teams can adopt, there are many best practices that parties external to the team, including universities and other research institutions and funders, can implement to help facilitate both the study and practice of team science.

Published analyses and the views of the experts who presented to the committee suggest that current policies in most academic institutions, including those surrounding tenure and promotion, authorship, costsharing, allowable costs, and resource-sharing, do not incentivize participating in team science or research on team science. Similarly, many research funders do not support the activities or research needed for team science. Science journals, too, follow guidelines and practices that do not support team science or research on team science.

Conclusion 4-1: Institutions wishing to foster the study of team science would benefit from reviewing the incentive structures that influence individuals' decisions about engaging in such research. Specifically, many policies and practices that are currently in place surrounding tenure and promotion, authorship, cost-sharing, allowable costs, and resource-sharing appear to discourage engagement in the study of team science and participation in collaborative science teams. Institutional processes can also reinforce disparities in team science by failing to properly recognize work in team science, including community-engaged research and mentorship.

Recommendation 4-1: Academic departments should adapt their promotion and tenure processes to acknowledge and reward the contributions of researchers who take on the additional professional responsibilities associated with participating in and studying team science by:

- recognizing and incorporating in tenure evaluations the valuable contributions made through different authorship roles, publications in interdisciplinary or nontraditional journals, and processoriented outcomes.
- b. ensuring that the demographic and disciplinary composition of promotion and tenure committees is both reflective and independent of

- the candidates they are reviewing to the extent possible to facilitate increased representation in science team leadership.
- c. revising criteria for selection to ensure fair consideration of candidates who research and team with underrepresented communities and/or do community-engaged research.
- d. considering candidates' involvement in committees, initiatives, and other activities, including engagement outside the scientific community, that offer value to science teams and encourage participation.
- e. reviewing the promotion and tenure process periodically, for instance every several years, to ensure continuous improvement and minimize inefficiencies.

Recommendation 4-2: Science journal editors should establish comprehensive systems and policies to build team science into the publishing mainstream, including:

- a. conducting a systematic assessment to identify barriers that may limit the incorporation of team science literature in their journal. These findings should be used to develop actionable strategies to address identified barriers.
- b. adopting clear policies and guidelines for authorship, allocation of credit, and contribution level, particularly when there are many contributing authors and when authors are contributing from different disciplines with different authorship practices. Policies can include strategies for addressing authorship disputes.
- c. allocating appropriate space to publish research on the science of team science.

The funders of scientific research can play a key role in fostering both conduct of team science and the study of its operation and outcomes.

Recommendation 4-3: Funders of team science, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should integrate team science needs into funding programs and policies and should remove barriers to team science efficacy by:

- including team science needs, such as team-building, travel and meetings, professional development and training, and resourcesharing in allowable costs.
- b. allowing for the inclusion of nonscientist team members and leaders in the project budget to allow for the compensation of their time.

Institutions that host team science initiatives can implement many measures to support their scientific collaborators directly.

SUMMARY 7

Recommendation 4-4: Institutions seeking to advance team science effectiveness should allocate resources to support science teams. Resource allocation may cover, but is not limited to the following:

- a. resources that mitigate the operational burden on science teams by investing in the support of administrative staff.
- b. training and mentorship for research administrative staff working on team science projects.
- c. resources to expand access to and the use of technologies that optimize team participation for geographically dispersed members.
- d. funding to address gaps in the physical, digital, and procedural environment. This could include applying universal design principles that accommodate the needs of all individuals.

# RESEARCH AND EVALUATION NEEDS

Despite the importance of team science, many questions about how it is best done and supported cannot yet be answered with full confidence. More robust collection of data on science teams and their work are needed, as well as evaluation of their operation. The committee found that the existing body of evaluation and research on team science tends to emphasize objective and archival measures of team performance output, especially the publications science teams produce, and tends to underemphasize evaluation of ongoing team functioning, the effect of team science on individual members, and other forms of team performance output.

Systematic evaluation of science teams can foster improvements in team science initiatives and best practices for science teams, although the evolving nature of science teams makes a one-size-fits-all approach to evaluation inadequate. However, data collection and evaluation of key elements will be important both for strengthening individual teams and for strengthening research on science team functioning.

Conclusion 5-1: Data collection and evaluation, supported by both institutions and science team leaders, are critical for answering questions about key features of science teams:

- How social processes on the team are unfolding (e.g., team members' perception of experiences of participation on the team, team member satisfaction, psychological safety, trust).
- What the team is producing (e.g., successfully completed team objectives, publications, patents and invention reports, research grant applications and awards, educational outcomes).
- What impact the team is having on individual members (e.g., well-being, skill and knowledge advancement, growth of professional networks, career success).

Recommendation 5-1: Funding agencies, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should require that the science teams they support develop an evaluation plan to assess their effectiveness and impact. The plans should incorporate team dynamics (e.g., social processes), team performance (e.g., bibliometric metrics), and impact on members (e.g., learning and development outcomes). Regular review periods should be established with the team to monitor and track progress and team effectiveness.

Researchers studying the science of team science continue to make progress in theoretical developments and are applying those theories to specific case studies. For the field to move forward, additional work is needed to develop a robust, empirical evidence base for identifying best practices and institutional supports. Many principles and concepts from the study of teams in general (e.g., from researchers in social and organizational psychology, organizational behavior, cognitive science) are likely to be useful in the context of team science, but the empirical basis for applying these insights to science teams is still limited. Addressing this gap is crucial, as science teams have unique features, raising concerns about the generalizability of these findings.

Conclusion 6-1: The research translating findings on the functioning of teams general to the functioning of science teams is incomplete for reasons that include insufficient funding and a lack of professional recognition and reward for the study of science teams. Specifically needed are:

- Studies focused on the science team context that explore the application of existing theories to the unique processes and dynamics that distinguish science teams from traditional organizational teams.
- Studies of team science competencies and interventions that allow for robust statistical analyses or pre- and post-testing to build the empirical evidence base for team science learning, training, and professional development in real time.
- Data-driven research, including longitudinal studies, to better understand team science outcomes.
- Research to identify and investigate institutional policies and practices that reinforce or serve as barriers to team science, such as the support structures universities can provide (e.g., research development professionals, team facilitators), how these affect team science processes and outcomes, and other incentives that influence the conduct of team science.

SUMMARY 9

• Research into the effectiveness of virtual collaboration tools and how these relate to the specific configuration of hybrid and virtual teams, including factors such as geographic distribution, temporal dynamics, and communication needs.

Recommendation 6-1: Funders interested in supporting the conduct of science should prioritize research on, and provide sufficient funding for, the application of findings from the broader study of teams to the science context. Areas of prioritization may include but are not limited to studies that incorporate both qualitative and quantitative data to build empirical evidence about the science context and research evaluating institutional policies and supports for science teams.



1

# Introduction

Scientific research has gravitated increasingly toward collaborative research, as contemporary scientific problems often require solutions that draw from multiple areas of expertise. Team science, defined as collaborative, interdependent research conducted by more than one individual, is becoming the norm as it often involves integrating scientific approaches to solve problems that are multifaceted, multidimensional, interdependent, and do not have a single answer. Critically, team science includes project teams and more complicated collections of researchers, such as multiteam systems that operate within research centers, initiatives, and networks. Team science also considers the engagement of input beyond researchers, such as community members and leaders, policymakers and other decision-makers, caregivers, health care providers, and research administrative professionals. To prepare a workforce that can participate effectively in team science, it is critical to equip researchers and their administrative team members with training on evidence-based tools and strategies for engaging in team science. The nascent field of the science of team science aims to generate and build on the empirical evidence base (e.g., National Research Council, 2015) and translate that knowledge to enhance the effectiveness of team science in practice.

Evidence suggests that team science approaches often lead to results with greater impact and innovation relative to single-investigator approaches (Lee et al., 2015; Wuchty et al., 2007). Boundary-spanning teams that cross disciplines, organization structures, or geographic distances are particularly effective, often yielding better outcomes, increased productivity, and greater scientific impact (Hall et al., 2018). Although there are many

benefits of collaborative research efforts that leverage the strengths and varied perspectives of scientists, team science approaches can also lead to significant challenges as researchers try to integrate effectively across fields that may have different approaches, methodologies, and concerns. Moreover, science teams are frequently spread across different locations, and team members may have different disciplinary or demographic backgrounds. Consequently, the *science of team science* has emerged as a research area in which scholars from various disciplines, such as psychology, organizational sciences, sociology, communication, and philosophy, contribute conceptually and empirically to understanding how science teams are organized and work together, how to best measure their effectiveness, and the implications of individual differences in team science.

Understanding how to best develop and facilitate the work of science teams is part of a broader motivation to understand science, scientists, and the context in which scientists practice. Many areas of scholarship share this motivation, ranging from the history and philosophy of science; science and technology studies; and, more recently, the science of science. These areas all examine different science concepts, processes, and outcomes. Some study science to add to the understanding of science, while others also seek out ways to improve the science ecosystem. To varying degrees, each has also studied collaboration in science, but rarely as a focal issue of inquiry. Because of this, the science of team science was developed specifically to pursue a scholarly examination of teamwork in science (Hall et al., 2008).

The goal of the science of team science is to improve the understanding of how scientists interact as members of a science team and how their collaboration helps build and integrate knowledge across disciplinary, professional, and institutional boundaries (Stokols et al., 2008). From this improved understanding, the science of team science aims to help science teams make full use of their integrative capacity—"the social and cognitive processes, along with emergent states, that shape a team's ability to combine diverse knowledge" (Salazar et al., 2012, p. 527). Fundamental to this goal is reframing science collaboration as a process of teamwork to be mastered (Fiore, 2008). In contrast to related fields, science teams are the focal area of study in the science of team science to improve fundamental understanding of the collaborative production of knowledge and develop new methods and models to improve teamwork on science teams.

To better conceptualize the collaborative component of team science, it is useful to follow a distinction developed in the organizational sciences that distinguishes between team and task competencies (Fiore et al., 2019). *Taskwork* refers to the activities associated with achieving a team's goals. Examples of taskwork in science include experimental design, data collection, and statistical analysis. *Teamwork*, on the other hand, involves the interactions among team members that are essential for effective

INTRODUCTION 13

collaboration. In science, this includes communicating clearly about complex ideas, managing coordination needs, and understanding and using teammate expertise. For science teams to succeed, they need to develop competencies relevant to both taskwork and teamwork. As such, a wide range of competencies encompassing the knowledge, skills, and attitudes that facilitate team science is necessary for success in scientific collaborations (Fiore et al., 2019).

#### STUDY BACKGROUND AND CHARGE TO THE COMMITTEE

In 2015, the National Academies of Sciences, Engineering, and Medicine (National Academies) released the report Enhancing the Effectiveness of Team Science (National Research Council, 2015). That consensus report synthesized findings from research on teams and factors that bear on the success of team science endeavors, offering recommendations for improving team science effectiveness and identifying several areas for further research. In the decade since, new developments have reshaped the landscape of team science. For example, the COVID-19 pandemic fostered new dimensions of virtual collaboration, given the need to shelter in place, which changed working conditions and arrangements across the globe. Emerging technologies and recent innovations in the realm of artificial intelligence (AI) also add a new dimension to team science. These tools can change the nature of collaboration processes and the dynamics of collaboration when collaborators are not limited to human team members (National Academies of Sciences, Engineering, and Medicine, 2022). Variations in human-machine teaming structures raise additional dimensions of team science to consider. Finally, the number of authors contributing on scientific journal articles has continued to increase in the years following the 2015 report, reflecting an increase in science team sizes (e.g., An et al., 2020; Paul et al., 2024).

Consequently, 17 institutes and centers at the National Institutes of Health<sup>1</sup> and the W. M. Keck Foundation requested a consensus study to explore the state of the science of team science in light of its growing relevance and the changing landscape in contemporary scientific endeavors. This

<sup>&</sup>lt;sup>1</sup> These include the Sexual & Gender Minority Research Office, National Institute of Neurological Disorders and Stroke (NINDS) BRAIN Initiative, NINDS, National Center for Advancing Translational Sciences, Environmental Influences on Child Health Outcomes Program, National Institute of Mental Health, National Cancer Institute, Chief Officer for Scientific Workforce Diversity, National Institute on Aging, Fogarty International Center, National Institute of General Medical Sciences, National Institute of Nursing Research, National Institute of Biomedical Imaging and Bioengineering, National Institute of Arthritis and Musculoskeletal and Skin Diseases, National Institute of Environmental Health Sciences, Office of Disease Prevention, and Office of Behavioral and Social Sciences Research.

consensus study (a) explores team science, including best practices, barriers, effects, and the role of virtual and hybrid environments; (b) develops a contemporary understanding of best practices in team science; (c) evaluates the growing role of virtual and hybrid teams; (d) identifies gaps in resources and training for team science; and (e) explores how to best measure the effectiveness of teams. The full statement of task for the committee can be found in Box 1-1.

## BOX 1-1 Statement of Task

The National Academies will convene an ad hoc, diverse committee of approximately 12–15 scholars, practitioners, and other experts to evaluate the current state of the science of team science and multidisciplinary collaborations across different scales and environments. The committee will meet, conduct public workshops, review the literature, deliberate, and then publish a consensus report of its findings, conclusions, and recommendations. Included will be forward-looking research recommendations (research gaps and infrastructure needs) and suggested applications and/or best practices for a variety of settings and scales. Following publication, dissemination activities, such as a public-facing webinar should take place to promote the report.

The committee will consider reviewing and synthesizing information related to the following topics:

- The effectiveness, benefits, and potential pitfalls of virtual and hybrid approaches for team science. Are there models of collaboration outside of science that might inform best practices? What collaborative technologies are best suited for virtual and hybrid collaboration and how might these vary depending on the dimensions of team science, including the scale of the team, types of collaborators, degree of disciplinary integration, proximity of team members, permeability of team boundaries, disciplinary and form of research, and domains of research involved? What are best practices for working in the hybrid, virtual, and in-person work environment(s) and how might these vary depending on the dimensions of team science?
- Evidence-based approaches and training that have been designed to enhance the effectiveness of team science and identify any gaps in resources and guidance. What methods have been employed and who was the target audience? What types of outcomes, methods, and measures are appropriate for evaluating team science training? What are best practices for training and/or optimizing teams that include various types of non-scientist team members?
- Evidence of best practices in team science and how to enhance those efforts. What are the current team science best practices? What research is needed to further evaluate and enhance these practices?

INTRODUCTION 15

To address this charge, the National Academies appointed an ad hoc committee with a broad range of expertise, including individuals with backgrounds in the science of team science, behavioral and social science, virtual platforms and collaborative tools, AI and emerging technologies, science policy and scientific research, individual differences, ethics and/or risk management, communication, business and management, industrial and organizational psychology, and related fields. Appendix A provides biographies of the 13 committee members.

## **COMMITTEE APPROACH**

The committee's approach to its charge consisted of a review of the evidence in the scientific literature and several other information-gathering activities. In reviewing the literature and formulating its conclusions and recommendations, the committee considered information from public presentations, targeted literature searches, and committee expertise. The committee drew from the team science literature when available, supplemented by team research in other organizational contexts including business and health care. The committee heard from multiple presenters on various topics related to the statement of task. These public information-gathering sessions included speakers from academic communities, research organizations, and funding agencies. The presentations covered topics such as participation and user-friendliness; ethics in team science; evaluation of team science approaches; and practical reflections from researchers, administrators, and funders in the team science ecosystem. To augment this report, the committee commissioned two papers, one on the topic of AI in team science and the other on the topic of disability and accessibility in team science.<sup>2</sup> To address the charge, the committee sought to understand developments in the study and practice of team science over the last decade, including increased awareness of team science practices in multiple fields and examining new programs for supporting team science. The committee chose to define a science team member broadly as including any person in a science team working toward a shared goal. This can include, but is not limited to, technicians, support staff, engineers, students, and participating members of the community. The committee also aimed to examine the role of individual differences in team science—identifying both enabling and inhibiting factors in the integration of individuals in teams. In addition, it sought to review team science best practices and their supporting evidence, how such practices are currently supported and could be strengthened, and the means of evaluating team science outcomes and impact.

<sup>&</sup>lt;sup>2</sup> Commissioned papers and more information are available at https://www.nationalacademies.org/our-work/research-and-application-in-team-science

#### **Identifying Interested Parties**

Identifying interested parties in team science is important for providing clarity and focus to team science initiatives, understanding how best to allocate resources, and building accountability and trust around the work conducted by team scientists, as well as for ethical considerations. *Interested parties* include individuals (e.g., researchers, scientists, administrators, policymakers, community members, patients and patient advocates) and institutions (e.g., funding agencies, academic institutions, national laboratories, nonprofit and industry partners; see Box 1-2). These interested parties form an interconnected ecosystem conducting and supporting science, and each may require different skills and competencies. Those conducting team science are focused on addressing pressing societal issues, advancing knowledge, and finding innovative solutions to problems that are too broad for individual researchers to tackle alone.

# BOX 1-2 Interested Parties Relevant to Team Science

- Researcher: The term researcher describes any member of a science team
  conducting and disseminating scientific research. This could be a faculty
  member running their own laboratory, a scientist in a private-sector research
  and development section, someone working at a national laboratory (e.g.,
  Sandia National Labs), or a community member. It could also include
  researchers in training, such as graduate students or postdocs.
- Science Team Leader: A science team leader could be a director of an academic laboratory or research institute, or anyone in a leadership position in the public or private sectors who oversees research projects. The setting of the research varies, ranging from wet labs to communities.
- Facilitator: A facilitator is a blanket term used to describe those who support
  various aspects of the science process. These individuals may provide episodic
  guidance on a specific project phase, such as ideation or proposal development, or they may engage with the science team continuously throughout the
  research process.
- Evaluator: The term evaluator encompasses those who have an interest in, or are responsible for, assessing the effectiveness of team science processes, outcomes, and contextual influences. Evaluators are often external to the team to give unbiased perspectives.
- Community Member: This report uses the term community member generally to describe those individuals on a team who may or may not be professionally trained scientists but participate in research—both in defining the problems to be solved and in developing solutions—requiring input and guidance from those directly affected, such as patients, family members, local business owners, and politicians.

INTRODUCTION 17

#### **BOX 1-2 Continued**

 Science Administrator: Science administrators might have some overlap in responsibilities with a science leader, but this report uses the term to differentiate those whose responsibilities do not necessarily involve direct oversight of research projects, such as institute directors and grant administrators.

- Institution: Institution refers to a variety of organizations, such as academic institutions, national laboratories, and industry research and development offices.
- Funding Agency: Funding agency refers to a public funding agency or a private philanthropic organization supporting science.
- Policymaker: Policymakers are individuals or groups responsible for creating, shaping, and implementing laws, regulations, and public policies that govern societies.
- Science of Team Science Scholar: Science of team science scholars use
  various research methods to study teamwork within, external to the science
  team context, or both; this contains, but is not limited to, social scientists
  conducting research to examine fundamental questions about science team
  processes, humanities scholars studying how different epistemological worldviews alter the questions cross-disciplinary teams develop, and those studying
  how educational programs or professional development interventions improve
  science team member competencies for collaboration.
- Research Translator: Research translators are those who bridge the gap
  between scientific innovation and its application to policy or practice. These
  professionals often work in the innovation offices of a research institute or
  enterprise entity, serving as the conduit through which research findings are
  transformed into patents, products, and other actionable solutions.

The science ecosystem comprises researchers, science team leaders, facilitators, evaluators, and science administrators who collaborate to address scientific questions. These professionals develop crucial team science competencies both during training and while leading collaborative initiatives.

Community members represent another vital component of the science ecosystem. These individuals and groups bring valuable lived experiences that inform research solutions and the problems to be solved, including both research questions and constraints (Corburn, 2007; Wandersman, 2003). Their contributions extend beyond simply providing data and context, as they actively participate in the research process and are affected directly by the outcomes of team science endeavors, whether those effects are positive or negative.

In this context, *institution* refers to a variety of organizations, including academic institutions, national laboratories, and industry research and

development offices. Institutional members can support team science initiatives by providing the resources, infrastructure, and administrative support necessary to conduct team science.

Another group of interested parties includes research translators who bridge the gap between scientific innovation and its application to policy and practice. These professionals often work in innovation offices, serving as the conduit through which research findings are transformed into patents, products, services, and other practical solutions or policy recommendations. Research translators possess a unique blend of scientific knowledge, business acumen, and communication skills, enabling them to facilitate communication between science teams and the intended recipients of the scientific discovery. They work closely with scientists, often as members of the research team, to comprehend the nuances of the research and to distill complex findings into accessible language that interested parties can readily understand. In some cases, this role involves guiding the research process to align with industry needs and navigating the patent landscape to secure intellectual property rights. In other situations, research translators may outline how the innovation can be used to evolve policy recommendations.

Funding agencies (e.g., government agencies, nonprofit foundations, private-sector entities) may provide financial support for science teams to conduct research, seek out innovative breakthroughs, advance science, and help society. It is important for funding agencies to understand what it takes to conduct team science so that they are aware of how best to structure funding mechanisms.

Policymakers use the findings produced by science teams to develop their reports and arguments for making changes to policy. With the research produced, they make informed, justifiable, and strategic decisions while also providing a foundation for accountability and evaluation. The research produced helps frame issues, assess risks and impacts, and build strong arguments by offering proof of existing problems or the success of interventions. Comparative studies guide policymakers in learning from other regions, while research also justifies budget allocations by showing where resources can be used most effectively.

Finally, science of team science scholars use various research methods to study teamwork within and external to the science team context.

# Terminology

Terminology in the science of team science is used inconsistently, often leading to confusion. For example, the term *team science* is often applied ambiguously, sometimes referring narrowly to the study of science teams and at other times, to the broader science of teams across disciplines. This

INTRODUCTION 19

lack of precision can lead to misunderstandings among researchers, practitioners, and funders who engage with the team science literature. Scholars in the field frequently report that this lack of clarity in terminology not only hampers communication but also potentially impedes interdisciplinary collaboration by creating differing expectations and interpretations of shared research goals.

Adding to this challenge is the inconsistent use of such terms as *cross-disciplinary*, *interdisciplinary*, *multidisciplinary*, and *transdisciplinary*. Although these terms each describe specific types of collaborative approaches in team science, they are often conflated or used interchangeably across disciplines. Within the science of team science, these terms have been defined to delineate distinct levels of integration, interdependence, communication, and coordination among team members from different disciplines. For example, *multidisciplinary teams* typically draw from various fields without integrating knowledge deeply, while *transdisciplinary teams* aim for seamless integration of disciplinary perspectives. However, in other scientific fields, these terms are not always used with such precision, leading to potential misalignment when teams from multiple disciplines collaborate. See Box 1-3 for these definitions and Appendix B for further discussion.

Inconsistent terminology can affect funding decisions, grant evaluation processes, and team formation strategies, potentially reducing the effectiveness of funded initiatives. By striving for consistency in the use of terms, team science scholars and funders can promote clearer communication and set uniform expectations for research crossing disciplines. Clear, standardized terminology will improve understanding within the field and contribute to more productive partnerships and innovative approaches in research that spans multiple scientific domains.

# BOX 1-3 Definitions of Collaborative Approaches

- Cross-disciplinary: Cross-disciplinary research refers to collaboration among multiple disciplines toward a shared objective.
- Interdisciplinary: In interdisciplinary research, concepts are blended or juxtaposed from different disciplines.
- Multidisciplinary: In multidisciplinary research, each discipline makes separate contributions in an additive way.
- Transdisciplinary: Transdisciplinary research involves the advancement and integration of discipline-specific theories, concepts, and methods.

#### ORGANIZATION OF THE REPORT

This report provides a comprehensive overview of team science and cross-disciplinary collaborations.

Chapter 2 delves into a review of the current state of the science of team science, beginning with a discussion of its definition and evolution. It highlights key findings from previous reports and assesses progress made since their publication. It also examines emerging changes in the team science landscape and practices, as well as unanticipated developments.

Chapter 3 considers best practices for team science and their effectiveness and supporting evidence base. The chapter begins by discussing education, training, and professional development. The second part of the chapter discusses best practices for supporting science teams, looking closely at the key areas of development, conceptualization, implementation, and translation.

Chapter 4 considers best practices for external support for team science. The chapter discusses exploring institutional policies related to tenure, promotion, and credit for team science contributions, alongside considerations for material and data sharing, ethical approvals, compliance, and staffing needs. The chapter also delves into infrastructure requirements, funding considerations, and incentives and reward structures within the broader community.

Chapter 5 focuses on appropriate ways to evaluate the processes and outcomes of team science and its effect on interested parties. The chapter delineates proximal and distal outcomes for individuals, teams, institutions, science, and society, emphasizing factors such as affective, behavioral, and cognitive processes, as well as broader societal impacts such as community engagement and solutions to pressing global challenges.

Chapter 6 focuses on identifying research gaps, synthesizing what the committee has learned and presenting its overarching conclusions and recommendations.

Finally, five appendices correspond to the content of the report: Appendix A includes committee member biosketches, Appendix B gives background on the science of team science, Appendix C gives an overview of the factors and characteristics of effective teams, Appendix D provides a sampling of survey-based assessments in the literature on teams, and Appendix E contains a glossary.

#### REFERENCES

An, J. Y., Marchalik, R. J., Sherrer, R. L., Baiocco, J. A., & Rais-Bahrami, S. (2020). Authorship growth in contemporary medical literature. SAGE Open Medicine, 8, https://doi.org/2050312120915399

INTRODUCTION 21

Corburn, J. (2007). Community knowledge in environmental health science: Co-producing policy expertise. *Environmental Science & Policy*, 10(2), 150–161. https://doi.org/10.1016/j.envsci.2006.09.004

- Fiore, S. M. (2008). Interdisciplinarity as teamwork: How the science of teams can inform team science. *Small Group Research*, 39(3), 251–277. https://doi.org/10.1177/1046496408317797
- Fiore, S. M., Gabelica, C., Wiltshire, T. J., & Stokols, D. (2019). Training to be a (team) scientist. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 421–444). Springer International Publishing. https://doi.org/10.1007/978-3-030-20992-6\_33
- Hall, K. L., Feng, A. X., Moser, R. P., Stokols, D., & Taylor, B. K. (2008). Moving the science of team science forward: Collaboration and creativity. *American Journal of Preventive Medicine*, 35(2), S243–S249. https://doi.org/10.1016/j.amepre.2008.05.007
- Hall, K. L., Vogel, A. L., Huang, G. C., Serrano, K. J., Rice, E. L., Tsakraklides, S. P., & Fiore, S. M. (2018). The science of team science: A review of the empirical evidence and research gaps on collaboration in science. *American Psychologist*, 73(4), 532–548. https://doi.org/10.1037/amp0000319
- Lee, Y.-N., Walsh, J. P., & Wang, J. (2015). Creativity in scientific teams: Unpacking novelty and impact. Research Policy, 44(3), 684–697. https://doi.org/10.1016/j.respol.2014.10.007
- National Academies of Sciences, Engineering, and Medicine (2022). Human-AI teaming: State-ofthe-art and research needs. The National Academies Press. https://doi.org/10.17226/26355
- National Research Council. (2015). Enhancing the effectiveness of team science. The National Academies Press. https://doi.org/10.17226/19007
- Paul, A., Segreti, M., Pani, P., Brunamonti, E., & Genovesio, A. (2024). The increasing authorship trend in neuroscience: A scientometric analysis across 11 countries. *IBRO Neuroscience Reports* 17, 52–57. https://doi.org/10.1016/j.ibneur.2024.05.012
- Salazar, M. R., Lant, T. K., Fiore, S. M., & Salas, E. (2012). Facilitating innovation in diverse science teams through integrative capacity. *Small Group Research*, 43(5), 527–558. https://doi.org/10.1177/1046496412453622
- Stokols, D., Hall, K. L., Taylor, B. K., & Moser, R. P. (2008). The science of team science: Overview of the field and introduction to the supplement. *American Journal of Preventive Medicine*, 35(2), S77–S89. https://doi.org/10.1016/j.amepre.2008.05.002
- Wandersman, A. (2003). Community science: Bridging the gap between science and practice with community-centered models. *American Journal of Community Psychology*, 31(3-4), 227–242. https://doi.org/10.1023/a:1023954503247
- Wuchty, S., Jones, B. F., & Uzzi, B. (2007). The increasing dominance of teams in production of knowledge. *Science*, 316(5827), 1036–1039.



2

# Review of Current Science of Team Science

Key findings from a previous National Academies of Sciences, Engineering, and Medicine (National Academies) consensus report, *Enhancing the Effectiveness of Team Science*, aimed to deepen understanding of science teamwork (National Research Council, 2015). This chapter reviews those findings and addresses changes in the team science landscape emerging since the previous report's release, drawing on materials from the committee's information-gathering sessions. The chapter also considers how key recommendations from the 2015 report have been implemented.

### KEY TAKEAWAYS FROM ENHANCING THE EFFECTIVENESS OF TEAM SCIENCE

The committee for the 2015 report was charged with conducting a consensus study to recommend strategies for enhancing the effectiveness of team science and exploring factors that affect science team dynamics, effectiveness, and productivity (National Research Council, 2015). Through a literature review and expert presentations, the committee examined factors at the team, center, and institute levels to understand their influence on science team effectiveness. In addition, the consensus study explored various management approaches and leadership styles that affect science team effectiveness. The committee also reviewed how tenure and promotion policies help or hinder academic researchers who participate in research teams. Finally, the committee considered organizational factors, such as human resources policies and cyberinfrastructure, as well as organizational structures, policies, and practices, that might impact the effectiveness of science teams.

At the outset, the National Academies committee identified a set of key features that pose challenges for science teams (National Research Council, 2015). These features include a highly varied membership, which can lead to differing perspectives and approaches that can sometimes complicate collaboration. Science teams often need deep knowledge integration arising from the goal of merging disparate expertise and disciplinary backgrounds (National Research Council, 2015).

Some science teams experience permeable boundaries, where membership may change and adapt depending on phases of the project. This can result in ambiguity regarding roles and responsibilities and in the amount each team member is expected to contribute to a given project phase. High task interdependence often requires close coordination and cooperation among team members to achieve shared objectives.

Furthermore, large science teams often face difficulties with complex coordination and communication (National Research Council, 2015). As the 2015 report noted, large science teams often better reflect structures known as multiteam systems in the field of organizational science (Zaccaro et al., 2011). *Multiteam systems* are interdependent sets of two or more component teams pursuing shared superordinate goals; they offer multiple benefits, including resource capacity, flexibility, and component team specialization (Mathieu et al., 2002; Shuffler & Carter, 2018). However, multiteam system coordination can be extremely challenging given the large size, complexity, and dynamism of these systems. Moreover, multiteam systems often experience goal misalignments within or across teams, which can hinder collaboration and cohesion across the research environment. Geographic dispersion adds another layer of complexity, requiring effective virtual communication and coordination mechanisms.

To address these types of challenges, the 2015 committee considered the broad body of literature in, and experts from, the social sciences to understand findings on teams and organizations (National Research Council, 2015). It highlighted the robust body of research on teamwork going back decades that shows how team processes relate to team effectiveness. The committee identified interventions, focused on team composition, development, and leadership, that support teamwork and provide a route to enhancing team effectiveness.

Other interventions can further enhance team science outcomes, such as improving virtual collaboration practices and technologies, revising promotion and tenure criteria to recognize team-based contributions, and increasing support from funding agencies to study the science of team science (National Research Council, 2015). By implementing the 2015 report recommendations, the broader scientific ecosystem can foster an environment more conducive to collaborative research and maximize the impact of team science.

The 2015 report provided nine recommendations covering the areas of team composition, team professional development, leadership for team science, supporting virtual collaboration, organizational supports for team science, funding for team science, and finally advancing research on the effectiveness of team science (National Research Council, 2015). Box 2-1 lists the recommended actions from the 2015 report, and Appendix B includes additional details about the key takeaways from the 2015 report.

# BOX 2-1 Recommendations from Enhancing the Effectiveness of Team Science

**Recommendation 1:** Team science leaders and others involved in assembling science teams and larger groups should consider making use of task analytic methods (e.g., task analysis, cognitive modeling, job analysis, cognitive work analysis) and tools that help identify the knowledge, skills, and attitudes required for effective performance of the project so that task-related diversity among team or group members can best match project needs. They should also consider applying tools such as research networking systems designed to facilitate assembly of science teams and partner with researchers to evaluate and refine these tools and task analytic methods.

**Recommendation 2:** Team-training researchers, universities, and science team leaders should partner to translate, extend, and evaluate the promising training strategies, shown to improve the effectiveness of teams in other contexts, to create professional development opportunities for science teams.

**Recommendation 3:** Leadership researchers, universities, and leaders of team science projects should partner to translate and extend the leadership literature to create and evaluate science leadership development opportunities for team science leaders and funding agency program officers.

Recommendation 4: Leaders of geographically dispersed science teams and larger groups should provide activities shown by research to help all participants develop shared knowledge (e.g., a common vocabulary and work style). These activities should include team professional development opportunities that promote knowledge sharing (see Recommendation #2 above). Leaders should also consider the feasibility of assigning some tasks to semi-independent units at each location to reduce the burden of constant electronic communication.

**Recommendation 5:** When selecting technologies to support virtual science teams or larger groups, leaders should carefully evaluate the needs of the project, and the ability of the individual participants to embrace new technologies. Organizations should promote human-centered collaboration technologies, provide technical staff, and encourage use of the technologies by providing ongoing training and technology support.

continued

#### **BOX 2-1 Continued**

**Recommendation 6:** Universities and disciplinary associations should proactively develop and evaluate broad principles and more specific criteria for allocating credit for team-based work to assist promotion and tenure committees in reviewing candidates.

**Recommendation 7:** Funders should work with the scientific community to encourage the development and implementation of new collaborative models, such as research networks and consortia; new team science incentives, such as academic rewards for team-based research (see Recommendation #6); and resources (e.g., online repositories of information on improving the effectiveness of team science and training modules).

**Recommendation 8:** Funders should require proposals for team-based research to present collaboration plans and provide guidance to scientists for the inclusion of these plans in their proposals, as well as guidance and criteria for reviewers' evaluation of these plans. Funders should also require authors of proposals for interdisciplinary or transdisciplinary research projects to specify how they will integrate disciplinary perspectives and methods throughout the life of the research project.

**Recommendation 9:** Public and private funders should support research on team science effectiveness through funding. As critical first steps, they should support ongoing evaluation and refinement of the interventions and policies recommended above and research on the role of scientific organizations (e.g., research centers, networks) in supporting science teams and larger groups. They should also collaborate with universities and the scientific community to facilitate researchers' access to key team science personnel and datasets.

SOURCE: Excerpted from National Research Council, 2015, pp. 8-13.

# IMPLEMENTING THE 2015 RECOMMENDATIONS: SUCCESSES AND CHALLENGES

The 2015 report applied research on team dynamics from various other disciplines, including organizational behavior and industrial-organizational psychology to the context of science teams (National Research Council, 2015). It thus laid the foundation for both scholars and practitioners to leverage this knowledge to enhance the effectiveness of science teams. Over the subsequent decade, the report has been downloaded over 31,000 times as of the summer of 2024 and by people in countries around the world, including from China, Canada, the United Kingdom, and Australia, demonstrating the international appeal of team science. The report's broad reach

<sup>&</sup>lt;sup>1</sup> Outreach data provided by National Academies Press, as of October 16, 2024 (National Research Council, 2015).

can also be seen in the actions of team science advocates, such as scholars, practitioners, funders, and administrators, operating at various levels in the science ecosystem.

Although the report spurred many changes, the committee for the present study identified factors that have shaped and influenced various aspects of team science over the past decade. As Chapter 1 noted, these include societal changes arising from the COVID-19 pandemic as well as increased use and sophistication of technologies such as virtual collaboration tools and generative artificial intelligence (AI). As such, the current report considers these additional factors.

#### **Indicators of Success**

One of the primary purposes of the 2015 report was to increase the recognition of team science in scientific collaborations (National Research Council, 2015). This has been one of the clearest indicators of that report's success, with influence on disciplines including biology (Full et al., 2015), health and medicine (Czajkowski et al., 2016; Hall et al., 2020; Little et al., 2017; Sargent et al., 2022), earth and environmental sciences (Gilligan, 2021; Lanier et al., 2018; Maxwell et al., 2019; Pennington et al., 2016; Wallen et al., 2019), organizational science (Burt et al., 2022; Fiore et al., 2018; Guise et al., 2017), psychology (Haynes et al., 2019; Tebes, 2018; Tebes & Thai, 2018), nursing (Conn et al., 2019), engineering (Roscoe et al., 2019), sustainability (Killion et al., 2018; von Wehrden et al., 2019), human factors (Gupta & Woolley, 2021), clinical and translational science (Pelfrey et al., 2021; Rolland et al., 2021; Vogel et al., 2021), and physics (Halford et al., 2023)—all have promoted team science principles as a mechanism for advancing their fields. This broad acceptance signifies a shift in perspective, where those conducting science acknowledge the value of cross-pollinating research. Although specialization remains crucial, scholars in many fields now recognize that significant scientific progress can be achieved through cross-disciplinary collaboration and by applying team science principles. Publications in multiple disciplines have emphasized the benefits of team science, which has helped to change the culture and approach to scientific research by promoting the integration of varied expertise and perspectives (Fiore et al., 2019; Lotrecchiano et al., 2023).

Another clear indicator of success is the growing recognition that tenure and promotion materials need to account for how the success metrics of effective collaboration differ from traditional guidelines (Brody et al., 2019; Cline et al., 2020; McHale et al., 2019; Meurer et al., 2023; Rohrbach & Genco, 2022). The engagement of junior scholars in cross-disciplinary team science can depend heavily on the incentive structures the university system provides. Without language that acknowledges participation in team

science—such as recognition for publishing in journals outside one's primary discipline, differences in authorship practices across disciplines, and participation as a co-investigator on large extramural funding efforts—junior scholars may be less inclined to engage in meaningful cross-disciplinary collaborations. Although there is still a long way to go in these efforts, recognizing that the reward structure for tenure and promotion needs to evolve is an important step forward.

The 2015 committee also noted that many research institutions had initiated efforts to promote applying team science principles (National Research Council, 2015). In the last decade, more institutions have created internal funding opportunities that promote and require translational research or research from cross-disciplinary teams.<sup>2</sup> In addition, several academic institutions have made significant investments in programs to improve collaboration within science teams. These prominent programs in clinical and translational science focus on equipping researchers with essential skills such as communication, conflict resolution, and leadership, often using the recommendations from the 2015 report in their training and guidance.<sup>3</sup> In addition, dedicated laboratories, such as the University of California, Irvine's Team Scholarship Acceleration Lab, 4 focus on promoting and applying team science principles. Such programs help science teams develop by providing talks and lunch-and-learn sessions, creating collaboration plans, offering team science training toolkits, and consulting on grant proposals. These efforts have enhanced collaboration within institutions and fostered a culture that values interdisciplinary research and teamwork.

The 2015 report was also successful in informing funding agencies, which have bolstered their support for team science initiatives (National Research Council, 2015). Whether directly or indirectly, team science is being promoted by those seeking to support complex collaborations. For example, center-level research at the National Science Foundation (NSF), including its Science and Technology Centers, Engineering Research Centers, Synthesis Centers, and Convergence Accelerators programs, has

<sup>&</sup>lt;sup>2</sup> For examples, see the Signature Research Initiative Program at American University (https://www.american.edu/research/sri.cfm), Global Grand Challenges at Cornell University (https://global.cornell.edu/global-grand-challenge/about), and funding initiatives at Virginia Commonwealth University (https://onevcuresearch.vcu.edu/funding/).

<sup>&</sup>lt;sup>3</sup> See, for example, the Center for Clinical & Translational Science & Training (https://www.cctst.org/team-science-training) and TeamMAPPS (https://teammapps.utmb.edu/).

<sup>&</sup>lt;sup>4</sup> More information about the Team Scholarship Acceleration Lab is available at https://tsal.uci.edu

<sup>&</sup>lt;sup>5</sup> See, for example, the Center for Chemical Currencies of a Microbial Planet (https://ccomp-stc.org/).

<sup>&</sup>lt;sup>6</sup> See, for example, the Center for Smart Streetscape (https://cs3-erc.org/).

<sup>&</sup>lt;sup>7</sup> See, for example, the National Synthesis Center for Emergence in the Molecular and Cellular Sciences (https://ncems.psu.edu/).

incorporated tenets of the science of team science. Many calls for proposals at NSF describe the requirements for a team-based approach to conducting center-level research, with an expectation that grant applications articulate roles clearly and describe the interdependencies of the research initiatives. Also relevant is NSF's Research Traineeship (NRT) program, which, for the past decade, has worked to improve education for graduate students. More recently, it has moved beyond only addressing the requisite knowledge, skills, and attitudes for science, technology, engineering, and mathematics (STEM) careers. Now it also emphasizes training graduate students so that they can collaborate across disciplines. The 2024 NRT call for proposals specifically refers to the 2015 National Academies report on team science effectiveness. As such, it encourages STEM workforce development that enables students to acquire competencies in collaboration as well as in their respective fields of study. This acknowledgment of team science emphasizes its practice more than its study.

Other programs, such as NSF's GERMINATION<sup>10</sup> and EAGER (Early-Concept Grants for Exploratory Research)<sup>11</sup> programs have also provided funding to those who study science teams at a more foundational level. NSF's Innovations in Graduate Education Program<sup>12</sup> has supported several initiatives aimed at improving collaboration and cross-disciplinarity in student learning. The Food and Drug Administration (FDA) has also been practicing tenets of the science of team science. For example, to improve translational science, FDA's Center for Clinical Trial Innovation<sup>13</sup> promotes team science to improve collaborative effectiveness. Similarly, the Department of Energy (DoE) has formally acknowledged the importance of teamwork in research activities. More recently, NSF and DoE have partnered in their Correctness for Scientific Computing Systems program.<sup>14</sup> This program recognizes that such research requires close collaboration between researchers and, in service of cross-disciplinarity, notes that submitters develop teams comprising experts from different disciplines.

<sup>&</sup>lt;sup>8</sup> More information about the Research Traineeship Program is available at https://new.nsf.gov/funding/initiatives/nrt

<sup>&</sup>lt;sup>9</sup> More information about the August 16, 2024, call for proposals is available at https://new.nsf.gov/funding/opportunities/us-national-science-foundation-research-traineeship-program/nsf24-597/solicitation

<sup>&</sup>lt;sup>10</sup> See https://www.nsf.gov/funding/opportunities/germination-germination-research-questions-addressing

<sup>&</sup>lt;sup>11</sup> See https://new.nsf.gov/funding/early-career-researchers#early-concept-grants-for-exploratory-research-eager-c8f

<sup>&</sup>lt;sup>12</sup> See https://new.nsf.gov/funding/initiatives/ige

 $<sup>^{13}</sup>$  See https://www.fda.gov/about-fda/center-drug-evaluation-and-research-cder/cder-center-clinical-trial-innovation-c3ti

<sup>&</sup>lt;sup>14</sup> See https://new.nsf.gov/funding/opportunities/cs2-correctness-scientific-computing-systems

A recent report on developing better scientific software notes the importance of cross-disciplinary teamwork in successful outcomes, specifically calling out a need to improve understanding on the variety of teams and individuals collaborating to develop scientific software (Heroux et al., 2023). In a section on cross-cutting themes, Heroux et al. (2023) note that social science research on teamwork can help "inform a variety of important challenges and opportunities in team-based scientific software development and use [...] Improving communication and interactions within and across teams can account for specific challenges in scientific software environments" (p. 17). The report goes on to discuss the importance of team interaction and managing conflict and communication across teams and multiteam systems (Heroux et al., 2023).

# Ongoing Challenges and the Evolution of the Scientific Landscape

The aforementioned progress is not universal, and several long-standing and emerging challenges continue to hinder the success of science teams. One of the more difficult challenges, funding for the scientific study of science teams, has not been addressed. While sponsors fund science teams to conduct research and fund research on the science of teams more generally—understanding team dynamics in organizations and the military, for example—there is a significant lack of funding for the science of team science research (i.e., the scholarly examination of teamwork in science teams). This is problematic in that it contributes to the growing gap between theory, evidence, and practice in team science. Conducting research on teams is inherently challenging for several reasons, including the complexity of capturing dynamic, multilevel interactions; the need for longitudinal data to understand team evolution; and the difficulty of isolating variables in a team setting (Delice et al., 2019; Kozlowski, 2015; Shuffler & Cronin, 2019). Logistical challenges also arise, such as finding times when all team members are available and recruiting the entire team to participate in a study. Furthermore, scientists are often reluctant to participate, with possible reasons including their busy schedules, their multiple team memberships, and their involvement in both research activities and other activities relevant to their jobs. In addition, they might be skeptical of social science methodologies, such as qualitative methods and survey research, questioning their relevance or validity (Ledford, 2015). Because of these challenges and the lack of funding to address them, there is an unfortunate scarcity of mid- to large-scale research on science teams. Hence, as will be apparent throughout the present report, much of the research on team science comes from studies that do not focus exclusively on science teams and/or that are qualitative in nature.

An additional remaining challenge is the lack of systematic training of team science competencies. Team science members receive training in their discipline during the time they complete their terminal degrees. Often, however, they do not receive formal training on how to collaborate or share leadership roles with others on a team. As such, while scientists have acquired competencies relevant to their scientific tasks, too often they can fall short on competencies needed to work as a team on those tasks. This can lead to problematic collaborations that hinder the effectiveness of science teams.

While work to develop and execute programs for training in team science competencies has been considerable since the 2015 report, these programs exhibit considerable differences in both the specific competencies they aim to cultivate and the formats they use. Training programs can range from brief 1-hour sessions to comprehensive semester-long courses, and from interactive workshops to informal lunch-time lectures. Furthermore, the evaluation methods for these programs' effectiveness are equally varied, relying on metrics such as connecting the attendance to training with the number of new grant applications funded or publications accepted by attendees; in other cases, no evaluation metrics are used at all. This lack of standardization results in inconclusive training outcomes, underscoring the need for a more cohesive and universally adopted approach to team science education.

#### COVID-19

The emergence of COVID-19 caused many changes in society, including lockdowns, social distancing, and working from home, all of which affected the way teams worked. The National Academies has released multiple reports on COVID-19 and its effects on education, transportation, community engagement, and other aspects of dealing with this major upheaval in society.<sup>15</sup>

From a team science perspective, arguably the largest impact during and following the pandemic has been in remote and hybrid workplace policies and the concomitant reliance on virtual communication tools to complete work (Karl et al., 2022; Woodruff et al., 2021; see also Rubinger et al., 2020, for a discussion regarding the changes to scientific research meetings and conferences). During the pandemic, work-from-home mandates brought to light associated work-life balance and accessibility gains that many employees now come to expect as part of evolving workplace benefits (Choudhury et al., 2022; Lake & Maidment, 2023). As the pandemic concluded, employers needed to balance the justification for bringing personnel back into the office with a heightened awareness of the benefits of working from home.

 $<sup>^{15}</sup>$  See https://nap.nationalacademies.org/search/?rpp=20&pp=4&ft=1&term=covid=15

As a compromise, employers established flexible hybrid work policies, but at the time of the present report, employers have been increasingly requiring employees to be present in the office—in some cases for the full work week—arguing that this arrangement is better for team cohesion, productivity, organizational culture, and innovation, among other things (Agovino, 2023; The White House, 2025). Employers may have internal data to draw upon to shape their evolving policies and practices, but there is limited scientific literature examining how work from the office, work from home, and hybrid work affect team performance and desired outcomes.

One study from 2024 analyzed the innovation activity of over 48,000 information technology professionals at an India-based firm, nearly all of whom had a minimum of a college degree in engineering (Gibbs et al., 2024). The study tracked a quantifiable internal innovation metric pre-pandemic (work from office), mid-pandemic (work from home), and post-pandemic (hybrid work). They found that while the quantity of submitted ideas did not change in transitioning from working at the office to at home, the average *quality* of submitted ideas declined (Gibbs et al., 2024).

In a related study, the authors found that productivity per hour decreased as well because of difficulties performing some work tasks in a virtual environment (Gibbs et al., 2023). As hybrid work in the postpandemic era began, the *quantity* of submitted ideas decreased<sup>16</sup> (Gibbs et al., 2024). The authors posit that these declines are, at least partially, a result of less spontaneous interactions that naturally occur in the office, contrasted against work from home and hybrid work, which require some level of coordination, thus limiting the ability to have "productive accidents" (Gibbs et al., 2023). This study suggests that innovation suffers when engineers, in this case, are not in the office at the same time. However, the authors make clear that the reduced innovation output needs to be balanced against employee productivity, satisfaction, and environmental implications (Gibbs et al., 2023). In addition, when properly supported, flexible and remote work options tend to better align with ergonomics and wellness needs (Beckel & Fisher, 2022), key considerations for promoting effective team science.

A McKinsey Global Survey of executives describes how COVID-19 had changed the digital landscape of companies:

In just a few months' time, the COVID-19 crisis has brought about years of change in the way companies in all sectors and regions do business. According to a new McKinsey Global Survey of executives, their companies have accelerated the digitization of their customer and supply-chain

<sup>&</sup>lt;sup>16</sup> This decrease in idea quantity was greater for females than for males, revealing a gender asymmetry with potential consequences for performance biases and career advancement.

interactions and of their internal operations by three to four years. And the share of digital or digitally enabled products in their portfolios has accelerated by a shocking seven years. Nearly all respondents say that their companies have stood up at least temporary solutions to meet many of the new demands on them, and much more quickly than they had thought possible before the crisis. What's more, respondents expect most of these changes to be long lasting and are already making the kinds of investments that all but ensure they will stick. In fact, when we asked executives about the impact of the crisis on a range of measures, they say that funding for digital initiatives has increased more than anything else—more than increases in costs, the number of people in technology roles, and the number of customers. (O'Toole et al., 2020, p. 2)

The acceptance of digital tools included using video conferencing for work, virtual national conferences, and social interaction with family and friends. During the pandemic, lockdown and continued social isolation also affected the way teams pursued science and how teams and research groups interacted. After lockdowns ended, a new work style was ushered in with "hybrid" rising in prevalence (Handke et al., 2024). In fact, since the COVID-19 pandemic, the adoption and integration of hybrid and virtual collaboration software have increased significantly across science teams (Kilcullen et al., 2021; Lematta et al., 2021; Miller, 2020; Paunov & Planes-Satorra, 2021). These tools offer various functions that enhance team communication, such as global video conferencing, text-to-speech capabilities, and large language models for summarizing meeting notes.

### Technology

Generative AI tools, in particular, which were not widely available until after the 2015 report, could help advance scientific collaboration by assisting in some of the tasks that may be precursors to scientific progress, such as drafting reports and generating basic data analysis code to assist individuals and teams. In this sense, generative AI applications or agents may be viewed eventually as valuable tools or teammates in team science (National Academies of Sciences, Engineering, and Medicine, 2021). However, the use of these tools raises several concerns, including potential biases or confabulations in AI-generated content that still require human oversight (Bryson et al., 2017) and ethical implications of AI-authored or AI-assisted publications (Schlagwein & Willcocks, 2023).

The 2015 report recommended using collaboration software (National Research Council, 2015). Although studies address technology use in virtual and hybrid teams, including hypothesized factors such as knowledge, skills, and attitudes that affect virtual teams and their technology use (Gibson et al., 2022; Kilcullen et al., 2021; Ofosu-Ampong & Acheampong, 2022;

Schulze & Krumm, 2017), it remains unclear how—or whether—the collaborative functions of these tools truly facilitate effectiveness in science teams. An additional research issue is whether these new capabilities are being implemented more broadly to improve aspects such as adherence to universal design principles in science teams, and, if so, whether they have produced the desired result.

In addition to possible beneficial outcomes of technology use, many collaboration challenges can result. To mitigate or reverse any potential negative effects of a team working virtually, key factors such as team trust and conflict management require thoughtful leadership and communication that will likely be mediated by technology design and use in virtual and hybrid teams (Schulze & Krumm, 2017).

# **Geopolitical Tensions**

Each of these areas underscores the significance of collaboration across geographic regions. However, with global collaborations, it is imperative to consider ever-changing geopolitical dynamics. Scientific progress and collaboration do not exist in a vacuum; they operate in the context of geopolitical forces that can both facilitate and potentially hinder collaboration among nations (Wagner & Cai, 2022). Considering the role of geopolitics in scientific progress and collaboration is not new. Ongoing research in Antarctica enabled by the Antarctic Treaty, for example, and in space as illustrated by the Outer Space Treaty and International Space Station, are examples of geopolitical challenges overcome for the advancement of science (Secretariat of the Antarctic Treaty, n.d.; United Nations Office for Outer Space Affairs, n.d.).

Nevertheless, many challenges persist. For instance, scientific collaborations between China and the United States began to decline in 2019, when political tensions around science, technology, and innovation arose, with the United States claiming that China was violating intellectual property norms (Wagner & Cai, 2022). Fostering sustained and productive international scientific collaborations will require continually addressing and navigating geopolitical complexities.

An increasingly important aspect of this geopolitical landscape is *research security*, which refers to protecting the integrity, confidentiality, and security of research activities, particularly those including human subjects, in a world where intellectual property theft, cyber threats, and the misuse of scientific knowledge are growing concerns (Office of the Director of National Intelligence, n.d.). As geopolitical tensions rise, particularly between major global powers, and as global scientific collaboration becomes more complex, safeguarding research from these threats is paramount (e.g., Gibney, 2024). National Security Presidential Memorandum-33 highlights

the importance of safeguarding federally funded research and development for ensuring U.S. national security and outlines specific research security requirements that all institutions receiving more than \$50 million in federal research and development funding are required to implement (The White House, 2021). This can include implementing robust cybersecurity measures; providing research security training; and ensuring compliance with international regulations related to export control and foreign research travel, as well as managing the risks associated with cross-border collaborations, especially in high-stakes research areas.

Geopolitical tensions can lead to restrictive policies on traveling to conferences, establishing collaborations, recruiting graduate students, buying and sharing technology, and data privacy and security. For example, the Florida "Countries of Concern" law for higher education, signed in 2023, imposes restrictions on Florida's public universities' interactions with China, Cuba, Venezuela, Russia, Iran, Syria, and North Korea (Florida Senate, 2022). The law prohibits Florida colleges from receiving grants, forming partnerships, or conducting research collaborations with institutions based in or controlled by these countries (State University System of Florida Board of Governors, 2023). Although the legislation was aimed to prevent potential risks related to national security, including intellectual property theft and influence by foreign governments, it also led Florida public institutions to close international programs in China and to end partnerships and reduce graduate student recruitment efforts with these nations (e.g., Knox, 2024; see also Rivero, 2024; Voice of America News, 2024). As a consequence of this law, faculty, researchers, staff, and student assistants at these institutions have to go through additional approval processes to travel to conferences and research events held in these countries.

In conclusion, significant growth and attention in the field of team science has led to numerous successes in the application of team science principles and recommendations from the previous report (National Research Council, 2015). Nonetheless, there is still room for improvement as the applications of some of the recommendations have not been fully realized and new challenges have arisen over the last decade.

#### REFERENCES

Agovino, T. (2023, July 17). The great compromise: Return-to-office policies are evolving. SHRM. https://www.shrm.org/topics-tools/news/employee-relations/the-great-compromise-return-to-office-policies-are-evolving

Beckel, J. L. O., & Fisher, G. G. (2022). Telework and worker health and well-being: A review and recommendations for research and practice. *International Journal of Environmental Research and Public Health*, 19(7), 3879. https://doi.org/10.3390/ijerph19073879

- Brody, A. A., Bryant, A. L., Perez, G. A., & Bailey, D. E. (2019). Best practices and inclusion of team science principles in appointment promotion and tenure documents in research intensive schools of nursing. *Nursing Outlook*, 67(2), 133–139. https://doi.org/10.1016/j. outlook.2018.11.005
- Bryson, J. J., Diamantis, M. E., & Grant, T. D. (2017). Of, for, and by the people: The legal lacuna of synthetic persons. *Artificial Intelligence and Law*, 25(3), 273–291. https://doi.org/10/gdwpkq
- Burt, B. A., Stone, B., Perkins, T., Polk, A., Ramirez, C., & Rosado, J. (2022). Team culture of community: Cultural practices for scientific team cohesion and productivity. *Small Group Research*, 53(6), 855–890. https://doi.org/10.1177/10464964221097699
- Choudhury, P., Khanna, T., Makridis, C., & Schirmann, K. (2022). *Is hybrid work the best of both worlds? Evidence from a field experiment* (HBS Technology & Operations Mgt. Unit Working Paper No. 22-063; HBS Strategy Unit Working Paper No. 22-063). Harvard Business School. https://doi.org/10.2139/ssrn.4068741
- Cline, H., Coolen, L., de Vries, S., Hyman, S., Segal, R., & Steward, O. (2020). Recognizing team science contributions in academic hiring, promotion, and tenure. *The Journal of Neuroscience*, 40(35), 6662–6663. https://doi.org/10.1523/JNEUROSCI.1139-20.2020
- Conn, V. S., McCarthy, A. M., Cohen, M. Z., Anderson, C. M., Killion, C., DeVon, H. A., Topp, R., Fahrenwald, N. L., Herrick. L. M., Benefield, L. E., Smith, C. E., Jefferson, U. T., & Anderson, E. A. (2019). Pearls and pitfalls of team science. Western Journal of Nursing Research, 41(6), 920–940. https://doi.org/10.1177/0193945918793097
- Czajkowski, S. M., Lynch, M. R., Hall, K. L., Stipelman, B. A., Haverkos, L., Perl, H., Scott, M. S., & Shirley, M. C. (2016). Transdisciplinary translational behavioral (TDTB) research: Opportunities, barriers, and innovations. *Translational Behavioral Medicine*, 6(1), 32–43. https://doi.org/10.1007/s13142-015-0367-3
- Delice, F., Rousseau, M., & Feitosa, J. (2019). Advancing teams research: What, when, and how to measure team dynamics over time. *Frontiers in Psychology*, 10, Article 1324. https://doi.org/10.3389/fpsyg.2019.01324
- Fiore, S. M., Gabelica, C., Wiltshire, T. J., & Stokols, D. (2019). Training to be a (team) scientist. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 421–444). Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6\_33
- Fiore, S. M., Graesser, A., & Greiff, S. (2018). Collaborative problem-solving education for the twenty-first-century workforce. *Nature Human Behaviour*, 2(6), 367–369. https://doi.org/10.1038/s41562-018-0363-y
- Florida Senate. (2022). 2022 Florida Statutes (including 2022C, 2022D, 2022A, and 2023B). https://www.flsenate.gov/laws/statutes/2022/286.101
- Full, R. J., Dudley, R., Koehl, M. A. R., Libby, T., & Schwab, C. (2015). Interdisciplinary laboratory course facilitating knowledge integration, mutualistic teaming, and original discovery. *Integrative and Comparative Biology*, 55(5), 912–925. https://doi.org/10.1093/ icb/icv095
- Gibbs, M., Mengel, F., & Siemroth, C. (2023). Work from home and productivity: Evidence from personnel and analytics data on information technology professionals. *Journal of Political Economy Microeconomics*, 1(1), 7–41. https://doi.org/10.1086/721803
- \_\_\_\_. (2024). Employee innovation during office work, work from home and hybrid work. Scientific Reports, 14(1). https://doi.org/10.1038/s41598-024-67122-6
- Gibney, E. (2024, September 18). CERN prepares to expel Russian scientists—But won't completely cut ties. *Nature*. https://www.nature.com/articles/d41586-024-02982-6

- Gibson, C. B., Dunlop, P. D., Majchrzak, A., & Chia, T. (2022). Sustaining effectiveness in global teams: The coevolution of knowledge management activities and technology affordances. *Organization Science*, 33(3), 1018–1048. https://doi.org/10.1287/orsc.2021.1478
- Gilligan, J. M. (2021). Expertise across disciplines: Establishing common ground in interdisciplinary disaster research teams. Risk Analysis, 41(7), 1171–1177. https://doi.org/10.1111/risa.13407
- Guise, J. M., Winter, S., Fiore, S. M., Regensteiner, J. G., & Nagel, J. (2017). Organizational and training factors that promote team science: A qualitative analysis and application of theory to the National Institutes of Health's BIRCWH career development program. *Journal of Clinical and Translational Science*, 1(2), 101–107. https://doi.org/10.1017/ cts.2016.17
- Gupta, P., & Woolley, A. W. (2021). Articulating the role of artificial intelligence in collective intelligence: A transactive systems framework. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 65(1), 670–674. https://doi.org/10.1177/1071181321651354c
- Halford, A. J., Bard, C. M., Burrell, A. G., McGranaghan, R. M., Wilson III, L. B., Jones Jr, M., Dong, C., Wang, L., Pulkkinen, T. I., Turner, N., Liemohn, M. W., & Klenzing, J. (2023). The importance of recruitment and retention in heliophysics: It's not just a pipeline problem. Frontiers in Astronomy and Space Sciences, 10. https://doi.org/10.3389/fspas.2023.1216449
- Hall, K. L., Vogel, A. L., & Croyle, R. T. (2020). Strategies for team science success. Springer. https://doi.org/10.1007/978-3-030-20992-6
- Handke, L., Aldana, A., Costa, P. L., & O'Neill, T. A. (2024). Hybrid teamwork: What we know and where we can go from here. *Small Group Research*, 55(5), 805–835. https://doi.org/10.1177/10464964241279078
- Haynes, N. J., Vandenberg, R. J., DeJoy, D. M., Wilson, M. G., Padilla, H. M., Zuercher, H. S., & Robertson, M. M. (2019). The workplace health group: A case study of 20 years of multidisciplinary research. *American Psychologist*, 74(3), 380–393. https://doi.org/10.1037/amp0000445
- Heroux, M. A., Bernholdt, D. E., McInnes, L. C., Cary, J. R., Katz, D. S., Raybourn, E. M., & Rouson, D. (2023). Basic research needs in the science of scientific software development and use: Investment in software is investment in science. Advanced Scientific Computing Research, U.S. Department of Energy. https://doi.org/10.2172/1846009
- Karl, K. A., Peluchette, J. V., & Aghakhani, N. (2022). Virtual work meetings during the COVID-19 pandemic: The good, bad, and ugly. Small Group Research, 53(3), 343–365. https://doi.org/10.1177/10464964211015286
- Kilcullen, M., Feitosa, J., & Salas, E. (2021). Insights from the virtual team science: Rapid deployment during COVID-19. *Human Factors*, 64(8), 1429–1440. https://doi.org/10.1177/0018720821991678
- Killion, A. K., Sterle, K., Bondank, E. N., Drabik, J. R., Bera, A., Alian, S., Goodrich, K. A., Hale, M., Myer, R. A., Phung, Q., Shew, A. M., & Thayer, A. W. (2018). Preparing the next generation of sustainability scientists. *Ecology and Society*, 23(4). https://www.jstor. org/stable/26796895
- Knox, L. (2024). Another Chinese partnership bites the dust. *Inside Higher Education*. https://www.insidehighered.com/news/global/us-colleges-world/2024/02/02/fiu-ends-chinese-partnerships-comply-state-law
- Kozlowski, S. W. J. (2015). Advancing research on team process dynamics: Theoretical, methodological, and measurement considerations. Organizational Psychology Review, 5(4), 270–299.
- Lake, B., & Maidment, D. W. (2023). "Is this a new dawn for accessibility?" A qualitative interview study assessing remote working experiences in adults with physical disabilities post COVID-19. WORK, 76(2), 437–451. https://doi.org/10.3233/wor-220622

- Lanier, A. L., Drabik, J. R., Heikkila, T., Bolson, J., Sukop, M. C., Watkins, D. W., Rehage, J., Mirchi, A., Engel, V., & Letson, D. (2018). Facilitating integration in interdisciplinary research: Lessons from a South Florida water, sustainability, and climate project. Environmental Management, 62, 1025–1037. https://doi.org/10.1007/s00267-018-1099-1
- Ledford, H. (2015). How to solve the world's biggest problems. *Nature*, 525(7569), 308–311. https://doi.org/10.1038/525308a
- Lematta, G. J., Corral, C. C., Buchanan, V., Johnson, C. J., Mudigonda, A., Scholcover, F., Wong, M. E., Ezenyilimba, A., Baeriswyl, M., Kim, J., Holder, E., Chiou, E. K., & Cooke, N. J. (2021). Remote research methods for human–AI–robot teaming. *Human Factors and Ergonomics in Manufacturing & Service Industries*, 32(1), 133–150. https://doi.org/10.1002/hfm.20929
- Little, M. M., St Hill, C. A., Ware, K. B., Swanoski, M. T., Chapman, S. A., Lutfiyya, M. N., & Cerra, F. B. (2017). Team science as interprofessional collaborative research practice: A systematic review of the science of team science literature. *Journal of Investigative Medicine*, 65(1), 15–22. https://doi.org/10.1136/jim-2016-000216
- Lotrecchiano, G. R., Bennett, L. M., & Vovides, Y. (2023). A framework for developing team science expertise using a reflective-reflexive design method (R2DM). *Humanities and Social Sciences Communications*, 10, 810. https://doi.org/10.1057/s41599-023-02298-2
- Mathieu, J. E., Marks, M. A., & Zaccaro, S. J. (2002). Multiteam systems. In N. Anderson, D. S. Ones, H. K. Sinangil, & C. Viswesvaran (Eds.), Handbook of industrial, work and organizational psychology. Vol. 2: Organizational psychology (pp. 289–313). Sage.
- Maxwell, K., Hubbell, B., & Eisenhauer, E. (2019). Institutional insights on integrating social and environmental science for solutions-driven research. *Environmental Science & Policy*, 101, 97–105. https://doi.org/10.1016/j.envsci.2019.08.003
- McHale, S. M., Ranwala, D. D., DiazGranados, D., Bagshaw, D., Schienke, E., & Blank, A. E. (2019). Promotion and tenure policies for team science at colleges/schools of medicine. *Journal of Clinical and Translational Science*, 3(5), 245–252. https://doi.org/10.1017/cts.2019.401
- Meurer, J. R., Fertig, J., Garrison, O., & Shaker, R. (2023). Team science criteria and processes for promotion and tenure of Health Science University Faculty. *Journal of Clinical and Translational Science*, 7(1), e27. https://doi.org/10.1017/cts.2022.523
- Miller, M. (2020, April 2). Zoom CEO says company reached 200 million daily users in March. *The Hill*. https://thehill.com/policy/cybersecurity/490794-zoom-ceo-says-company-reached-200-million-daily-users-in-march/
- National Academies of Sciences, Engineering, and Medicine. (2021). *Human-AI teaming: State of the art and research needs*. The National Academies Press. https://doi.org/10.17226/26355
- National Research Council. (2015). Enhancing the effectiveness of team science. The National Academies Press. https://doi.org/10.17226/19007
- O'Toole, C., Schneider, J., Smaje, K., & LaBerge, L. (2020, October). How COVID-19 has pushed companies over the technology tipping point—and transformed business forever. *McKinsey & Company*. https://www.mckinsey.com/capabilities/strategy-and-corporate-finance/our-insights/how-covid-19-has-pushed-companies-over-the-technology-tipping-point-and-transformed-business-forever#/
- Office of the Director of National Intelligence. (n.d.). Research security. *The National Counter-intelligence and Security Center*. https://www.dni.gov/index.php/safeguarding-science/research-security
- Ofosu-Ampong, K., & Acheampong, B. (2022). Adoption of contactless technologies for remote work in Ghana post-COVID-19: Insights from technology-organisation-environment framework. *Digital Business*, 2(2), 100023. https://doi.org/10.1016/j.digbus.2022.100023

- Paunov, C., & Planes-Satorra, S. (2021). What future for science, technology and innovation after COVID-19? (OECD Science, Technology and Industry Policy Papers, No. 107). Organisation for Economic Co-Operation and Development. https://doi.org/10.1787/ de9eb127-en
- Pelfrey, C. M., Goldman, A. S., & DiazGranados, D. J. (2021). What does team science look like across the CTSA consortium? A qualitative analysis of the Great CTSA Team Science Contest submissions. *Journal of Clinical and Translational Science*, 5(1), e154. https:// doi.org/10.1017/cts.2021.812
- Pennington, D., Bammer, G., Danielson, A., Gosselin, D., Gouvea, J., Habron, G., Hawthorne, D., Parnell, R., Thompson, K., Vincent, S., & Wei, C. (2016). The EMBeRS project: Employing model-based reasoning in socio-environmental synthesis. *Journal of Environmental Studies and Sciences*, 6, 278–286. https://doi.org/10.1007/s13412-015-0335-8
- Rivero, D. (2024). 'Those folks are not welcome': How a DeSantis priority is changing Florida universities. NPR. https://www.wusf.org/education/2024-03-02/desantis-2023-law-florida-universities-fiu-china-cuba-international-education
- Rohrbach, A., & Genco, C. A. (2022). Rewarding team science in tenure and promotion practices: An operational imperative for the academic research enterprise of the twenty-first century. In D. Lerner, M. E. Palm, T. W. Concannon (Eds.), *Broadly engaged team science in clinical and translational research* (pp. 231–241). Springer, Cham. https://doi.org/10.1007/978-3-030-83028-1\_25
- Rolland, B., Hohl, S. D., & Johnson, L. J. (2021). Enhancing translational team effectiveness: The Wisconsin Interventions in Team Science framework for translating empirically informed strategies into evidence-based interventions. *Journal of Clinical and Translational Science*, 5(1), e158. https://doi.org/10.1017/cts.2021.825
- Roscoe, R. D., Becker, D. V., Branaghan, R. J., Chiou, E. K., Gray, R., Craig, S. D., Gutzwiller, R. S., & Cooke, N. J. (2019). Bridging psychology and engineering to make technology work for people. *American Psychologist*, 74(3), 394–406. https://doi.org/10.1037/amp0000444
- Rubinger, L., Gazendam, A., Ekhtiari, S., Nucci, N., Payne, A., Johal, H., Khanduja, V., & Bhandari, M. (2020). Maximizing virtual meetings and conferences: A review of best practices. *International Orthopaedics*, 44, 1461–1466. https://doi.org/10.1007/s00264-020-04615-9
- Sargent, L., Slattum, P., Brooks, M., Gendron, T., Mackiewicz, M., Diallo, A., Waters, L., Winship, J., Battle, K., Ford, G., Falls, K., Chung, J., Zanjani, F., Pretzer-Abott, I., Price, E. T., Prom-Wormley, E., Parsons, P., & iCubed Health and Wellness in Aging Transdisciplinary Core. (2022). Bringing transdisciplinary aging research from theory to practice. *The Gerontologist*, 62(2), 159–168. https://doi.org/10.1093/geront/gnaa214
- Schlagwein, D., & Willcocks, L. (2023). 'ChatGPT et al.': The ethics of using (generative) artificial intelligence in research and science. *Journal of Information Technology*, 38(3), 232–238. https://doi.org/10/gtm4f4
- Schulze, J., & Krumm, S. (2017). The "virtual team player": A review and initial model of knowledge, skills, abilities, and other characteristics for virtual collaboration. Organizational Psychology Review, 7(1), 66–95. https://doi.org/10.1177/2041386616675522
- Secretariat of the Antarctic Treaty. (n.d.). The Antarctic Treaty. https://www.ats.aq/e/antarctictreaty.html
- Shuffler, M. L., & Carter, D. R. (2018). Teamwork situated in multiteam systems: Key lessons learned and future opportunities. *American Psychologist*, 73(4), 390.
- Shuffler, M. L., & Cronin, M. A. (2019). The challenges of working with "real" teams: Challenges, needs, and opportunities. *Organizational Psychology Review*, 9(4), 211–218.

- State University System of Florida Board of Governors. (2023, October). Activity with foreign countries of concern guidance document for State University System Institutions. https://www.flbog.edu/wp-content/uploads/2023/10/Foreign-Influence-Guidance-Document\_101923.pdf
- Tebes, J. K. (2018). Team science, justice, and the co-production of knowledge. *American Journal of Community Psychology*, 62(1–2), 13–22. https://doi.org/10.1002/ajcp.12252
- Tebes, J. K., & Thai, N. D. (2018). Interdisciplinary team science and the public: Steps toward a participatory team science. *American Psychologist*, 73(4), 549–562. https://doi.org/10.1037/amp0000281
- The White House. (2021, January 14). Presidential memorandum on United States government-supported research and development national security policy. https://trumpwhitehouse.archives.gov/presidential-actions/presidential-memorandum-united-states-government-supported-research-development-national-security-policy/
- \_\_\_\_. (2025, January 20). Return to in-person work. https://www.whitehouse.gov/presidential-actions/2025/01/return-to-in-person-work/
- United Nations Office for Outer Space Affairs. (n.d.). Treaty on principles governing the activities of states in the exploration and use of outer space, including the moon and other celestial bodies. https://www.unoosa.org/oosa/en/ourwork/spacelaw/treaties/introouterspacetreaty.html
- Vogel, A. L., Knebel, A. R., Faupel-Badger, J. M., Portilla, L. M., & Simeonov, A. (2021). A systems approach to enable effective team science from the internal research program of the National Center for Advancing Translational Sciences. *Journal of Clinical and Translational Science*, 5(1), e163. https://doi.org/10.1017/cts.2021.811
- Voice of America News. (2024, January 3). 'Countries of concern': Florida law raises questions about student recruitment from China. https://www.voanews.com/a/7405245.html
- von Wehrden, H., Guimarães, M. H., Bina, O., Varanda, M., Lang, D. J., John, B., Gralla, F., Alexander, D., Raines, D., White, A., & Lawrence, R. J. (2019). Interdisciplinary and transdisciplinary research: Finding the common ground of multi-faceted concepts. *Sustainability Science*, 14, 875–888. https://doi.org/10.1007/s11625-018-0594-x
- Wagner, C. S., & Cai, X. (2022). Changes in co-publication patterns among China, the European Union (28) and the United States of America, 2016–2021. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.4035897
- Wallen, K. E., Filbee-Dexter, K., Pittman, J. B., Posner, S. M., Alexander, S. M., Romulo, C. L., Bennett, D. E., Clark, E. C., Cousins, S. J. M., Dubik, B. A., Garcia, M., Haig, H. A., Koebele, E. A., Qiu, J., Richards, R. C., Symons, C. C., & Zipper, S. C. (2019). Integrating team science into interdisciplinary graduate education: An exploration of the SESYNC Graduate Pursuit. *Journal of Environmental Studies and Sciences*, 9, 218–233. https://doi.org/10.1007/s13412-019-00543-2
- Woodruff, P., Wallis, C. J., Albers, P., & Klaassen, Z. (2021). Virtual conferences and the COVID-19 pandemic: are we missing out with an online only platform? *European Urology*, 80(2), 127. https://doi.org/10.1016/j.eururo.2021.03.019
- Zaccaro, S. J., Marks, M. A., & DeChurch, L. A. (2011). Multiteam systems: An introduction. In S. J. Zaccaro, M. A. Marks, & L. DeChurch (Eds.), *Multiteam systems* (pp. 3–32). Routledge. https://doi.org/10.4324/9780203814772

3

# **Best Practices**

This chapter addresses the questions outlined in the statement of task (see Box 1-1 in Chapter 1) concerning best practices for supporting team science, including those related to training and education, virtual collaboration, and incorporating nonscientist team members. Drawing from the literature, committee expertise, and expert presentations, the chapter summarizes the current understanding of which best practices to implement within a science team, how to implement them, and when they could be applied to enhance team effectiveness. The committee defines *best practice* as an activity or strategy that can enable a science team to collaborate effectively to achieve its goals; it is synonymous with the term *team development intervention* in the broader literature on team effectiveness (Shuffler et al., 2018).

The chapter begins with an overview of best practices related to education, training, and professional development opportunities to support team science. These opportunities form the foundation of a healthy team science ecosystem and prepare individuals to collaborate effectively as members of science teams. The chapter also articulates a series of best practices that can be implemented throughout the life cycle of a science team to potentially improve team effectiveness. Table 3-2 at the end of the chapter outlines suggested best practices discussed throughout.

#### EDUCATION, TRAINING, AND PROFESSIONAL DEVELOPMENT

As noted in Chapter 1, the success of a science team depends on the technical expertise of its members, known as taskwork competencies, and on their ability to collaborate effectively, reflected in teamwork competencies.

Thus, team science competencies, or the knowledge, skills, and abilities required for team science success, extend beyond scientific expertise to include competence in communication, coordination, conflict resolution, and the ability to work across disciplinary and organizational boundaries. By developing both taskwork and teamwork competencies, members of science teams can better navigate the complex dynamics of scientific collaboration.

Teamwork and taskwork competencies can be either context-specific or transferable (Cannon-Bowers et al., 1995; Cannon-Bowers & Salas, 1998). Context-driven competencies are necessary for a specific team working on a specific scientific task. These competencies are particularly relevant to teams with consistent membership that perform similar tasks over time. Team-contingent competencies are those specific to a particular team, but the competencies are more generic in that they can be applied across different scientific tasks. These competencies are particularly relevant to science teams that work together regularly but handle a variety of projects. Taskcontingent competencies are specific to a type of task but can be applied across different teams, and they are valuable when tackling a scientific problem that may involve different teams. Finally, transportable competencies are both team and task generic in that they can be applied across a range of tasks and teams. These general competencies are crucial for effective collaboration in any scientific context. Therefore, interdisciplinary graduate programs and professional development for team science often emphasize them (Fiore et al., 2019). While many existing programs and professional development opportunities focus on those earlier in their careers, including students, it is important to note that seasoned faculty could also benefit from participating in training opportunities. Faculty have cited time demands, a lack of incentive, and a lack of career-stage-appropriate training as barriers to their participation in professional development activities (Brownell & Tanner, 2017; Vela et al., 2023).

Developing team science competencies, especially those transportable across tasks and teams, is essential for supporting the team science ecosystem. Building these equips individuals to contribute meaningfully, collaborate productively with others, and ultimately function as members of effective science teams, thereby driving the success of collaborative scientific endeavors.

Team science competencies are developed in various ways. Although many team science competencies are acquired through "on-the-job" learning, more structured efforts can cultivate these skills (see Fiore et al., 2019). In addition to the availability of online resources, such as those available through the International Network for the Science of Team Science<sup>1</sup>

 $<sup>^{\</sup>rm 1}$  To learn more about the International Network for the Science of Team Science and view available resources, please visit https://www.inscits.org

BEST PRACTICES 43

and the University of California, Irvine's Team Scholarship Acceleration Laboratory,<sup>2</sup> these efforts include workshops ranging from a few hours to several days or longer, courses focused on interdisciplinary topics within academic curricula, or short courses offered to faculty. The *Enhancing the Effectiveness of Team Science* (National Research Council, 2015) report spurred the development of the training program TeamMAPPS, which was based on identified knowledge, skills, and activities integral to successful team science work (Bisbey et al., 2021b). In addition, some programs incorporate team science competencies into their mentoring initiatives. The sections that follow provide examples of programs designed to develop team science competencies.

# Workshops

Workshops are organized, structured interventions designed to bring together individuals to engage in collaborative learning, skill development, and reflection over a few hours to a few days. Typically, workshops focus on specific learning objectives, such as fostering interdisciplinary understanding, enhancing competencies related to collaboration, and bringing together unique ideas to generate innovative and transdisciplinary frameworks. These objectives are accomplished through various interactive activities, such as discussion, practical exercises, and assessment. These activities can provide participants with a safe environment to explore different perspectives, practice their team science skills, and develop a greater epistemological awareness (Audi, 2010). Workshops can vary in both length and format, from multiday immersive programs to half-day informational sessions. This section continues by describing the results of several studies on the outcomes of science team training workshops.

One study, for example, explored how workshops can enable science and engineering doctoral students to develop a better understanding of the various perspectives held by experts from different disciplines (Gosselin et al., 2020). This study of dispositional and epistemological characteristics found that a 9-day workshop intervention exposed students to different ways of thinking. Gosselin et al. (2020) considered factors such as communicating different perspectives and valuing insights from others. Using retrospective pre- and post-module evaluation and data from questionnaires and group reflections, they assessed participants' behavioral and dispositional differences, finding that the workshop helped students increase awareness about their own values associated with science knowledge and how these values differ from those of others on their team. The authors suggest that

<sup>&</sup>lt;sup>2</sup> To learn more about the Team Scholarship Acceleration Laboratory and access resources, please visit https://tsal.uci.edu

such workshops help students begin "their journey to understanding the importance of learning to navigate and negotiate dispositional distances and other forms of compositional diversity as part of collaborative processes" (Gosselin et al., 2020, p. 321). Although the assessments and interventions were proprietary in nature, these workshop formats could be followed with more open-source measures that examine similar competencies.

In support of ideation and knowledge integration, innovation labs are a type of workshop designed to support learning about others and developing new ideas for transdisciplinary research (Hawk et al., 2024). In one study, facilitators supported early career scholars in a creative problem-solving process and assessed participants on competencies such as collaboration readiness and interest in starting new research projects. This study also examined team formations and writing research grants, new publications, or both based on activities from the workshop. Compared with scientists who engaged in normal institute activities, Hawk et al. (2024) found little to no differences in attitudes toward collaboration or research productivity. However, they found increases in the intent to submit grant proposals among innovation lab participants. What is most useful about this approach is the attention to outcomes over a longer duration, with the researchers examining several factors over time, including collaboration readiness and collaboration network size at 12 months and 21 months postintervention, respectively (Hawk et al., 2024).

Another study examined learning and attitude change for faculty members who received seed funding awarded to support research idea development in interdisciplinary teams (Morgan et al., 2021). Attendees at this workshop were a mix of junior- and senior-level faculty in several disciplines, including social and life sciences, engineering, and humanities. The workshop, comprising two half-days, had several sessions targeting a set of team competencies meant to develop collaborative capabilities of team members. These ranged from team science knowledge, such as best practices, collaboration skills, including communication and interpersonal collaboration, and team regulation type concepts, such as goal setting and meeting coordination. Activities such as group discussion, quizzes, and templates for teamwork provided practice in these different competencies. To measure outcomes, the investigators used pre- and post-test change measures, along with subjective reactions to the variety of content in the workshop. In general, the participants' subjective reactions to the workshop sessions were that they were useful. Among participants who attended at least one of the workshop sessions, there was evidence of some attitude changes, including rating themselves as more ready to collaborate, as well as increases in behavioral trust. However, there were no changes in teamfocused concepts around clarity of roles, goals, or processes (Morgan et al., 2021).

BEST PRACTICES 45

#### Courses

Courses can be leveraged to develop team science competencies over a longer time period. By balancing lecture- and activity-based learning, courses provide opportunities to understand concepts and ideas in theory and then give participants opportunities to apply these learned competencies in practical, real-world contexts. Assessing learned competencies in this format may include reflective writing, oral presentations, team projects, declarative knowledge exams, and practical application of skills.

One study of a course designed specifically to improve team science competencies such as mentoring and debriefing found some increases in leadership self-efficacy and improvements in various facets of collaboration (Tumilty et al., 2022; see also Appendix B). This study measured leadership self-efficacy using the Kane-Baltes Leadership self-efficacy test, which assesses a person's beliefs that they can perform successfully as a leader. The course, which used authentic learning activities, focused on interprofessional training for transdisciplinary teams, examining how to acquire specific (e.g., leadership training) and general collaborative competencies (e.g., grant writing, interactions with experts such as biostatisticians). Finally, the participants received formative feedback regarding their work on proposal writing and oral presentations, for example. This course emphasized team science competencies, including collaborative behaviors such as monitoring and reflection, perspective seeking, and inquiring about and probing research ideas. Also included were tools such as interactive team contracts, which showed benefits such as addressing issues in the team. Participants showed beneficial effects on communication, integrating certain practices into their writing, such as open communication styles, empathic practices, and collaborative workflows (Tumilty et al., 2022). Overall, the methods used here can provide an effective template for both developing team-science relevant courses and evaluating their effectiveness.

One group of investigators developed coursework for promoting interdisciplinary education, with an equal focus on faculty instructors and students (Corbacho et al., 2021). They examined teaching practices that could trigger academic motivation and how different teaching teams used a common strategy to create a collaborative interdisciplinary environment, leading to consistent student experiences. The investigators also examined what distinguishes interdisciplinary courses from traditional ones. Using tools such as the Student Course Experience Questionnaire (Ginns et al., 2007) and the eMpowerment, Usefulness, Success, Interest, and Caring Model of Academic Motivation (Jones, 2009), Corbacho et al. (2021) assessed how well students recognized key components of interdisciplinary teamwork. In the context of teaching, this study identified some challenges, including teachers feeling out of their comfort zones when applying teambuilding practices and creating open-ended problems.

In 1998, the National Science Foundation (NSF) created the Integrative Graduate Education and Research Traineeship (IGERT) program to both develop interdisciplinary competencies in students and support changes in scholarly culture at participating universities. Although NSF ended IGERT in 2013—NSF recently created the NSF Research Traineeship (NRT) program to similarly support interdisciplinary, evidence-based traineeships (National Science Foundation, 2024)—researchers have examined its effects and highlighted critical findings. For example, one report discussed the overall purpose of the program and summarized the formal and informal ways IGERT awardees developed interdisciplinary competence (Van Hartesveldt, 2016). Awardees of IGERT funding created new courses in a collaborative way to ensure that faculty with varied perspectives contributed to an interdisciplinary curriculum. From these courses, graduate students learned about other disciplines and each other as they engaged in research team projects.

One of the primary goals of these courses was to increase multidirectional communication skills, including communicating ideas effectively, especially with those from other disciplines. IGERT projects included informal mechanisms such as boot camps or summer sessions that provided more experiential forms of learning, as well as an opportunity for focused learning and interaction. More broadly, IGERT included crossmentoring, where faculty guided students from different disciplines. In general, IGERT outcomes promoted students' attitudes toward learning and cross-disciplinary research (Van Hartesveldt, 2016).

The same report described findings suggesting that IGERT students tended to produce dissertations that incorporated more disciplines (Van Hartesveldt, 2016). IGERT students were also more active in conducting interdisciplinary research and creating interdisciplinary programs. Additionally, the report highlighted longer-term analyses in which graduates of IGERT programs said they were still drawing on their experience and competencies gained while in their graduate program (Van Hartesveldt, 2016).

## **Mentoring Programs**

Mentoring programs are a great resource for developing team science competencies in a less formal environment over prolonged periods of time. *Mentoring programs* are structural initiatives that assign experts—mentors—to support their protégés' development in specific activities that align with the mentoring program's purpose.<sup>3</sup> Mentoring programs use a

<sup>&</sup>lt;sup>3</sup> While informal mentor–protégés relations that develop more organically exist, they are not the focus of this section.

BEST PRACTICES 47

variety of frameworks (e.g., Montgomery & Page, 2018), such as a traditional hierarchical structure consisting of one mentor and one protégé, as well as nondyadic structures, including one mentor and one team, or team members mentoring each other. In the context of team science, these programs aim to enhance both teamwork (e.g., improving collaborative competencies and cross-disciplinary understanding) and taskwork (e.g., technical and research competencies) through such activities as direct mentoring interactions, experiential training, and collaborative exercises (National Academies of Sciences, Engineering, and Medicine, 2019; Rodríguez et al., 2021).

The federal government has been active in supporting mentoring programs for graduate students and early career professionals. The National Cancer Institute programs sometimes focus on developing both teamwork and taskwork by promoting an appreciation of different perspectives on scientific issues. These provide multiple mentors spanning a range of competencies that need to be developed based upon mentor needs. For example, the Transdisciplinary Research on Energetics and Cancer Centers initiative (Gehlert et al., 2015; Vogel et al., 2012) has postdoctoral scholars identify and select mentors that span disciplines with the goal of enhancing inter- and intrapersonal competencies. The Academic Learning Health System Scholars Program also assigns multiple mentors, including a primary mentor, co-mentor, Translational Research Training Program core faculty mentor, peer mentor, and health system mentor (Woodside et al., 2021). Along with this set of mentors, the postdoctoral scholars also participate in collaborative forms of training, including transdisciplinary research courses, journal clubs, and workshops to develop skills in collaborative writing. Even if not specifically identified as such, a primary objective of these programs is to improve teamwork and taskwork, as they are designed to expose trainees to multiple disciplinary perspectives, to increase research competencies, and to enhance attitudes about collaboration. The overall goal is greater scholarly productivity based on becoming a more teamoriented scientist.

The National Institutes of Health (NIH) Building Interdisciplinary Research Careers in Women's Health program also includes a mentoring component.<sup>4</sup> This program is designed for junior faculty interested in advancing cross-disciplinary research in women's health; it provides multiple perspectives on a range of scientific and career issues. Nagel et al. (2013) evaluated this program using success indicators such as improved idea generation based on more grant submissions in women's health. This program

<sup>&</sup>lt;sup>4</sup> More information about the BIRCWH program is available at https://www.mayo.edu/research/centers-programs/womens-health-research-center/education/building-interdisciplinary-research-careers-womens-health-program

identified a set of recommendations or best practices for ensuring success in team-based mentoring, including development of a written contract or agreement between participants to manage mentoring expectations (Guise et al., 2012) and clearly articulating roles for the mentoring team, such that some focus on career issues while others focus on scientific content (see also Guise et al., 2017).

More recently, the NIH Clinical and Translational Science Awards (CTSA) Program has developed an approach that focuses on teams of trainees. CTSA combines didactic and experiential training in team science that includes cross-disciplinary mentoring (McCormack & Strekalova, 2021). Team composition is also a focus; mentors are assigned two students pursuing PhDs or dual degrees (e.g., MD-PhD). In addition, these mentee teams need to be pursuing their doctoral degree in different disciplines and in different colleges. Thus far, there is evidence of an increase in collaborative activities of both the CTSA team trainees and mentors (McCormack & Strekalova, 2021). Some attitudinal changes are also evident: two-thirds of mentees noted that they plan to continue collaborations after training and to continue using the support tools, such as collaboration plans and author agreements, to which they were introduced in the CTSA program (McCormack & Strekalova, 2021).

In a recent review of team mentoring approaches, whereby mentees are assigned multiple mentors with the goal of providing a variety of disciplinary and professional perspectives, one report described how team mentoring can improve transdisciplinary science (Shah & Fiore, 2022). The authors noted that team mentors support professional development by providing career guidance and direction and improving exposure to critical people; these help mentees gain insights about political culture, organizational culture, or both. Furthermore, the mentoring team acts as advocates on behalf of the mentees and can help identify opportunities for research and critical resources needed for early career success (Shah & Fiore, 2022).

The same review provided a set of guidelines for team mentoring to help identify targets for collaborative competencies, such as active listening (Shah & Fiore, 2022). For example, mentors can demonstrate active listening with their team by asking follow-up questions when discussing complicated issues. As a complement, mentees can practice active listening by discussing research with those from other disciplines. Shah & Fiore (2022) pointed out that mentors can enhance competencies in assertive communication by creating a safe environment for arguing around research topics, where mentees can readily address their opinions and ideas in task-focused ways, ensuring that all ideas are being considered.

Shah & Fiore (2022) discussed guidelines for team mentoring in cultivating interpersonal competencies, including understanding coordination. For example, mentors could create artifacts delineating roles and goals on

BEST PRACTICES 49

the team and expectations for different mentors. In addition, mentees can provide feedback to mentors when they are unclear about roles or goals while also learning how to safely ask for assistance as needed. This coordination and delineation of goals becomes particularly important for multiteam systems, where teams have to balance both local and shared goals (Shuffler et al., 2015).

Another important interpersonal competency is cultivating an appreciation of varied perspectives (Shah & Fiore, 2022). For example, mentors can model respect for theories, methods, or both that come from different fields during team meetings. Mentees can learn how to better attend to, and maintain awareness of, any experience of positive or negative commentary about other disciplines, such as disdain for a particular field, and be comfortable discussing why such attitudes are problematic.

#### BEST PRACTICES FOR SUPPORTING SCIENCE TEAMS

The past century of research on team effectiveness across industries has demonstrated that key *teamwork processes*—such as coordination, information-sharing, and conflict management—and *emergent psychological states*—such as cohesion, shared mental models, and transactive memory systems—are linked closely to achieving team objectives (Bell et al., 2018; Kozlowski & Chao, 2018; Mathieu et al., 2018). Moreover, research on team effectiveness has identified numerous best practices and team development interventions that teams and team leaders can deploy to support critical teamwork processes, emergent states, and team goal accomplishment. In this section, the committee articulates how to apply these best practices in the context of team science.

The committee framed its discussion of team science best practices according to the four stages of transdisciplinary research (Hall et al., 2012). The first stage, *development*, involves establishing a shared understanding of a scientific problem space and group mission. During the second stage, *conceptualization*, the team develops research questions, conceptual frameworks, and research designs to integrate and expand approaches from multiple disciplines. In the third stage, *implementation*, the team focuses on initiating, executing, and refining the planned research. Finally, *translation* involves applying research findings to develop innovative solutions for real-world problems.

Although these four phases are conceptualized as sequential, they are highly interconnected and are nonlinear for many teams (Hall et al., 2012). For instance, a team may cycle between the conceptualization and implementation phases many times before progressing to translation. This aligns with the broader teams literature, which emphasizes that teams cycle repeatedly through *transition phases*, marked by activities such as planning

and mission analysis, and *action phases* involving tasks such as coordination and back-up behavior (Marks et al., 2001). In addition, not all teams will engage in all four phases, particularly if transdisciplinarity is not a primary research goal. However, the overall organization of this section broadly mirrors the evolving and dynamic nature of science teams and how a team's challenges, opportunities, and needs will shift over time, depending on the goals of the science. Basic science teams, for example, still engage in these early phases.

The remainder of this chapter outlines best practices for supporting a science team corresponding to the four phases of transdisciplinary research. Some best practices, such as those pertaining to leadership or virtual collaboration, can transcend all four team stages and are thus afforded their own sections. Understanding the phase within which a science team is situated and leveraging corresponding best practices during that phase improves the likelihood that the team will produce its desired outcomes.

# Phase I: Development

In the development phase, a group of potential collaborators comes together to define the scientific or societal problem of interest, including its complexities and boundaries (Hall et al., 2012). Collaborators might be selected based on their relevant disciplines and perspectives to address the problem comprehensively. One framework, for example, identified several teamwork processes that are likely to be essential to team success during the development phase. The authors suggested that, during development, science teams can work to generate a shared mission and goal and ensure all members are aligned and motivated to pursue a common purpose. Another important process during the development phase is fostering critical awareness, where team members gain a broad understanding of the problem space and the different perspectives and expertise each team member brings (Hall et al., 2012). For example, as discussed in Chapter 1, community members and/or nonscientist team members may be instrumental in defining the problem space.

The authors also argue that the development phase can involve externalizing group cognition, which involves making explicit the team's collective knowledge and thought processes (see Fiore & Schooler, 2004). Externalizing cognition plays a functional role in science teams in that it supports discussion and elaboration and helps teams identify points of agreement and confusion (Fiore & Wiltshire, 2016). *Psychological safety*, defined as the belief that a group is safe to engage in interpersonal risktaking, is essential during the development phase, given that psychological safety encourages open communication, risk-taking, and the sharing of ideas that are critical for generating new knowledge as a science team

BEST PRACTICES 51

(Edmondson & Lei, 2014). It is important because it helps teammates feel safe to ask questions, challenge ideas, and share perspectives without facing consequence or judgments. Together, these processes lay the foundation for effective collaboration and set the stage for successful research or problemsolving efforts. The following best practices during the development phase can help achieve these developmental processes and emergent states.

## Best Practice: Team Assembly

Being intentional, strategic, and discerning when composing a team is a critical step toward effective team science, with the team's purpose, needs, and tasks front and center. Team assembly includes (a) team task analysis, or thinking through the demands of the team; (b) team composition, or understanding the individual characteristics needed within the team; and (c) team member recruitment, or identifying and recruiting potential teammates.

Conduct a Team Task Analysis. A team task analysis is defined as "the process by which the major work behaviors and associated [knowledge, skills, and abilities] that are required for successful job or task performance are identified" (Arthur et al., 2005, p. 654). This analysis can be undertaken during and immediately after team assembly. Team task analysis involves articulating the key tasks the team will accomplish and determining how many individuals will be needed to perform those tasks; the complexity and interdependence of tasks; their frequency; and the specific knowledge, skills, and abilities required to complete them (Shuffler et al., 2018). A team task analysis can reveal which academic disciplines or areas of expertise are essential to achieve project goals. It can also help teams take a broader view of how tasks, especially those that are cross-disciplinary, may or may not align, potentially pointing to the need to prioritize generalist team members who can facilitate communication and collaboration across boundaries (Bammer, 2013). Multiteam systems face even greater challenges in aligning team goals, managing cultural differences, and coordinating tasks; identifying individuals who can facilitate this communication and collaboration can help with information exchange, trust-building, alignment of objectives, and maintaining system-wide cohesion (Carter et al., 2019; Kotarba et al., 2023; Zaccaro et al., 2020).

Following a team task analysis, the context, problem of interest, goals, timeline, funding, and other factors determine the appropriate size of a team (Hall et al., 2018). It is important that team size be commensurate with both the quantity and quality of tasks identified as necessary to solve a problem or test a hypothesis. Bibliometric analyses have found that larger teams tend to focus on more rapid development, adding incremental advances,

whereas smaller teams produce more innovative outcomes that take time to make an impact (Wu et al., 2019). Thus, team sizes will vary depending on the scientific objectives of the research. It is worth noting that larger teams have higher coordination costs due to the need to align more individuals, institutions, time zones, and so on (Berntzen et al., 2021; Faraj & Sproull, 2000; Forscher et al., 2023; Pendharkar & Rodger, 2009). Stronger efforts may also need to be made to ensure larger teams with weaker ties do not splinter into siloed objectives, tasks, and outcomes (Jeske & Olson, 2024).

Decisions around team size could also benefit from considering team freshness, or how incorporating new members and perspectives might impact an existing team's performance. In an analysis based on article citations as a proxy for impact, for example, smaller teams were found to be more negatively affected when new members were added (Liu et al., 2022), potentially resulting from a more concentrated effect of being unfamiliar with one another's work-related experiences, skills, styles, and values. The effect of a new member on team performance has been shown to differ depending on how similar the new member is to existing members in relational and taskrelated characteristics (Liu et al., 2023b). New members that differ from existing members in characteristics related to accomplishing work tasks are associated in positive reactions from existing team members, while those who differ in aspects such as trait likability are associated with negative reactions (Liu et al., 2023b). These findings are at odds with traditional research on teams that show how turnover can improve innovation (Levine et al., 2003) but may depend on how experts are coordinated (Newton et al., 2019). Such findings illustrate why more research is needed on science teams to identify what is extendable or not from the science of teams.

Consider Team Composition. The team task analysis and identification of knowledge, skills, and abilities feed directly into team composition, or selecting individuals to form a team. Scholars have often emphasized that team selection decisions ought to consider both taskwork and teamwork competencies. For example, there is increasing support for selecting team members who demonstrate collaborative attitudes and behaviors, rather than those who exhibit more individualistic styles of thinking and working (Kilcullen et al., 2023). This concept, known as team orientation, encompasses both an individual's preference for working in a team versus working alone, for example, and their belief in teamwork as an effective means to get work done and accomplish goals. Individuals with a high level of team orientation tend to contribute positively to team functioning by promoting higher levels of cooperation and creating an environment where the team collectively believes in its ability to succeed (Kilcullen et al., 2023). Trust and cohesion, emergent psychological states that have been shown to be critical for team effectiveness, are also enhanced when team members

are oriented toward teamwork (Kilcullen et al., 2023). The literature on team science recognizes the importance of team orientation, with scholars emphasizing that team-oriented individuals are more likely to engage in the collaborative processes necessary for scientific innovation (Fiore et al., 2019; National Research Council, 2015). In addition, team orientation is a key component of collaboration readiness (Hall et al., 2008).

Science teams might also prioritize including potential team members who exhibit a *transdisciplinary orientation*, a characteristic that develops over the course of an individual's career, reflecting their values, attitudes, and beliefs as well as the behaviors necessary for effective cross-disciplinary collaboration (Misra et al., 2015). Going beyond merely appreciating or valuing interdisciplinary work, transdisciplinary orientation encompasses the cultivation of the specific competencies required for transdisciplinary work and a demonstrated history of engaging in such collaborations. However, as with valuing interdisciplinary collaboration, transdisciplinary orientation is underresearched.

Many other individual characteristics—such as agreeableness, conscientiousness, openness to experience, collectivism, and preference for teamwork—have been linked with team effectiveness in other contexts (e.g., Barrick et al., 1998; Bell, 2007). Moreover, science teams are likely to reap innovation and creativity benefits when team members with different identities and lived experiences are supported appropriately (Smith-Doerr et al., 2017). For example, research has shown that variety across demographic factors improves team performance and scientific impact (Hall et al., 2018). These teams can also be more effective at problem-solving, decision-making, and innovating, including when adopting participatory approaches that invite nonscientist members to inform research need, relevancy, feasibility, and outcomes (Tebes & Thai, 2018; Wallerstein et al., 2019). For example, gender has been the focus of statistical analysis using quantitative methods for studying patterns in scientific outputs such as publications and grants since the 2015 report (Aksnes et al., 2019; Fox et al., 2018; Huang et al., 2020; Jadidi et al., 2018; Kwiek & Roszka, 2021, 2022; Liu et al., 2023a; Nielsen et al., 2018; Smith-Doerr et al., 2017). Yang et al. (2022) found that science teams that have both men and women produce more novel and highly cited papers than single-gender teams. This advantage increases with greater gender balance and on average is consistent across team sizes and subfields and does not depend on the team leader's gender; it applies to all science fields over the past 20 years (Yang et al., 2022).

Research in team science has also focused on examining aspects of heterogeneity beyond gender and discipline to include nationality and how that influences the collaboration process and research output of teams. Specht & Crowston (2022) examined scientific working groups that varied in terms of gender, discipline, and nationality and how team composition

influenced collaboration and outcomes. They found that key aspects of team composition had positive effects on the interdisciplinarity of the collaboration process, defined as the range of the disciplines of the journals the teams published in and those they cited in their publications. Specifically, Specht & Crowston (2022) found that teams with more women had greater cited discipline variety and teams with more discipline variety had greater cited discipline and publication records. In a related study, Smith-Doerr et al. (2017) found a positive correlation between the participation of more women in disciplines traditionally dominated by men and expanded scientific research agendas. However, there was a negative relationship between publication variety and the presence of different nationalities on the team (Specht & Crowston, 2022). Outcomes were measured by the number of publications and the impact of the group's work by computing the median number of citations. This study also assessed team satisfaction and perceived team effectiveness and found that individual level of satisfaction was positively related to the number of publications the teams produced and with the proportion of women on the team. Moreover, the investigators found a negative relationship between country and work practices—as measured by the level of variety of articles sourced, for example—on personal satisfaction and perceived effectiveness of the team (Specht & Crowston, 2022).

However, increasing differences across demographic factors on teams has been described as a "double-edged sword" because of its potential for both positive and negative effects (Milliken & Martins, 1996, p. 403; Smith-Doer et al., 2017; Verwijs & Russo, 2024). Empirical evidence indicates the effect of heterogeneity on team outcomes is complex and may vary by how it is measured (Horwitz & Horwitz, 2007). Understanding these nuances is essential for leveraging individual differences to improve team performance. Several theories can help explain the potential for positive and negative effects associated with team variability. Information processing theory suggests that heterogeneous teams benefit from a wider range of information, knowledge, and perspectives, which can enhance decision-making and problem-solving (van Knippenberg & Schippers, 2007; Williams & O'Reilly, 1998). One area of organizational science research introduced the term faultlines, which are ways in which teams may split into subgroups based on one or more attributes (Lau & Murnighan, 1998). Research on faultlines has shown they are related positively to conflict and negatively to information elaboration, team performance, and team satisfaction (Thatcher et al., 2024). Social categorization theory posits individuals can view dissimilar team members as "out-group" members, potentially causing communication difficulties, conflict, faultlines, and reduced cohesion (Verwijs & Russo, 2024; Williams & O'Reilly, 1998). Several meta-analyses have grouped demographic variables into those that are highly job-related

or less job-related, as well as task-oriented (related to completing tasks) or relation-oriented (related to building relationships; e.g., Joshi & Roh, 2009; Webber & Donahue, 2001). One meta-analysis (Bell et al., 2011) examined variability and team performance relationship by focusing on specific variables rather than grouping variables, as was done in past meta-analyses. Bell et al. (2011) found that having team members who represented the different functional areas in an organization, such as marketing and finance, was consistently related positively to team performance, while demographic variables such as education-level, team tenure, and organizational tenure consistently demonstrated no relationship between variability and team performance.

To determine the extent to which potential team members possess the requisite knowledge, skills, and abilities for a specific science team, scholars have suggested conducting detailed interviews that collect information on disciplinary expertise, as well as values and competencies such as interactional expertise, behaviors, and team orientation (Knott et al., 2022). The team science literature offers guiding questions to gauge individual and team readiness—such as how much the team trusts one another to make individual compromises in favor of an integrated team approach, how available an individual is to collaborate, and how feasible it will be to work across the constraints of different participants and institutions (e.g., O'Rourke et al., 2019, p. 32).

Scientifically derived and empirically tested team readiness instruments represent another means by which a team can make decisions about its members. Both the *Motivation Assessment for Team Readiness, Integration, and Collaboration* instrument (Mallinson et al., 2016) and the *Collaboration Success Wizard* (Bietz et al., 2012) use survey responses to assess how individual motivations and orientations may coalesce into team-level opportunities and vulnerabilities. Another example, the Interdisciplinary Perspectives Index, measures the degree to which individuals value interdisciplinary work and captures their attitudes and emotional responses toward the significance of this work (Misra et al., 2009).

Recruit Team Members. The process of assembling a science team is often complex and informal. First, there is the challenge of finding the right individuals. Although it might be clear what expertise or knowledge is needed for the team, locating individuals who possess those skills and are willing to participate in the team can be difficult. This process is often driven by informal networking, where researchers rely on personal contacts, recommendations from other scientists, recommendations from nonscientist partners, or discussions at academic conferences to identify potential collaborators (Lungeanu et al., 2014, 2018; Zajdela et al., 2022). Such approaches, while often effective, are also unpredictable and may leave team leaders with

limited options if they lack the necessary connections within their fields or across disciplines.

Indeed, recruiting science team members is unlike team assembly in many industries, where team formation can follow a traditional, structured process—for example, a leader identifies a clear goal, posts job ads, interviews candidates, and selects the best fits. Instead, researchers may initiate collaborations based on mutual interests, long-standing professional relationships, or chance encounters at conferences or workshops. Many science teams are formed first without a fully developed goal or project in mind, and the task itself may only take shape once the team members are assembled (Wang & Hicks, 2015). This informal process is further complicated by the fact that many scientific collaborations start with a vague idea or broad topic of interest, and only after initial discussions do the collaborators flesh out specific goals and methodologies. As a result, team members are often chosen not because they fit neatly into a predefined role, but because they bring a unique perspective or area of expertise that helps shape the project as it evolves. In these cases, the goals and tasks are developed alongside the team composition, rather than preceding it, and team composition can shift or expand as the project becomes clearer.

Moreover, even when scientists have a well-defined task or research goal in mind, assembling a team with the right mix of competencies and expertise is not easy. When team members lack extensive networks or are looking to expand into unfamiliar disciplines, technological resources, such as third-party scientific expertise databases, can facilitate collaborator searches (National Research Council, 2015). Artificial intelligence (AI) models can also be trained to analyze research networks and identify relevant experts (Sourati & Evans, 2023; Xu, 2025). It is important to note, however, that several AI systems have perpetuated biases in hiring decisions, including in aspects such as race and sex (Dastin, 2018; Lee et al., 2019). Nonetheless, technological resources are less helpful for identifying individuals who possess the necessary combinations of teamwork and taskwork skills.

An additional challenge is convincing identified individuals to join a team. This is no small task in the world of science, where researchers are often stretched thin across multiple projects, grants, teaching responsibilities, and institutional roles. Even if a scientist is interested in the proposed project, they might not have the time to take on additional work. Moreover, team members need to be personally motivated to participate, whether because they find the research question compelling, see potential for career advancement, or value the opportunity to collaborate with particular individuals (Adler & Chen, 2011; Bennett & Gadlin, 2012). Without sufficient intrinsic or extrinsic motivations, it can be difficult to persuade highly capable scientists and other potential team members to commit their time and energy to a new collaboration.

Convincing potential collaborators to join a science team often involves articulating a compelling vision for the project. Team leaders or initiators may need to communicate the scientific value of the project as well as how participation will benefit each team member. Whether it is through the promise of publications, opportunities for professional networking, or the ability to contribute to cutting-edge research, individuals need to feel that their involvement is worthwhile (Bennett & Gadlin, 2012). In addition, clear expectations around roles, time commitments, and contributions are crucial to ensuring that potential team members feel confident about joining. Without this, even highly motivated individuals may be hesitant to sign on. Thus, the process of science team assembly is not only about finding and recruiting the right people, but also about creating the right environment, one where potential team members see value in the project, feel motivated to contribute, and have clarity about what their roles will entail (Bennett & Gadlin, 2012).

#### Best Practice: Team Onboarding and Building

After team assembly, onboarding team members in ways that increase both work-related and personal familiarity is important for team effectiveness (Tannenbaum et al., 2023). Onboarding can include sending welcome letters with essential information, such as team member biographies and key points of contact, and facilitated discussions to promote mutual understanding around a project's central components and highlight individuals' strengths (e.g., the Toolbox Dialogue Initiative; see Bennett et al., 2014; Hubbs et al., 2020). As discussed in the next phase, collaborative agreements (i.e., team charters) can establish a shared set of expectations and synchronize team members into a cohesive vision. Team membership is often not static over time, meaning onboarding activities may need to occur during any phase in which new members are added to build and maintain shared mental models of the collaborative work. For example, teams could appoint an onboarding buddy to help a new member integrate into a team (Tannenbaum et al., 2023). Onboarding strategies can focus on providing sensitivity training, co-creating a team culture that supports work-life balance, implementing universal design principles, and fostering mentorprotégé relationships within the team (Arslan et al., 2025; Behar-Horenstein & Prikhidko, 2017; Mosca & Merkle, 2024; Wellemeyer & Williams, 2019).

Onboarding strategies can incorporate team-building as a proven means of enhancing interpersonal relationships (Klein et al., 2009). Interpersonal relationships help to strengthen social ties on a science team, which can help both launch and sustain collaborations over the long term (Smith et al., 2016). By strengthening team bonds, team-building can foster team cohesion and trust and a team climate that promotes perseverance

and constructive conflict resolution (National Research Council, 2015). A recent case study highlighted the positive connection between interpersonal relationships and science team productivity (Love et al., 2021).

Research suggests that to be most effective, team-building efforts can be tailored to the team's unique challenges, context, and needs (Klein et al., 2009; Lacerenza et al., 2018; Shuffler et al., 2018). A scientifically based team-building intervention capable of helping a team overcome its obstacles will focus on one or more of the following: goal-setting, interpersonal relationships, role clarification, and problem-solving (Shuffler et al., 2018). One meta-analysis showed that all four of these focus areas engender positive effects on team outcomes, with goal-setting and role clarification exerting the strongest effect (Klein et al., 2009). All members of a science team can play an active role in designing a team-building intervention, including identifying and communicating any challenges to be addressed. If, for example, a team determines that improving interpersonal relationships is the priority, a team-building strategy could include creating time for informal, social experiences, such as sharing a meal between or after working sessions. In a presentation to the committee, Emily Ackerman (Harvard Medical School) emphasized that team-building interventions can prioritize participation, considering as needed the physical aspects of a location, the timing and cost of the activity, and the social complexity and sensory levels involved.<sup>5</sup>

# Best Practice: Developing a Shared Language

Effective communication and information-sharing are the core of successful teamwork. Communication can allow team members to exchange and integrate their varying knowledge, skills, and expertise, which can be critical in addressing the complex, interdisciplinary problems that science teams often face. Communication can facilitate team coordination, problem-solving, and goal alignment, ensuring that all members contribute meaningfully to the team's objectives. Effective communication facilitates critical team processes, such as team learning behaviors, where members actively share knowledge, ask questions, and clarify misunderstandings (Harvey et al., 2022; Wiese et al., 2022). In addition, communication plays a pivotal role in whether critical key emergent states, such as psychological safety, emerge sufficiently (Frazier et al., 2017). In this way, communication drives task execution and fosters the conditions necessary for sustained team success. Communication is essential throughout all team phases and tasks, but given the importance of generating a shared language during the development phase, relevant evidence is presented here. Without clear

<sup>&</sup>lt;sup>5</sup> Presentation to the committee April 10, 2024.

communication, the flow of critical information can be disrupted, leading to inefficiencies, misunderstandings, and ultimately ineffective performance.

To communicate effectively, teams can develop a shared team language and vocabulary, especially for inter- and transdisciplinary efforts. Crosstraining efforts within the team are helpful to this end, where members share key concepts and terminology from their respective disciplines that are relevant to the project (Falcone et al., 2019; Henson et al., 2020). The use of analogies may be particularly suitable for cross-disciplinary teams, as they can demonstrate individual team members' knowledge application in more widely relevant ways (Graff & Clark, 2018) and uncover and bridge differences in how a team understands a shared project (Paletz et al., 2013). Similarly, integrating keywords from individual team members' areas of study into a common team glossary can facilitate improved communication by ensuring team members either are using the same words in the same ways or are aware of discrepancies. In case of discrepancies, teams can benefit from collectively redefining terms or arriving at an alternative vocabulary. When possible, identifying and avoiding discipline-specific jargon can reduce frustrations and misunderstandings and contribute to team effectiveness (Henson et al., 2020).

Modes of communication are a key consideration throughout all project stages. Promoting communication practices such as the use of qualified and vetted sign language interpreters and real-time captioning and embracing different communication styles can help science teams unlock the full array of information and lived experiences to advance knowledge production and innovation. AI technology can also help facilitate communication that can increase participation, including through real-time translation tools (Johnson et al., 2017). When participating in cross-training activities, teams can strive to use plain language summaries. Visuals can also be used to support text and speech, but not as the only means of communication.

# Development Phase Summary

The development phase focuses on critical team processes such as generating shared goals, developing critical awareness, externalizing group cognition, and establishing a psychologically safe environment. Careful attention to best practices pertaining to team assembly, team onboarding and building, and developing a shared language can set team members up for success and help them understand and respect one another's different perspectives, lived experiences, and values; engage in co-learning; and find common ground. This is crucial for effective knowledge integration during the conceptualization phase.

#### Phase II: Conceptualization

In the conceptualization phase of a research project, science teams develop research questions and hypotheses, build conceptual models, and agree on research designs and plans (Hall et al., 2012). As such, this phase establishes the directions a science team's research will take. Essential team phenomena during this stage include developing shared mental models regarding taskwork and teamwork, continuing to generate a shared language, and fostering team norms and values (Hall et al., 2012). These processes ensure that team members have a common understanding of the project's team, methods, and expected outcomes, which is essential for effective collaboration.

#### Best Practice: Team Charters

A team charter is a "formal document written by team members at the outset of a team's life cycle that specifies acceptable behaviors in the team" (Courtright et al., 2017, p. 1462). Also known as a science prenup (Bennett et al., 2018), charters establish the foundation for effective team science by clarifying what teams will achieve, how they will accomplish their work, who will do what, and when the team will accomplish deliverables. Charters fit within the broader space of collaboration planning, a concept encompassing many factors that can affect taskwork and teamwork (Hall et al., 2019; Woolley et al., 2008) and are recommended as an activity to support teams (Shuffler et al., 2018). When formulated at the beginning of team interaction, ideally before the team has performed any work, charters operate as a psychological contract by setting mutual expectations and operating procedures and outlining what success looks like (Egeland & Schei, 2015). By helping the team agree on conflict resolution strategies and plan for potential challenges, such as authorship disputes, before they occur, misunderstandings regarding work effort and quality can be eliminated or reduced. Therefore, reviews in the broader team literature (Shuffler et al., 2018), the science team literature (Hall et al., 2019), and the translational science team literature (e.g., Begerowski et al., 2021) regard team charters as an evidence-based team development intervention for improving performance trajectories. Although typically implemented in smaller teams, research has also identified team charters as a best practice in multiteam systems (i.e., teams of teams; Carter et al., 2019).

Although widely implemented in practice, empirical research on team charters is limited in volume and sample, with most studies using non-science teams and students (e.g., Aaron et al., 2014; Courtright et al., 2017; Egeland et al., 2017; Johnson et al., 2022; Mathieu & Rapp, 2009; McDowell et al., 2011). However, existing research concurs that team

charters positively predict team functioning and processes and are associated with higher levels of communication, effort, mutual support, cohesion (Aaron et al., 2014; McDowell et al., 2011), member satisfaction (Aaron et al., 2014), motivation, team efficacy (Johnson et al., 2022), and information integration (Woolley et al., 2008).

Although team charters tend not to influence team performance directly (e.g., Courtright et al., 2017; Johnson et al., 2022), studies have demonstrated team performance enhancements when team charters were combined with task-appropriate expertise (Woolley et al., 2008) and taskwork performance strategies (Mathieu & Rapp, 2009). Task cohesion has been found to be a key mediator of the team charter quality-team performance relationship (Courtright et al., 2017). In addition, charters boosted team performance by enhancing teams' ability to navigate disruptions (Egeland et al., 2017). Team charters can potentially further promote psychological safety, particularly for heterogeneous science teams, through methods such as establishing clear policies around including those with individual differences, codifying flexible schedules and remote work options to accommodate different working styles and needs of all team members, designating specific channels or individuals as points of contact for team members to feel safe disclosing their concerns or needs, and establishing not only clear ground rules for operating as a team but also protocols for how to address grievances or rectify harm when rules are not followed.

Team charter creation can diverge in significant ways, with the final product being a written document that outlines the team's purpose, objectives, responsibilities, and operational procedures. In some studies, team members prepared responses independently prior to group discussion (e.g., Mathieu & Rapp, 2009). In addition, some teams used trained facilitators to guide members through topics, probe for deeper insight, and request participation from all members (e.g., Rolland et al., 2021b). Topics discussed vary substantially, given differences in the type of team and the nature of team tasks (e.g., Egeland et al., 2017; Hall et al., 2019; Mathieu & Rapp, 2009). To illustrate, one group identified ten components to address in a collaboration plan for science teams (Hall et al., 2019):

- Rationale for team approach—is a team needed—and team composition in terms of what expertise and how many members need to be included
- 2. Individual, team, and institutional collaboration readiness
- 3. Technological readiness
- 4. Team functioning
- 5. Communication and coordination
- 6. Leadership, management, and administration
- 7. Conflict prevention and management

- 62
- 8. Training
- 9. Quality improvement activities, including metrics
- 10. Budget and resource allocation. (pp. 591–601)

Interdisciplinary science teams also need guidance on authorship policies, data and information management, and conflict management (Rolland et al., 2021a). Many science teams qualify as multiteam systems (Carter et al., 2019), necessitating that multiteam charters include identifying boundary spanners between component teams, determining inter-team leadership, agreeing on how team goals will be aligned, and anticipating inter-team friction (Asencio et al., 2012).

Although conflict is often seen as something to avoid, it is an inevitable aspect of teamwork and even a necessary component of collaboration, especially in scientific contexts where various viewpoints are essential for innovation and problem-solving (Fiore et al., 2015). Conflict arises when individuals bring unique perspectives and experiences to the table, leading to differing interpretations of the problem space (Weingart et al., 2015). These differences create perception gaps, where team members have varying understandings of the task at hand. Research on conflict resolution in science teams is scarce; however, a first step toward developing a conflict management protocol is distinguishing between the three types of conflict types identified in the literature: task, relationship, and procedural/process (Jehn, 1997; O'Neill et al., 2013).

Task conflict refers to disagreements related to the content and outcomes of the team's work, focusing on ideas, viewpoints, and strategies. When facing task conflict, fostering mutual understanding across team members, such as through facilitated discussions guided by the Toolbox Dialogue Initiative, as mentioned in the development phase, could be effective (Hubbs et al., 2020). Relationship conflict, on the other hand, involves personal disagreements and interpersonal tensions that are not task-related, often stemming from personality clashes or emotional incompatibilities. Highly intentional approaches to team member selection that work to surface individual orientations, values, and personalities during team assembly could help mitigate potential relationship conflicts; however, an impartial external facilitator may best handle relationship conflicts that emerge in later stages of teaming. Finally, process conflict arises from disagreements about how the team's work, such as task delegation and timelines, will be carried out. A strong team charter can specify roles, responsibilities, and timelines to which every member of the team is asked to agree.

Whereas some studies have examined the simple existence of a charter (e.g., Egeland et al., 2017; Rolland et al., 2021a), others have emphasized the importance of charter quality (Courtright et al., 2017; Mathieu & Rapp, 2009). High charter quality is characterized as being detailed; broad

in scope, covering more aspects of team functioning; and high on professionalism, where the "content of the team charter is clearly and consistently laid out and is presented in a form that makes its content understandable" (Courtright et al., 2017, p. 1464). In addition to quality, a team's engagement with its charter throughout its life cycle is an important consideration. In a study of nearly 1,900 teams, one group found that implementing a charter at the beginning of a team's life cycle can help improve its processes (Johnson et al., 2022).

Regardless of the exact form a team charter takes or how it is developed, science teams will benefit when all members, including late-joining ones, are included in the charter development process; the discussion fosters trust, commitment, and buy-in for decisions that affect subsequent team interactions (Rolland et al., 2021a). All members of a science team can review and sign off on the written product, ensuring a mutual agreement and commitment to the charter's principles (Byrd & Luthy, 2010). Science teams can both adhere to the content of their team charter and treat it like a living document that requires periodic review and updating to evolve with the team (Mathieu & Rapp, 2009).

### Best Practice: Team Planning and Project Design

When establishing a team charter and developing a project design, science teams need to engage in extensive team planning. Team planning refers to the processes through which a team sets goals, defines roles, outlines tasks, and organizes resources collaboratively to achieve a shared objective. It might involve creating a strategy that aligns the team's efforts and ensures all members are working toward common outcomes. For instance, developing collaborative research questions and hypotheses may require revisiting the team's inventory of expertise, interests, capacities, resources, and timelines to identify points of integration among team members.

Unfortunately, many teams struggle to plan effectively. The broader literature on team functioning emphasizes that teams often exhibit critical process losses during planning that can severely limit the effectiveness of team plans (Montoya et al., 2015). For instance, team planning can be hindered by a tendency for teams to focus on and discuss shared information all team members know, rather than discussing unique information that only one or a few members possess (Stasser & Titus, 1985). This shared information bias can limit a team's ability to make optimal decisions, as critical insights may be overlooked. Prediscussion preferences can also create challenges, as team members often enter discussions with preconceived ideas, making it difficult to integrate new information. This can result in decisions being based on entrenched opinions rather than logical evaluation of all available data.

Uneven participation in discussion is another common issue, particularly in larger teams or those with individuals who have dominant personalities. Another problem is *group escalation*, which refers to the tendency of teams, especially highly cohesive ones, to commit to poor decisions resulting from pressures for conformity (Liao et al., 2004). Group escalation can lead to overconfidence and a failure to explore alternative options thoroughly. Moreover, teams often experience planning aversion, where they skip the planning phase entirely, particularly when under pressure to act quickly (Montoya et al., 2015). A lack of structured planning leaves teams unprepared when faced with unexpected challenges (Montoya et al., 2015). Each of these issues highlights the need for careful management and strategies to ensure effective team planning and decision-making.

To mitigate the typical problems associated with team planning, teams may need to implement a series of interventions. For example, to mitigate the tendencies for prediscussion preferences or shared information to dominate the conversation, leaders might frame the planning process as a problem-solving task rather than a judgment-based one. When teams focus on solving a problem, they are more likely to exchange novel insights and explore new solutions rather than merely validating existing opinions (Griffin & Guez, 2014). This approach encourages teams to gather all relevant information before making decisions, similar to methods used in creativity research (Wang & Nickerson, 2017).

Matching approaches, disciplines, and methodologies to the problem's demands, rather than to any individual's attachments, is important for designing a study capable of addressing the complex problem or question identified. Because they are engaged in complex problem-solving, science teams often rely on boundary objects to scaffold their understanding of a problem and its elements (Star & Griesemer, 1989). Scholars originally conceived of *boundary objects* as "inhabit[ing] several intersecting social worlds and satisfy[ing] the informational requirements of each of them" (Star & Griesemer, 1989, p. 393). Boundary objects can be used to align individual work in multidisciplinary teams, for instance, or to produce novel pathways for more engaged collaboration outside an existing field or discipline.

For example, systems mapping can help a team match a collaboration to a problem of interest by identifying and displaying the interlinkages of relevant components, connections, and interested parties and then deciding what is in versus out (e.g., Braithwaite et al., 2018). A systems map can be seen as a type of boundary object through which members of a science team understand their project's collective goals, how they contribute to those goals, and how their efforts intersect with those of others on the team. In collaborative problem-solving, these types of co-constructed artifacts have taken different names, including process-based descriptions, such as

model-based reasoning (Pennington et al., 2021), external representations (Fiore & Schooler, 2004), cognitive artifacts (Fiore et al., 2010; Hutchins, 1999), and coordination artifacts (Schmidt & Wagner, 2004).

To foster knowledge integration in science teams, some intervention research has incorporated cognitive artifacts with teamwork processes. One group developed a program based on the cognitive and learning sciences, where science team members worked to integrate internal mental models of a problem with externalized or visual forms co-created with their team (Pennington, 2016; Pennington et al., 2016, 2021). Although there has been some evidence of attitudinal changes following this intervention, it has yet to be tested rigorously. Any material or processual artifact could be considered a boundary object, so long as it holds meaning for each individual member of a team and facilitates collaboration through the shared mental models or understanding it promotes across the team. As such, boundary objects can support inclusive design for team science (Star, 2010).

Another approach science teams may use to leverage collective knowledge and strengths into integrative project designs is *perspective-taking*, where individuals adopt other team members' viewpoints on a topic to better appreciate differences across backgrounds and fields and integrate disparate perspectives into a cohesive team vision (Hoever et al., 2012). Boundary objects could represent a useful starting point for perspective-taking activities, as members of a science team try to reflect on the same object from an intersecting lens other than their own.

Open dialogue around assumptions and disciplinary standards is a proven means of fostering integration in science teams (Piso et al., 2016). It is also important to practice and promote epistemic humility, which involves encouraging team members to be open about the strengths and limitations of their disciplines and to develop respect for and engage with new ways of knowing (Boix Mansilla, 2006, 2017). This practice can be informative to other team members, who may be unaware of the constraints present in other disciplines and other organizations (Castillo et al., 2024). Boundary objects, cross-training, and dialogue during team planning and project design can together help science teams avoid *disciplinary capture*, or scenarios in which one of the collaborating disciplines overshadows the others in terms of decisions made and research directions taken, hampering the integrative potential of a team science approach (Brister, 2016).

Planning the taskwork is integral to improving team processes and outcomes (Shuffler et al., 2018). Although sources in the literature recommend certain task-based strategies—such as building in task interdependencies to encourage team trust and cohesion (De Jong et al., 2016) and holding regular team check-ins to foster information exchange and team learning—team members' preferences and needs (e.g., working style, platforms, and interdependence) need to be taken into account for motivation to remain

high (Park et al., 2013). In the committee's experience, failure to account for such preferences during team planning and project design can result in frustration and even team member attrition.

To improve decision-making during planning, teams could also allocate time to discuss alternative strategies rather than focusing solely on initial preferences. This increases the likelihood of identifying effective solutions and avoiding decisions based on entrenched views (Tschan et al., 2009). Including members with conflicting preferences can stimulate richer discussions, as dissenting opinions push teams to engage in deeper, more critical evaluations (Schweiger et al., 1989). Ensuring that team members develop accurate cognitive models of the task and environment also helps, particularly in reactive situations. Leaders can play a key role in guiding teams to consider different perspectives and avoid fixating on initial assumptions.

Small groups naturally promote participation, as individuals feel a sense of responsibility to contribute (Dening et al., 2022). However, large groups may require different strategies. For example, leveraging asynchronous virtual communication platforms, such as message boards or chat rooms, can provide quieter members an opportunity to contribute, reducing the dominance of more vocal team members. In addition, counteracting the tendency to exert less energy when working in a group than as an individual and uneven participation is important (Dening et al., 2022). Teams can address motivation losses by increasing cohesion, identifying individual contributions, and holding members accountable for their input (Berengüí et al., 2021; Braun & Avital, 2006; Cady et al., 2018; Stewart et al., 2023; Whitworth & Biddle, 2007).

Finally, it is essential for teams to allocate sufficient time for planning. Without designated planning time, teams may skip this critical phase and resort to reactive decision-making. Using formal planning tools, such as charters or strategy outlines as discussed in the previous section, can ensure that key points are addressed during discussions. Teams under heavy performance demands can use low-workload periods to engage in planning, ensuring that preparation is not neglected when pressure mounts. Promoting a cohesive climate and setting challenging goals can further motivate teams to invest in detailed planning, recognizing the complexity of tasks ahead and the need for thorough strategies.

### Conceptualization Phase Summary

The conceptualization phase emphasizes the importance of laying strong, synergistic foundations for both teamwork and taskwork before any team science project begins. Best practices surrounding team charter development, team planning, and project design represent opportunities to build on team efforts in the development phase, strengthen shared mental

models, and provide a clear path forward. This foundation is crucial for effective operationalization during the implementation phase.

## Phase III: Implementation

In the implementation phase, science teams execute the project plans established during earlier stages (Hall et al., 2012). Even when science teams have diligently followed best practices during the development and conceptualization phases, the transition to implementation is rarely seamless. During this phase, teams need to engage in several complex processes, such as experiment execution, that require integrating teamwork and taskwork. It is important for the team to periodically revisit and ensure alignment with its shared goals and vision. Effectively engaging in the implementation phase requires regular interactions among team members, sustained effort, and continuous monitoring of progress toward scientific goals. This includes communicating openly, providing backup support when needed, giving constructive feedback, and coordinating actions according to task demands (Hall et al., 2012; Marks et al., 2001). In the following sections, the committee provides a detailed overview of two critical best practices—project management and team debriefs—during the implementation phase.

### Best Practice: Project Management

Once a team enters the implementation phase of its scientific work, managing the project effectively becomes essential. *Project management* involves the systematic and deliberate application of knowledge, tools, and expertise to ensure the successful completion of complex projects in a timely and efficient manner (Sutton et al., 2019; Wuchty et al., 2007). A cornucopia of project management practices has been shown to work outside of science teams, and programs are being developed for implementing these practices in the science team context (Brasier et al., 2023a; Steiner et al., 2023; Sutton et al., 2019). Project management, however, is not limited to a single individual within the team. Multiple members can engage in project management activities, contributing to task organization, communication, and coordination across the team. In the following, the committee highlights some of the best practices most applicable to science teams.

Team Meetings. Over the past 2 decades, an entire science devoted to understanding team meetings has developed (Rogelberg, 2019; Rogelberg et al., 2006; Wolf et al., 2024). Meetings can enhance collaboration, facilitate exchanging information, and improve decision-making by providing a structured platform for communication and coordination (Allen & Rogelberg, 2013). When managed properly, meetings can foster employee engagement,

promote team cohesion, and align individual tasks with broader team goals (Mroz et al., 2018). Moreover, well-facilitated meetings can create a space for shared leadership and help balance individual and team-wide objectives, which are particularly important in multiteam systems (Wolf et al., 2024).

On the other hand, poorly run team meetings can lead to disengagement, frustration, and wasted time. And if meetings are too frequent, too long, or lack clear objectives, they can contribute to meeting fatigue and reduce productivity (Allen et al., 2012; Mroz et al., 2018). In multiteam systems, an overemphasis on meetings without sufficient inter-team task interdependence can be counterproductive, hindering performance and collaboration (Wolf et al., 2024). In addition, meetings that do not encourage participation or are dominated by a few voices can lead to missed opportunities for varying input and innovation (Allen & Rogelberg, 2013). Meetings and discussions can also be exclusionary if they do not incorporate accessible solutions such as sign language interpreters, real-time captioning, or assistive listening devices when needed.

Several strategies can help science teams maximize the effectiveness of their meetings. Key practices include having a clear agenda, starting and ending meetings on time, and ensuring that only essential personnel are present (Mroz et al., 2018). These strategies not only help meetings run efficiently but also avoid wasting time for individuals whose presence may not be required. In addition, those responsible for managing meetings can regularly assess whether a meeting is truly necessary. If there is no pressing need to meet, even for regularly scheduled meetings, it is entirely appropriate to cancel them (Rogelberg, 2019). Those managing the meeting can also play a crucial role in creating a welcoming environment where all participants feel psychologically safe to share their ideas and perspectives (Allen & Rogelberg, 2013). This promotes open communication and collaboration. In multiteam systems, managing meetings can be more complex, but maintaining a balance between team-specific and system-wide goals is essential for sustaining productivity (Wolf et al., 2024). Finally, regular feedback and follow-up on meeting outcomes, such as distributing meeting minutes and reviewing action items, are critical for ensuring accountability, maintaining momentum, and maximizing the effectiveness of team meetings (Rogelberg et al., 2006; Wolf et al., 2024). Artificial intelligence (AI) tools, such as Otter.ai, ChatGPT, and Fireflies.ai, can be effective in transcribing meetings and summarizing key points and discussion for later access by all team. In the context of science teams acting as multiteam systems, it is recommended that meeting preparation and planning be incorporated into the multiteam charter (Asencio et al., 2012).

Facilitation. Facilitators, whether internal or external to a collaboration, can help science teams make the most of their meetings together. With a

focus on both the activities and interactions of teams (Bens, 2012), facilitators integrate teamwork and taskwork holistically to achieve meeting goals (Wróbel et al., 2021). Science team meetings can benefit from facilitation in a more traditional sense, whereby a facilitator consults on meeting design, tracks progress and time, fosters equitable and inclusive conversations, and establishes the right meeting tone and environment (Wardale, 2013; Wróbel et al. 2021). Conventional facilitation can further help science teams structure and guide crucial processes such as problem-solving, reaching consensus, decision-making, conflict resolution, and navigating different stages of a team (e.g., forming, norming, storming, performing; Kaner, 2014; Tuckman, 1965). Through thoughtful and intentional team interactions, facilitation can enhance idea generation (Kramer et al., 2001) and engagement (Parker, 2020) as important contributors to overall team performance. However, recent work has pointed to the need for facilitation practices designed specifically for boundary-spanning science teams (Cravens et al., 2022; Graef et al., 2021a,b).

One group proposed that science facilitation uniquely intersects collaborative science expertise with interpersonal expertise to align with the distinct purpose and challenges of knowledge-producing teams (Cravens et al., 2022). Meetings, then, are opportunities for science facilitators to create enabling conditions for "resolving many of the interpersonal and conceptual challenges of interdisciplinarity" (Graef et al., 2021b, p. 109). Challenges unique to science teams include navigating the ambiguity of novel cross-disciplinary collaborations, reconciling epistemic discrepancies, partnering with nonscientists, and negotiating disparate scientific priorities and approaches into a cohesive project capable of solving a real-world problem. Science facilitators can integrate these team science challenges into meeting activities and design (Graef et al., 2021a), and thus capitalize on meetings' immense potential to contribute to the collaborative research process (Graef et al., 2021b).

For example, science facilitators can guide scientific visioning in a team meeting by creating a shared conceptual framework. In doing so, they can help science teams bridge cross-disciplinary communication, identify points of scientific contention, and prompt for linkages and uncertainties until achieving an inspiring, integrated vision that represents and holds meaning for every individual on a team. A science facilitator can also help a team match its meeting sessions to its phase in the scientific process. During conceptualization, a science facilitator may prioritize sessions featuring whole-group discussions where everyone can share their perspective with and learn from the full array of collaborators. A science facilitator can, however, recognize that session formats during the implementation phase may need to shift to small breakouts based on tasks or even to individual quiet working time. In these ways, science facilitation can operationalize

the intersection between collaborative science and interpersonal dynamics to advance team science (Hall et al., 2018).

Communication Management. As indicated throughout this chapter, communication is vitally important for the success of science teams. However, no matter how much planning is done in the development and conceptualization phases, communication challenges are inevitable when the implementation phase begins. Factors such as disciplinary silos, power dynamics, differing communication styles, and coordination across time zones and locations can create communication challenges for science teams (e.g., O'Rourke et al., 2023). Still, actions can be taken during the implementation phase to mitigate some of these challenges.

One general best practice is fostering a structured dialogue that creates an environment conducive to purposeful and meaningful exchanges among team members. Teams can prioritize cultivating a positive communication culture where all members feel safe and encouraged to share their perspectives, as this leads to more successful outcomes (Cason et al., 2020; Hubbs et al., 2020; O'Rourke et al., 2023). In addition, structured feedback mechanisms, such as regular updates or debriefs (discussed in greater detail later), are essential for maintaining productive communication during the implementation phase. Even in teams with generally effective communication, it is important to monitor interactions for acute challenges that, if left unchecked, can develop into chronic communication issues (O'Rourke et al., 2023).

A particularly effective practice is *turn-taking*, which ensures more even distribution of speaking opportunities in team meetings, allowing team members to share the floor rather than letting a few individuals dominate the conversation. Any team member, whether the leader, a participant, or an external facilitator, can implement turn-taking. One case study of using an external facilitator found that teams practicing even turn-taking tended to have higher success rates because it promoted the balanced exchange of ideas and enhanced collective problem-solving (Love et al., 2022). Although interactional best practices, such as turn-taking and creating space for social time, can be incorporated into virtual team settings, the consensus in the literature is that in-person meetings with face-to-face communication lead to more effective team science, especially during times of problem-solving and trust-building (Henson et al., 2020; Zajdela et al., 2025).

As the project matures, teams may consider developing information-exchange protocols that consider team members' preferences to stipulate when, how, and to whom project updates or information will be communicated (Zajac et al., 2021), with quality of communication prioritized over quantity (Marlow et al., 2018). During check-in meetings, it is important for team members to discuss more than surface-level updates and instead

share honest responses around successes, failures, and unknowns. Allowing other members of the team to process such information can lead to improved understanding of where another member of the team can be deployed to help solve a problem, for example. In addition to discussions, it is also important to determine specific next steps to be taken between regular check-ins to ensure forward momentum is maintained.

Monitoring and Assessment. Ongoing monitoring and assessment during the implementation phase is a crucial best practice (see also Chapter 5). Having a clear understanding of how the team is performing at any given time is beneficial, but there are specific considerations regarding what is being monitored, when it is monitored, and how the monitoring is conducted. First, it is essential to target specific attributes of the team to assess. Teams can be evaluated on various factors such as performance outcomes (e.g., quality, accuracy, and timeliness of the work produced) or on process and emergent state metrics (e.g., satisfaction, trust, cohesion; Rosen et al., 2008; Shuffler et al., 2018). When deciding what to measure, those managing a project need to consider whether the assessed factors are related to the team's overall goals directly. For example, if the goal is to foster innovation, the team could measure factors such as creativity, the generation of new ideas, and disciplinary integration rather than focusing solely on traditional productivity metrics such as the number of papers published or the speed of task completion.

It is equally important to assess when to measure these dynamics. Some emergent states, such as trust, psychological safety, and cohesion, might be more meaningful when measured after significant events such as major project milestones, decision points, or conflict resolution (Carter et al., 2018; Kozlowski & Chao, 2018). Measuring team cohesion after an intense grant submission or research presentation, for example, could provide valuable insights into how the team responds to high-pressure situations and whether support mechanisms need adjustment. Other metrics, such as communication flow, may be appropriate to assess regularly throughout the project.

Finally, how a team monitors performance and process is critical for gathering actionable information. Different sources of feedback provide different perspectives, making it important to collect data from various team members, including principal investigators, junior researchers, and support staff. This variety ensures a more accurate and comprehensive team performance assessment, as each member may experience and interpret team dynamics differently (Wiese et al., 2015). Relying on a single source of feedback risks missing key insights, while integrating multiple data sources can highlight patterns, uncover hidden challenges, and support more targeted interventions.

Project Management Tools. Although the published research in the science team domain is limited, several tools have been identified for enhancing the project management process significantly. *Project management tools* are platforms, software, and structured methods that help organize, monitor, and manage project tasks, timelines, resources, and communications efficiently. These tools aim to streamline operations, reduce bottlenecks, and ensure that team efforts are well-coordinated for achieving project goals. Anecdotally, science teams often utilize such tools, but research on their specific applications in this context is still emerging, with recent studies highlighting their potential effectiveness (e.g., Gaffney et al., 2019; Steiner et al., 2023; Timóteo et al., 2021)

For example, one study examined the use of project management tools in a large, multisite lung cancer screening consortium to facilitate collaboration, efficiency, and productivity (Steiner et al., 2023). The tools employed in this case study included platforms such as SmartSheet for managing work plans, tracking data acquisition timelines, and maintaining project deliverables. In addition, the project used SharePoint and Microsoft Teams for document management and team communication, providing a centralized system for all sites to access essential documents and updates in real time. These tools allowed the team to maintain transparency; track progress; and coordinate complex, multisite research activities effectively.

Efficient resource-sharing is another key component of project management. Sharing resources such as data, software, and findings in real time can reduce duplication of effort, accelerate the research process, and ensure that all team members have access to the most current information (Santos et al., 2012). Leveraging collaborative document tools, such as Google Docs and Microsoft Teams, can allow team members to work on shared documents in real time, reducing email overload and ensuring everyone has access to the most up-to-date information.

Many project management platforms include integrated systems for documenting and archiving communications and decisions. This capability can help science teams maintain a comprehensive record of their projects' development, which can support team communication, transparency, reproducibility, and onboarding of new members. Implementing a consistent file management system with clear folder structures and naming conventions helps everyone find information easily, further supporting efficient collaboration. Teams can additionally take steps to ensure that all documentation is in an accessible format.

AI-powered platforms can improve taskwork and teamwork by automating processes and improving coordination. These platforms streamline activities such as scheduling, task assignment, literature reviews, and document organization (e.g., Benchling, Connected Papers, ScopusAI). AI-driven project management tools can assign tasks, track progress, and automate

workflows so teams can focus on high-priority tasks (Jackson, 2022). Virtual workspaces such as GrantedAI and Kudos support specific stages of a project, including grant writing and research dissemination. AI tools can foster collaboration and efficiency by automating activities and coordinating efforts.

AI can also improve data and knowledge management systems (Jarrahi et al., 2023), enabling team members to access and share information easily, while intelligent collaboration platforms can optimize task allocation and track progress (Chen et al., 2017). AI tools, such as retrieval-augmented generation (Lewis et al., 2020), can organize, index, and retrieve relevant information from large datasets, providing team members with access to resources. AI tools can also support data anonymization and privacy management (Abay et al., 2019) and directly allow global model learning without sharing sensitive information across teams (McMahan et al., 2017).

### Best Practice: Team Debriefs

A team debrief is a formal team development intervention that "turns a recent event into a learning opportunity through a combination of task feedback, reflection, and discussion" (Keiser & Arthur, 2021, p. 1008). Conducted periodically after completing significant team activities, team debriefs are structured sessions for reviewing and analyzing a team's performance (e.g., Shuffler et al., 2018). The primary goal is to reflect on what happened, identify what went well and what went poorly, and determine ways to improve in the future (Tannenbaum & Cerasoli, 2013). A wide swath of industries, including manufacturing, education, information technology, aviation, the military, and health care, employs team debriefs (e.g., Chen et al., 2018; Duff et al., 2024; Eddy et al., 2013). The committee uses the term team debrief because it has been used most often in reviews of the literature (e.g., Keiser & Arthur, 2021; Shuffler et al., 2018; Tannenbaum & Cerasoli, 2013), including literature on science teams (Begerowski et al., 2021). However, other names include after-action reviews (e.g., Department of the Army, 1993), post-mortem evaluations (e.g., Kasi et al., 2008), huddles (e.g., Reiter-Palmon et al., 2015), reflexivity (e.g., Tesler et al., 2018), and guided team self-correction (Smith-Jentsch et al., 2008).

Research on debriefs has been conducted primarily on nonscience teams. Multiple meta-analyses demonstrate that team debriefs are associated with higher team performance (e.g., Keiser & Arthur, 2021, 2022; Tannenbaum & Cerasoli, 2013). Compared with no-debrief teams, debrief teams improved team performance and that of the individuals within teams by approximately 25% (Tannenbaum & Cerasoli, 2013). A later meta-analysis found even higher performance effects after increasing the number of included studies (Keiser & Arthur, 2021). Mediators of the team

debrief-performance relationship include workload-sharing (Vashdi et al., 2012).

In addition to task performance, team debriefs can influence and improve attitudes, task knowledge, and team processes (Keiser & Arthur, 2021). Research has shown that debriefs enhance team adaptation (Abrantes et al., 2022) and leadership development (DeRue et al., 2012), and reduce decision time (Oudrat-Ullah, 2007). Meta-analyses have demonstrated the efficacy of debriefs in small (2 members), medium (3-5 members), and large (6-16 members) teams (e.g., Keiser & Arthur, 2022). In addition, debriefs are effective in both geographically dispersed and face-to-face settings (Keiser & Arthur, 2022). Furthermore, team debriefs are especially useful in high-complexity and ambiguous task environments that offer no intrinsic feedback (Keiser & Arthur, 2022). Therefore, in addition to action teams—in the military and health care—debriefs can be effective for project and decision-making tasks (Keiser & Arthur, 2022). Given the strength and generalizability of these results, debriefs are a straightforward, inexpensive, and easily implemented way to facilitate team effectiveness (e.g., Shuffler et al., 2018; Tannenbaum & Cerasoli, 2013). Based on the research conducted in the broader team literature, one analysis identified team debriefs as a central team development intervention for science teams (Begerowski et al., 2021). However, research on debriefing in science teams is needed.

Fundamental components of team debriefs include feedback, reflection, and discussion about specific performance events (Keiser & Arthur, 2021). Note that debriefs extend beyond feedback in that they are collaborative and ideally include all team members rather than just the team leader offering comments (Tannenbaum & Cerasoli, 2013). Additionally, a defining feature of team debriefs is self-learning, in which team members are actively involved in self-discovery rather than receiving feedback passively (Tannenbaum & Cerasoli, 2013). In addition, whereas traditional feedback emphasizes outcomes, the focus of debriefs is the processes that contributed to successes and failures (Allen et al., 2018; Tannenbaum & Cerasoli, 2013). Because debriefs are developmental in intent and are carried out to promote self- and team-learning, as opposed to evaluative for administrative decision-making (Tannenbaum & Cerasoli, 2013), outcome feedback regarding success or failure can be provided after debriefs (Salas et al., 2008).

Despite these similar components, there is substantial variation in how debriefs are conducted across studies and organizational settings (e.g., Eddy et al., 2013; Smith-Jentsch et al., 2008; Smith-Jentsch & Sierra, 2023). However, meta-analytic findings support using the original structure for debriefs developed by the U.S. Army (Department of the Army, 1993; Keiser & Arthur, 2022). This structure includes reviewing intended objectives, actual outcomes, effective actions, ineffective actions including near misses,

intended future objectives, and the strategy for achieving the intended future objectives (Department of the Army, 1993; Keiser & Arthur, 2022). Debriefs can include asking open-ended questions, including: What were we trying to accomplish? What happened in the team event? Where did we succeed in meeting our goals? Where did we fail to meet our goals? What caused our results? What can we start, stop, and continue doing? What are the important takeaways and lessons learned? (e.g., Keiser & Arthur, 2022). Identifying actionable next steps is critical to debriefing effectively (e.g., Chen et al., 2018; Salas et al., 2008). Following the group discussion, the debrief can be paired with outcome feedback and individual or team training as needed (Salas et al., 2008).

Researchers have provided additional evidence-based best practices in response to the variability in debrief implementation (e.g., Keiser & Arthur, 2021, 2022; Salas et al., 2008; Tannenbaum & Cerasoli, 2013). For example, team debriefs can be conducted as soon as possible after completing a significant team event, such as achieving a major deliverable; time period, such as after a shift; or training, such as after a simulation (Tannenbaum & Greilich, 2023) to avoid members forgetting important details with the passage of time (Salas et al., 2008). Levels of analysis can be aligned such that tasks, training, and criteria match for individuals and teams (Keiser & Arthur, 2021; Tannenbaum & Cerasoli, 2013).

Instead of a general performance overview, debriefs can provide specific examples of competencies and deficiencies (Tannenbaum & Cerasoli, 2013). As such, it is critical to establish and maintain high psychological safety so that members feel comfortable sharing errors and mistakes in a supportive rather than judgmental environment (e.g., Kolbe et al., 2020; Salas et al., 2008). It is also advised that feedback be supported with objective data, such as text or video recordings, rather than relying solely on memory (Keiser & Arthur, 2021; Salas et al., 2008).

Although one study found that using trained facilitators in debriefing activities enhanced team performance (Tannenbaum & Cerasoli, 2013), a later meta-analysis concluded the picture was more complex (Keiser & Arthur, 2021). Specifically, individual tasks benefited the most from having a facilitator, but team tasks benefited from a self-led approach, especially when combined with objective review media (Keiser & Arthur, 2021). Additional debrief characteristics interacted with one another to affect team outcomes, requiring more nuanced implementation guidelines (Keiser & Arthur, 2021). To illustrate, a highly structured debrief was more effective for action teams, such as those in the military, whereas less structure was needed in health care and other industries (Keiser & Arthur, 2021). Moreover, shorter debriefs of a maximum of 20 minutes were best for teams, especially in health care, whereas longer debriefs of a minimum of 20 minutes may be needed for individuals to allow sufficient time to cover

all the feedback (Keiser & Arthur, 2021, 2022). These results highlight the importance of considering debriefing characteristics in combination rather than implementing them in isolation (Keiser & Arthur, 2021).

#### Implementation Phase Summary

The implementation phase provides an opportunity for the team to sustain or reignite the initial enthusiasm for the scientific collaboration, which can help sustain long-term progress. To achieve team goals, team members need to actively interact, coordinate, and adapt. This requires project management, where actions are taken to ensure teams engage productively and stay on track, alongside regular team debriefs, which allow the team to reflect on progress, identify challenges, and adjust strategies as needed. Maintaining trust through transparency and reliability is essential during implementation. Trust fosters cooperation, promotes the sharing of resources and knowledge, and enhances the team's motivation to achieve their objectives. High levels of achievement and trust will help teams continue their work together in the translation phase.

#### Phase IV: Translation

Research translation is the process of moving the findings of scientific research from the laboratory environment to human studies and ultimately into policy and practical applications that affect society directly. Because research translation requires understanding both the basic science and the context in which the innovation will ultimately be applied, science teams can strategically include team members—scientists, community advocates, patients, policymakers, and industry partners—with a range of expertise, experience, and reach.

U.S. government agencies have long affirmed the role of science teams in closing the gap between scientific discoveries and the application of those discoveries into tangible outcomes. For instance, in 2011, NIH established the National Center for Advancing Translational Science (NCATS)<sup>6</sup> to support developing and implementing innovative processes, technologies, and methods for addressing the spectrum of human diseases and conditions affecting society. The NCATS approach is based on enabling the work of science teams that include government agencies, private-sector companies, patient advocacy groups, and other members of the scientific community. With a core value of collaborative team science culture, NCATS's strategic plan outlines goals for collaboration and partnerships in each of its four

<sup>&</sup>lt;sup>6</sup> For more information, see http://ncats.nih.gov/about/about-translational-science and http://ncats.nih.gov/about/about-translational-science/spectrum

programmatic pillars. And in 2022, the CHIPS and Science Act, signed into law by President Joe Biden, affirmed this federal commitment by establishing the NSF Technology, Innovation and Partnerships (TIP) Directorate, the first new directorate in 30 years (Afful & Meiksin, 2022). TIP programs advance national competitiveness and strengthen societal impact by supporting partnerships between and among teams of researchers, practitioners, and users, enabling the co-creation of scientific discovery and thus closing the gap between discovery and societal impact (Glasgow & Emmons, 2007; Huebschmann et al., 2019).

Translational science teams face many of the same challenges of any science team and, as such, team leaders can prioritize team-building (Begerowski et al., 2021) and development (Stokols, 2010) using the best practices outlined elsewhere in this chapter.

## Best Practice: Focus on Both External and Internal Validity

A well-documented challenge in public health research is the limited attention to replicating studies to explore generalizability of public health interventions across contexts (Huebschmann et al., 2019). Public health researchers have prioritized internal validity over external validity, thus limiting the ability to translate research into practice.

## Best Practice: Connecting with Community Members

Science teams oriented toward research translation are described using a variety of terms. Research translation can include components of community-based research, participatory research, community engagement, empowerment evaluation, participatory or community-based action research, and engaged research, all of which partner scientist and nonscientist team members. Not surprisingly, including nonscientist members on science teams has varied success in producing outcomes that are readily applicable to the target community. Mercer et al. (2008) reliability tested an evaluation tool for measuring the effectiveness of community-based participatory research. Their review of the literature revealed five best practices for effective engagement with community members, including forming an advisory board, establishing research agreements that outline roles and responsibilities, leveraging group facilitation techniques and designated facilitators, holding regular meetings to keep team members engaged, and hiring team members from the community. While Mercer et al. (2008) focused specifically on mechanisms for facilitating research translation, these best practices are used in other contexts as well.

## **Cross-Cutting Best Practices**

Certain best practices, including collaborative technologies and team leadership, transcend any one phase of a science team. As such, relevant evidence for these two best practices follows, separate from the development, conceptualization, implementation, and translation phases but pertinent to all four.

### Best Practice: Incorporating Virtual Collaborative Technologies

With the advent of advanced information and communication technologies and the proliferation of their adoption that accompanied the COVID-19 lockdowns, the ways science teams are collaborating have significantly changed. As noted in Chapter 2, teams can operate entirely virtually, entirely in person, or as a hybrid team. Hybrid teams can be characterized by three key dimensions: geographic distribution, temporal dynamics, and communication richness. *Geographic distribution* refers to how team members are physically dispersed across different locations, ranging from fully collocated teams to those in which members work in entirely separate geographic areas (Handke et al., 2024). *Temporal dynamics* concern the timing and coordination of work, particularly how team members' schedules and time zones align or differ (Handke et al., 2024). Finally, *communication richness* refers to the capacity of the communication medium to carry information (Kirkman & Mathieu, 2005).

It is crucial to recognize that hybrid teams can take on various configurations depending on these characteristics. For instance, a science team studying climate change may have field researchers collecting data in remote locations while others are in laboratory settings analyzing the data in real time. This team might have some members who meet in person periodically for intensive problem-solving, while the rest of the team participates remotely from different time zones, coordinating work through collaborative platforms. Another example could be an interdisciplinary research team in which experimental biologists work in a laboratory, while computational modelers collaborate from other institutions, contributing to the project through cloud-based data sharing platforms and virtual meetings. For teams that have a large component team distance, referring to large geographical, cultural, functional, or disciplinary distance, these complexities can be particularly relevant. Virtual multiteam systems that are geographically dispersed, have communication barriers, and work in different time zones can struggle to navigate these configurations (Ingersoll et al., 2024). Developing communication norms, holding in-person meetings, and creating cross-team coordination roles could potentially help bridge these divides.

As virtual collaboration tools have evolved to facilitate science teamwork, it is important to consider the team's specific configuration and how this configuration may change over time when assessing virtual collaboration tools' use and effectiveness (e.g., Gibson et al., 2022). Compared with in-person teams, virtual and hybrid teams have different challenges when it comes to achieving team effectiveness, many of which are mediated by or dependent on virtual collaboration tools (Brucks & Levav, 2022; Handke et al., 2024; Purvanova & Kenda, 2022). When applied thoughtfully, virtual collaboration tools can help enhance communication, foster collaboration across geographic and temporal boundaries, and support flexible, adaptable, and efficient team functioning, regardless of how the team is configured.

# Technology-Team Member Interactions

When adopting, configuring, or implementing a technological tool, it is crucial to consider how team members may interact with the tool to best facilitate team dynamics. As suggested above, while these tools offer several benefits, the success of their implementation depends largely on how well they align with team members' skills, needs, and comfort levels (e.g., Larson & DeChurch, 2020). Misalignment can lead to inefficiencies and frustration, mitigating the likelihood that these tools will facilitate scientific collaboration, and potentially contributing to suboptimal virtual team outcomes (e.g., Larson & DeChurch, 2020; Waizenegger et al., 2020).

One key factor to consider is team members' familiarity and comfortability with technology. While many constructs and scales are available (Holcomb et al., 2004; Martínez-Córcoles et al., 2017; Mason et al., 2014; Merritt et al., 2013; Montag et al., 2023; Sindermann et al., 2021; Sinkovics et al., 2002), the shared premise across these constructs is understanding an individual's comfortability and, consequently, propensity to engage with technologies. Individual dispositions toward technology, as well as learned or situational factors, can play a critical role in how effectively people adopt and use these technologies within the team (Hoff & Bashir, 2015). When team members are comfortable with the technology, they are more likely to engage with it fully, taking advantage of its features to enhance team communication and coordination (e.g., Colbert et al., 2016; Kilcullen et al., 2022). On the other hand, a lack of familiarity can create barriers, preventing team members from adopting new tools or leading to inefficient use of existing ones.

For example, in a science team working across multiple institutions, unfamiliarity with new collaboration software can create significant challenges. If team members struggle to use key features such as document editing or data management, this can lead to miscommunication about research

findings, delays in decision-making, or even missed opportunities for collaboration. Instead of facilitating effective communication and collaboration, the software becomes a barrier, requiring extra time to troubleshoot. This disruption can detract from the team's focus on critical scientific tasks, leading to frustration and negatively impacting overall team performance and morale.

Thus, it is critical for team members to feel comfortable using technological tools. This can be achieved through targeted training sessions that familiarize members with the tools they will be using and allow them to practice in a low-pressure environment. Exposure to technological tools can facilitate their use of these technologies and decrease anxiety surrounding them (e.g., Sherrill et al., 2022). That is, training not only helps individuals gain competence but also promotes collective confidence in using the tools effectively.

Another crucial consideration when selecting and using technological tools is their accessibility for virtual and hybrid work. While many tools claim to support seamless remote collaboration, not all are accessible to all team members, particularly team members with disabilities<sup>7</sup> (Doush et al., 2023; Hersh et al., 2024). For instance, video conferencing software that lacks closed captioning can present barriers for D/deaf or hard-of-hearing team members. Similarly, tools with complex interfaces or high bandwidth requirements may exclude team members with limited technical capabilities or unreliable internet connections. Another accessibility challenge comes to bear when considering those who are blind or have low vision. Tools that are incompatible with screen readers, for example, can make it difficult for blind or low-vision team members to contribute effectively (Leporini et al., 2023). In a related vein, tools that require extensive use of a computer mouse may not be suitable for people with different physical abilities (e.g., Marchant et al., 2005; Trewin & Pain, 1999). As such, ensuring that all tools are accessible to all team members—showing consideration for their individual and differing needs and enabling their participation without placing undue burden—is a key component of fostering an effective team environment.

# Establishing Norms

Technological tools often include a wide range of features—such as screen sharing, instant messaging, hand raising, and file sharing—designed to facilitate communication and collaboration. While these are useful,

<sup>&</sup>lt;sup>7</sup> The committee acknowledges that identity-related terminology is complex, personal, and continuously evolving, and that the language used to refer to disabled communities (i.e., person-first vs. identity-first) differs by community and by individual.

without clear norms and expectations about how such functions will be used, teams may experience confusion, inefficiency, or frustration in their collaboration (Kilcullen et al., 2022).

For example, teams may end up violating data privacy agreements when norms surrounding these technologies are not established. Consider a laboratory using a multifunction project management platform that includes features for chat, video conferencing, file sharing, and data storage. Although the technology is convenient and could technically handle file sharing, it may not meet the data protection standards required for the sensitive research the team is conducting (see Chapter 4). Without established file sharing norms, team members could share sensitive files through this platform, inadvertently violating data privacy regulations.

Relatedly, establishing clear norms around the use of these tools is essential for promoting effective collaboration for everyone on the team (e.g., Gibson et al., 2014; Kirkman et al., 2002; Kirkman & Stoverink, 2021). This is particularly important in the context of team science, which frequently involves researchers from different disciplines, cultures, and language abilities (especially for non-native speakers of a primary team language) who bring unique skills and knowledge to a project. Virtual collaboration tools, such as speech-to-text features, can help foster communication and understanding by bridging language barriers or assisting team members who are D/deaf or hard of hearing. For D/deaf and hard-of-hearing team members, these tools can be tremendously beneficial in facilitating participation (e.g., Alshawabkeh et al., 2021; Ang et al., 2022). However, it is important to note that while automated speech-to-text technologies can be helpful, their use in transcripts is not considered compliant with the Americans with Disabilities Act (2010) if the content differs from the audioonly content; teams ought to be mindful of such limitations when setting expectations for accessibility. While speech-to-text technologies may work sufficiently to increase participation ease for D/deaf and hard-of-hearing team members, they are not as effective as live interpreters, who can more effectively convey aspects such as tone and nuance (Secară & Perez, 2022).

Beyond immediate accessibility, these tools can integrate different view-points by providing structured platforms for communication, data sharing, and collaborative problem-solving. The flexibility these tools offer can allow geospatially distributed team members to meet synchronously, and contribute asynchronously, making it possible for teams who may not have had the opportunity otherwise to work together effectively (Meluso et al., 2022). Importantly, it is not just the tools themselves that can foster this, but the thoughtful design, implementation, and norms that guide their use. By co-creating and adopting norms, the team can ensure that everyone has an equal opportunity to contribute to the team.

#### Data Security and Use

Data security and use is a critical aspect of information technology, and data sharing is a unique challenge for team science.

Addressing these barriers requires a nuanced understanding of how data security practices can inadvertently exclude certain groups or reinforce existing inequities, and many teams may be ill-equipped and may not understand or anticipate these issues when building teams (e.g., Law, 2023). Governance structures that include different perspectives are more likely to develop comprehensive security policies that address the needs of the population (Grindstaff & Mascarenhas, 2019). Data security protocols often assume a one-size-fits-all approach that may not consider the cultural and linguistic differences of users. For example, security warnings and protocols that are not localized or adapted to different languages and cultural contexts may lead to misunderstandings and noncompliance, posing a significant barrier. Data security measures, while crucial, often enhance complexity in system design, which can disproportionately affect those with disabilities or those who are less technologically proficient. For instance, complex authentication processes can be a hurdle for users with cognitive disabilities or who are blind or have low vision. Studies have highlighted the need for adaptive technologies that comply with security standards while also being accessible, ensuring that security enhancements do not hinder usability (Wentz et al., 2011).

The development of security technologies, such as biometric authentication systems, has raised concerns about built-in biases. Studies have shown, for instance, that facial recognition technologies have lower accuracy rates for women and people of color than for White males (Buolamwini & Gebru, 2018). This raises concerns and questions about the fairness of security measures. Some groups often face greater risks of surveillance and privacy violations, which can deter their participation in digital platforms where their data might be insecure (Bacchini & Lorusso, 2019). Ensuring that data security measures protect all users is crucial, particularly for populations who may be disproportionately affected by data gathering and data breaches (Goldshtein et al., 2024).

In addition to these concerns, there are several emerging questions regarding technology and data use, and the ways in which they may function as a barrier to team inclusion. For example, AI technology is increasingly included in many personnel decisions, such as applicant screening. Some evidence suggests that algorithms may introduce biases, and additional research is needed to understand the full effects of possible biases impacts on teams (Albaroudi et al., 2024; Tilmes, 2022). Furthermore, in instances where demographic data are involved, careful consideration can be taken into the security of these data, particularly when their public disclosure could cause the individual harm (Calabro, 2018). This can be applicable for demographic

data in relation to both research subjects and the researchers themselves. To address internal issues, institutions can provide clear definitions of harassment and equip offices, such as offices of research or general council, to raise awareness of policies regarding reporting, investigations, and remediation of harassment claims. Some universities have implemented policies and provided resources to support researchers who are victims of doxing, or the publication of private or identifying information on the internet (e.g., Columbia University's Resources to Assist Ater Online Targeting/Doxing,<sup>8</sup> University of Illinois Urbana-Champaign's Trolling and Doxxing Attacks on Scholars – Executive Officer Action<sup>9</sup>). Professional societies like the American Association for University Professors have also issued guidance for faculty members who have been targeted by online harassment.<sup>10</sup> Notably, some of the forementioned resources for addressing online harassment may only be effective for addressing internal threats, and external threats may need to be addressed differently.

Advanced data security measures and data use can be resource-intensive, requiring modern infrastructure and sophisticated hardware or software that may not be accessible in underresourced schools, small community hospitals, low-income areas, or partners from low- and middle-income countries (Onoja & Ajala, 2022). This digital divide can prevent individuals in these areas from accessing secure services, thereby secure, yet affordable, technology solutions are essential to bridge this gap.

By addressing the specific barriers that data security can present, institutions can create a more secure, equitable, and inclusive digital environment. Effective data policies may include:

- Designing with accessibility in mind, ensuring that security measures do not hinder usability for people with disabilities.
- Investing in technology that is accessible and affordable.
- Adapting security measures to diverse cultural and linguistic contexts to ensure broad usability and understanding.
- Actively working to eliminate biases in security technologies to ensure fairness and equity.
- Enhancing privacy protections to build trust among marginalized populations.
- Involving diverse groups in the policymaking and governance processes concerning data security to ensure that these measures are equitable and inclusive.

<sup>&</sup>lt;sup>8</sup> See https://universitylife.columbia.edu/doxing-resources

<sup>9</sup> See https://provost.illinois.edu/faculty-affairs/faculty-resources/trolling-attacks-on-scholars-executive-officer-action/

Nee https://www.aaup.org/issues/targeted-harassment/what-you-can-do-about-targeted-online-harassment

#### Physical Security

Implementing physical security policies and protocols is also critical to protecting the safety of team members. Security risks can extend beyond threats to personnel to also include damage to physical spaces, equipment, and information technology systems. Institutions participating in team science, as well as science teams themselves, could benefit from being aware of and prepared for potential security issues that can arise, particularly when their research focuses on unpopular, politicized, or controversial topics. For example, some scientists working on projects related to COVID-19 have faced harassment and instances of physical violence (National Academies of Sciences, Engineering, and Medicine, 2023; Nogrady, 2021). Additionally, researchers who work with animal subjects have long been the target of extreme attacks, including bombings and arson (Collier, 2014). Along with the risks to the physical security of researchers, researchers who work on sensitive topics or are the targets of perceived or actual threats may also have additional needs to maintain psychological safety (Paterson et al., 1999; Williamson & Burns, 2014).

The National Research Council has previously issued guidance that research institutions and science teams can use to protect their physical safety (National Research Council, 2011). For institutions, this can include the use of security systems and the control of access to facilities, including through locks and access cards. Science teams can encourage individual members to increase their situational awareness and report suspicious behavior. They can also provide all members with training on what to do if a security emergency arises and avenues for reporting any incidents. Additionally, teams may need to take care in sharing scientific results when there are concerns regarding content and venue (Beechey, 2024). Teams can create dissemination and communications plans and data management plans to safely share results, which may include actions such as avoiding social media and anonymizing data.

Institutions have also issued crisis toolkits to provide resources to targeted individuals. For example, the University of Massachusetts Amherst's Academic Freedom Crisis Toolkit provides resources and outlines the responsibilities of both faculty and administrators in responding to a security crisis. <sup>11</sup> This stresses that the onus of responsibility lies not only with faculty and team members, but also with the institutions.

Finally, it is important to note that security can pose a particular challenge for team science, as science teams can include community members who may not have access to or knowledge of institutional resources. Multi-institutional teams can also face challenges with differences in institutional

<sup>&</sup>lt;sup>11</sup> See https://www.umass.edu/faculty-development/resources/academic-freedom-crisis-toolkit

practices and protocols, making team communication at the outset of projects important in deciding what practices and protocols will be used and creating a cohesive strategy for how the team will respond to security issues.

### Adapt to Changing Configurations

Importantly, virtual, hybrid, and in-person teams rarely remain in the same configuration over time. For instance, a group initially composed of principal investigators at the same university might expand to include international researchers or transition from mostly in-person meetings to fully remote interactions. As teams evolve over time, it is necessary for the ways technological tools are used to also evolve to accommodate new needs and challenges. As team compositions and structures change, the affordances of technological tools (e.g., video conferencing software, shared databases, project management platforms) may need to be revisited. Affordances refers to the potential actions or functions that a tool offers based on how it is used (e.g., Gibson et al., 2022). For instance, a team may have decided initially to use a tool for ad hoc communication but subsequently realized a need for handling more complex coordination tasks, such as scheduling across time zones or managing larger datasets.

As team configurations evolve, it becomes essential to revisit the accessibility norms and expectations that were initially set for the use of these technologies. A tool that worked well for a small, collocated group may no longer be effective for a dispersed or growing team. Similarly, accessibility needs may shift as team members from different regions or with different technological proficiencies join. Regularly reviewing these norms and conducting additional training when necessary ensures that all team members remain on the same page and can fully leverage the tools at their disposal.

## Best Practice: Team Leadership

Leadership has been a major focus of research in the organizational sciences for the past century (Carton, 2022; Yammarino et al., 2005; Yukl et al., 2002). However, empirical research on leadership in science teams specifically is relatively limited. Therefore, this section draws primarily from the broader leadership literature, especially the literature on team leadership (e.g., Zaccaro et al., 2001), to discuss leadership for science teams. When applicable, we incorporate findings from studies of science team leadership. The committee emphasizes the need for further research investigating science team leadership, particularly research focused on how to develop and support science team leaders.

*Leadership* is broadly defined as "the process of influencing others to understand and agree about what needs to be done and how to do it, and the

process of facilitating individual and collective efforts to accomplish shared objectives" (Yukl, 2006, p. 8). Whether this influence process emerges and proves effective depends on multiple factors, including the characteristics of leaders and followers and their interactions, as well as situational elements such as timing, group history, goals, and the scope of leadership activities (e.g., within a single team vs. across multiple interdependent teams). Moreover, leadership in science teams might be exerted by people who occupy formal positions of authority (e.g., principal investigators) or informally, by people without official leader positions. Leadership in science teams is often shared, distributed, or rotated over time, particularly when teams are large and include multiple co–principal investigators.

Team leadership can be understood through the lens of functional leadership theory, which argues that leadership effectiveness is the extent to which leaders and leadership processes support team effectiveness (Zaccaro et al., 2001). Team effectiveness is multifaceted and includes (a) team dynamics, or how effectively team processes and emergent states contribute to performance while fostering team learning and future collaborations; (b) the degree to which the team's performance outputs meet or exceed performance expectations; and (c) the extent to which individual members benefit personally from being part of the team (see also Chapter 5). Thus, effective leadership in science teams refers to how well leaders and leadership processes support these core aspects of science team effectiveness. Notably, it is important to consider team leadership as a dynamic process that plays a critical role throughout the team life cycle. While the team performs activities related to achieving its goals, leaders can perform functions such as managing team boundaries, challenging the team, and providing resources (Morgeson et al., 2010).

Effective team leaders prioritize supporting team dynamics (Zaccaro et al., 2001). To do so, leaders can use the best practices discussed in this chapter to establish key enabling conditions for team effectiveness (Hackman, 2012): a real team, a compelling purpose, the right people, clear norms of conduct, a supportive organizational environment, and positive leadership styles such as coaching.

The first enabling condition, a real team, is an intact social system with clear boundaries that distinguish members from nonmembers. Team members work interdependently toward shared goals, with collective accountability for outcomes. Science team leaders can use tools such as team charters, team debriefs, and team communication, and they can clarify team membership and patterns of interdependence.

The second condition is a compelling purpose that inspires and intellectually simulates the team. A well-defined purpose can motivate team members, align their efforts, and engage their talents. Establishing a clear purpose early on is crucial, as it influences team structure and determines the necessary

organizational support. Science team leaders could articulate and reinforce a compelling purpose during initial team-building activities and throughout all phases of the team's life cycle.

Third, leaders need to ensure the team is made up of the right people. This means carefully selecting members based on their skills and suitability for collaboration and ensuring that team members sufficiently understand one another's areas of expertise. To this end, leaders could leverage best practices related to team assembly and carefully consider aspects of team composition.

Fourth, leaders can establish and reinforce clear norms of conduct through regular communication, meetings, feedback, and debriefs. Setting clear behavioral expectations minimizes the need for managing team member behavior and allows the team to focus on performance. These norms also encourage continuous evaluation of the team's environment and the adoption of appropriate strategies. Leaders who articulate high-performance goals, provide constructive feedback, and model effective strategies help support the team's affective and motivational states, such as trust, cohesion, psychological safety, and collective efficacy. Furthermore, leaders maintain the emotional climate by keeping conflicts task-focused and managing stress during high-pressure situations. Leaders also guide team coordination processes, ensuring that actions and resources are synchronized. For example, leaders can facilitate the integration of individual contributions and establish communication norms that foster flexibility and adaptability in dynamic environments.

Fifth, leaders can help ensure that the team has necessary resources and that the organizational environment remains supportive of team activities by engaging in boundary-spanning activities with the external environment. Boundary-spanning activities might include securing resources and information, building external relationships, advocating for the team, transferring knowledge, or coordinating with other teams (Marrone, 2010; Marrone et al., 2007). As articulated in Chapter 4, support from the broader scientific ecosystem can be essential for team science success.

Lastly, effective leaders provide team coaching and adopt positive leadership styles, such as transformational leadership (Hall et al., 2018) and inclusive leadership. Transformational leadership, characterized by communicating an inspiring vision, encoring innovation, and promoting personal and collective growth, has been linked to enhanced team performance and creativity (Bass, 1999; Schaubroeck et al., 2007; Wang et al., 2011). Inclusive leadership behaviors including taking time to learn about and act upon the strengths, needs, and preferences of each member of the team; inviting different points of views on a topic; and practicing humility and limiting power dynamics, such as through shared leadership and decision-making (Nishii & Leroy, 2022; Roberson & Perry, 2021; Shore &

Chung, 2021). Inclusive leadership promotes psychological safety, which in turn leads to greater risk-taking and innovation (Brasier et al., 2023b). Team coaching, particularly when conducted by formal leaders, can have a positive effect on team processes and performance (Bisbey et al., 2021b; Shuffler et al., 2018; Traylor et al, 2020). Coaching behaviors can evolve as the team progresses, shifting from inspiring vision during development to offering moral support, problem-solving, or coordinating resources during implementation (Hackman & Wageman, 2005; Reich et al., 2009).

In conclusion, effective leadership is crucial for the success of science teams, particularly in fostering team dynamics and ensuring the achievement of shared goals. According to the broader literature on team leadership, science team leaders need to strive to create the enabling conditions necessary for team effectiveness. This involves establishing a real team with a compelling goal, assembling the right mix of team members, setting and reinforcing behavioral norms, and maintaining a supportive emotional climate. This also involves engaging in boundary-spanning activities (e.g., securing resources, aligning the team with the external environment) and exhibiting positive leadership behavioral styles (e.g., transformational and inclusive leadership, team coaching). These leadership practices are important throughout the life cycle of the team. Given the unique demands of science teams, future research on effective leadership in science teams will be essential for improving their collaborative efforts.

# **FUTURE RESEARCH QUESTIONS**

This chapter has outlined best practices for supporting science teams, situating them within the framework of the four stages of transdisciplinary research (Hall et al., 2012) and highlighting cross-cutting practices. Although the best practices discussed in this chapter are widely recognized as important for enhancing team science effectiveness, the committee emphasizes several key caveats. First, not all best practices will be suitable for every team or scenario. Science teams are variable in their structures, composition, and purposes—ranging from small, focused groups working on narrow and specialized problems to large cross-disciplinary teams tackling broad, complex issues. Moreover, the ideal timing and implementation of a best practice can vary depending on many factors, including team size, purpose, goals, challenges, or level of virtuality. It is important for teams to carefully assess their unique context and needs to determine which practices will be most beneficial and when they need to be applied.

The committee's second key caveat is that the empirical evidence base for team science best practices is still evolving. Although the broader literature on teamwork has identified key strategies that science team leaders, facilitators, or members could apply to enhance team effectiveness, the

specific application of these practices in science teams requires further investigation. Some of the best practices discussed in this chapter, such as team charters and team debriefs, are well established and supported by a strong empirical foundation (e.g., Courtright et al., 2017; Mathieu & Rapp, 2009), whereas other best practices are based on the committee's expert judgment. When possible, the committee recommends best practices with empirical evidence. However, when this evidence does not exist, the committee recommends best practices based on committee and other expert experience. Moreover, with regard to the best practices with a stronger empirical foundation, much of the research has been conducted in organizational contexts outside of scientific collaboration. Since science teams share many essential characteristics with teams outside of science, the best practices identified in the broader literature on team effectiveness are likely to be relevant within the context of team science. However, science teams may also exhibit unique characteristics that are relatively underexplored in the broader literature on team effectiveness. To the extent that science teams differ from other types of teams, some best practices drawn from the general teamwork literature may not be fully applicable. This creates an evidence gap specific to team science that highlights the need for further research on team science best practices and development of tailored approaches for optimizing science teams.

The critical research questions in Table 3-1 are intended to encourage research on the best practices discussed in this chapter. Although not exhaustive, these questions provide researchers with a foundation for exploring unanswered areas within the science of team science. Table 3-2 presents a summary of the best practices described in this chapter, for easy reference.

TABLE 3-1 Research Questions for Understanding Effective Science Teams

	Research Questions
Development	
Team assembly	What are the most important characteristics (beyond expertise) for selecting science team members to maximize collaboration and scientific output?  How do team members' orientations toward teamwork and interdisciplinary science influence the development of attitudes (e.g., trust, cohesion) in newly formed science teams?  How do structured onboarding processes (e.g., tailored team charters, mentorship programs) affect team science attitudes (e.g., psychological safety), behaviors (e.g., communication across disciplines), cognition (e.g., knowledge integration, role clarity, shared mental models), and productivity? Do these vary by experience or seniority (e.g., new graduate students, postdoctoral researchers) within established science teams?
Team onboarding and building	How do onboarding practices contribute to fostering a sense of belonging and engagement among members of science teams?  How do belonging, engagement, and psychological safety affect team processes and outcomes?  At what stage in the team's developmental life cycle do team-building activities contribute most to the success of science team success?  How do nontask team-building activities (e.g., social events such as meals) affect team processes and outcomes?  How do task-based team-building activities (e.g., writing retreats) or team-based initiatives (e.g., conflict resolution workshops) affect team processes and outcomes?  Of the above task and nontask team-building activities, how do they differentially influence emergent states such as psychological safety, trust, and cohesion within science teams?
Team charter	To what extent do onboarding strategies, such as the use of team charters, enhance alignment and mutual understanding during the initial stages of science team collaboration?  Which components of a team charter (e.g., roles and responsibilities, communication protocols, conflict resolution, team goals) have the greatest impact on the long-term success of science teams?  How does the inclusion of a detailed conflict resolution process in a team charter influence the ability of science teams to handle disagreements and maintain productivity?

# TABLE 3-1 Continued

	Research Questions
Shared goals and understanding	What are the key factors that facilitate the development of shared mental models in newly formed science teams, and how do these factors vary across interdisciplinary and single-discipline teams? How do cognitive process interventions (e.g., systems mapping, boundary objects) affect knowledge and behavioral coordination? How does the process of collaborative goal formulation and the negotiation of task interdependence during the early stages of team formation contribute to the emergence of shared mental models? How does the frequency and quality of team communication influence the development of transactive memory systems in science teams? What are the challenges stemming from differences in expertise that prevent the science team from developing shared cognitive states (e.g., shared mental models, transactive memory systems)?
Conceptualization	
Team communication	How does the quality of team communication (e.g., clarity, openness, frequency) influence the effectiveness of scientific collaboration in interdisciplinary science teams?  How do individual and organizational cultural and disciplinary differences affect communication patterns within interdisciplinary science teams, and what strategies are most effective in overcoming communication barriers?  How does long-term communication scaffolding between senior and junior team members (e.g., graduate students, postdoctoral researchers) foster the development of expertise and support the gradual integration of new knowledge into the team's research output?
Project design	How does the use of boundary objects or physical artifacts influence knowledge coordination (e.g., integrative team planning, project design) in science teams?  How does practicing epistemic humility among team members influence team processes (e.g., concept integration) and project outcomes (e.g., novel ideas) in interdisciplinary collaborations?  What strategies are most effective in promoting even participation among team members in larger or heterogeneous science teams?
Implementation	
Team meetings	How does the frequency and structure of team meetings influence knowledge coordination (e.g., shared mental models) and behavioral coordination (e.g., participative decision-making) in science teams?  How can team meetings best be leveraged to ensure that the methods and policies that set team norms and collaborative dynamics remain relevant as the team evolves?  What meeting strategies (e.g., turn-taking) promote team learning and integration of different perspectives in science teams?

continued

TABLE 3-1 Continued

	Research Questions
Project management	Does real-time resource-sharing (e.g., data, software, documents) impact science team collaborative processes such that the team works better together and is more efficient in developing shared cognitive states (e.g., shared mental models)?  What is the best strategy for employing regular team check-ins, where science team members discuss successes, failures, and unknowns, for virtual and hybrid teams to facilitate the development of team learning?  What type of training is needed to facilitate the use of project management software in science teams to ensure that it is the most effective?
Team debriefs	How does the use of structured team debriefs impact the long-term performance and learning outcomes of science teams, particularly in high-complexity or interdisciplinary projects?  What role does psychological safety play in enhancing the effectiveness of team debriefs in science teams, and how does it affect the willingness of members to share mistakes and areas for improvement?  To what extent should text or video recordings be included in team debriefs and to what extent does their use enhance feedback accuracy and increase the acceptance of feedback among science team members?
Translation	
Open science	How does the implementation of open science practices (e.g., sharing data and methodologies) influence the speed and breadth of research translation into societal benefits?  How does the use of open science practices foster interdisciplinary collaboration, and how does this collaboration contribute to more effective translation of research findings across different sectors?
Other	What strategies can science teams use to ensure that their findings are replicable and applicable across different societal, cultural, and geographic settings (i.e., external validity)?  How does the focus on external validity during the research process impact the development of team dynamics and the eventual translation of scientific findings?  At what stage is it best to include individuals directly impacted by the scientific research into the collaborative process to best facilitate the translation of findings?  What communication strategies are most effective for ensuring scientific findings are understood and utilized by nonacademic partners, such as policymakers and community leaders?  How does early planning for research translation influence the long-term success of translational efforts, and what are the key factors that contribute to effective planning across different phases of research? To what degree do translations goals need to be incorporated into ongoing team evaluations to best ensure that research findings are effectively moved into policy and practice?

TABLE 3-1 Continued

	Research Questions
Leadership	
Enabling conditions	How do science team leaders establish the enabling conditions for team effectiveness in a cross-disciplinary team context, and how quickly can these conditions be established?  Can the responsibility for setting and reinforcing behavioral norms be distributed among all members of a science team, rather than resting solely on the formal leader, and can this shared responsibility be established immediately after team formation, or does it need to develop over time?
Boundary-spanning	Which boundary-spanning behaviors are most critical for team success throughout the scientific process, and does the importance of these behaviors change depending on the team's current scientific phase? To what extent will leaders engage in boundary-spanning activities to align team goals with the broader scientific ecosystem, and how transparent will they be with their team about these efforts?
Shared leadership	How does shared or distributed leadership influence team dynamics, accountability, and effectiveness in large science teams with multiple co–principal investigators?  How does the rotating or shared leadership model in science teams affect leadership effectiveness, and what best practices can be developed to support this structure?
Collaborative Techn	ologies
Team configuration	How does the geographic distribution of science team members in hybrid and virtual science teams affect the use and effectiveness of virtual collaboration tools for sharing knowledge-related research materials? To what degree are cultural and language-based barriers to scientific collaborations in international science teams mitigated when using technological tools?
Technology-team interaction	What impact does exposure to and familiarity with virtual collaboration tools have on reducing anxiety and enhancing productivity in interdisciplinary scientific collaborations?  To what degree does establishing norms for using virtual collaboration tools influence the quality of communication, engagement in effective team processes, and development of important emergent states in hybrid and virtual science teams?
Tool accessibility	How does the accessibility of virtual collaboration tools, particularly for scientists with disabilities, affect their participation and contribution to team science projects?  What are the potential risks of data breaches or violations of privacy when science teams use virtual collaboration tools, and how can these risks be mitigated?
Adapting to changing configurations	What critical events determine when the use of technological tools needs to be revisited (e.g., adding new team members, changes in geographic distribution)?

TABLE 3-2 Summary of Best Practices for Science Teams

	Best Practice
Development	
Team assembly	<ul> <li>Conduct a team task analysis to determine what is needed for the project, including team size and needed knowledge, skills, and abilities</li> <li>Consider team composition, including teamwork and taskwork competencies of team members</li> <li>Recruit team members by articulating a compelling vision for the project</li> </ul>
Team onboarding and building	<ul> <li>Onboard in ways that increase both work-related and personal familiarity</li> <li>Design team-building activities that focus on goal-setting, interpersonal relationships, role clarification, and/or problem-solving</li> </ul>
Developing a shared language	<ul> <li>Ensure all team members have a clear understanding of how terminology will be used, particularly when terminology is used differently in different disciplines</li> <li>Use modes of communication that are accessible to all team members</li> </ul>
Conceptualization	
Team charters	<ul> <li>Use team charters to clarify team planning aspects such as goals, processes, roles, and timelines</li> <li>Include all team members in the development process of a team charter</li> <li>Treat team charters as living documents that require periodic review and revision</li> </ul>
Team planning and project design	<ul> <li>Frame team planning interventions as problem-solving</li> <li>Use boundary objects to support team understanding of a problem</li> <li>Promote open dialogue and the consideration of different perspectives during planning</li> <li>Allocate sufficient time to planning</li> </ul>
Implementation	
Project management	<ul> <li>Maximize the effectiveness of meetings by having a clear agenda, being mindful of time, and creating an environment where participants feel psychologically safe</li> <li>Use facilitators to optimize team meetings</li> <li>Foster structured dialogue between team members (e.g., implementing turn-taking during meetings)</li> <li>Make use of existing project management platforms and programs</li> </ul>
Team debriefs	• Periodically review and reflect on team performance and progress, including a discussion of both aspects that are going well and those that are going poorly
Translation	
Connecting with community members	Establish clear communication with community member partners on research aspects including roles and responsibilities

TABLE 3-2 Continued

	Best Practice
Cross-Cutting	
Technology	<ul> <li>Consider team configuration and how that may evolve when assessing virtual collaboration tools</li> <li>Ensure that technological tools align well with team member skills, needs, and comfort levels</li> <li>Select tools that are accessible to all team members, considering their individual and differing needs</li> <li>Establish clear expectations around use of technological tools</li> <li>Carefully craft data security and use protocols that consider aspects including accessibility</li> <li>Revisit technology norms and expectations periodically as teams evolve</li> </ul>
Team leadership	<ul> <li>Prioritize supporting team dynamics by establishing key enabling conditions for team effectiveness</li> <li>Provide team coaching and positive leadership styles</li> <li>Exercise inclusive leadership and limit power dynamics by practicing shared leadership</li> </ul>

#### CONCLUSION AND RECOMMENDATION

Conclusion 3-1: The following strategies offer the potential to improve science team performance and outcomes, if adapted to specific contexts and circumstances.

- Development stage: careful team assembly that reflects a task analysis, consideration of the team composition, attention to the orientation of new members, and development of a shared language.
- Conceptualization stage: development of a team charter, deliberative team planning and project design, and attention to a shared mental model of the team's work.
- Implementation stage: systematic project management, regular team debriefs to identify what went well and what went poorly at each stage and to determine the best ways for the next project stage.
- Translation stage: working with community members and attention to external as well as internal validity, that is, understanding of the generalizability of the research findings.

Recommendation 3-1: Research funders, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should provide resources enabling the study of team development, conceptualization, implementation, and translation.

#### REFERENCES

- Aaron, J. R., McDowell, W. C., & Herdman, A. O. (2014). The effects of a team charter on student team behaviors. *Journal of Education for Business*, 89(2), 90–97. https://doi.org/10.1080/08832323.2013.763753
- Abay, N. C., Zhou, Y., Kantarcioglu, M., Thuraisingham, B., & Sweeney, L. (2019). Privacy preserving synthetic data release using deep learning. In Machine learning and knowledge discovery in databases: European Conference, ECML PKDD 2018, Dublin, Ireland, September 10–14, 2018, Proceedings, Part I 18 (pp. 510–526). Springer International Publishing.
- Abrantes, A. C., Passos, A. M., Cunha, M. P. E., & Santos, C. M. (2022). Getting the knack for team-improvised adaptation: The role of reflexivity and team mental model similarity. *The Journal of Applied Behavioral Science*, 58(2), 281–315. https://doi.org/10.1177/00218863211009344
- Adler, P. S., & Chen, C. X. (2011). Combining creativity and control: Understanding individual motivation in large-scale collaborative creativity. *Accounting, Organizations and Society*, 36(2), 63–85. https://doi.org/10.1016/j.aos.2011.02.002
- Afful, H. Q., & Meiksin, J. (2022). US National Science Foundation opens technology directorate. MRS Bulletin, 47(11), 1074–1076. https://doi.org/10.1557/s43577-022-00450-y
- Aksnes, D. W., Piro, F. N., & Rørstad, K. (2019). Gender gaps in international research collaboration: A bibliometric approach. *Scientometrics*, 120, 747–774.
- Albaroudi, E., Mansouri, T., & Alameer, A. (2024). A comprehensive review of AI techniques for addressing algorithmic bias in job hiring. *AI*, 5(1), 383–404.
- Allen, J. A., & Rogelberg, S. G. (2013). Manager-led group meetings: A context for promoting employee engagement. *Group & Organization Management*, 38(5), 543–569. https://doi.org/10.1177/1059601113503040
- Allen, J. A., Reiter-Palmon, R., Crowe, J., & Scott, C. (2018). Debriefs: Teams learning from doing in context. *American Psychologist*, 73(4), 504–516. https://doi.org/10.1037/amp0000246
- Allen, J. A., Sands, S. J., Mueller, S. L., Frear, K. A., Mudd, M., & Rogelberg, S. G. (2012). Employees' feelings about more meetings: An overt analysis and recommendations for improving meetings. *Psychology Faculty Publications*, 95. https://digitalcommons. unomaha.edu/psychfacpub/95
- Alshawabkeh, A. A., Woolsey, M. L., & Kharbat, F. F. (2021). Using online information technology for deaf students during COVID-19: A closer look from experience. *Heliyon*, 7(5), e06915. https://doi.org/10.1016/j.heliyon.2021.e06915
- Americans with Disabilities Act. (2010). 2010 ADA Standards for Accessible Design. https://www.ada.gov/law-and-regs/design-standards/2010-stds/\_
- Ang, J. R. X., Liu, P., McDonnell, E., & Coppola, S. (2022). "In this online environment, we're limited": Exploring inclusive video conferencing design for signers. Proceedings of the 2022 CHI Conference on Human Factors in Computing Systems, New Orleans, LA. https://doi.org/10.1145/3491102.3517488
- Arslan, C., Chang, E. H., Chilazi, S., Bohnet, I., & Hauser, O. P. (2025). Behaviorally designed training leads to more diverse hiring. *Science*, 387(6732), 364–366. https://doi.org/doi:10.1126/science.ads5258
- Arthur Jr, W., Edwards, B. D., Bell, S. T., Villado, A. J., & Bennett Jr, W. (2005). Team task analysis: Identifying tasks and jobs that are team based. *Human Factors*, 47(3), 654–669. https://doi.org/10.1518/001872005774860087
- Asencio, R., Carter, D. R., DeChurch, L. A., Zaccaro, S. J., & Fiore, S. M. (2012). Charting a course for collaboration: A multiteam perspective. *Translational Behavioral Medicine*, 2(4), 487–494. https://doi.org/10.1007/s13142-012-0170-3

Audi, R. (2010). Epistemology: A contemporary introduction to the theory of knowledge (3rd ed.). Routledge.

- Bacchini, F., & Lorusso, L. (2019). Race, again: How face recognition technology reinforces racial discrimination. *Journal of Information, Communication and Ethics in Society*, 17(3), 321–335.
- Bammer, G. (2013). Disciplining interdisciplinarity: Integration and implementation sciences for researching complex real-world problems. ANU Press.
- Barrick, M. R., Stewart, G. L., Neubert, M. J., & Mount, M. K. (1998). Relating member ability and personality to work-team processes and team effectiveness. *Journal of Applied Psychology*, 83(3), 377–391. https://doi.org/10.1037/0021-9010.83.3.377
- Bass, B. M. (1999). Two decades of research and development in transformational leadership. European Journal of Work and Organizational Psychology, 8(1), 9–32. https://doi. org/10.1080/135943299398410
- Beechey, K. (2024). How can researchers share sensitive data openly? *Gates Open Research*. https://blog.gatesopenresearch.org/2024/02/22/how-can-researchers-share-sensitive-data-openly/
- Begerowski, S. R., Traylor, A. M., Shuffler, M. L., & Salas, E. (2021). An integrative review and practical guide to team development interventions for translational science teams: One size does not fit all. *Journal of Clinical and Translational Science*, 5(1), e198. https://doi.org/10.1017/cts.2021.832
- Behar-Horenstein, L. S., & Prikhidko, A. (2017). Exploring mentoring in the context of team science. *Mentor Tutoring*, 25(4), 430–454. https://doi.org/10.1080/13611267.2017.1403579
- Bell, S. T. (2007). Deep-level composition variables as predictors of team performance: A metaanalysis. *Journal of Applied Psychology*, 92(3), 595–615.
- Bell, S. T., Brown, S. G., Colaneri, A., & Outland, N. (2018). Team composition and the ABCs of teamwork. *American Psychologist*, 73(4), 349–362.
- Bell, S. T., Villado, A. J., Lukasik, M. A., Belau, L., & Briggs, A. L. (2011). Getting specific about demographic diversity variable and team performance relationships: A meta-analysis. *Journal of Management*, 37(3), 709–743. https://doi.org/10.1177/0149206310365001
- Bennett, L. M., & Gadlin, H. (2012). Collaboration and team science: From theory to practice. *Journal of Investigative Medicine*, 60(5), 768–775. https://doi.org/10.2310/JIM.0b013e318250871d
- Bennett, L. M., Gadlin, H., & Marchand, C. (2018). Collaboration team science: Field guide. U.S. Department of Health & Human Services and National Institutes of Health, National Cancer Institute.
- Bennett, L. M., Maraia, R., & Gadlin, H. (2014). The 'welcome letter': A useful tool for laboratories and teams. *Journal of Translational Medicine & Epidemiology*, 2(2), 1035.
- Bens, I. (2012). Facilitating with ease! Core skills for facilitators, team leaders and members, managers, consultants, and trainers (3rd ed). Jossey-Bass.
- Berengüí, R., Carralero, R., Castejón, M. A., Campos-Salinas, J. A., & Cantón, E. (2021). Values, motivational orientation and team cohesion amongst youth soccer players. *International Journal of Sports Science & Coaching*, 17(5), 1049–1058. https://doi.org/10.1177/17479541211055690
- Berntzen, M., Stray, V., & Moe, N. B. (2021). Coordination strategies: Managing inter-team coordination challenges in large-scale agile. *Agile Processes in Software Engineering and Extreme Programming*.
- Bietz, M. J., Abrams, S., Cooper, D. M., Stevens, K. R., Puga, F., Patel, D. I., Olson, G. M., & Olson, J. S. (2012). Improving the odds through the Collaboration Success Wizard. *Translational Behavioral Medicine*, 2(4), 480–486. https://doi.org/10.1007/s13142-012-0174-z
- Bisbey, T., Traylor, A., & Salas, E. (2021a). Transforming teams of experts into expert teams: Eight principles of expert team performance. *Journal of Expertise*, 4(2), 190–207.

- Bisbey, T. M., Wooten, K. C., Salazar Campo, M., Lant, T. K., & Salas, E. (2021b). Implementing an evidence-based competency model for science team training and evaluation: TeamMAPPS. *Journal of Clinical and Translational Science*, 5(1), e142. https://doi.org/10.1017/cts.2021.795
- Boix Mansilla, V. (2006). Assessing expert interdisciplinary work at the frontier: An empirical exploration. *Research Evaluation*, 15(1), 17–29. https://doi.org/10.3152/147154406781776075
- ... (2017). Interdisciplinary learning: A cognitive-epistemological foundation. In R. Frodeman (Ed.), The Oxford handbook of interdisciplinary (2nd ed., pp. 261–275) https://doi.org/10.1093/oxfordhb/9780198733522.013.22
- Braithwaite, J., Churruca, K., Long, J. C., Ellis, L. A., & Herkes, J. (2018). When complexity science meets implementation science: A theoretical and empirical analysis of systems change. *BMC Medicine*, 16, 1–14.
- Brasier, A. R., Burnside, E. S., & Rolland, B. (2023a). Competencies supporting high-performance translational teams: A review of the SciTS evidence base. *Journal of Clinical and Translational Science*, 7(1), e62. https://doi.org/10.1017/cts.2023.17
- Brasier, A. R., Casey, S. L., Hatfield, P., Kelly, P. W., Sweeney, W. A., Schweizer, M., Liu, B., & Burnside, E. S. (2023b). A leadership model supporting maturation of high-performance translational teams. *Journal of Clinical and Translational Science*, 7(1), e171. https://doi.org/10.1017/cts.2023.598
- Braun, F., & Avital, M. (2006). The role of accountability in motivating knowledge sharing among team members in information technology projects. In *Americas Conference on Information Systems* 2006 proceedings (p. 454). AIS Electronic Library.
- Brister, E. (2016). Disciplinary capture and epistemological obstacles to interdisciplinary research: Lessons from central African conservation disputes. *Studies in History and Philosophy of Science Part C: Studies in History and Philosophy of Biological and Biomedical Sciences*, 56, 82–91. https://doi.org/10.1016/j.shpsc.2015.11.001
- Brownell, S. E., & Tanner, K. D. (2017). Barriers to faculty pedagogical change: Lack of training, time, incentives, and...tensions with professional identity? *CBE—Life Sciences Education*, 11(4), 339–346. https://doi.org/10.1187/cbe.12-09-0163
- Brucks, M. S., & Levav, J. (2022). Virtual communication curbs creative idea generation. *Nature*, 605(7908), 108–112. https://doi.org/10.1038/s41586-022-04643-y
- Buolamwini, J., & Gebru, T. (2018). Gender shades: Intersectional accuracy disparities in commercial gender classification. In *Proceedings of the 1st Conference on Fairness, Accountability and Transparency, Proceedings of Machine Learning Research*. https://proceedings.mlr.press/v81/buolamwini18a.html
- Byrd, J. T., & Luthy, M. R. (2010). Improving group dynamics: Creating a team charter. *Academy of Educational Leadership Journal*, 14(1), 13–26.
- Cady, S. H., Michelle, B., & Parker, N. (2018). When a team is more like a group: Improving individual motivation by managing integrity through team action processes. *Public Integrity*, 21(1), 86–103. https://doi.org/10.1080/10999922.2017.1419052
- Calabro, S. M. (2018). From the message board to the front door: Addressing the offline consequences of race- and gender-based doxxing and swatting. *Suffolk University Law Review*, 55.
- Cannon-Bowers, J. A., & Salas, E. (1998). Individual and team decision making under stress: Theoretical underpinnings. In J. A. Cannon-Bowers & E. Salas (Eds.), *Making decisions under stress: Implications for individual and team training* (pp. 17–38). American Psychological Association. https://doi.org/10.1037/10278-001
- Cannon-Bowers, J. A., Tannenbaum, S. I., Salas, E., & Volpe, C. E. (1995). Defining competencies and establishing team training requirements. In R. Guzzo, E. Salas, & Associates (Eds.), Team effectiveness and decision making in organizations (pp. 333–380). Jossey-Bass.

Carter, D. R., Asencio, R., Trainer, H. M., DeChurch, L. A., Kanfer, R., & Zaccaro, S. J. (2019). Best practices for researchers working in multiteam systems. In K. Hall, A. Vogel, & R. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 391–400). https://doi.org/10.1007/978-3-030-20992-6\_29

- Carter, N. T., Carter, D. R., & DeChurch, L. A. (2018). Implications of observability for the theory and measurement of emergent team phenomena. *Journal of Management*, 44(4), 1398–1425. https://doi.org/10.1177/0149206315609402
- Carton, A. M. (2022). The science of leadership: A theoretical model and research agenda. Annual Review of Organizational Psychology and Organizational Behavior, 9, 61–93.
- Cason, M., Sessions, L. C., Nemeth, L., Catchpole, K., & Bundy, D. G. (2020). Components of team science—What contributes to success? *Journal of Interprofessional Education & Practice*, 18, 100298. https://doi.org/10.1016/j.xjep.2019.100298
- Castillo, F., Pearson, Y., Frizell, S., Skaggs, S., & Bisbey, T. (2024). Understanding organizational cultural influences in multisector multi-team systems. Paper presented at 2024 ASEE Annual Conference & Exposition, Portland, Oregon. https://doi.org/10.18260/1-2--48193
- Chen, C., Xing, Z., & Liu, Y. (2017). By the community & for the community: A deep learning approach to assist collaborative editing in Q&A sites. *Proceedings of the ACM on Human-Computer Interaction*, 1(CSCW), 1–21.
- Chen, J., Bamberger, P. A., Song, Y., & Vashdi, D. R. (2018). The effects of team reflexivity on psychological well-being in manufacturing teams. *Journal of Applied Psychology*, 103(4), 443–462.
- Colbert, A., Yee, N., & George, G. (2016). The digital workforce and the workplace of the future. *Academy of Management Journal*, 59(3), 731–739. https://doi.org/10.5465/amj.2016.4003
- Collier, L. (2014). Defending animal research. *Monitor on Psychology*, 45(7), 40. https://www.apa.org/monitor/2014/07-08/defending-research
- Corbacho, A. M., Minini, L., Pereyra, M., González-Fernández, A. E., Echániz, R., Repetto, L., Cruz, P., Fernández-Damonte, V., Lorieto, A., & Basile, M. (2021). Interdisciplinary higher education with a focus on academic motivation and teamwork diversity. *International Journal of Educational Research Open*, 2, Article 100062. https://doi.org/10.1016/j.ijedro.2021.100062
- Courtright, S. H., McCormick, B. W., Mistry, S., & Wang, J. (2017). Quality charters or quality members? A control theory perspective on team charters and team performance. *Journal of Applied Psychology*, 102(10), 1462–1470. https://doi.org/10.1037/apl0000229
- Cravens, A. E., Jones, M. S., Ngai, C., Zarestky, J., & Love, H. B. (2022). Science facilitation: Navigating the intersection of intellectual and interpersonal expertise in scientific collaboration. *Humanities and Social Sciences Communications*, 9, Article 256. https://doi.org/10.1057/s41599-022-01217-1
- Dastin, J. (2018). Insight Amazon scraps secret AI recruiting tool that showed bias against women. *Reuters*. https://www.reuters.com/article/us-amazon-com-jobs-automation-insight/amazon-scraps-secret-ai-recruiting-tool-that-922showed-bias-against-women-idUSKCN1MK08G/
- De Jong, B. A., Dirks, K. T., & Gillespie, N. (2016). Trust and team performance: A metaanalysis of main effects, moderators, and covariates. *Journal of Applied Psychology*, 101(8), 1134–1150. https://doi.org/10.1037/apl0000110
- Dening, L., Xiaoyan, L., & Shuyi, W. (2022). A literature review of diffusion of responsibility phenomenon. *Proceedings of the 2022 8th International Conference on Humanities and Social Science Research*.

- Department of the Army. (1993). A leaders guide to after-action reviews (Training Circular No. 25–20).
- DeRue, D. S., Nahrgang, J. D., Hollenbeck, J. R., & Workman, K. (2012). A quasi-experimental study of after-event reviews and leadership development. *Journal of Applied Psychology*, 97, 997–1015. http://dx.doi.org/ 10.1037/a0028244
- Doush, I. A., Al-Jarrah, A., Alajarmeh, N., & Alnfiai, M. (2023). Learning features and accessibility limitations of video conferencing applications: Are people with visual impairment left behind. *Universal Access in the Information Society*, 22(4), 1353–1368. https://doi.org/10.1007/s10209-022-00917-4
- Duff, J. P., Morse, K. J., Seelandt, J., Gross, I. T., Lydston, M., Sargeant, J., Dieckmann, P., Allen, J. A., Rudolph, J. W., & Kolbe, M. (2024). Debriefing methods for simulation in healthcare: A systematic review. Simulation in Healthcare, 19(1S), S112–S121.
- Eddy, E. R., Tannenbaum, S. I., & Mathieu, J. E. (2013). Helping teams to help themselves: Comparing two team-led debriefing methods. *Personnel Psychology*, 66(4), 975–1008. https://doi.org/10.1111/peps.12041
- Edmondson, A. C., & Lei, Z. (2014). Psychological safety: The history, renaissance, and future of an interpersonal construct. *Annual Review of Organizational Psychology and Organizational Behavior*, 1, 23–43. https://doi.org/https://doi.org/10.1146/annurevorgpsych-031413-091305
- Egeland, T., & Schei, V. (2015). "Cut me some slack": The psychological contracts as a foundation for understanding team charters. *The Journal of Applied Behavioral Science*, 51(4), 451–478. https://doi.org/10.1177/0021886314566075
- Egeland T., Schei, V., & Tjølsen, Ø. A. (2017). Expecting the unexpected: Using team charters to handle disruptions and facilitate team performance. *Group Dynamics: Theory, Research, and Practice*, 21(1), 53–59. http://dx.doi.org/10.1037/gdn0000059
- Falcone, M., Loughead, J., & Lerman, C. (2019). The integration of research from diverse fields: Transdisciplinary approaches bridging behavioral research, cognitive neuroscience, pharmacology, and genetics to reduce cancer risk behavior. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 69–80). Springer, Cham.
- Faraj, S., & Sproull, L. (2000). Coordinating expertise in software development teams. Management Science, 46(12), 1554–1568. https://doi.org/10.1287/mnsc.46.12.1554.12072
- Fiore, S. M., & Schooler, J. W. (2004). Process mapping and shared cognition: Teamwork and the development of shared problem models. In E. Salas & S. M. Fiore (Eds.), *Team cognition: Understanding the factors that drive process and performance* (pp. 133–152). American Psychological Association. https://doi.org/10.1037/10690-007
- Fiore, S. M., & Wiltshire, T. J. (2016). Technology as teammate: Examining the role of external cognition in support of team cognitive processes. *Frontiers in Psychology*, 7, 1531.
- Fiore, S. M., Carter, D. R., & Asencio, R. (2015). Conflict, trust, and cohesion: Examining affective and attitudinal factors in science teams. In E. Salas, A. X. Estrada, & W. B. Vessey (Eds.), *Team cohesion: Advances in psychological theory, methods and practice* (pp. 271–301). Emerald Group Publishing Limited.
- Fiore, S. M., Gabelica, C., Wiltshire, T. J., & Stokols, D. (2019). Training to be a (team) scientist. In K. L., Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 421–444). Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6\_33
- Fiore, S. M., Rosen, M. A., Smith-Jentsch, K. A., Salas, E., Letsky, M., & Warner, N. (2010). Toward an understanding of macrocognition in teams: Predicting processes in complex collaborative contexts. *Human Factors*, 52(2), 203–224.

Forscher, P. S., Wagenmakers, E.-J., Coles, N. A., Silan, M. A., Dutra, N., Basnight-Brown, D., & IJzerman, H. (2023). The benefits, barriers, and risks of big-team science. *Perspectives on Psychological Science*, 18(3), 607–623. https://doi.org/10.1177/17456916221082970

- Fox, C. W., Ritchey, J. P., & Paine, C. T. (2018). Patterns of authorship in ecology and evolution: First, last, and corresponding authorship vary with gender and geography. *Ecology and Evolution*, 8(23), 11492–11507.
- Frazier, M. L., Fainshmidt, S., Klinger, R. L., Pezeshkan, A., & Vracheva, V. (2017). Psychological safety: A meta-analytic review and extension. *Personnel Psychology*, 70(1), 113–165. https://doi.org/10.1111/peps.12183
- Gaffney, S. G., Ad, O., Smaga, S., Schepartz, A., & Townsend, J. P. (2019). GEM-NET: Lessons in multi-institution teamwork using collaboration software. ACS Central Science, 5(7), 1159–1169. https://doi.org/10.1021/acscentsci.9b00111
- Gehlert, S., Carothers, B. J., Lee, J. A., Gill, J., Luke, D., & Colditz, G. (2015). A social network analysis approach to diagnosing and improving the functioning of transdisciplinary teams in public health. *Transdisciplinary Journal of Engineering & Science*, 6, 11–22. https://doi.org/10.22545/2015/00070
- Gibson, C. B., Dunlop, P. D., Majchrzak, A., & Chia, T. (2022). Sustaining effectiveness in global teams: The coevolution of knowledge management activities and technology affordances. *Organization Science*, 33(3), 1018–1048. https://doi.org/10.1287/orsc.2021.1478
- Gibson, C. B., Huang, L., Kirkman, B. L., & Shapiro, D. L. (2014). Where global and virtual meet: The value of examining the intersection of these elements in twenty-first-century teams. *Annual Review of Organizational Psychology and Organizational Behavior*, 1, 217–244. https://doi.org/10.1146/annurev-orgpsych-031413-091240
- Ginns, P., Prosser, M., & Barrie, S. (2007). Students' perceptions of teaching quality in higher education: The perspective of currently enrolled students. *Studies in Higher Education*, 32(5), 603–615. https://doi.org/10.1080/03075070701573773
- Glasgow, R. E., & Emmons, K. M. (2007). How can we increase translation of research into practice? Types of evidence needed. *Annual Review of Public Health*, 28(1), 413–433.
- Goldshtein, M., Chiou, E. K., & Roscoe, R. D. (2024). 'I just don't trust them': Reasons for distrust and non-disclosure in demographic questionnaires for individuals in STEM. Societies, 14(7), 105. https://doi.org/10.3390/soc14070105
- Gosselin, D. C., Thompson, K., Pennington, D., & Vincent, S. (2020). Learning to be an interdisciplinary researcher: Incorporating training about dispositional and epistemological differences into graduate student environmental science teams. *Journal of Environmental Studies and Sciences*, 10, 310–326. https://doi.org/10.1007/s13412-020-00605-w
- Graef, D., J., Kramer, J. G., & Motzer, N. (2021a). Facilitating interdisciplinary meetings: A practical guide. National Socio-Environmental Synthesis Center. https://www.sesync.org/sites/default/files/2022-01/Facilitating%20Interdisciplinary%20Meetings%20Guide.pdf
- Graef, D. J., Motzer, N., & Kramer, J. G. (2021b). The value of facilitation in interdisciplinary socio-environmental team research. Socio-Ecological Practice Research, 3(2), 109–113. https://doi.org/10.1007/s42532-021-00082-7
- Graff, D., & Clark, M.A. (2018). Clear as a bell: The influence of analogies on the development of cross-understanding in design teams. *Team Performance Management*, 24(7/8), 396–410. https://doi.org/10.1108/TPM-04-2018-0028
- Griffin, A. S., & Guez, D. (2014). Innovation and problem solving: A review of common mechanisms. *Behavioural Processes*, 109, 121–134. https://doi.org/https://doi.org/10.1016/j. beproc.2014.08.027
- Grindstaff, K., & Mascarenhas, M. (2019). "No one wants to believe it": Manifestations of white privilege in a STEM-focused college. *Multicultural Perspectives*, 21(2), 102–111. https://doi.org/10.1080/15210960.2019.1572487

- Guise, J. M., Nagel, J. D., & Regensteiner, J. G. (2012). Best practices and pearls in interdisciplinary mentoring from Building Interdisciplinary Research Careers in Women's Health Directors. *Journal of Women's Health*, 21(11), 1114–1127. https://doi.org/10.1089/ jwh.2012.3788
- Guise, J. M., Winter, S., Fiore, S. M., Regensteiner, J. G., & Nagel, J. (2017). Organizational and training factors that promote team science: A qualitative analysis and application of theory to the NIH's BIRCWH career development program. *Journal of Clinical and Translational Science*, 1(2), 101–107. https://doi.org/10.1017/cts.2016.17
- Hackman, J. R. (2012). From causes to conditions in group research. *Journal of Organizational Behavior*, 33(3), 428–444. https://doi.org/10.1002/job.1774
- Hackman, J. R., & Wageman, R. (2005). A theory of team coaching. *The Academy of Management Review*, 30(2), 269–287. https://doi.org/10.5465/amr.2005.16387885
- Hall, K. L., Stokols, D., Moser, R. P., Taylor, B. K., Thornquist, M. D., Nebeling, L. C., Ehret, C. C., Barnett, M. J., McTiernan, A., Berger, N. A., Goran, M. I., & Jeffery, R. W. (2008). The collaboration readiness of transdisciplinary research teams and centers: Findings from the National Cancer Institute's TREC year-one evaluation study. *American Journal of Preventive Medicine*, 35(2), \$161–\$172. https://doi.org/10.1016/j.amepre.2008.03.035
- Hall, K. L., Vogel, A. L., & Crowston, K. (2019). Comprehensive collaboration plans: Practical considerations spanning across individual collaborators to institutional supports. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 587–612). Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6
- Hall, K. L., Vogel, A. L., Huang, G. C., Serrano, K. J., Rice, E. L., Tsakraklides, S. P., & Fiore, S. M. (2018). The science of team science: A review of the empirical evidence and research gaps on collaboration in science. *American Psychologist*, 73(4), 532–548. https://doi.org/10.1037/amp0000319
- Hall, K. L., Vogel, A. L., Stipelman, B. A., Stokols, D., Morgan, G., & Gehlert, S. (2012). A four-phase model of transdisciplinary team-based research: Goals, team processes, and strategies. *Translational Behavioral Medicine*, 2(4), 415–430. https://doi.org/10.1007/s13142-012-0167-y
- Handke, L., Aldana, A., Costa, P. L., & O'Neill, T. A. (2024). Hybrid teamwork: What we know and where we can go from here. Small Group Research, 55(5), 805–835. https://doi.org/10.1177/10464964241279078
- Harvey, J.-F., Bresman, H., Edmondson, A. C., & Pisano, G. P. (2022). A strategic view of team learning in organizations. Academy of Management Annals, 16(2). https://doi. org/10.5465/annals.2020.0352
- Hawk, L. W., Murphy, T. F., Hartmann, K. E., Burnett, A., & Maguin, E. (2024). A randomized controlled trial of a team science intervention to enhance collaboration readiness and behavior among early career scholars in the Clinical and Translational Science Award network. *Journal of Clinical and Translational Science*, 8(1), e6. https://doi.org/10.1017/cts.2023.692
- Henson, V. R., Cobourn, K. M., Weathers, K. C., Carey, C. C., Farrell, K. J., Klug, J. L., Sorice, M. G., Ward, N. K., & Weng, W. (2020). A practical guide for managing interdisciplinary teams: Lessons learned from coupled natural and human systems research. *Social Sciences*, 9(7), 119. https://doi.org/10.3390/socsci9070119
- Hersh, M., Leporini, B., & Buzzi, M. (2024). A comparative study of disabled people's experiences with the video conferencing tools Zoom, MS Teams, Google Meet and Skype. *Behaviour & Information Technology*, 43(15), 1–20. https://doi.org/10.1080/01 44929X.2023.2286533

Hoever, I. J., van Knippenberg, D., van Ginkel, W. P., & Barkema, H. G. (2012). Fostering team creativity: Perspective taking as key to unlocking diversity's potential. *Journal of Applied Psychology*, 97(5), 982–996. https://doi.org/10.1037/a0029159

- Hoff, K. A., & Bashir, M. (2015). Trust in automation: Integrating empirical evidence on factors that influence trust. *Human Factors*, 57(3), 407–434. https://doi.org/10.1177/0018720814547570
- Holcomb, L. B., King, F. B., & Brown, S. W. (2004). Student traits and attributes contributing to success in online courses: Evaluation of university online courses. *The Journal of Interactive Online Learning*, 2(3), 1–17.
- Horwitz, S. K., & Horwitz, I. B. (2007). The effects of team diversity on team outcomes: A meta-analytic review of team demography. *Journal of Management*, 33(6), 987–1015. https://doi.org/10.1177/0149206307308587
- Huang, J., Gates, A. J., Sinatra, R., & Barabási, A. L. (2020). Historical comparison of gender inequality in scientific careers across countries and disciplines. *Proceedings of the National Academy of Sciences*, 117(9), 4609–4616.
- Hubbs, G., O'Rourke, M., & Orzack, S. H. (Eds.). (2020). The Toolbox Dialogue Initiative: The power of cross-disciplinary practice. CRC Press.
- Huebschmann, A. G., Leavitt, I. M., & Glasgow, R. E. (2019). Making health research matter: A call to increase attention to external validity. *Annual Review of Public Health*, 40(1), 45–63.
- Hutchins, E. (1999). Cognitive artifacts. In F. C. Keil (Ed.), MIT encyclopedia of the cognitive sciences. MIT Press.
- Ingersoll, B., Espinel, A., Nauman, J., Broder-Fingert, S., Carter, A. S., Sheldrick, R. C., Stone, W. L., & Wainer, A. L. (2024). Using virtual multiteam systems to conduct a multisite randomized clinical trial in the part C early intervention system: Benefits, challenges, and lessons learned. Contemporary Clinical Trials, 143, 107585. https://doi.org/10.1016/j.cct.2024.107585
- Jackson, G. (2022). The AI-enhanced future of health care administrative task management. NEJM Catalyst Innovations in Care Delivery.
- Jadidi, M., Karimi, F., Lietz, H., & Wagner, C. (2018). Gender disparities in science? Dropout, productivity, collaborations and success of male and female computer scientists. Advances in Complex Systems, 21(03n04), 1750011. https://doi.org/10.1142/S0219525917500114
- Jarrahi, M. H., Askay, D., Eshraghi, A., & Smith, P. (2023). Artificial intelligence and knowledge management: A partnership between human and AI. *Business Horizons*, 66(1), 87–99. https://doi.org//https://doi.org/10.1016/j.bushor.2022.03.002
- Jehn, K. A. (1997). A qualitative analysis of conflict types and dimensions in organizational groups. Administrative Science Quarterly, 42(3), 530–557. https://doi.org/10.2307/2393737
- Jeske, D., & Olson, D. (2024). Silo mentality in teams: Emergence, repercussions and recommended options for change. *Journal of Work-Applied Management*.
- Johnson, M., Schuster, M., Le, Q. V., Krikun, M., Wu, Y., Chen, Z., Thorat, N., Viégas, F., Wattenberg, M., Corrado, G., Hughes, M., & Dean, J. (2017). Google's multilingual neural machine translation system: Enabling zero-shot translation. *Transactions of the Association for Computational Linguistics*, 5, 339–351. https://doi.org/10.1162/ tacl\_a\_00065
- Johnson, W. H., Baker, D. S., Dong, L., Taras, V., & Wankel, C. (2022). Do team charters help team-based projects? The effects of team charters on performance and satisfaction in global virtual teams. *Academy of Management Learning & Education*, 21(2), 236–260. https://doi.org/10.5465/amle.2020.0332
- Jones, B. D. (2009). Motivating students to engage in learning: The music model of academic motivation. *International Journal of Teaching and Learning in Higher Education*, 21(2), 272–285.

- Joshi, A., & Roh, H. (2009). The role of context in work team diversity research: A meta-analytic review. Academy of Management Journal, 52(3), 599–627. https://doi. org/10.5465/amj.2009.41331491
- Kaner, S. (2014). Facilitator's guide to participatory decision-making (3rd ed.). Jossey-Bass.
- Kasi, V., Keil, M., Mathiassen, L., & Pedersen, K. (2008). The post mortem paradox: A Delphi study of IT specialist perceptions. *European Journal of Information Systems*, 17(1), 62–78. https://doi.org/10.1057/palgrave.ejis.3000727
- Keiser, N. L., & Arthur Jr, W. (2021). A meta-analysis of the effectiveness of the after-action review (or debrief) and factors that influence its effectiveness. *Journal of Applied Psychology*, 106(7), 1007–1032. https://doi.org/10.1037/apl0000821
- ... (2022). A meta-analysis of task and training characteristics that contribute to or attenuate the effectiveness of the after-action review (or debrief). *Journal of Business and Psychology*, 37(5), 953–976. https://doi.org/10.1007/s10869-021-09784-x
- Kilcullen, M., Bisbey, T. M., Rosen, M., & Salas, E. (2023). Does team orientation matter? A state-of-the-science review, meta-analysis, and multilevel framework. *Journal of Organizational Behavior*, 44(2), 355–375. https://doi.org/10.1002/job.2622
- Kilcullen, M., Feitosa, J., & Salas, E. (2022). Insights from the virtual team science: Rapid deployment during COVID-19. *Human Factors*, 64(8), 1429–1440. https://doi.org/10.1177/0018720821991678
- Kirkman, B. L., & Mathieu, J. E. (2005). The dimensions and antecedents of team virtuality. Journal of Management, 31(5), 700–718. https://doi.org/10.1177/0149206305279113
- Kirkman, B. L., & Stoverink, A. C. (2021). Building resilient virtual teams. *Organizational Dynamics*, 50(1), 100825. https://doi.org/10.1016/j.orgdyn.2020.100825
- Kirkman, B. L., Rosen, B., Gibson, C. B., Tesluk, P. E., & McPherson, S. O. (2002). Five challenges to virtual team success: Lessons from Sabre, Inc. *Academy of Management Perspectives*, 16(3), 67–79. https://doi.org/10.5465/ame.2002.8540322
- Klein, C., DiazGranados, D., Salas, E., Le, H., Burke, C. S., Lyons, R., & Goodwin, G. F. (2009). Does team building work? Small Group Research, 40(2), 181–222. https://doi.org/10.1177/1046496408328821
- Knott, E., Rao, A. H., Summers, K., & Teeger, C. (2022). Interviews in the social sciences. Nature Reviews Methods Primers, 2, Article 73. https://doi.org/10.1038/s43586-022-00150-6
- Kolbe, M., Eppich, W., Rudolph, J., Meguerdichian, M., Catena, H., Cripps, A., Grant, V., & Cheng, A. (2020). Managing psychological safety in debriefings: A dynamic balancing act. BMJ Simulation and Technology Enhanced Learning, 6(3), 164–171. https://doi.org/10.1136/bmjstel-2019-000470
- Kotarba, J. A., Molldrem, S., Smith, E., Spratt, H., Bhavnani, S. K., Farroni, J. S., & Wooten, K. (2023). Exploring team dynamics during the development of a multi-institutional cross-disciplinary translational team: Implications for potential best practices. *Journal of Clinical and Translational Science*, 7(1), e220. https://doi.org/10.1017/cts.2023.640
- Kozlowski, S. W. J., & Chao, G. T. (2018). Unpacking team process dynamics and emergent phenomena: Challenges, conceptual advances, and innovative methods. *American Psychologist*, 73(4), 576–592. https://doi.org/10.1037/amp0000245
- Kramer, T. J., Fleming, G. P., & Mannis, S. M. (2001). Improving face-to-face brainstorming through modeling and facilitation. *Small Group Research*, 32(5), 533–557. https://doi.org/10.1177/104649640103200502
- Kwiek, M., & Roszka, W. (2021). Gender disparities in international research collaboration: A study of 25,000 university professors. *Journal of Economic Surveys*, 35(5), 1344–1380.
- \_\_\_\_. (2022). Are female scientists less inclined to publish alone? The gender solo research gap. *Scientometrics*, 127(4), 1697–1735.

Lacerenza, C. N., Marlow, S. L., Tannenbaum, S. I., & Salas, E. (2018). Team development interventions: Evidence-based approaches for improving teamwork. *American Psychologist*, 73(4), 517–531. https://doi.org/10.1037/amp0000295

- Larson, L., & DeChurch, L. A. (2020). Leading teams in the digital age: Four perspectives on technology and what they mean for leading teams. *The Leadership Quarterly*, 31(1), 101377. https://doi.org/10.1016/j.leaqua.2019.101377
- Lau, D. C., & Murnighan, J. K. (1998). Demographic diversity and faultlines: The compositional dynamics of organizational groups. The Academy of Management Review, 23(2), 325–340. https://doi.org/10.5465/amr.1998.533229
- Law, S. K. (2023). Barriers experienced by individuals with IDD when interacting with digital technology and online content. OCAD University. https://openresearch.ocadu.ca/id/eprint/4126/1/Law\_Sandra\_2023\_MDes\_INCD\_MRP.pdf
- Lee, N. T, Resnick, P., & Barton, G. (2019) Algorithmic bias detection and mitigation: Best practices and policies to reduce consumer harms. *Brookings Institute*. https://www.brookings.edu/articles/algorithmic-bias-detection-and-mitigation-best-practices-and-policies-to-reduce-consumer-harms/
- Leporini, B., Buzzi, M., & Hersh, M. (2023). Video conferencing tools: Comparative study of the experiences of screen reader users and the development of more inclusive design guidelines. ACM Transactions on Accessible Computing, 16(1), 1–36. https://doi.org/10.1145/3573012
- Levine, J. M., Choi, H.-S., & Moreland, R. L. (2003). Newcomer innovation in work teams. In P. Paulus & B. Nijstad (Eds.), Group creativity: Innovation through collaboration (pp. 202–224). Oxford University Press. https://doi.org/10.1093/acprof: oso/9780195147308.003.0010
- Lewis, P., Perez, E., Piktus, A., Petroni, F., Karpukhin, V., Goyal, N., Küttler, H., Lewis, M., Yih, W-t., Rocktäschel, T., Riedel, S., & Kiela, D. (2020). Retrieval-augmented generation for knowledge-intensive NLP tasks. Advances in Neural Information Processing Systems, 33, 9459–9474. https://proceedings.neurips.cc/paper/2020/file/6b493230205f780e1bc26945d f7481e5-Paper.pdf
- Liao, W. M., Finley, D. R., & Shafer, W. E. (2004). Effects of responsibility and cohesiveness on group escalation decisions. *Advances in Management Accounting*, 13, 245–259.
- Liu, F., Holme, P., Chiesa, M., AlShebli, B., & Rahwan, T. (2023a). Gender inequality and self-publication are common among academic editors. *Nature Human Behaviour*, 7(3), 353–364.
- Liu, M., Jaiswal, A., Bu, Y., Min, C., Yang, S., Liu, Z., Acuña, D., & Ding, Y. (2022). Team formation and team impact: The balance between team freshness and repeat collaboration. *Journal of Informetrics*, 16(4), 101337. https://doi.org/10.1016/j.joi.2022.101337
- Liu, Y., Song, Y., Trainer, H., Carter, D., Zhou, L., Wang, Z., & Chiang, J. T. (2023b). Feeling negative or positive about fresh blood? Understanding veterans' affective reactions toward newcomer entry in teams from an affective events perspective. *Journal of Applied Psychology*, 108(5), 728–749. https://doi.org/10.1037/apl0001044
- Love, H. B., Cross, J. E., Fosdick, B., Crooks, K. R., VandeWoude, S., & Fisher, E. R. (2021). Interpersonal relationships drive successful team science: An exemplary case-based study. *Humanities and Social Sciences Communications*, 8, 106. https://doi.org/10.1057/s41599-021-00789-8
- Love, H. B., Fosdick, B. K., Cross, J. E., Suter, M., Egan, D., Tofany, E., & Fisher, E. R. (2022). Towards understanding the characteristics of successful and unsuccessful collaborations: A case-based team science study. *Humanities and Social Sciences Communications*, 9, Article 371. https://doi.org/10.1057/s41599-022-01388-x

- Lungeanu, A., Carter, D. R., DeChurch, L. A., & Contractor, N. S. (2018). How team interlock ecosystems shape the assembly of scientific teams: A hypergraph approach. *Computational Methods for Communication Science*, 12(2-3), 174–198. https://doi.org/10.1080/ 19312458.2018.1430756
- Lungeanu, A., Huang, Y., & Contractor, N. S. (2014). Understanding the assembly of inter-disciplinary teams and its impact on performance. *Journal of Informetrics*, 8(1), 59–70. https://doi.org/10.1016/j.joi.2013.10.006
- Mallinson, T., Lotrecchiano, G. R., Schwartz, L. S., Furniss, J., Leblanc-Beaudoin, T., Lazar, D., & Falk-Krzesinski, H. J. (2016). Pilot analysis of the Motivation Assessment for Team Readiness, Integration, and Collaboration (MATRICx) using Rasch analysis. *Journal of Investigative Medicine*, 64(7), 1186–1193. https://doi.org/10.1136/jim-2016-000173
- Marchant, T. D., Tiernan, T. M., & Mann, W. C. (2005). Computer accessibility issues for older adults with disabilities: A pilot study. *Occupational Therapy Journal of Research*, 25(2), 55–65. https://doi.org/10.1177/153944920502500203
- Marks, M. A., Mathieu, J. E., & Zaccaro, S. J. (2001). A temporally based framework and taxonomy of team processes. *The Academy of Management Review*, 26(3), 356–376. https://doi.org/10.2307/259182
- Marlow, S. L., Lacerenza, C. N., Paoletti, J., Burke, C. S., & Salas, E. (2018). Does team communication represent a one-size-fits-all approach?: A meta-analysis of team communication and performance. Organizational Behavior and Human Decision Processes, 144, 145–170. https://doi.org/10.1016/j.obhdp.2017.08.001
- Marrone, J. A. (2010). Team boundary spanning: A multilevel review of past research and proposals for the future. *Journal of Management*, 36(4), 911–940. https://doi.org/10.1177/0149206309353945
- Marrone, J. A., Tesluk, P. E., & Carson, J. B. (2007). A multilevel investigation of antecedents and consequences of team member boundary-spanning behavior. *Academy of Management Journal*, 50(6), 1423–1439. https://doi.org/10.5465/amj.2007.28225967
- Martínez-Córcoles, M., Teichmann, M., & Murdvee, M. (2017). Assessing technophobia and technophilia: Development and validation of a questionnaire. *Technology in Society*, *51*, 183–188. https://doi.org/10.1016/j.techsoc.2017.09.007
- Mason, O. J., Stevenson, C., & Freedman, F. (2014). Ever-present threats from information technology: the Cyber-Paranoia and Fear Scale. *Frontiers in Psychology*, *5*, 1298. https://doi.org/10.3389/fpsyg.2014.01298
- Mathieu, J. E., & Rapp, T. L. (2009). Laying the foundation for successful team performance trajectories: The roles of team charters and performance strategies. *Journal of Applied Psychology*, 94(1), 90–103. https://doi.org/10.1037/a0013257
- Mathieu, J. E., Wolfson, M. A., & Park, S. (2018). The evolution of work team research since Hawthorne. *American Psychologist*, 73(4), 308–321.
- McCormack, W. T., & Strekalova, Y. A. L. (2021). CTS teams: A new model for translational team training and team science intervention. *Journal of Clinical and Translational Science*, 5(1), e183. https://doi.org/10.1017/cts.2021.854
- McDowell, W. C., Herdman, A. O., & Aaron, J. (2011). Charting the course: The effects of team charters on emergent behavioral norms. *Organization Development Journal*, 29(1), 79–88.
- McMahan, B., Moore, E., Ramage, D., Hampson, S., & y Arcas, B. A. (2017, April). Communication-efficient learning of deep networks from decentralized data. In *Artificial intelligence and statistics* (pp. 1273–1282). PMLR.
- Meluso, J., Johnson, S., & Bagrow, J. (2022). Flexible environments for hybrid collaboration: Redesigning virtual work through the four orders of design. *Design Issues*, *38*(1), *55–69*. https://doi.org/10.1162/desi\_a\_00670

Mercer, S. L., Green, L. W., Cargo, M., Potter, M. A., Daniel, M., Olds, S., & Reed-Gross, E. (2008). Appendix C: Reliability-tested guidelines for assessing participatory research projects. In N. W. Meredith Minkler (Ed.), Community-based participatory research for health: From process to outcomes (2nd ed., pp. 407–433). Jossey-Bass.

- Merritt, S. M., Heimbaugh, H., LaChapell, J., & Lee, D. (2013). I trust it, but I don't know why: Effects of implicit attitudes toward automation on trust in an automated system. *Human Factors*, 55(3), 520–534. https://doi.org/10.1177/0018720812465081
- Milliken, F. J., & Martins, L. L. (1996). Searching for common threads: Understanding the multiple effects of diversity in organizational groups. *The Academy of Management Review*, 21(2), 402–433. https://doi.org/10.5465/amr.1996.9605060217
- Misra, S., Harvey, R. H., Stokols, D., Pine, K. H., Fuqua, J., Shokair, S. M., & Whiteley, J. M. (2009). Evaluating an interdisciplinary undergraduate training program in health promotion research. *American Journal of Preventive Medicine*, 36(4), 358–365. https://doi.org/10.1016/j.amepre.2008.11.014
- Misra, S., Stokols, D., & Cheng, L. (2015). The Transdisciplinary Orientation scale: Factor structure and relation to the integrative quality and scope of scientific publications. *Journal of Translational Medicine & Epidemiology*, 3(2), 1042.
- Montag, C., Kraus, J., Baumann, M., & Rozgonjuk, D. (2023). The propensity to trust in (automated) technology mediates the links between technology self-efficacy and fear and acceptance of artificial intelligence. *Computers in Human Behavior Reports*, 11, Article 100315. https://doi.org/10.1016/j.chbr.2023.100315
- Montgomery, B. L., & Page, S. C. (2018). [Mentoring beyond hierarchies: Multi-mentor systems and models]. Paper commissioned by the Committee on Effective Mentoring in STEMM at the National Academies of Sciences, Engineering, and Medicine. https://nap.nationalacademies.org/resource/25568/Montgomery%20and%20Page%20-%20 Mentoring.pdf
- Montoya, A. C., Carter, D. R., Martin, J., & DeChurch, L. A. (2015). The five perils of team planning: Regularities and remedies. In M. D. Mumford & M. Frese (Eds.), *The psychology of planning in organizations* (pp. 166–185). Routledge.
- Morgan, S. E., Mosser, A., Ahn, S., Harrison, T. R., Wang, J., Huang, Q., Reynolds, A., Mao, B., & Bixby, J. L. (2021). Developing and evaluating a team development intervention to support interdisciplinary teams. *Journal of Clinical and Translational Science*, 5(1), e166. https://doi.org/10.1017/cts.2021.831
- Morgeson, F. P., DeRue, D. S., & Karam, E. P. (2010). Leadership in teams: A functional approach to understanding leadership structures and processes. *Journal of Management*, 36(1), 5–39.
- Mosca, J. B., & Merkle, J. F. (2024). Strategic onboarding: Tailoring Gen Z transition for workplace success. *Journal of Business Diversity*, 24(1). https://doi.org/10.33423/jbd. v24i1.6852
- Mroz, J. E., Allen, J. A., Verhoeven, D. C., & Shuffler, M. L. (2018). Do we really need another meeting? The science of workplace meetings. *Current Directions in Psychological Science*, 27(6), 484–491. https://doi.org/10.1177/0963721418776307
- Nagel, J. D., Koch, A., Guimond, J. M., Galvin, S., & Geller, S. (2013). Building the women's health research workforce: Fostering interdisciplinary research approaches in women's health. Global Advances in Health and Medicine, 2(5), 24–29. https:// doi.org/10.7453/gahmj.2013.049
- National Academies of Sciences, Engineering, and Medicine. (2019). The science of effective mentorship in STEMM. The National Academies Press. https://doi.org/10.17226/25568
- \_\_\_. (2023). Attacks on scientists and health professionals during the pandemic: Proceedings of a symposium in brief. The National Academies Press. https://doi.org/10.17226/26936

- National Research Council. (2011). Prudent practices in the laboratory: handling and management of chemical hazards, updated version. The National Academies Press. https://doi.org/10.17226/12654
- \_\_\_\_. (2015). Enhancing the effectiveness of team science. The National Academies Press. https://doi.org/10.17226/19007
- National Science Foundation. (2024). U.S. National Science Foundation research traineeship program. https://new.nsf.gov/funding/opportunities/us-national-science-foundation-research-traineeship-program
- Newton, O. B., Fiore, S. M., & Song, J. (2019). Expertise and complexity as mediators of knowledge loss in open source software development. In *Proceedings of the Human Factors and Ergonomics Society annual meeting* (pp. 1580–1584). SAGE Publications.
- Nielsen, M. W., Bloch, C. W., & Schiebinger, L. (2018). Making gender diversity work for scientific discovery and innovation. *Nature Human Behaviour*, 2(10), 726–734. https:// doi.org/10.1038/s41562-018-0433-1
- Nishii, L. H., & Leroy, H. (2022). A multi-level framework of inclusive leadership in organizations. *Group & Organization Management*, 47(4), 683–722. https://doi.org/10.1177/10596011221111505
- Nogrady, B. (2021). 'I hope you die': How the COVID pandemic unleashed attacks on scientists. *Nature*, 598, 250–253. https://doi.org/10.1038/d41586-021-02741-x
- O'Neill, T. A., Allen, N. J., & Hastings, S. E. (2013). Examining the "pros" and "cons" of team conflict: A team-level meta-analysis of task, relationship, and process conflict. *Human Performance*, 26(3), 236–260. https://doi.org/10.1080/08959285.2013.795573
- Onoja, J. P., & Ajala, O. A. (2022). Innovative telecommunications strategies for bridging digital inequities: A framework for empowering underserved communities. *GSC Advanced Research and Reviews*, 13(01), 210–217.
- O'Rourke, M., Crowley, S., Laursen, B., Robinson, B., & Vasko, S.E. (2019). Disciplinary diversity in teams: Integrative approaches from unidisciplinarity to transdisciplinarity. In K. Hall, A. Vogel, & R. Croyle (Eds.), *Strategies for team science success*. Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6\_2
- O'Rourke, M., Rinkus, M. A., Cardenas, E., & McLeskey, C. (2023). Communication practice for team science. In D. Gosselin (Ed.), *A practical guide for developing cross-disciplinary collaboration skills* (pp. 83–102). Springer. https://doi.org/10.1007/978-3-031-37220-9\_5
- Paletz, S. B. F., Schunn, C. D., & Kim, K. H. (2013). The interplay of conflict and analogy in multidisciplinary teams. Cognition, 126(1), 1–19. https://doi.org/10.1016/j.cognition.2012.07.020
- Park, G., Spitzmuller, M., & DeShon, R. P. (2013). Advancing our understanding of team motivation: Integrating conceptual approaches and content areas. *Journal of Manage*ment, 39(5), 1339–1379. https://doi.org/10.1177/0149206312471389
- Parker, P. (2020) The art of gathering: How we meet and why it matters. Riverhead Books.
- Paterson, B., Gregory, D., & Thorne, S. (1999). A protocol for researcher safety. *Qualitative Health Research*, 9(2), 259–269. https://doi.org/10.3399/bjgp14X679480
- Pendharkar, P. C., & Rodger, J. A. (2009). The relationship between software development team size and software development cost. *Communications of the ACM*, 52(1), 141–144. https://doi.org/10.1145/1435417.1435449
- Pennington, D. (2016). A conceptual model for knowledge integration in interdisciplinary teams: Orchestrating individual learning and group processes. *Journal of Environmental Studies and Sciences*, 6, 300–312.
- Pennington, D., Bammer, G., Danielson, A., Gosselin, D., Gouvea, J., Habron, G., Hawthorne, D., Parnell, R., Thompson, K., Vincent, S., & Wei, C. (2016). The EMBeRS project: Employing model-based reasoning in socio-environmental synthesis. *Journal of Environmental Studies and Sciences*, 6, 278–286.

Pennington, D., Vincent, S., Gosselin, D., & Thompson, K. (2021). Learning across disciplines in socio-environmental problem framing. *Socio-Environmental Systems Modelling*, 3, 17895.

- Piso, Z., O'Rourke, M., & Weathers, K. C. (2016). Out of the fog: Catalyzing integrative capacity in interdisciplinary research. *Studies in History and Philosophy of Science Part A*, 56, 84–94. https://doi.org/10.1016/j.shpsa.2016.01.002
- Purvanova, R. K., & Kenda, R. (2022). The impact of virtuality on team effectiveness in organizational and non-organizational teams: A meta-analysis. *Applied Psychology*, 71(3), 1082–1131. https://doi.org/10.1111/apps.12348
- Qudrat-Ullah, H. (2007). Debriefing can reduce misperceptions of feedback: The case of renewable resource management. *Simulation & Gaming*, 38(3), 382–397. https://doi.org/10.1177/1046878107300669
- Reich, Y., Ullmann, G., Van der Loos, M., & Leifer, L. (2009). Coaching product development teams: A conceptual foundation for empirical studies. *Research in Engineering Design*, 19(4), 205–222. https://doi.org/10.1007/s00163-008-0046-1
- Reiter-Palmon, R., Kennel, V., Allen, J. A., Jones, K. J., & Skinner, A. M. (2015). Naturalistic decision making in after-action review meetings: The implementation of and learning from post-fall huddles. *Journal of Occupational and Organizational Psychology*, 88(2), 322–340. https://doi.org/10.1111/joop.12084
- Roberson, Q., & Perry, J. L. (2021). Inclusive leadership in thought and action: A thematic analysis. *Group & Organization Management*, 47(4), 755–778. https://doi.org/10.1177/10596011211013161
- Rodríguez, D. C., Jessani, N. S., Zunt, J., Ardila-Gómez, S., Muwanguzi, P. A., Atanga, S. N., Sunguya, B., Farquhar, C., & Nasuuna, E. (2021). Experiential learning and mentorship in global health leadership programs: Capturing lessons from across the globe. *Annals of Global Health*, 87(1), 61. https://doi.org/10.5334/aogh.3194
- Rogelberg, S. G. (2019). The surprising science of meetings: How you can lead your team to peak performance. Oxford University Press.
- Rogelberg, S. G., Leach, D. J., Warr, P. B., & Burnfield, J. L. (2006). "Not another meeting!" Are meeting time demands related to employee well-being? *Journal of Applied Psychology*, 91(1), 83–96. https://doi.org/10.1037/0021-9010.91.1.83
- Rolland, B., Scholl, L., Suryanarayanan, S., Hatfield, P., Judge, K., Sorkness, C., Burnside, E., & Brasier, A. R. (2021a). Operationalization, implementation, and evaluation of collaboration planning: A pilot interventional study of nascent translational teams. *Journal of Clinical and Translational Science*, 5(1), e23. https://doi.org/10.1017/cts.2020.515
- Rolland, B., Hohl, S. D., & Johnson, L. J. (2021b). Enhancing translational team effectiveness: The Wisconsin Interventions in Team Science framework for translating empirically informed strategies into evidence-based interventions. *Journal of Clinical and Translational Science*, 5(1), e158. https://doi.org/10.1017/cts.2021.825
- Rosen, M. A., Salas, E., Wilson, K. A., King, H. B., Salisbury, M., Augenstein, J. S., Robinson, D. W., & Birnbach, D. J. (2008). Measuring team performance in simulation-based training: Adopting best practices for healthcare. Simulation in Healthcare, 3(1), 33–41. https://doi.org/10.1097/SIH.0b013e3181626276
- Salas, E., Klein, C., King, H., Salisbury, M., Augenstein, J. S., Birnbach, D. J., Robinson, D. W., & Upshaw, C. (2008). Debriefing medical teams: 12 evidence-based best practices and tips. The Joint Commission Journal on Quality and Patient Safety, 34(9), 518–527. https://doi.org/10.1016/S1553-7250(08)34066-5
- Santos, V. R., Soares, A. L., & Carvalho, J. Á. (2012). Knowledge sharing barriers in complex research and development projects: An exploratory study on the perceptions of project managers. *Knowledge and Process Management*, 19(1), 27–38.

- Schaubroeck, J., Lam, S. S. K., & Cha, S. E. (2007). Embracing transformational leadership: Team values and the impact of leader behavior on team performance. *Journal of Applied Psychology*, 92(4), 1020–1030. https://doi.org/10.1037/0021-9010.92.4.1020
- Schmidt, K., & Wagner, I. (2004). Ordering systems: Coordinative practices and artifacts in architectural design and planning. Computer Supported Cooperative Work, 13, 349–408
- Schweiger, D. M., Sandberg, W. R., & Rechner, P. L. (1989). Experiential effects of dialectical inquiry, devil's advocacy and consensus: Approaches to strategic decision making. *Academy of Management Journal*, 32(4), 745–772. https://doi.org/10.5465/256567
- Secară, A., & Perez, E. (2022). Addressing content, technical and collaboration concerns in providing access to the D/deaf and hard of hearing audience: Integrated theatre captioning and theatre sign language interpreting. *InTRAlinea: Online Translation Journal*, 24.
- Shah, D. T., & Fiore, S. M. (2022). Building effective mentoring team using team science competencies. In A. Fornari & D. T. Shah (Eds.), Mentoring in health professions education: Evidence-informed strategies across the continuum (pp. 13–21). Springer, Cham. https://doi.org/10.1007/978-3-030-86935-9
- Sherrill, A. M., Wiese, C. W., Abdullah, S., & Arriaga, R. I. (2022). Overcoming clinician technophobia: What we learned from our mass exposure to telehealth during the COVID-19 pandemic. *Journal of Technology in Behavioral Science*, 7(4), 547–553. https:// doi.org/10.1007/s41347-022-00273-3
- Shore, L. M., & Chung, B. G. (2021). Inclusive leadership: How leaders sustain or discourage work group inclusion. *Group & Organization Management*, 47(4), 723–754. https://doi.org/10.1177/1059601121999580
- Shuffler, M. L., DiazGranados, D., Maynard, M. T., & Salas, E. (2018). Developing, sustaining, and maximizing team effectiveness: An integrative, dynamic perspective of team development interventions. *Academy of Management Annals*, 12(2), 688–724. https://doi.org/10.5465/annals.2016.0045
- Shuffler, M. L., Jiménez-Rodríguez, M., & Kramer, W. S. (2015). The science of multiteam systems: A review and future research agenda. *Small Group Research*, 46(6), 659–699. https://doi.org/10.1177/1046496415603455
- Sindermann, C., Sha, P., Zhou, M., Wernicke, J., Schmitt, H. S., Li, M., Sariyska, R., Stavrou, M., Becker, B., & Montag, C. (2021). Assessing the attitude towards artificial intelligence: Introduction of a short measure in German, Chinese, and English language. Künstliche Intelligenz, 35, 109–118. https://doi.org/10.1007/s13218-020-00689-0
- Sinkovics, R. R., Stöttinger, B., Schlegelmilch, B. B., & Ram, S. (2002). Reluctance to use technology-related products: Development of a technophobia scale. *Thunderbird International Business Review*, 44(4), 477–494. https://doi.org/10.1002/tie.10033
- Smith, A. M., Lai, S. Y., Bea-Taylor, J., Hill, R. B., & Kleinhenz, N. (2016). Collaboration and change in the research networks of five Energy Frontier Research Centers. Research Evaluation, 25(4), 472–485. https://doi.org/10.1093/reseval/rvw006
- Smith-Doerr, L., Alegria, S., & Sacco, T. (2017) How diversity matters in the US science and engineering workforce: A critical review considering integration in teams, fields, and organizational contexts. *Engaging Science, Technology, and Society*, 3(2017). https://doi.org/10.17351/ests2017.142
- Smith-Jentsch, K. A., Cannon-Bowers, J. A., Tannenbaum, S. I., & Salas, E. (2008). Guided team self-correction: Impacts on team mental models, processes, and effectiveness. Small Group Research, 39(3), 303–327. https://doi.org/10.1177/1046496408317794
- Smith-Jentsch, K. A., & Sierra, M. J. (2023). Houston we have a problem: How debriefing method impacts open communication and the depth of team reflexivity. *Journal of Business and Psychology*, 38(6), 1211–1232. https://doi.org/10.1007/s10869-023-09912-9

Sourati, J., & Evans, J. A. (2023). Accelerating science with human-aware artificial intelligence. *Nature Human Behaviour*, 7(10), 1682–1696. https://doi.org/10.1038/s41562-023-01648-z

- Specht, A., & Crowston, K. (2022). Interdisciplinary collaboration from diverse science teams can produce significant outcomes. *PLoS One*, 17(11), e0278043. https://doi.org/10.1371/journal.pone.0278043
- Star, S. L. (2010). This is not a boundary object: Reflections on the origin of a concept. Science, Technology, & Human Values, 35(5), 601–617. https://doi.org/10.1177/0162243910377624
- Star, S. L., & Griesemer, J. R. (1989). Institutional ecology, `translations' and boundary objects: Amateurs and professionals in Berkeley's Museum of Vertebrate Zoology, 1907-39. *Social Studies of Science*, 19(3), 387–420. https://doi.org/10.1177/030631289019003001
- Stasser, G., & Titus, W. (1985). Pooling of unshared information in group decision making: Biased information sampling during discussion. *Journal of Personality and Social Psychology*, 48(6), 1467–1478
- Steiner, J. S., Blum-Barnett, E., Rolland, B., Kraus, C. R., Wainwright, J. V., Bedoy, R., Martinez, Y. T., Alleman, E. R., Eibergen, R., Pieper, L. E., Carroll, N. M., Hixon, B., Sterrett, A., Rendle, K. A., Saia, C., Vachani, A., Ritzwoller, D. P., & Burnett-Hartman, A. (2023). Application of team science best practices to the project management of a large, multi-site lung cancer screening research consortium. *Journal of Clinical and Translational Science*, 7(1), e145. https://doi.org/10.1017/cts.2023.566
- Stewart, V. R., Snyder, D. G., & Kou, C.-Y. (2023). We hold ourselves accountable: A relational view of team accountability. *Journal of Business Ethics*, 183(3), 691–712. https://doi.org/10.1007/s10551-021-04969-z
- Stokols, D. (2010). Cross-disciplinary team science initiatives: Research, training, and translation. In R. Frodemen, J. T. Klein, C. Mitcham, & J. B. Holbrook (Eds.), Oxford handbook of interdisciplinarity. https://escholarship.org/uc/item/4wp2g8dm
- Sutton, L., Berdan, L. G., Bolte, J., Califf, R. M., Ginsburg, G. S., Li, J. S., McCall, J., Moen, R., Myers, B. S., Rodriquez, V., Veldman, T., & Boulware, L. E. (2019). Facilitating translational team science: The project leader model. *Journal of Clinical and Translational Science*, 3(4), 140–146. https://doi.org/10.1017/cts.2019.398
- Tannenbaum, S., Fernández Castillo, G., & Salas, E. (2023). How to overcome the nine most common teamwork barriers. *Organizational Dynamics*, 52, 101006. https://doi.org/10.1016/j.orgdyn.2023.101006
- Tannenbaum, S. I., & Cerasoli, C. P. (2013). Do team and individual debriefs enhance performance? A meta-analysis. *Human Factors*, 55(1), 231–245. https://doi.org/10.1177/0018720812448394
- Tannenbaum, S. I., & Greilich, P. E. (2023). The debrief imperative: Building teaming competencies and team effectiveness. *BMJ Quality & Safety*, 32(3), 125–128. https://doi.org/10.1136/bmjqs-2022-015259
- Tebes, J. K., & Thai, N. D. (2018). Interdisciplinary team science and the public: Steps toward a participatory team science. *The American Psychologist*, 73(4), 549–562. https://doi.org/10.1037/amp0000281
- Tesler, R., Mohammed, S., Hamilton, K., Mancuso, V., & McNeese, M. (2018). Mirror, mirror: Guided storytelling and team reflexivity's influence on team mental models. *Small Group Research*, 49(3), 267–305. https://doi.org/10.1177/1046496417722025
- Thatcher, S. M., Meyer, B., Kim, Y., & Patel, P. C. (2024). A meta-analytic integration of the faultlines literature. *Organizational Psychology Review*, 14(2), 238–281. https://doi.org/10.1177/20413866231225064
- Tilmes, N. (2022). Disability, fairness, and algorithmic bias in AI recruitment. *Ethics and Information Technology*, 24(2), 21. https://doi.org/10.1007/s10676-022-09633-2

- Timóteo, M., Lourenço, E., Brochado, A. C., Domenico, L., da Silva, J., Oliveira, B., Barbosa, R., Montemezzi, P., Mourão, C. F. de A. B., Olej, B., & Alves, G. (2021). Digital management systems in academic health sciences laboratories: A scoping review. *Healthcare*, 9(6), 739. https://doi.org/10.3390/healthcare9060739
- Traylor, A., Stahr, E., & Salas, E. (2020). Team coaching: Three questions and a look ahead: A systematic literature review. *International Coaching Psychology Review*, 15, 54–67. https://doi.org/10.53841/bpsicpr.2020.15.2.54
- Trewin, S., & Pain, H. (1999). Keyboard and mouse errors due to motor disabilities. *International Journal of Human-Computer Studies*, 50(2), 109–144. https://doi.org/10.1006/ijhc.1998.0238
- Tschan, F., Semmer, N. K., Gurtner, A., Bizzari, L., Spychiger, M., Breuer, M., & Marsch, S. U. (2009). Explicit reasoning, confirmation bias, and illusory transactive memory: A simulation study of group medical decision making. *Small Group Research*, 40(3), 271–300. https://doi.org/10.1177/1046496409332928
- Tuckman, B. W. (1965). Developmental sequence in small groups. *Psychological Bulletin*, 63(6), 384–399. https://doi.org/10.1037/h0022100
- Tumilty, E., Spratt, H., Cestone, C., Wooten, K., Aronson, J., Hommel, J., Hellmich, M. R., & Chao, C. (2022). Developing future translational scientists through authentic learning and assessments. *International Journal of Educational Research Open*, 3, 100151. https://doi.org/10.1016/j.ijedro.2022.100151
- Van Hartesveldt, C. J. (2016). Integrative graduate education and research. In W. Bainbridge & M. Roco (Eds.), *Handbook of science and technology convergence* (pp. 1045–1058). Springer, Cham. https://doi.org/10.1007/978-3-319-07052-0\_61
- van Knippenberg, D., & Schippers, M. C. (2007). Work group diversity. *Annual Review of Psychology*, 58(1), 515–541. https://doi.org/10.1146/annurev.psych.58.110405.085546
- Vashdi, D. R., Bamberger, P. A., & Erez, M. (2012). Can surgical teams ever learn? The role of coordination, complexity, and transitivity in action team learning. *Academy of Management Journal*, 56, 945–971. http://dx.doi.org/10.5465/amj.2010.0501
- Vela, C., Menchaca, V. D., & Silva, H. (2023). University faculty perceptions of professional development: Impact and effectiveness. *Journal of Educational Leadership in Action*.
- Verwijs, C., & Russo, D. (2024). The double-edged sword of diversity: How diversity, conflict, and psychological safety impact software teams. *IEEE Transactions on Software Engineering*, 50(1), 141–157. https://doi.org/10.1109/TSE.2023.3339881
- Vogel, A., Feng, A., Oh, A., Hall, K., Stipelman, B., Stokols, D., Okamoto, J., Perna, F., Moser, R., & Nebeling, L. (2012). Influence of a National Cancer Institute transdisciplinary research and training initiative on trainees' transdisciplinary research competencies and scholarly productivity. *Translational Behavioral Medicine*, 2(4), 459–468. https://doi.org/10.1007/s13142-012-0173-0
- Waizenegger, L., McKenna, B., Cai, W., & Bendz, T. (2020). An affordance perspective of team collaboration and enforced working from home during COVID-19. European Journal of Information Systems, 29(4), 429–442. https://doi.org/10.1080/0960085X.2020.1800417
- Wallerstein, N., Calhoun, K., Eder, M., Kaplow, J., & Wilkins, C.H. (2019). Engaging the community: Community-based participatory research and team science. In K. Hall, A. Vogel, & R. Croyle (Eds.), Strategies for team science success. Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6\_9
- Wang, G., Oh, I.-S., Courtright, S., & Colbert, A. (2011). Transformational leadership and performance across criteria and levels: A meta-analytic review of 25 years of research. *Group & Organization Management*, 36, 223–270. https://doi.org/10.1177/1059601111401017
- Wang, J., & Hicks, D. (2015). Scientific teams: Self-assembly, fluidness, and interdependence. *Journal of Informetrics*, 9(1), 197–207. https://doi.org/https://doi.org/10.1016/j.joi.2014.12.006

Wang, K., & Nickerson, J. V. (2017). A literature review on individual creativity support systems. *Computers in Human Behavior*, 74, 139–151. https://doi.org/10.1016/j. chb.2017.04.035

- Wardale, D. (2013). Towards a model of effective group facilitation. Leadership & Organization Development Journal, 34(2), 112–129. https://doi.org/10.1108/01437731311321896
- Webber, S. S., & Donahue, L. M. (2001). Impact of highly and less job-related diversity on work group cohesion and performance: A meta-analysis. *Journal of Management*, 27(2), 141–162. https://doi.org/10.1016/S0149-2063(00)00093-3
- Weingart, L. R., Behfar, K. J., Bendersky, C., Todorova, G., & Jehn, K. A. (2015). The directness and oppositional intensity of conflict expression. *The Academy of Management Review*, 40(2), 235–262. https://doi.org/10.5465/amr.2013.0124
- Wellemeyer, D., & Williams, J. (2019). Onboarding 2.0: Methods of designing and deploying effective onboarding training for academic libraries. Nova Science Publishers, Inc.
- Wentz, B., Jaeger, P. T., & Lazar, J. (2011). Retrofitting accessibility: The legal inequality of after-the-fact online access for persons with disabilities in the United States. *First Monday*, 16(11). https://doi.org/10.5210/fm.v16i11.3666
- Whitworth, E., & Biddle, R. (2007). Motivation and cohesion in agile teams. Lecture Notes in Computer Science, 4536. https://doi.org/10.1007/978-3-540-73101-6\_9
- Wiese, C. W., Burke, C. S., Tang, Y., Hernandez, C., & Howell, R. (2022). Team learning behaviors and performance: A meta-analysis of direct effects and moderators. *Group & Organization Management*, 47(3), 571–611. https://doi.org/10.1177/10596011211016928
- Wiese, C. W., Shuffler, M. L., & Salas, E. (2015). Teamwork and team performance measurement. In J. D. Wright (Ed.), *International encyclopedia of the social & behavioral sciences* (2nd ed., pp. 96–103). https://doi.org/10.1016/B978-0-08-097086-8.22017-5
- Williams, K. Y., & O'Reilly, III, C. A. (1998). Demography and diversity in organizations: A review of 40 years of research. *Research in Organizational Behavior*, 20, 77–140.
- Williamson, A., & Burns, N. (2014). The safety of researchers and participants in primary care qualitative research. *British Journal of General Practice*, 64(621), 198–200. https://doi.org/10.3399/bjgp14X679480
- Wolf, A. V., Hendrick, K. N., Kramer, W. S., & Shuffler, M. L. (2024). More teams, more meetings? Toward an understanding of multiteam system meeting design, facilitation, and effectiveness. *Organizational Psychology Review*. https://doi.org/10.1177/20413866241264121
- Woodside, R., Rosenthal, G., & Olivier, C. (2021). Implementing the innovative academic Learning Health System Scholars (aLHSS) Postdoctoral Training Program (TL1) at Wake Forest University Health Sciences (WFUHS). *Journal of Clinical and Translational Science*, 5(s1), 132–133. https://doi.org/10.1017/cts.2021.739
- Woolley, A. W., Gerbasi, M. E., Chabris, C. F., Kosslyn, S. M., & Hackman, J. R. (2008). Bringing in the experts: How team composition and collaborative planning jointly shape analytic effectiveness. Small Group Research, 39(3), 352–371. https://doi.org/10.1177/1046496408317792
- Wróbel, A. E., Lomberg, C., & Cash, P. (2021). Facilitating design: Examining the effects of facilitator's neutrality on trust and potency in an exploratory experimental study. *Design Science*, 7(e6). https://doi.org/10.1017/dsj.2021.5
- Wu, L., Wang, D., & Evans, J. A. (2019). Large teams develop and small teams disrupt science and technology. *Nature*, 566(7744), 378–382. https://doi.org/10.1038/s41586-019-0941-9
- Wuchty, S., Jones, B. F., & Uzzi, B. (2007). The increasing dominance of teams in production of knowledge. *Science*, 316(5827), 1036–1039. https://doi.org/10.1126/science.1136099
- Xu, H. (2025). [Exploring the use of artificial intelligence to facilitate team science]. Paper commissioned by the Committee on Research and Application in Team Science at the National Academies of Sciences, Engineering, and Medicine.

- Yammarino, F. J., et al. (2005). Leadership and levels of analysis: A state-of-the-science review. *The Leadership Quarterly*, 16(6), 879–919.
- Yang, Y., Tian, T. Y., Woodruff, T. K., Jones, B. F., & Uzzi, B. (2022). Gender-diverse teams produce more novel and higher-impact scientific ideas. *Proceedings of the National Academy of Sciences*, 119(36), e2200841119. https://doi.org/10.1073/pnas.2200841119 Yukl, G. (2006). *Leadership in organizations*, 9/e. Pearson India.
- Yukl, G., Gordon, A., & Taber, T. (2002). A hierarchical taxonomy of leadership behavior: Integrating a half century of behavior research. *Journal of Leadership & Organizational Studies*, 9(1), 15–32.
- Zaccaro, S. J., Dubrow, S., Torres, E. M., & Campbell, L. N. P. (2020). Multiteam systems: An integrated review and comparison of different forms. *Annual Review of Organizational Psychology and Organizational Behavior*, 7(1), 479–503. https://doi.org/10.1146/annurev-orgpsych-012119-045418
- Zaccaro, S. J., Rittman, A. L., & Marks, M. A. (2001). Team leadership. *The Leadership Quarterly*, 12(4), 451–483.
- Zajac, S., Woods, A., Tannenbaum, S., Salas, E., & Holladay, C. L. (2021). Overcoming challenges to teamwork in healthcare: A team effectiveness framework and evidence-based guidance. *Frontiers in Communication*, *6*, 606445.
- Zajdela, E. R., Huynh, K., Feig, A. L., Wiener, R. J., & Abrams, D. M. (2025). Face-to-face or face-to-screen: A quantitative comparison of conferences modalities. *PNAS Nexus*, 4(1). https://doi.org/10.1093/pnasnexus/pgae522
- Zajdela, E. R., Huynh, K., Wen, A. T., Feig, A. L., Wiener, R. J., & Abrams, D. M. (2022). Dynamics of social interaction: Modeling the genesis of scientific collaboration. *Physical Review Research*, 4(4), L042001. https://doi.org/10.1103/PhysRevResearch.4.L042001

4

# Institutional and External Supports for Team Science

When scientists share their experiences about working in teams, they frequently refer to ways that their working environment hinders teamwork. Complaints range from the trivial—the difficulties of adding an external collaborator to a messaging system—to the existential—the bureaucratic obstacles involved in sending funds or data to collaborators at other institutions (e.g., Byers-Heinlein et al., 2020). Principal investigators also point to structures that disincentivize participation in large projects because of the emphasis placed on first- or final-author publications rather than contributions to manuscripts with many authors (e.g., Forscher et al., 2023). Members of science teams often worry that their collaborative work will not be acknowledged during their institutional tenure and promotion processes or that their work may be less likely to find a "home" in the most prestigious journals of their field (Forscher et al., 2023).

These concerns highlight that team science takes place in an overlapping set of contexts that together are important determinants of success or failure. These contexts include the institutions—academic and nonacademic—in which research takes place and their infrastructure and policies. They also include funding for team science and policies associated with specific funders. Finally, they include the scientific culture as a whole and scientific incentives that motivate research, including intersections with journals and scientific societies. Each of these contexts includes qualities that can help or hinder teams. Therefore, to best answer the question in the statement of task pertaining to best practices, the committee elected to include best practices and potential gaps that are external to the team or are situated at the institutional level. Specifically, this chapter draws from the literature to

cover institutional infrastructure, culture and policy, funders, and scientific incentives.

#### INSTITUTIONAL INFRASTRUCTURE

Because most researchers work within institutions (academic or nonacademic), they rely on these institutions to provide infrastructure to support their work. This infrastructure can range from the basic (e.g., space, connectivity, administrative support) to the highly specialized (e.g., research computing support). As researchers collaborate in larger teams that cross traditional boundaries-including laboratory groups, disciplines, and institutions—or teams that span sectors, the friction caused by infrastructural barriers (or the lack of support by a home institution) can hinder progress substantially (Forscher et al., 2023; Mirel & Harris, 2015). In this section, the committee highlights the roles that institutions play in providing physical infrastructure; technological resources, defined broadly as software, data storage, and computation; and human resources, including personnel and project management support for teams. Critically, through legislation such as the Americans with Disabilities Act (2010), institutions are required to make reasonable accommodations for individuals with disabilities.

# Physical Infrastructure

Physical environments can manifest a critical barrier for individuals with disabilities on teams (e.g., Lindsay & Fuentes, 2022). Despite advancements in accessibility regulations and standards, many scientific institutions and workplaces still lag in providing fully accessible physical spaces and equipment (Jeannis et al., 2020). For example, scientific laboratories are often designed without consideration for individuals with physical disabilities—for example, laboratory benches and equipment may be positioned at heights that are unreachable for scientists who use wheelchairs (Heidari, 1996; Hilliard et al., 2013). In addition, narrow or unclear aisles and nonadjustable furniture can further limit accessibility (Jeannis et al., 2020). The inaccessibility of research tools and equipment can also limit full participation in scientific research teams (Devitz, 2023). Inaccessible tools not only limit inclusivity within labs but also can prevent researchers from engaging in field research and participating fully in the workplace.

Making accessible practices systematic rather than ad hoc is crucial (Americans with Disabilities Act, 2010). For disabled scientists, this means creating an environment where their participation is seamlessly integrated into all aspects of team operations (Anderson et al., 2022; Persson et al.,

2015). Most importantly, these practices can be proactive rather than reactive (Mamboleo et al., 2020). Instead of waiting for a team member to request disability access needs or raise concerns, teams can anticipate and plan for their specific needs from the start. For example, laboratories or institutions might restructure their lab protocols to be inherently accessible by creating protocols that can be executed effectively regardless of whether someone is standing, sitting, or using assistive devices (e.g., Burgstahler, 2012). Ensuring that lab spaces are designed with adjustable workstations, sufficient maneuvering space, and accessible equipment is crucial for enabling disabled scientists to conduct their research independently (e.g., Massachusetts Institute of Technology, n.d.).

One method for addressing the need for accessible tools and the associated costs, which can often be an insurmountable barrier even when the need is recognized, is to establish an equipment repository (Devitz, 2023). The National Science Foundation (NSF) has also provided funding through Facilitation Awards for Scientists and Engineers with Disabilities to support the engagement of researchers with disabilities. Furthermore, a survey by the Job Accommodation Network (2023) found that many workplace accommodations were low cost.

Conference rooms and collaborative spaces can also overlook the needs of disabled individuals. Accessible meeting spaces ought to include features such as ramps or elevators, wide doorways, adequate lighting, and seating arrangements that cater to individuals with mobility impairments (Americans with Disabilities Act, 2010; Smith & Dropkin, 2018). Furthermore, the availability of assistive technologies, such as hearing loops and accessible presentation equipment, can enhance these spaces significantly (National Institute on Deafness and Other Communication Disorders, 2019). Solutions such as using a live interpreter rather than closed captioning when possible can optimize the effectiveness of accessibility solutions. Beyond laboratories and meeting rooms, public areas such as restrooms, cafeterias, and communal spaces also need to comply with accessibility standards (Americans with Disabilities Act, 2010). Inaccessible amenities can severely affect the overall experience and daily functioning of disabled researchers, thus indirectly affecting their productivity and sense of belonging within the team (Lindsay & Fuentes, 2022).

Notably, by using universal design principles, creating accessible environments benefits all team members, regardless of ability. Universal design principles include providing all users the same provisions, creating designs

 $<sup>^1\,</sup>$  For more information about the NSF Facilitation Awards for Scientists and Engineers with Disabilities, please see https://www.nsf.gov/funding/opportunities/dcl-persons-disabilities-stemengagement-access-pwd-sea

that are flexible in nature, making designs easy to use and understand, and ensuring that designs are usable with little physical effort for all users.<sup>2</sup>

#### **Technological Resources**

Technological resources necessary to perform team science include software, data storage, and computational resources. Choices around these resources are critical enablers of progress in team science. Too often, teams take a piecemeal approach to information management, resulting in conflict that hinders their progress (Kelly et al., 2023). Furthermore, in the case of data sharing—or even sharing materials such as presentations—funders may make specific mandates regarding whether and how sharing proceeds; for example, they may regulate sharing based on specific regulations such as the Defense Federal Acquisition Regulation Supplement (2020) or export controls.

### Collaboration Software

As teams grow larger and more distributed across time, space, and sector, the need for collaboration, communication, and project management tools increases dramatically (see Chapter 3). Practically speaking, these tools are instantiated in commercial or open-source software products (de Vreede et al., 2016). Open-source tools are freely accessible to all collaborators. While many commercial tools, such as word processing software and conference and chat platforms (e.g., Google Docs, Zoom, Slack, or Microsoft Teams), offer free versions, they are often limited in functionality and may not meet the needs of larger projects. As a result, institutions typically purchase the full versions of these tools. Site licenses for these products can be very helpful for facilitating intra-institutional collaboration—for example, by setting up shared messaging platforms or by sharing documents through institutional storage. But they can also be extremely restrictive. For example, if a cross-institution team, such as those across academia and industry, would like to set up a message board, they may not be able to—even if many or all researchers have access individually to the appropriate software (e.g., Microsoft Teams, 2025). When faced with this challenge, teams have the option to either use free versions of proprietary software or operate outside their institutions' recommendations to reduce collaboration barriers.

The challenge of proprietary collaboration software can be addressed by both institutional and community solutions. Institutions can recognize the

<sup>&</sup>lt;sup>2</sup> To learn more about universal design, see the Centre for Excellence in Universal Design at https://universaldesign.ie/

need for teams to access shared software platforms and allocate resources to support collaborative access, by, for example, providing guest access to important collaboration platforms (e.g., Emerson College, 2025). Alternatively, teams can consider using exclusively free and open-source software for collaboration. While many of these tools (e.g., the open-source version control package "Git") are extremely powerful, they require substantial training to use, especially for scientists who do not have formal education in software engineering (Braga et al., 2023; Perez-Riverol et al., 2016). Indeed, recognizing this need, organizations such as The Carpentries<sup>3</sup> train scientists in the basic open-source tools that power software engineering (Wilson, 2013).

Relatedly, one barrier to effective participation in team science can include the lack of accessible communication (Isaacson et al., 2011; Persson et al., 2015; Rizzo et al., 2024). This includes issues with verbal and written communication, as well as the use of digital tools and platforms. For example, individuals who are D/deaf or hard of hearing often face significant challenges in environments where verbal communication predominates (Adler, 2025; Gehret et al., 2017; Marchetti et al., 2024). Written materials, such as research papers, meeting agendas, and collaborative documents, may not always be available in formats that are accessible to individuals with learning disabilities or who are blind or have low vision. For instance, scientific documents that are incompatible with screen readers or lack alternative text for images and graphs can prevent individuals from fully understanding and contributing to the material (Kumar & Wang, 2024; Singh Chawla, 2024). In addition, the increasing reliance on digital communication tools for virtual and hybrid meetings may pose further accessibility challenges (Bercaru & Popescu, 2024). Platforms that do not comply with accessibility standards can be difficult or impossible for some individuals to navigate. Features such as screen reader compatibility, keyboard shortcuts, and adaptable text sizes can help make these tools accessible to all users. Furthermore, including options for text-to-speech and speech-to-text capabilities can accommodate various disabilities in real-time communication tools, such as chat functions (Bercaru & Popescu, 2024).

#### Data Storage

Increasingly, science is data intensive (Kitchin, 2014), and many teams require data storage and sharing infrastructure. At smaller scales, these needs can be met by user- and enterprise-focused commercial platforms, such as Dropbox or Google Drive (with all the licensing challenges described above). At larger scales and for more complex datasets, bespoke solutions are often required.

<sup>&</sup>lt;sup>3</sup> More information about The Carpentries is available at https://carpentries.org

In the best-case scenario, datasets are unencumbered by legal or ethical restrictions and can be shared openly. For example, when datasets (as well as working documents, protocols, and materials) are shared in open repositories, it is far easier for collaborators to access them across a wide range of platforms and geographic locations (Baumgartner et al., 2023). This greater accessibility through openness highlights the importance of open science practices in facilitating collaboration (see Box 4-1). Data can be shared

# BOX 4-1 The Role of Open Science in Facilitating Team Science

Open science describes a broad range of practices and policies emerging from the central role of transparency in scientific progress (Crüwell et al., 2019; Klein et al., 2018; Spellman et al., 2018). Open-science practices can include sharing code and data (when permissible based on privacy and other constraints), preregistration of hypotheses, distribution of scholarship through posting preprints, and open-access publication. These practices together are intended to increase the reproducibility, replicability, and rigor of research by providing all the materials necessary for independent researchers to verify published research. They also aim to increase the impact of research by broadening the set of research products that can be reused (Piwowar & Vision, 2013; Wilkinson et al., 2016). Finally, they are intended to decrease barriers to inclusion in science by ensuring that anyone can engage with scientific products, regardless of background or affiliation (Grahe et al., 2020).

Although it is not always possible to do so, making data, analysis, and administrative documents (including team charters [see Chapter 3]) open dramatically decreases barriers for joining teams (Dai et al., 2018). Especially when teams have a "grassroots" strategy, open materials can encourage engagement from researchers, including those from nonacademic backgrounds (Baumgartner et al., 2023). Furthermore, openness around scientific products can signal a science team's interest in broad engagement.

Open-science tools and approaches can vastly simplify team operations, especially when they span institutions and countries. Tools for data and materials sharing—especially in the context of institutionally licensed products—can require complex and cumbersome security measures. These measures are warranted when there are regulatory or legal constraints on data sharing. In the absence of such constraints, however, making research materials open can be the easiest way to ensure that all team members can access them (Foster & Deardorff, 2017). For example, the Open Science Framework (OSF)<sup>a</sup> is a software platform for sharing and managing research products and facilitating open collaboration across science teams. Unlike commercial file-sharing platforms, research materials shared through the OSF receive digital object identifiers and are guaranteed to be accessible in perpetuity (Alter & Gonzalez, 2018).

<sup>&</sup>lt;sup>a</sup> More information about the Open Science Framework is available at http://osf.io

through institutional repositories such as Dataverse; cross-institutional platforms such as Zenodo, Figshare, Data Dryad, the Open Science Framework and others; and field- or dataset-specific repositories (e.g., OpenNeuro, <sup>4</sup> a free and open database of brain imaging data).

Challenges multiply, however, when teams deal with datasets that are encumbered by legal or ethical restrictions—as is often the case for data from human participants in the social and biomedical sciences. For example, the overlapping regulatory constraints on sharing data from clinical trials are complex (Gudi et al., 2022). In their study of translational teams, Kelly et al. (2023) highlighted how the complex security requirements for data sharing hindered their attempts to translate research findings between university collaborators and clinical or community partners. This research demonstrates how complications may be amplified when attempting to streamline coordination of team members across sectors.

Ensuring that stored data and information are accessible is also important. For example, scientific information, whether in the form of research papers, datasets, or educational resources, can be available in accessible formats (e.g., Rizzo et al., 2024; Wu et al., 2022). This includes providing documents that are compatible with screen readers, offering braille versions, and utilizing large print formats. Utilizing plain language summaries can also enhance understanding for individuals with cognitive or learning disabilities. Visual data representations, such as charts, graphs, and infographics, play a significant role in scientific research. However, these visual tools can be challenging for individuals who are blind or have low vision. Providing alternative text descriptions, tactile graphics, and audio explanations can help make this information accessible (e.g., Vidal-Verdú et al., 2007).

Online platforms and repositories where scientific knowledge are shared also need to adhere to web accessibility standards. This includes ensuring that websites are navigable via keyboard, providing text alternatives for multimedia content, and ensuring that interactive elements are accessible. In addition, enabling customizable viewing options can help individuals tailor their browsing experience to suit their specific needs. Mandates exist, such as Section 508 of the Rehabilitation Act (29 U.S.C. § 794d; General Services Administration, 2024). This is a federal law that requires agencies to provide individuals with disabilities equal access to electronic information and data comparable with those who do not have disabilities, unless an undue burden would be imposed on the agency. However, while compliance with standards such as Section 508 is necessary, it is merely the starting point for creating inclusive environments. Current standards can sometimes fall short of being accessible (Pearson & Alexander, 2020). Institutions can aspire to

<sup>&</sup>lt;sup>4</sup> More information about OpenNeuro is available at https://openneuro.org

go beyond compliance, adopting a proactive stance toward inclusion and equity where access is seamlessly integrated and viewed as essential, not burdensome (Humphrey et al., 2020; McDaniels & Asiedu, 2023).

#### Computational Resources

With the rise of artificial intelligence (AI) and machine learning methods, team science projects increasingly require high-performance computing resources, including access to clusters of graphics processing units (GPUs) or other specialized hardware (e.g., Besiroglu et al., 2024; Fujinuma et al., 2022; Ho et al., 2022; Puertas-Martín et al., 2020). There are current worries about the scarcity of access to GPUs (Griffith, 2023), which could decrease innovation in universities (Ho et al., 2022). For example, availability of computational resources has been cited as a major reason why industry is taking a larger role in recent research on AI (Ahmed et al., 2023). Thus, institutional access to high-performance computing resources enables science teams to thrive (Apon et al., 2010; Vecchiola et al., 2009). In the context of cross-institutional or cross-sector teams, the ability of collaborators to offer computational resources can also be an important incentive for collaboration.

#### **Human Resources**

As teams grow larger and more complex, the human resources required to ensure their success have increased significantly. Reimagining who is considered a member of the science team and establishing appropriate funding mechanisms can help foster more inclusive and innovative research (National Science Foundation, 2023; Specht & Crowston, 2022). Much of the diversity surrounding scientists can be found in the adjacent ecosystems, including staff members, study participants, and community partners (Bergeron, 2021; Passmore et al., 2022; Swartz et al., 2019). This extends beyond simply adding more personnel; specialized expertise is needed for developing and managing team science initiatives, particularly those spanning multiple disciplines or institutions. As team science projects increasingly cross disciplines, institutions, sectors, states, and countries, grant proposals for these projects have become more challenging to prepare initially and implement successfully after receiving funding. Institutions play an important role in human resources for team science, as they can provide training in team science skills and access to designated research personnel, including individuals with skills ranging from proposal development to financial administration and compliance. Historically, NSF responded to this challenge with its Growing Research Access for Nationally Transformative Equity and Diversity (GRANTED)

program<sup>5</sup> in 2023, which focused on bolstering the support ecosystem. In this context, the following sections review evidence for the role institutional personnel play in enhancing team science. Most of the works discussed are theoretical, anecdotal, or case based; more robust empirical investigations are needed.

#### Research Development Professionals

Research development professionals represent a relatively new but rapidly expanding profession with potential for advancing team science (Carter et al., 2019; Chedin, 2024; Hunt, 2019; Preuss et al., 2018). In the view of the National Organization of Research Development Professionals (NORDP, n.d.), research development "encompasses a set of strategic, catalytic, and capacity-building activities that advance research, especially in higher education" (p. 1). Research development grew out of research administration, coinciding with more competitive funding environments and additionally the rise of higher-dollar, longer-term team science opportunities from major funders that placed new demands on research ecosystem personnel to actively and intentionally support budding science teams (Levin, 2011; Mason & Learned, 2006; Mulfinger et al., 2016). As discussed in Chapter 3, most factors that influence team science success fall outside technical, disciplinary areas that comprise the bulk of scientists' training and expertise—that is, outside task competence. Building on this history, NORDP's (n.d.) four research development pillars are:

- 1. enhancement of collaboration and team science,
- 2. strategic research advancement,
- 3. communication of research and research opportunities, and
- 4. proposal development.

Given its relative newness, the role of research development professionals is evolving in the university ecosystem. The exact roles they play depend on their placement within an institution, including whether they are located within a central office or spread across smaller units, such as colleges, medical schools, and research centers. Overall, though, research development professionals' contributions can cut across all four of the NORDP components. For example, they may deliver grant training that can lead to more collaborative proposals; convene research groups and networking opportunities that can build cross-disciplinary, cross-sector, and cross-institutional bridges; disseminate funding opportunities that can initiate a

<sup>&</sup>lt;sup>5</sup> More information about the GRANTED program is available at https://new.nsf.gov/funding/initiatives/broadening-participation/granted

team effort; and provide extensive proposal development and interpersonal coaching that can produce more competitive funding proposals, including for large research centers. For institutions that offer internal team science seed-funding opportunities, it is common for research development professionals to lead or otherwise facilitate these programs. For example, Stanford University's Research Development Office (2024) is staffed by a group of research development professionals with experience in team science.

Survey data collected from NORDP members showed that research development professionals view "proposal development support for large, multi-investigator project grants" as their most important activity, outranking "grant team project management (coordination of meetings, proposal development deadlines, shared documents, etc.)" (Ross et al., 2019, p. 118). Stephens et al. (2024) were the first to empirically (albeit retrospectively) test what were previously only descriptive linkages between the support of research development professionals and funding outcomes. In the context of a U.S. medical school, they found that coaching in teamwork provided by research development professionals is positively associated with an awarded outcome, with notable gains for new teams and those with both research and clinical responsibilities (Stephens et al., 2024), suggesting the value these professionals can add for teams when familiarity and bandwidth among team members are low. A related, vet distinct area of potential support for a team includes funding for Research Specialist Awards, such as those offered by the National Cancer Institute (NCI). Similar to research development professionals, these awards are designed to support the research with personnel who do not serve as independent investigators (National Cancer Institute, 2024).

#### Research Administrators

While research development professionals support the formation, ideation, resource acquisition, and proposal development of science teams, research administrators support team science from a distinct yet equally important lens of compliance, monitoring, reporting, and financial support services (National Council of University Research Administrators, n.d.). Research administrators can be in a main sponsored programs office or distributed across an institution. They are well positioned to mitigate documented scientific costs of multi-institutional collaborations (e.g., Cummings & Kiesler, 2007), including helping teams spend less time and attention navigating institutional differences in appointment structures, salary payments, and policies.

Additionally, anecdotal evidence suggests that research administrators can be essential when science teams partner with other community members or organizations outside academia. Even when sponsors such as the National Institutes of Health (NIH) encourage a community-engaged approach, specific guidance around how to fund such engagement in team science can be sparse. Moreover, the proposal guidelines and policies that do exist are designed for use by academic actors (Seidel, 2022). Research administrators can bridge this gap between funders, science teams, and community partners by helping to administer subawards or subcontracts to community partner organizations and by ensuring necessary assurances, certifications, and protections are in place. From a fiscal standpoint, and relevant to the following discussion of nonscientist collaborators, research administrators can also bridge compliance gaps between a funder's allowable costs and a community partner's budgetary and compensatory needs, which may look different from what is traditionally found in a research proposal budget and for which no standard rates apply (Seidel, 2022).

# Facilitators and Integration Experts

Scholars increasingly call for greater recognition of and support for specialized professionals who operate at the nexus of research, relationships, and operations to help teams navigate the inherent challenges of collaboration and maximize potential benefits (e.g., Barker Scott & Manning, 2024). The specific roles and terms used to describe them (e.g., interdisciplinary executive scientist, facilitator, integration expert, coach) may differ, but there is significant conceptual similarity among them.

Hendren & Ku (2019) identified the role of *interdisciplinary executive scientist* as one who coordinates and connects into an integrated whole the scientific, administrative, and interpersonal pieces of a science team. These professionals, they argue, operate within the liminal spaces between roles and disciplines to actively ensure desired outcomes, such as synthesis and translation, are achieved rather than simply assuming they will emerge (Hendren & Ku, 2019).

Duke University's Clinical and Translational Science Institute (CTSI), funded in part by NIH, instituted the similar role of project leader after recognizing that asking principal investigators to lead the project, team, and science efforts at once is unrealistic and unsustainable, especially if their training has been only scientific (Sutton et al., 2019). As an extension of the principal investigator, CTSI project leaders pair their scientific understanding with team science best practices and tools, including team charters and project development plans, to achieve collaborative goals (Sutton et al., 2019). Vogel et al. (2021) described how the Division of Preclinical Innovation within NIH's National Center for Advancing Translational Sciences has institutionalized roles to support its large, heterogeneous science teams as well. These roles include, for example, project analysts, who track team activities and schedule meetings, and

project managers, who take leadership roles in scientific discussions and mitigating conflict.

Another role in team studies is that of *facilitator*, or one who supports collaboration for research teams, with a key distinction being how external they are to the team (see Widdowson et al., 2020). Facilitators might be engaged to oversee discussions (Clutterbuck, 2007) or manage the collaborative process (Hawkins, 2018), while not necessarily being part of the team. In recent years, the role of facilitator has been tailored for science teams and established as a semiformal position within a team. This role is typically filled by someone skilled in scientific collaboration practices who might also possess interactional expertise (Cravens et al., 2022). *Interactional expertise* involves a certain understanding of scientific tasks, enabling the facilitator to communicate relevant scientific practices (Bammer et al., 2020; Collins & Evans, 2019). However, others suggest that science facilitation may focus simply on tasks such as organizing scientists and their meetings or coordinating projects (Jiang et al., 2023).

A closely related but more specialized role is that of an *integration expert*, whose focus is on helping scientists combine knowledge across various disciplines. Hoffmann et al. (2022) described integration experts as scholars who "lead, administer, manage, monitor, assess, accompany, and/ or advise others" (p. 3) during interdisciplinary teamwork. This can also involve some form of coaching.

Although the roles described here are defined by different competencies, there is overlap; facilitation, integration, and coaching frequently intersect in these discussions. For instance, Cravens et al. (2022) mentioned that cofacilitators can act as coaches to support reflective learning, while Hoffmann et al. (2022) highlighted *advising* as a competency for integration experts, which they define as including "accompanying, supporting, or coaching others" in leading integrative efforts and achieving integrated outcomes (p. 3). As another example, several scholars have highlighted how facilitation is key for guiding effective team processes, designing productive interactions, fostering communication, promoting participatory decision-making processes, and encouraging knowledge exchange (Cravens et al., 2022; Graef et al., 2021; Kaner, 2014).

A substantial issue requiring more study is the fact that professionals inhabiting such roles are most often tied to temporary project funding. Since many science teams begin as unfunded collaborations or use institutional seed funding, support for these roles may not be available. Furthermore, even when personnel with expertise relevant to team science are engaged, they are more likely to be misunderstood, overlooked, and undervalued because of their boundary-spanning position (Bammer et al., 2020; Hendren & Ku, 2019; Hoffmann et al., 2022; Lyall, 2019). As a result, sustainable, long-term investments in personnel are needed as critical team science

infrastructure (Hendren & Ku, 2019; Hoffmann et al., 2022; Rolland et al., 2017).

Regardless of whether individuals are engaging in unfunded, institutionally funded, or sponsored research, the personnel discussed in this section are critical to making teams work and achieving their scientific goals. Much about this process remains understudied, however. As discussed, if the members of a science team are broadened to include research development professionals who are involved in the conception, development, implementation, and dissemination of team science inputs, processes, and outcomes, institutions may be able to better develop metrics to help assess the efficacy of these roles in contributing to team science. Importantly, measuring efficacy can be done at multiple levels, from surveying team members to assessing their connection to broader team science metrics (see Chapter 5). Periodic assessments can inform continuous improvement of individuals, their roles, or departmental processes, and can quantify and characterize how research development personnel, or related professionals, contribute to team science outcomes at their institutions (Bennett & Gladin, 2012). For example, there is little research on which project management tools, models, or frameworks are being adopted by individuals within these roles and whether these approaches address the administrative, coordinative, and generative workload that can accompany team science (e.g., innovation and project management research [Davies et al., 2018]).

#### Other Nonscientist Team Members

Several applied science fields offer approaches and frameworks that more directly address including nonscientist team members as part of the science team. These include community-based participatory research (CBPR) and human factors or ergonomics (HFE) science, in part because their research objectives often cannot be achieved without input from nonscientist team members (Giardullo, 2023; McDermott, 2022). Whereas CBPR grew out of health research domains (Leung et al., 2004) and HFE from addressing human-technology performance concerns during World War II (Meister, 1999), both fields are concerned with improving systems and testing interventions. In CBRP and HFE, success is measured by the actual adoption of the intervention, not just a demonstration in a laboratory that it works. Thus, for both CBPR and HFE, it is practically impossible to achieve their research objectives without including nonscientist teammates as study participants or localized subject matter experts and community partners, who may play a critical role as recruiters, maintainers, improvers, problem definers, problem solvers, and implementers of parts of a project that are necessary to realize the project's scientific or research objectives (Gopalan et al., 2020).

While including nonscientists as team members may not be as critical for all scientific fields, it is increasingly important for scientists and researchers aiming to tackle complex problems that may require complex solutions. The effective and ethical implementation and adoption of AI in society is one example of a complex societal problem that will increasingly need to involve marginalized communities that have been traditionally excluded from AI research and development teams (Parthasarathy & Katzman, 2024). Agricultural development research (Ingram, 2014) and sustainability (Diaz-Reviriego et al., 2019; Ernst et al., 2017; Osinski, 2021) also frequently involve and rely on people who are not career scientists in their research work and knowledge implementation. In general, these subject matter experts comprise the very population that the researchers are intending to help (Smikowski, 2009; World Health Organization, 2023). Scholars have discussed the importance of not only involving nonscientist team members but also implementing routine practices that foster trusting relationships. This can include the ethical and safety considerations of data sharing practices for example, with those who have historically not been on science teams (Kraft & Mittendorf, 2024; Sabatello et al., 2022).

#### INSTITUTIONAL CULTURE AND POLICY

Institutional policies and informal cultures can support and motivate team science; alternatively, they can create barriers that stifle scientific collaboration and the production of interdisciplinary research. To support team science, institutions can carefully consider the nature of their policies and cultures and, as needed, revise policies to streamline collaboration and encourage and reward teamwork (National Research Council, 2015). For example, hiring, tenure, promotion, and other reward systems (e.g., internal institutional awards, small grants) may need to be revised to reward team science. Policies related to materials and data sharing, ethical approvals, and staffing support may need to be revised to reduce unnecessary barriers to team science. This section will consider the impact of institutional policies and cultural features on scientific collaboration and suggest strategies for creating institutional environments that are conducive for team science.

# Material Transfers, Data Sharing, and Intellectual Property Policies and Procedures

Policies related to materials and data sharing are often cited as significant barriers to team science (Borgman, 2012; Kowalczyk & Shankar, 2011). Indeed, because science teams are more likely to need to share materials and data, they are more likely to need to work with personnel who have expertise in these policies (e.g., research administrators, other

team science–related personnel). One hurdle for research teams is the complexities involved in negotiating material transfer agreements and data use agreements, which may deter researchers from sharing valuable resources (Mello et al., 2020). These agreements are designed to protect intellectual property and ensure ethical use of materials and data, but they can become overly burdensome and time consuming. Challenges related to these agreements can be overcome through the open release of data and materials in cases where no privacy or intellectual property constraints exist (see Box 4-1). When such concerns exist, however, institutional policies can be a determinant of whether sharing is possible (Bubela et al., 2015).

#### Barriers to Collaboration

A primary challenge for team science related to material transfer and data use agreements is the intricate legal and administrative processes they often entail (Mello et al., 2020). These agreements are often reviewed extensively by legal departments, which can lead to prolonged negotiations. Researchers often navigate multiple layers of approval, from their home institution to the receiving institution, each with its own set of requirements and protocols. This bureaucratic maze can result in significant delays, sometimes stretching into months, thus stalling scientific progress (Bubela et al., 2015; Mello et al., 2020).

The lack of standardized templates and procedures for material transfer and data use agreements across institutions further complicates the process. Each institution may have its own unique agreement terms and conditions, leading to a situation where each new collaboration necessitates the drafting of a bespoke agreement. This lack of uniformity not only increases the time required to finalize agreements but also may add to the costs involved (Mello et al., 2020). Institutions and researchers may find themselves in prolonged negotiations over specific terms, such as intellectual property rights, confidentiality clauses, and usage limitations.

Although the protection of intellectual property is a legitimate concern, the stringent terms often embedded in material transfer and data use agreements can be counterproductive (Bubela et al., 2015). For example, clauses that limit the sharing of derivative works or impose strict usage restrictions can inhibit the free flow of information and resources necessary for collaborative research. Researchers might hesitate to share their materials or data, fearing that they may lose control over their intellectual contributions or that their work might be misappropriated.

Similar barriers exist in collaborations among industry, academia, and/ or national laboratories, where intellectual property and publication terms are agreed upon in advance of any collaboration. In such arrangements, industry benefits from expanding their capabilities and subject matter expertise to answer fundamental research questions that may be beyond the scope of what their organization is able to address alone, and academia benefits from additional funding sources, direct access to industrially relevant research challenges, and expanded networks for students who may be interested in a career in industry following their academic studies (Chai & Shih, 2016; Stuart & Ding, 2006).

# Impact on Collaboration

In combination, complex legal and administrative processes, a lack of standardization, and stringent limitations on the sharing of intellectual property can be a significant deterrent to team science collaboration (Mello et al., 2020). Researchers, especially those studying time-sensitive topics in fields such as medicine or environmental science, may opt to work independently rather than engage in the cumbersome process of securing material transfer and data use agreements. This fragmentation of effort can lead to duplicated work, inefficiencies, and a slower pace of scientific advancement (Bubela et al., 2015).

In the field of biomedical research, for example, sharing biological samples, such as cell lines, tissues, and genetic material, is critical. However, the negotiation of transfer agreements for these materials can be particularly arduous. For example, a researcher attempting to obtain a unique cell line from another institution might face months of negotiations over the terms of use, publication rights, and potential commercial applications. This delay can hinder timely research advancements and potentially delay the development of new therapies or diagnostic tools.

Similarly, environmental scientists often rely on data collected from different geographical locations and multiple research teams. The negotiation of data use agreements for accessing these datasets can be fraught with challenges. Institutions may have different data protection policies, leading to lengthy negotiations. For instance, a researcher seeking to combine climate data from different sources to study global patterns may face significant delays due to the need to secure multiple data use agreements, each with specific requirements and restrictions.

#### Potential Solutions

One effective strategy for mitigating these barriers is the standardization of material transfer and data use agreements (Bubela et al., 2015). Developing common templates that are accepted across institutions can streamline the negotiation process. Federal funding organizations can spearhead efforts to create standardized agreements that balance the need for intellectual property protection with the facilitation of scientific collaboration

(e.g., Bennett et al., 2010). Furthermore, it may be worth exploring whether having specialized teams that understand the intricacies of these agreements can expedite the review and approval process (Steiner et al., 2023).

Creating clear and flexible guidelines for the use of materials and data can also help. Institutions can provide researchers with frameworks that outline the necessary ethical considerations without imposing overly stringent requirements. Providing researchers with training and resources to understand and comply with these guidelines can further reduce the compliance burden. Often teams can pursue an "open by design" strategy to foresee and avoid complex regulatory situations (National Academies, 2018). Institutions and funding agencies can also create incentives for sharing materials and data, by, for example, recognizing such sharing in their metrics for research products. U.S. federal funders now require data sharing (Nelson, 2022), and grant proposals could be evaluated favorably if they include proactive plans for both scientific data sharing and public-facing dissemination. Recognizing and rewarding researchers who contribute to shared resources can also foster a culture of openness and collaboration (Nosek et al., 2015; Shaw et al., 2022). Lastly, advanced technologies can provide solutions for some of these challenges. Online platforms that facilitate the sharing of materials and data, equipped with built-in compliance and tracking features, may have the potential to streamline these processes (Wegner et al., 2024).

In conclusion, although policies related to materials and data sharing are essential for protecting intellectual property and ensuring ethical research practices, they can become significant barriers to team science if not managed effectively. The complexities and time-consuming nature of negotiating material transfer and data use agreements can deter researchers from engaging in collaborative efforts, thereby slowing scientific progress and innovation. By pursuing open-science policies when possible, standardizing agreements, providing centralized legal support, creating clear guidelines, offering incentives, and leveraging technology, institutions can reduce these barriers and promote a more collaborative and productive research environment.

# **Ethical Approvals**

For research involving human participants, ethical approval from institutional review boards (IRBs) is one of the most important—and sometimes one of the most challenging—administrative tasks (Oakes, 2002). This generalization can often be especially true for practitioners of team science, for whom the challenge of obtaining ethics approval in a single discipline and institution is multiplied by navigating multiple IRBs across institutions (McWilliams et al., 2003; Peek et al., 2021). Furthermore, this situation can

result in substantial variation in ethics review outcomes, which means that participants may be protected to greater or lesser degrees depending on the vagaries of local review (Caulfield et al., 2011). This situation has improved in recent years but can remain challenging in the context of international or non-university collaborators.

Prior to 2018, researchers engaged in cross-institutional research were typically required to pursue independent review by local IRBs. This situation required substantial duplication of effort and could easily lead to difficult situations in which different IRBs required conflicting alterations to a protocol. NIH addressed this issue in 2018 by revising the Common Rule to require a single IRB to be designated as the primary site for review of multisite projects involving human participants (U.S. Department of Health and Human Services, 2018). IRBs for other participating institutions are required to create a reliance agreement with the primary IRB. These agreements substantially decrease the burden of review by independent sites but still require a significant investment of resources by participating investigators (Resnik et al., 2018). Evidence from a similar policy instituted earlier by NCI indicates that a single-IRB process led to substantially more efficient review and higher satisfaction for investigators (Massett et al., 2018). Consistent with this evidence, a qualitative study of single-IRBs in clinical trials regulated by the Food and Drug Administration showed generally high satisfaction with the process (Corneli et al., 2021). Unfortunately, single-IRB policies driven by the Common Rule do not apply to international research with human participants; hence, multisite international collaborations can remain quite challenging from the perspective of ethical approvals.

# Hiring, Tenure, Promotion, and Reward Policies and Procedures

Historically, academic success has been measured by individual achievements such as the number of first-authored publications, personal citation counts, and individual grant awards. These criteria are often used in hiring, tenure, and promotion decisions to assess a candidate's contributions and impact within their field. Although these metrics can provide a quantifiable measure of an individual's academic performance, hiring, tenure, and promotion policies and procedures that focus exclusively on these metrics can disincentivize collaborative efforts (Bouwma-Gearhart et al., 2021). Indeed, the focus on individual achievements can create a competitive environment where researchers are less likely to engage in team science (National Academies, 2020). Collaborative projects, which often involve shared credit and coauthorship, may be viewed as less valuable during tenure and promotion evaluations. As a result, researchers might avoid interdisciplinary collaborations and/or team-based research out of fear that their contributions will not be recognized nor rewarded. To foster a culture of team science,

universities may need to revise these policies to encourage and reward collaboration (Klein & Falk-Krzesinski, 2017).

## Hiring Practices

University hiring policies and procedures can support interdisciplinary collaboration by emphasizing the value of differing expertise and teamwork in the recruitment process. For example, job descriptions could explicitly state the importance of interdisciplinary research and collaborative skills, ensuring that candidates understand the institution's commitment to team science. Hiring committees can be composed of members from various disciplines to evaluate applicants' potential for cross-disciplinary work effectively. Additionally, universities can prioritize candidates with a proven track record of successful collaborations, assessing their ability to integrate different perspectives and contribute to multifaceted projects (Klein & Falk-Krzesinski, 2017). Joint appointments across departments or faculties, for instance through interdisciplinary research centers, can further encourage interdisciplinary engagement by encouraging researchers to bridge gaps between fields and foster a culture of cooperation (Hart & Mars, 2008; Yang et al., 2020). Providing resources and support for collaborative research, such as funding for interdisciplinary projects and access to shared facilities, can also enhance the appeal of the institution to prospective hires. A useful tool to support collaboration across units is a memorandum of understanding (MOU). These can be established between participating units to ensure transparency. Like a team charter, MOUs can help establish a mutual understanding on roles and responsibilities and financial commitments. They can also specify the administrative contacts, level of expected effort, and the general scope of the work (Platt et al., 2024). Klein & Falk-Krzesinski (2017) added that MOUs not only define expectations but also can specify what is required for teaching and service across the participating departments, as well as percentages of time the employee dedicates to each. By integrating these elements into hiring policies and procedures, universities can create an environment that attracts and nurtures researchers committed to interdisciplinary collaboration.

From an industry perspective, having a proven track record of being able to successfully collaborate across teams in different institutions and geographies may provide candidates with an advantage over those with a more siloed research experience. Industry considers the ability to effectively collaborate across disciplines and business groups (e.g., supply chain, manufacturing, sales, marketing) as part of a baseline set of skills that is critical to success (Schwartz et al., 2019).

In March 2022, the U.S. House and Senate passed the America COMPETES (America Creating Opportunities to Meaningfully Promote

Excellence in Technology, Education, and Science) Act, which was a reauthorization of the COMPETES Act of 2007. The basis of this act was to enhance U.S. economic and technological leadership by investing in "innovation through research and development, and to improve the competitiveness of the United States" (America COMPETES Act, 2007). One key provision of the act called for investing in research and development that emphasized science, technology, engineering, mathematics, and medicine (STEMM) education and workforce development. The act acknowledges the need for a more representative workforce in STEMM fields and calls for measures to broaden participation.

Systemic barriers can be found in hiring, promotion, and recruitment practices, as well as resource allocation and power distribution (Bhalla, 2019; Li & Koedel, 2017; National Academies, 2023). These biases may undermine initiatives such as America COMPETES. For example, Milkman et al. (2015) found significant biases in professors' responses to prospective doctoral students, with fewer responses going to applicants who were women or from a minoritized group. The same study illustrated the need to not focus solely on one part of the barrier, but in the case of hiring and recruitment, that attention be given to both increasing individual differences on the supply side (e.g., by focusing on improving the heterogeneity of the applicant pool) and reducing bias on the demand side (e.g., by addressing personally held biases within the hiring and research context). Sege et al. (2015) summarized their preliminary findings in a research letter in which they found that female early career faculty received significantly less start-up support than their male counterparts. Nguyen et al. (2023) found disparities in super principal investigator representation across gender and racial lines, with significant gaps for women and Black researchers (super principal investigator was defined as investigators who hold three or more concurrently active research grants). Despite a threefold increase in super principal investigators between 1991 and 2020, women and Black super principal investigators remained significantly underrepresented, even after controlling for factors such as career stage and educational background (Nguyen et al., 2023). The intersection of race and gender creates compounded disadvantages, most notably for Black women principal investigators, who were 71 percent less likely to achieve super principal investigator status than White male principal investigators (Nguyen et al., 2023). This stark disparity suggests that systemic barriers go beyond applicant pool issues and points to deeper institutional challenges in advancement opportunities, mentoring systems, resource allocation, and institutional support for underrepresented scientists. As a result, long-standing gender imbalances can create additional challenges for women and racially minoritized individuals seeking to advance in their careers (Casad et al., 2021; Charlesworth & Banaji, 2019; National Academies, 2023).

#### Tenure and Promotion

Tenure and promotion policies and procedures can significantly support team science by recognizing and rewarding collaborative efforts alongside individual achievements (Carter et al., 2021). Promotion and tenure systems are often criticized for their lack of transparency and potential biases. Traditional metrics for evaluation, such as publication count, journal impact factors, and grant money, may inadvertently disadvantage certain groups of individuals, such as women (Misra et al., 2011). To foster a culture of interdisciplinary collaboration, universities could revise evaluation criteria to include contributions to team-based research projects, coauthored publications, and collaborative grant awards (Klein & Falk-Krzesinski, 2017; Meurer et al., 2023) and provide clarity for the evaluators and investigators on the criteria (Potter et al., 2024). For example, Meurer et al. (2023) outlined criteria that can be used to evaluate how well a faculty member helps with team effectiveness (e.g., "develop consensus around shared research goals," "provide participatory, inclusive, and empowering leadership" [p. 4]), as well as how they contribute to the scientific process (e.g., assist in developing research questions, engage in data analyses and interpretation). Klein & Falk-Krzesinski (2017) further noted that existing policy documents related to tenure and promotion need to be evaluated to ensure they do not penalize or marginalize interdisciplinarity and team science. As such, evaluation criteria would consider investigators who may contribute a high level of expertise in a methodological or theoretical approach (e.g., qualitative/quantitative methods or community engagement), who are sought out as collaborators on multiple projects and grants for their unique skills and may or may not have their own line of research. Metrics can be expanded to assess the impact and quality of collaborative work, rather than focusing solely on the quantity of solo-authored outputs (Meurer et al., 2023). Allowing tenure and promotion candidates to submit narrative statements that detail their role in team science, the significance of their collaborative contributions, and the outcomes of such efforts can provide a more comprehensive evaluation of their achievements. Additionally, recognizing leadership roles within collaborative projects, such as coordinating research teams or managing interdisciplinary initiatives, can highlight the importance of teamwork in achieving significant scientific advancements. By incorporating these elements into tenure and promotion criteria, universities can incentivize researchers to engage in team science, ultimately enhancing the institution's research capabilities and addressing complex, multifaceted problems through a collaborative approach. For example, the University of Southern California (2022) policy on promotion and tenure provides explicit guidelines on the use of external letters and the candidate's research statement to clarify unique contributions to team science.

Academic departments can inadvertently create barriers to team science based on the journals they privilege for recognition and reward. When departments exclusively reward publications in disciplinary journals, they often send a message that only research contributions within that single discipline are valuable for their employees to pursue (Mäkinen et al., 2024). These metrics can discourage constituent scientists from pursuing interdisciplinary collaborations and/or publishing in journals that reach broader audiences, as these endeavors may be seen as less prestigious or beneficial for career advancement. On the other hand, departments have the potential to foster collaboration and interdisciplinary research by rewarding publications based on their overall quality, regardless of the specific journal (Frank, 2019). By recognizing and valuing publications across a range of disciplines, departments can encourage scientists to engage in team science, share knowledge across fields, and contribute to more comprehensive and innovative research. This approach not only supports the career growth of individual researchers but also has the potential to enhance the overall quality and societal impact of the scientific work produced (Arnold et al., 2021; Carter et al., 2021; Mazumdar et al., 2015; Moher et al., 2018).

To value and support team science, tenure and promotion policies and procedures may need to be expanded to include collaborative leadership, interdisciplinary contributions, and the effective management of research teams. This may mean recognizing the critical roles that researchers play in team-based projects, from coordinating large, multifaceted studies to facilitating communication and cooperation among team members. It also may mean recognizing the longer timescale of many team science projects (Hall et al., 2012). Incorporating diversity into the promotion and tenure criteria can incentivize faculty to engage in activities that promote values such as mentoring diverse students or leading community-based projects (Stewart & Valian, 2018). This also recognizes and gives credit for the valuable mentoring and community-engaged work (Sotto-Santiago et al., 2023). This multifaceted approach could be considered more akin to how industry treats promotions, where promotions are generally based on technical, business, and organizational impact, all of which may require strong collaboration skills, interpersonal effectiveness, and teamwork.

Additionally, tenure committees could understand and appreciate the unique dynamics of team science, which often involve substantial time investments in team formation, coordination, and ongoing communication (Klein & Falk-Krzesinski, 2017). Revising tenure criteria to reflect these aspects can help institutions foster a culture that not only incentivizes but also rewards collective scientific efforts. By doing so, they can promote an environment where collaborative research is valued, leading to greater scientific innovation and impact. This approach not only benefits individual researchers by recognizing their contributions to team efforts but also can

advance the institution's overall research mission by encouraging interdisciplinary and high-impact projects.

#### **FUNDERS**

Major funders of research within the United States and globally have long understood the value of team science for addressing priority challenges and accelerating innovation. For instance, NSF funded the National Socio-Environmental Synthesis Center (SESYNC)<sup>6</sup> in 2011 as a first-ofits kind center focused on growing researchers' team science capacities to address interdisciplinary problems collaboratively at the interface of humans and the environment. As another example, the NIH National Institute on Minority Health and Health Disparities (2024) launched the Transdisciplinary Collaborative Centers for Health Disparities Research Program in 2012 to support "collaboration at the regional level [...] because it provides opportunities for institutions and organizations to achieve a broader reach than is possible with isolated local efforts while combining expertise and resources in an era of constrained budgets" (Section I).

As described in Chapter 2, interest in funding team science since the publication of the 2015 report has continued (National Research Council, 2015). In 2017, NSF amplified its support of team science by prioritizing foundation-wide investments in cross-disciplinary collaboration and creative cross-sectoral partnerships, establishing "Growing Convergence Research" as one of its 10 Big Ideas. Also important to consider is whether funding agencies' signaled interests in and support for team science efforts align with other aspects of the research funding landscape, such as proposal solicitations and review and project funding.

## Proposal Solicitations and Review

How funders solicit and review research proposals, including funding opportunity language and requirements, can have implications for team science effectiveness. For example, NSF's (2024) call for Science and Technology Centers (STCs) allows for the inclusion of team science–related requirements at all levels, including in the proposal content (e.g., "Highlight the unique assets and strengths, including the diversity of experiences and perspectives, of the proposing team compared to other groups working in related areas" and "Identify specific activities and mechanisms that will enable cross-organizational and cross-sector integration of the team") (p. 11)

<sup>&</sup>lt;sup>6</sup> More information about SESYNC is available at https://www.sesync.org

 $<sup>^7</sup>$  More information about NSF's 10 Big Ideas is available at <code>https://www.nsf.gov/news/special\_reports/big\_ideas/index.jsp</code>

and solicitation-specific review criteria (e.g., "Is the team of partner organizations and personnel assembled for the proposed Center appropriate, essential and consistent with the solicitation? Is the role of each participant clear? Does the partnership have unique strengths?") (p. 20).8

NSF's Engineering Research Center (ERC) program has prioritized team science in its solicitations. For example, in 2018, for applicants who made it past the preproposal stage for ERCs, NSF funded attendance at a team science workshop led by SESYNC, which featured sessions and talks led by team science experts from across the country with goals of elevating and strengthening team science in ERC proposals. Actions taken by programs such as the STC and ERC encourage proposal teams to factor team science principles and best practices into their project design (and budget) from inception; they also allow NSF as the funder to weigh team science-specific concerns in its decision-making processes.

The degree to which the STC and ERC calls explicitly prioritize and integrate team science is not typical of proposal solicitations from funding agencies; however, more subtle proposal requirements, resources, and policies can still influence team science outcomes. For example, NIH allows for a multiple principal investigator option on grant applications. This option requires that applicants include a leadership plan that addresses the responsibilities of each principal investigator, as well as publication policies and procedures for resolving conflicts.9 To aid in plan development, the National Center for Advancing Translational Sciences offers agreement templates and discussion questions<sup>10</sup> to guide collaborators (see also the discussion on team charters in Chapter 3). Similarly, at NSF, team proposals are increasingly required to develop a collaboration plan that reflects on the rationale for the collaboration while detailing team plans for communication, governance, integration, project management, and more (e.g., see NSF I-Corps Teams<sup>11</sup>). NSF's invitation to submit collaborative proposals, where multiple institutions submit as equal partners, may additionally spur more productive collaborations as this option can save both time and money by eliminating subawards and equalizing power dynamics between collaborating entities.

Although some aspects of the proposal development process have changed in recent years in response to the rise of team science, the committee discussed that the proposal review process has largely stayed the

<sup>&</sup>lt;sup>8</sup> More information about the program solicitation is available at https://new.nsf.gov/funding/opportunities/science-technology-centers-integrative-partnerships/nsf24-594/solicitation

 $<sup>^9</sup>$  For more information about NIH's option for multiple principal investigators, see https://grants.nih.gov/grants-process/plan-to-apply/consider-your-idea-resources-and-collaborators/multiple-principal-investigators

<sup>&</sup>lt;sup>10</sup> See https://ncats.nih.gov/research/alliances/forms-and-model-agreements

<sup>&</sup>lt;sup>11</sup> See https://www.nsf.gov/funding/initiatives/i-corps/about-teams

same. Some programs, such as the STC, have begun to include additional solicitation-specific review criteria and thus ask for evaluation of team science aspects of the proposed activities. Largely, however, proposals to major funding agencies are not assessed based on team science activities. If team science is not included in review criteria, investigators will not be incentivized to consider team planning and organization, in turn potentially decreasing the effectiveness of their teams.

# **Project Funding**

Which proposals are funded, and how recipients are permitted to use those funds, can also have implications for team science. Support for team science training, for example, is most readily available in large center grants or training grants (e.g., National Institute of General Medical Sciences, 2025); however, team science approaches are important for teams and projects of all sizes (Hall et al., 2018). There is evidence showing that, compared with investigator-initiated research grants, NIH-funded transdisciplinary center grants faced an initial lag in productivity; however, they saw overall advantages for productivity and collaboration (Hall et al., 2012). The presence of such a lag suggests that the typical 3- to 5-year length of research awards may be too short to allow for proper planning, trust, relationships, shared vision, integration, dissemination or translation of project results, and other key aspects of team science to occur. Furthermore, planning grants, seed funding, travel for face-to-face meetings, and project coordinators can benefit science teams, yet such funding may not always be available or allowable. With the 26% federal cap on indirect administrative costs, which was implemented in 1991 (University of California, n.d.), budgetary restrictions on costs that are positively linked with team science outcomes means that award recipients may take on greater financial burdens to keep pace with the needs and costs of conducting team science effectively (Hall et al., 2018).

Beyond lack of support for team science components of funded proposals, a lack of funding for *research* on science teams is also problematic. More specifically, funding is lacking tremendously for those who want to study team science effectiveness. This creates a research-to-practice gap, in that the prevalence of science teamwork outpaces the ability to guide it. Research associated with science teams can often be viewed in a supporting or evaluative role rather than as a necessary research field in and of itself. As has been made clear throughout this report, however, there is a lack of strong evidence linking team science interventions or best practices to positive team science outcomes. With little funding made available for research on science teams, funding agencies may continue to rely upon guidance for funding and supporting science teams that is not fully informed by rigorous research (e.g., Berg, 2017; Kaiser, 2017).

Finally, funding agencies can directly or indirectly disincentivize team science through other policies as well. For example, tightening research security restrictions—such as the NIH (2023) laboratory notebook policy that requires foreign grant subrecipients to provide yearly copies of all lab notebooks and documentation to the primary grant recipient—can impose significant administrative burdens and potential privacy concerns that could dissuade international collaboration. As another example, programs supported by the 2018 Farm Bill limit total indirect costs for an entire award to 30%, meaning that all collaborating institutions would be in a position to recover costs from a single small pool of funds rather than independently at their federally negotiated indirect cost rate (U.S. Department of Agriculture, 2018). If collaborations cost institutions more than they benefit them, advances in team science are likely to stagnate.

#### SCIENTIFIC INCENTIVES

As discussed throughout this chapter, researchers are highly responsive to scientific incentives as imposed by institutions, funders, and their broader field. To the extent that these incentives reward team science, researchers will feel empowered to participate in team science; in cases where incentives are unfavorable, they may not be able to participate. This section discusses incentives that (dis-)incentivize contribution to team science.

# Authorship, Contributorship, and Credit

Scientific publications serve as the primary currency in academia, where a researcher's career trajectory often hinges on the number of peer-reviewed articles listing them as an author. Scientometric indices, such as the h-index or the i10, index further quantify researchers' impact based on citation rates of their publications (Mingers & Leydesdorff, 2015). The importance of these authorship metrics in hiring, tenure, and grant decisions is well documented (e.g., see Moher et al., 2018, for a review). Yet these metrics do not reflect the reality that authors contribute in very different ways to projects. According to the International Committee of Medical Journal Editors (n.d.), listed authors are those who contributed to writing the paper and to the research being reported, approved the final manuscript, and agreed to be held accountable for the work; however, authorship standards vary widely across fields and even from journal to journal (Yale University, n.d.).

The authors on a paper are given as a list of names, but parsing out the contributions of each name is very difficult (e.g., Venkatraman, 2010). In standard biomedical authorship conventions, first authorship indicates primary responsibility for the scientific work described in the article (and,

typically, true "authorship"—that is, having written the first draft of much or all the text of the article). In contrast, "senior" or last authorship typically indicates supervisory responsibility. For others, the order indicates significance of contributions. In fields following this convention, first-author positions are especially coveted, sometimes leading to the presence of "multiple first authors," as indicated by a note that several authors have contributed equally. In still other fields, such as economics, authorship order is typically alphabetical, and the assumption is that all authors made equal contributions. For external readers, especially those with limited knowledge of a particular field, it can be mystifying to decode the contributions of individual authors from the list.

Notions of authorship are even more challenging in team science (Coles et al., 2023). Increasingly, scientific papers have hundreds or even thousands of authors, a situation termed *hyperauthorship* (Cronin, 2001; Nogrady, 2023). One wonders how to apportion credit and responsibility among so many. In addition, the threshold for what constitutes authorship can be challenging to delineate within a science team when participants' contributions differ so widely. To address these issues, in 2015, the CRediT (Contributor Roles Taxonomy) introduced a taxonomy of contributions to scholarly work, ranging from conceptualization to funding acquisition (Brand et al., 2015). The intent of this taxonomy was to allow researchers to designate the role or roles that each contributor played in a project, obviating guesswork about how contributions were distributed across an author list. Since its introduction, a wide variety of journals have adopted CRediT (Allen et al., 2019), and in 2022, it was adopted as an American National Standards Institute (2022) standard (Z39.104-2022).

Another challenge for science teams is considering how team members can contribute to a multiauthor paper. Borer et al. (2023) provided a model for a collaborative process for writing papers with large numbers of authors, in which the lead author(s) establishes a "storyboard" for the paper and solicits specific and targeted contributions—for example, providing citations, creating figures, considering alternative hypotheses, and other moderately sized tasks—in an iterative, deadline-driven process. Frassl et al. (2018) provided a complementary set of rules for large teams to create a collaborative and productive writing process.

Synthesizing this set of issues, science teams navigating questions of authorship need to establish clear rules for what constitutes authorship, how to track the different kinds of contributions that authors may make to a project, and how to establish opportunities for team members to contribute to the written product (e.g., Baumgartner et al., 2023). Considerations for fairness and accountability should be made (International Committee of Medical Journal Editors, n.d.), to ensure that all participating authors such as those who are early career or nontraditional team members can

be included. Returning to the guidance of Chapter 3, science teams could consider adopting such policies in their team charters.

#### Journals, Societies, and the Broader Field

Journals are the venues in which credit and authorship are determined, and their policies can provide powerful incentives and disincentives for team science. For example, a journal's requirement for individual author data to be entered into a web portal is not onerous for a small team, but for a team with hundreds or thousands of authors, this requirement can be prohibitive for submission. Allowing batch upload of name, affiliation, and Open Researcher and Contributor ID (ORCID) or other identifiers can make a journal accessible to larger teams. In addition, many journals ask for individual authors to provide explicit confirmation for submission (e.g., by way of email notification). This well-intentioned confirmation can inadvertently create a difficult situation for hyperauthored papers, in which a single contributor whose contact information changed can prevent a submission from going forward.

One potential solution for larger science teams is to pursue "group authorship" (also known as "consortium authorship"), where a single group name is given in place of a list of individual authors. This solution appears appealing but can have a variety of unintended consequences (Hosseini et al., 2024). Group authorship can pose technical challenges in associating the manuscript with the contributors' citation records, which can decrease incentives for contribution. In addition, issues of legal and ethical responsibility for the manuscript can be more problematic when individuals are not listed as authors (Hosseini et al., 2024). Thus, group authorship may not be a panacea; instead, journals ought to create policies that facilitate large teams in navigating the submission and publication process.

Scholarly societies provide another venue for recognition of scientific contributions, which can potentially facilitate team science. Many societies control important journals in their field and set their authorship policies. In addition, they often provide recognition of their members' accomplishments through awards. If these awards can be given to teams (e.g., as in the Society for the Improvement of Psychological Science Mission Award<sup>12</sup>) or to team leaders, this simple step can promote participation and change incentives for team participation.

Finally, embracing open-science values (Christensen et al., 2020; National Academies, 2018) can provide positive support for team science.

<sup>&</sup>lt;sup>12</sup> More information about the the SIPS Mission Award is available at https://improving-psych.org/mission/awards

Many large teams use open practices (see Box 4-1) and many teams also produce shared resources that benefit their field as a whole (Nogrady, 2023). Support—and recognition for the contributions of—open science can facilitate recognition of the many contributions that teams make beyond specific publications. Box 4-2 highlights a project that used open-science practices in a team science context to track research products and incentivize sharing of code, data, and materials.

# BOX 4-2 Using Open Products and Resource-Tracking to Align Team Science Incentives

The Aligning Science Across Parkinson's (ASAP) initiative is an international, foundation-funded collective of researchers across institutions whose goal is to accelerate discoveries about Parkinson's disease. As part of this initiative, the group mandates compliance with a range of open-science policies, including posting of preprints, sharing of all research products (including materials, code, and data) to appropriate repositories, and using research identifiers such as digital object identifiers and research resource IDs for tracking contributions (Riley & Schekman, 2021). These practices facilitate collaboration within the network, ensure rapid dissemination of all products of network grants, and ensure that originators of the materials receive credit for their contributions.

Dumanis et al. (2023) described this move to open-science practices as initially challenging, in part because many investigators and trainees had limited education about how to curate and deposit research products. To facilitate compliance, they pursued a two-part approach. First, they used automated tools to track all manuscripts that were funded by the initiative and to identify whether resources were deposited and documented appropriately. By applying these tools before manuscripts were submitted, they were able to give feedback about what steps authors needed to take to achieve compliance with ASAP policies. Second, they developed training resources to ensure that participating laboratories had guidance in curating and depositing their research products.

Critically, these efforts have allowed for open data from funded Parkinson's investigators to be consolidated using team science approaches, accelerating discoveries about the disease while at the same time creating routes for individual investigators to receive credit (Junker et al., 2021). As this case study shows, open-science practices require support for science teams—through education and compliance tracking—but they can help to alleviate some of the disincentives for team science.

#### CONCLUSIONS AND RECOMMENDATIONS

Conclusion 4-1: Institutions wishing to foster the study of team science would benefit from reviewing the incentive structures that influence individuals' decisions about engaging in such research. Specifically, many policies and practices that are currently in place surrounding tenure and promotion, authorship, cost-sharing, allowable costs, and resource-sharing appear to discourage engagement in the study of team science and participation in collaborative science teams. Institutional processes can also reinforce disparities in team science by failing to properly recognize work in team science, including community-engaged research and mentorship.

Recommendation 4-1: Academic departments should adapt their promotion and tenure processes to acknowledge and reward the contributions of researchers who take on the additional professional responsibilities associated with participating in and studying team science by:

- a. recognizing and incorporating in tenure evaluations the valuable contributions made through different authorship roles, publications in interdisciplinary or nontraditional journals, and processoriented outcomes.
- b. ensuring that the demographic and disciplinary composition of promotion and tenure committees is both reflective and independent of the candidates they are reviewing to the extent possible to facilitate increased representation in science team leadership.
- c. revising criteria for selection to ensure fair consideration of candidates who research and team with underrepresented communities and/or do community-engaged research.
- d. considering candidates' involvement in committees, initiatives, and other activities, including engagement outside the scientific community, that offer value to science teams and encourage participation.
- e. reviewing the promotion and tenure process periodically, for instance every several years, to ensure continuous improvement and minimize inefficiencies.

Recommendation 4-2: Science journal editors should establish comprehensive systems and policies to build team science into the publishing mainstream, including:

a. conducting a systematic assessment to identify barriers that may limit the incorporation of team science literature in their journal. These findings should be used to develop actionable strategies to address identified barriers.

- b. adopting clear policies and guidelines for authorship, allocation of credit, and contribution level, particularly when there are many contributing authors and when authors are contributing from different disciplines with different authorship practices. Policies can include strategies for addressing authorship disputes.
- c. allocating appropriate space to publish research on the science of team science.

Recommendation 4-3: Funders of team science, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should integrate team science needs into funding programs and policies and should remove barriers to team science efficacy by:

- a. including team science needs, such as team-building, travel and meetings, professional development and training, and resource-sharing in allowable costs.
- b. allowing for the inclusion of nonscientist team members and leaders in the project budget to allow for the compensation of their time.

Recommendation 4-4: Institutions seeking to advance team science effectiveness should allocate resources to support science teams. Resource allocation may cover, but is not limited to the following:

- a. resources that mitigate the operational burden on science teams by investing in the support of administrative staff.
- b. training and mentorship for research administrative staff working on team science projects.
- c. resources to expand access to and the use of technologies that optimize team participation for geographically dispersed members.
- d. funding to address gaps in the physical, digital, and procedural environment. This could include applying universal design principles that accommodate the needs of all individuals.

#### REFERENCES

- Adler, H. J. (2025). Language complexities for deaf and hard of hearing individuals in their pursuit of a career in science, technology, engineering, mathematics, and medicine: Perspectives from an LSL/ASL user. *Ear and Hearing*. https://doi.org/10.1097/aud.000000000001637
- Ahmed, N., Wahed, M., & Thompson, N. C. (2023). The growing influence of industry in AI research. *Science*, 379(6635), 884–886. https://doi.org/10.1126/science.ade2420
- Allen, L., O'Connell, A., & Kiermer, V. (2019). How can we ensure visibility and diversity in research contributions? How the Contributor Role Taxonomy (CRediT) is helping the shift from authorship to contributorship. *Learned Publishing*, 32(1), 71–74. https://doi.org10.1002/leap.1210

- Alter, G., & Gonzalez, R. (2018). Responsible practices for data sharing. *American Psychologist*, 73(2), 146–156. https://doi.org/10.1037/amp0000258
- America COMPETES Act, P. L. 110-69, 121 Stat. 572. (2007). https://www.congress.gov/110/plaws/publ69/PLAW-110publ69.pdf
- American National Standards Institute. (2022). ANSI/NISO Z39.104-2022, CRediT, Contributor Roles Taxonomy. https://www.niso.org/publications/z39104-2022-credit
- Americans with Disabilities Act. (2010). 2010 ADA Standards for Accessible Design. https://www.ada.gov/law-and-regs/design-standards/2010-stds/
- Anderson, J., Anderson, Z., Beaton, K., Bhandari, S., Bultinck, E., Ching, J., Clark, H., Ho, L., Holloway, R., Hopping, L., Hrosz, M., Hrvojevic, D., Huneycutt, A., Iglesias, J., Jogopulos, J., Joshi, S., King, T., Klug, M., LaMonaca, G., ... Duerstock, B. S. (2022). Challenges in inclusiveness for people with disabilities within STEM learning and working environments. Purdue. https://docs.lib.purdue.edu/cgi/viewcontent.cgi?article=1004&context=ugcw
- Apon, A., Ahalt, S., Dantuluri, V., Gurdgiev, C., Limayem, M., Ngo, L., & Stealey, M. (2010). High performance computing instrumentation and research productivity in US universities. *Journal of Information Technology Impact*, 10, 87–98.
- Arnold, A., Cafer, A., Green, J., Haines, S., Mann, G., & Rosenthal, M. (2021). Perspective: Promoting and fostering multidisciplinary research in universities. *Research Policy*, 50(9), 104334. https://doi.org/10.1016/j.respol.2021.104334
- Bammer, G., O'Rourke, M., O'Connell, D., Neuhauser, L., Midgley, G., Klein, J. T., Grigg, N. J., Gadlin, H., Elsum, I. R., Bursztyn, M., Fulton, E. A., Pohl, C., Smithson, M., Vilsmaier, U., Bergmann, M., Jaeger, J., Merkz, F., Baptista, B. V., Burgman, M. A., Walker, D. H., Young, J., Bradbury, H., Crawford, L., Haryanto, B., . . . Richardson, G. P. (2020). Expertise in research integration and implementation for tackling complex problems: When is it needed, where can it be found and how can it be strengthened? Palgrave Communications, 6, Article 5. https://doi.org/10.1057/s41599-019-0380-0
- Barker Scott, B. A., & Manning, M. R. (2024). Designing the collaborative organization: A framework for how collaborative work, relationships, and behaviors generate collaborative capacity. *The Journal of Applied Behavioral Science*, 60(1), 149–193. https://doi.org/10.1177/00218863221106245
- Baumgartner, H. A., Alessandroni, N., Byers-Heinlein, K., Frank, M. C., Hamlin, J. K., Soderstrom, M., Voelkel, J. G., Willer, R., Yuen, F., & Coles, N. A. (2023). How to build up big team science: A practical guide for large-scale collaborations. *Royal Society Open Science*, 10(6), 230235. https://doi.org/10.1098/rsos.230235
- Bennett, L., Gadlin, H., & Levine-Finley, S. (2010). Collaboration and team science: A field guide. National Institutes of Health. https://www.cancer.gov/about-nci/organization/crs/research-initiatives/team-science-field-guide/collaboration-team-science-guide.pdf
- Bennett, L. M., & Gadlin, H. (2012). Collaboration and team science: From theory to practice. *Journal of Investigative Medicine*, 60(5), 768–775. https://doi.org/10.2310/jim.0b013e318250871d
- Bercaru, V., & Popescu, N. (2024). A systematic review of accessibility techniques for online platforms: Current trends and challenges. *Applied Sciences*, 14(22), 10337. https://doi. org/10.3390/app142210337
- Berg, J. (2017, June 6). Research output as a function of grant support: The scatter matters. *Science*. https://www.science.org/content/blog-post/research-output-function-grant-support-scatter-matters
- Bergeron, E. (2021, June 14). The historical roots of mistrust in science. *American Bar Association*. https://www.americanbar.org/groups/crsj/publications/human\_rights\_magazine\_home/the-truth-about-science/the-historical-roots-of-mistrust-in-science/

- Besiroglu, T., Bergerson, S. A., Michael, A., Heim, L., Luo, X., & Thompson, N. (2024). The compute divide in machine learning: A threat to academic contribution and scrutiny? *arXiv*. https://arxiv.org/abs/2401.02452
- Bhalla, N. (2019). Strategies to improve equity in faculty hiring. *Molecular Biology of the Cell*, 30(22), 2744–2749. https://doi.org/10.1091/mbc.E19-08-0476
- Borer, E. T., MacDougall, A. S., Stevens, C. J., Sullivan, L. L., Wilfahrt, P. A., & Seabloom, E. W. (2023). Writing a massively multi-authored paper: Overcoming barriers to meaningful authorship for all. *Methods in Ecology and Evolution*, 14(6), 1432–1442. https://doi.org/10.1111/2041-210X.14096
- Borgman, C. L. (2012). The conundrum of sharing research data. *Journal of the American Society for Information Science and Technology*, 63(6), 1059–1078.
- Bouwma-Gearhart, J., Carter, R., & Mundorff, K. (2021). A call for promoting faculty innovation and entrepreneurship. *Change: The Magazine of Higher Learning*, 53(2), 18–24. https://doi.org/10.1080/00091383.2021.1883973
- Braga, P. H. P., Hébert, K., Hudgins, E. J., Scott, E. R., Edwards, B. P. M., Sánchez Reyes, L. L.,
  Grainger, M. J., Foroughirad, V., Hillemann, F., Binley, A. D., Brookson, C. B., Gaynor,
  K. M., Shafiei Sabet, S., Güncan, A., Weierbach, H., Gomes, D. G. E., & Crystal-Ornelas,
  R. (2023). Not just for programmers: How GitHub can accelerate collaborative and reproducible research in ecology and evolution. *Methods in Ecology and Evolution*, 14(6),
  1364–1380. https://doi.org/10.1111/2041-210x.14108
- Brand, A., Allen, L., Altman, M., Hlava, M., & Scott, J. (2015). Beyond authorship: Attribution, contribution, collaboration, and credit. *Learned Publishing*, 28(2), 151–155.
- Bubela, T., Guebert, J., & Mishra, A. (2015). Use and misuse of material transfer agreements: Lessons in proportionality from research, repositories, and litigation. *PLoS Biology*, 13(2), e1002060. https://doi.org/10.1371/journal.pbio.1002060
- Burgstahler, S. (2012) Making science labs accessible to students with disabilities. DO-IT, University of Washington. https://www.washington.edu/doit/sites/default/files/atoms/files/Making-Science-Labs-Accessible-Students-Disabilities.pdf
- Byers-Heinlein, K., Bergmann, C., Davies, C., Frank, M. C., Hamlin, J. K., Kline, M., Kominsky, J., F., Kosie, J. E., Lew-Williams, C., Mastroberardino, M., Singh, L., Waddell, C. P. G., & Soderstrom, M. (2020). Building a collaborative psychological science: Lessons learned from ManyBabies 1. Canadian Psychology/Psychologie Canadienne, 61(4), 349–363.
- Carter, R. G., Mundorff, K., Risien, J., Bouwma-Gearhart, J., Bratsch-Prince, D., Brown, S. A., Campbell, A L., Hartman, J. C., Hasemann, C. A., Hollenbeck, P. J., Lupiani, B., McCarty, O. J. T., McClure, I. D., Mealey, K., Mimura, C., Romero, A. J., Sztajn, P., & Van Egeren, L. (2021). Innovation, entrepreneurship, promotion, and tenure. *Science*, 373(6561), 1312–1314. https://doi.org/10.1126/science.abj2098
- Carter, S., Carlson, S., Crockett, J., Falk-Krzesinski, H. J., Lewis, K., & Walker, B. E. (2019). The role of research development professionals in supporting team science. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), Strategies for team science success: Handbook of evidence-based principles for cross-disciplinary science and practical lessons learned from health researchers (pp. 375–388). Springer, Cham.
- Casad, B. J., Franks, J. E., Garasky, C. E., Kittleman, M. M., Roesler, A. C., Hall, D. Y., & Petzel, Z. W. (2021). Gender inequality in academia: Problems and solutions for women faculty in STEM. *Journal of Neuroscience Research*, 99(1), 13–23. https://doi.org/https://doi.org/10.1002/jnr.24631
- Caulfield, T., Ries, N., & Barr, G. (2011). Variation in ethics review of multi-site research initiatives. *Amsterdam Law Forum*, 3(1), 85–100.
- Chai, S., & Shih, W. (2016). Bridging science and technology through academic–industry partnerships. *Research Policy*, 45(1), 148–158. https://doi.org/10.1016/j.respol.2015.07.007

- Charlesworth, T. E. S., & Banaji, M. R. (2019). Gender in science, technology, engineering, and mathematics: Issues, causes, solutions. *The Journal of Neuroscience*, 39(37), 7228–7243. https://doi.org/10.1523/jneurosci.0475-18.2019
- Chedin, E. M. (2024). The critical role of grants facilitation for supporting individual and team science. In J. Markovac, K. E. Barrett, & H. Garrison (Eds.), *Life science careers:* Perspectives in physiology. Springer, Cham. https://doi.org/10.1007/978-3-031-50694-9\_17
- Christensen, G., Wang, Z., Levy Paluck, E., Swanson, N., Birke, D., Miguel, E., & Littman, R. (2020). *Open science practices are on the rise: The state of social science (3S) survey* (Working Paper Series No. WPS-106). Center for Effective Global Action, University of California, Berkeley.
- Clutterbuck, D. (2007). Coaching the team at work. Nicholas Brealey Publishing.
- Coles, N. A., DeBruine, L. M., Azevedo, F., Baumgartner, H. A., & Frank, M. C. (2023). 'Big team' science challenges us to reconsider authorship. *Nature Human Behaviour*, 7(5), 665–667. https://doi.org/10.1038/s41562-023-01572-2
- Collins, H., & Evans, R. (2019). Rethinking expertise. University of Chicago Press.
- Corneli, A., Dombeck, C. B., McKenna, K., & Calvert, S. B. (2021). The single IRB review process: Stakeholder experiences and recommendations for guidance by the U.S. Food and Drug Administration. *Ethics & Human Research*, 43(3), 26–36. https://doi.org/10.1002/eahr.500092
- Cravens, A. E., Jones, M. S., Ngai, C., Zarestky, J., & Love, H. B. (2022). Science facilitation: navigating the intersection of intellectual and interpersonal expertise in scientific collaboration. *Humanities and Social Sciences Communications*, 9, 256. https://doi.org/10.1057/s41599-022-01217-1
- Cronin, B. (2001). Hyperauthorship: A postmodern perversion or evidence of a structural shift in scholarly communication practices? *Journal of the American Society for Information Science and Technology*, 52(7), 558–569. https://doi.org/10.1002/asi.1097
- Crüwell, S., van Doorn, J., Etz, A., Makel, M. C., Moshontz, H., Niebaum, J. C., Orben, A., Parsons, S., & Schulte-Mecklenbeck, M. (2019). Seven easy steps to open science. Zeitschrift für Psychologie, 227(4), 237–248. https://doi.org/10.1027/2151-2604/a000387
- Cummings, J. N., & Kiesler, S. (2007). Coordination costs and project outcomes in multiuniversity collaborations. *Research Policy*, 36, 1620–1634. https://doi.org/10.1016/j. respol.2007.09.001
- Dai, Q., Shin, E., & Smith, C. (2018). Open and inclusive collaboration in science: A framework (OECD Science, Technology and Industry Working Papers No. 2018/07). Organisation for Economic Co-operation and Development Publishing. https://doi.org/10.1787/2dbff737-en
- Davies, A., Manning, S., & Söderlund, J. (2018). When neighboring disciplines fail to learn from each other: The case of innovation and project management research. *Research Policy*, 47(5), 965–979. https://doi.org/10.1016/j.respol.2018.03.002
- Defense Federal Acquisition Regulation Supplement. (2020). DFARS. https://www.acquisition.gov/dfars
- Devitz, A. C. (2023). Equipment repositories for accessibility: A model for improving access in field science. *Integrative and Comparative Biology*, 63(1), 98–107. https://doi.org/10.1093/icb/icad024
- de Vreede, G.-J., Antunes, P., Vassileva, J., Gerosa, M. A., & Wu, K. (2016). Collaboration technology in teams and organizations: Introduction to the special issue. *Information Systems Frontiers*, 18, 1–6. https://doi.org/10.1007/s10796-016-9632-3
- Díaz-Reviriego, I., Turnhout, E., & Beck, S. (2019). Participation and inclusiveness in the intergovernmental science–policy platform on biodiversity and ecosystem services. *Nature Sustainability*, 2(6), 457-464. https://doi.org/10.1038/s41893-019-0290-6

- Dumanis, S. B., Ratan, K., McIntosh, S., Shah, H. V., Lewis, M., Vines, T. H., Schekman, R., & Riley, E. A. (2023). From policy to practice: Lessons learned from an open science funding initiative. *PLoS Computational Biology*, 19(12), e1011626. https://doi.org/10.1371/journal.pcbi.1011626
- Emerson College. (2025). Inviting an external guest to a Zoom meeting. https://support.emerson.edu/hc/en-us/articles/360054483931-Inviting-an-external-guest-to-a-Zoom-meeting
- Ernst, A., Fischer-Hotzel, A., & Schumann, D. (2017). Transforming knowledge for sustainability: Insights from an inclusive science-practice dialogue on low-carbon society in Germany. *Energy Research & Social Science*, 29, 23–35. https://doi.org/10/g8km4x
- Forscher, P. S., Wagenmakers, E. J., Coles, N. A., Silan, M. A., Dutra, N., Basnight-Brown, D., & IJzerman, H. (2023). The benefits, barriers, and risks of big-team science. *Perspectives on Psychological Science*, 18(3), 607–623. https://doi.org/10.1177/17456916221082970
- Foster, E. D., & Deardorff, A. (2017). Open science framework (OSF). *Journal of the Medical Library Association*, 105(2), 203. https://doi.org/10.5195/jmla.2017.88
- Frank, M. C. (2019). N-best evaluation for academic hiring and promotion. *Trends in Cognitive Sciences*, 23(12), 983–985.
- Frassl, M. A., Hamilton, D. P., Denfeld, B. A., de Eyto, E., Hampton, S. E., Keller, P. S., Sharma, S., Lewis, A. S. L., Weyhenmeyer, G. A., O'Reilly, C. M., Lofton, M. E., & Catalán, N. (2018). Ten simple rules for collaboratively writing a multi-authored paper. *PLoS Computational Biology*, 14(11), e1006508. https://doi.org/10.1371/journal.pcbi.1006508
- Fujinuma, N., DeCost, B., Hattrick-Simpers, J., & Lofland, S. E. (2022). Why big data and compute are not necessarily the path to big materials science. *Communications Materials*, 3, 59. https://doi.org/10.1038/s43246-022-00283-x
- Gehret, A. U., Trussell, J. W., & Michel, L. V. (2017). Approaching undergraduate research with students who are deaf and hard-of-hearing. *Journal of Science Education for Students with Disabilities*, 20(1), 20–35. https://doi.org/10.14448/jsesd.08.0002
- General Services Administration. (2024). Section 508 of the Rehabilitation Act, as amended. https://www.section508.gov/manage/laws-and-policies/section-508-law/
- Giardullo, P. (2023). Non-experts' participation in processes of scientific knowledge creation: The case of citizen science. *Sociology Compass*, 17(9). https://doi.org/10.1111/soc4.13100
- Gopalan, G., Bunger, A. C., & Powell, B. J. (2022). Skills for developing and maintaining community-partnerships for dissemination and implementation research in children's behavioral health: Implications for research infrastructure and training of early career investigators. *Administration and Policy in Mental Health*, 47(2), 227–243. https://doi.org/10.1007/s10488-019-00930-5
- Graef, D. J., Motzer, N., & Kramer, J. G. (2021). The value of facilitation in interdisciplinary socio-environmental team research. Socio-Ecological Practice Research, 3(2), 109–113. https://doi.org/10.1007/s42532-021-00082-7
- Grahe, J. E., Cuccolo, K., Leighton, D. C., & Cramblet Alvarez, L. D. (2020). Open science promotes diverse, just, and sustainable research and educational outcomes. *Psychology Learning & Teaching*, 19(1), 5–20. https://doi.org/10.1177/1475725719869164
- Griffith, E. (2023, Aug. 16). The desperate hunt for the A.I. boom's most indispensable prize. The New York Times. https://www.nytimes.com/2023/08/16/technology/ai-gpu-chips-shortage.html
- Gudi, N., Kamath, P., Chakraborty, T., Jacob, A. G., Parsekar, S. S., Sarbadhikari, S. N., & John, O. (2022). Regulatory frameworks for clinical trial data sharing: Scoping review. *Journal of Medical Internet Research*, 24(5), e33591. https://doi.org/10.2196/33591

- Hall, K. L., Stokols, D., Stipelman, B. A., Vogel, A. L., Feng, A., Masimore, B., Morgan, G., Moser, R. P., Marcus, S. E., & Berrigan, D. (2012). Assessing the value of team science: A study comparing center-and investigator-initiated grants. *American Journal of Preventive Medicine*, 42(2), 157–163.
- Hall, K. L., Vogel, A. L., Huang, G. C., Serrano, K. J., Rice, E. L., Tsakraklides, S. P., & Fiore, S. M. (2018). The science of team science: A review of the empirical evidence and research gaps on collaboration in science. *The American Psychologist*, 73(4), 532–548. https://doi.org/10.1037/amp0000319
- Hart, J., & Mars, M. M. (2009). Joint Appointments and the professoriate: Two houses but no home? *Innovative Higher Education*, 34(1), 19–32. https://doi.org/10.1007/s10755-008-9092-x
- Hawkins, P. (2018). Coaching supervision. In E. Cox, T. Bachkirova, & D. Clutterbuck (Eds.), *The complete handbook of coaching* (pp. 381–393). Sage.
- Heidari, F. (1996). Laboratory barriers in science, engineering, and mathematics for students with disabilities (ERIC No. ED397583). *ERIC Institute of Education Sciences*. https://eric.ed.gov/?id=ED397583
- Hendren, C. O., & Ku, S. T. (2019). The interdisciplinary executive scientist: Connecting scientific ideas, resources and people. In K. Hall, A. Vogel, & R. Croyle (Eds.), Strategies for team science success. Springer, Cham. https://doi.org/10.1007/978-3-030-20992-6\_27
- Hilliard, L., Dunston, P., McGlothlin, J., & Duerstock, B. S. (2013, June 20–24). Designing beyond the ADA-creating an accessible research laboratory for students and scientists with physical disabilities [Conference session]. RESNA Conference, Bellevue, WA. https://www.resna.org/sites/default/files/legacy/conference/proceedings/2013/JEA/Hilliard.html
- Ho, D. E., King, J., Wald, R. C., & Wan, C. (2022). The centrality of data and compute for AI innovation: A blueprint for the National Research Cloud. Notre Dame Journal on Emerging Technologies, 3(2), 71–228.
- Hoffmann, S., Deutsch, L., Klein, J. T., & O'Rourke, M. (2022). Integrate the integrators! A call for establishing academic careers for integration experts. *Humanities and Social Sciences Communications*, 9, Article 147. https://doi.org/10.1057/s41599-022-01138-z
- Hosseini, M., Holcombe, A. O., Kovacs, M., Zwart, H., Katz, D. S., & Holmes, K. (2024). Group authorship, an excellent opportunity laced with ethical, legal and technical challenges. *Accountability in Research*, 1–23. https://doi.org/10.1080/08989621.2024.2322557
- Humphrey, M., Conley, Q., & Dunlap, J. C. (2020). Moving beyond compliance and broadening the discussion. *Quarterly Review of Distance Education*, 21(2), 1–21.
- Hunt, C. (2019, August 14). The role of a research development professional in team science. Society of Research Administrators International. https://www.srainternational.org/blogs/srai-news/2019/08/14/the-role-of-a-research-development-professional-in
- Ingram, J. (2014). Farmer-scientist knowledge exchange. In P. B. Thompson & D. M. Kaplan (Eds.), *Encyclopedia of agricultural and food ethics* (pp. 722–729). Springer, Dordrecht. https://doi.org/10.1007/978-94-007-0929-4
- International Committee of Medical Journal Editors. (n.d.). Defining the role of authors and contributors. https://www.icmje.org/recommendations/browse/roles-and-responsibilities/defining-the-role-of-authors-and-contributors.html
- Isaacson, M., Schleppenbach, D., & Lloyd, L. L. (2011). Increasing STEM accessibility in students with print disabilities through MathSpeak. *Journal of Science Education for Students with Disabilities*, 14(1), 25–32. http://dx.doi.org/10.14448/jsesd.03.0002
- Jeannis, H., Goldberg, M., Seelman, K., Schmeler, M., & Cooper, R. A. (2020). Barriers and facilitators to students with physical disabilities' participation in academic laboratory spaces. Disability and Rehabilitation: Assistive Technology, 15(2), 225–237. https://doi. org/10.1080/17483107.2018.1559889

- Jiang, G., Boghrat, D., Grabmeier, J., & Cross, J. E. (2023). Complexity leadership in action: A team science case study. *Frontiers in Research Metrics and Analytics*, 8.
- Job Accommodation Network. (2023). A to z: Costs and benefits of accommodation. https://askjan.org/topics/costs.cfm#publications
- Junker, J., Lange, L. M., Vollstedt, E. J., Roopnarain, K., Doquenia, M. L. M., Annuar, A. A., Avenali, N., Bardien, S., Bahr, N., Ellis, M., Galandra, C., Gasser, T., Heutink, P., Illarionova, A., Kanana, Y., Keller Sarmiento, I J., Kumar, K. K., Lim, S.-Y., Madoev, H., Mata, I F., Mencacci, N. E., Nalls, M., Padmanabhan, S., Shambetova, C., Solle, J. E., . . . Global Parkinson's Genetics Program (GP2). (2024). Team science approaches to unravel monogenic Parkinson's disease on a global scale. Movement Disorders, 39(10), 1868–1873. https://doi.org/10.1002/mds.29925
- Kaiser, J. (2017, June 7). Critics challenge NIH finding that bigger labs aren't necessarily better. ScienceInsider. https://www.science.org/content/article/critics-challenge-nih-finding-bigger-labs-aren-t-necessarily-better
- Kaner, S. (2014). Facilitator's guide to participatory decision-making (3rd ed). Jossey-Bass.
- Kelly, P. W., Chladek, J., & Rolland, B. (2023). Toward a translational team science hierarchy of needs: Exploring the information management challenges of team science. *Journal of Clinical and Translational Science*, 7(1), e210. https://doi.org/10.1017/cts.2023.614
- Kitchin, R. (2014). The data revolution: Big data, open data, data infrastructures and their consequences. Sage.
- Klein, J. T., & Falk-Krzesinski, H. J. (2017). Interdisciplinary and collaborative work: Framing promotion and tenure practices and policies. *Research Policy*, 46(6), 1055–1061. https:// doi.org/10.1016/j.respol.2017.03.001
- Klein, O., Hardwicke, T. E., Aust, F., Breuer, J., Danielsson, H., Mohr, A. H., IJzerman, H., Nilsonne, G., Vanpaemel, W., & Frank, M. C. (2018). A practical guide for transparency in psychological science. *Collabra: Psychology*, 4(1), 20. https://doi.org/10.1525/ collabra.158
- Kowalczyk, S., & Shankar, K. (2011). Data sharing in the sciences. *Annual Review of Information Science and Technology*, 45(1), 247–294.
- Kraft, S. A., & Mittendorf, K. F. (2024). Can open science advance health justice? Genomic research dissemination in the evolving data-sharing landscape. *Hastings Center Report*, 54, S73–S83.
- Kumar, A., & Wang, L. L. (2024, October). Uncovering the new accessibility crisis in scholarly PDFs: Publishing model and platform changes contribute to declining scholarly document accessibility in the last decade. In *Proceedings of the 26th International ACM SIGACCESS Conference on Computers and Accessibility* (pp. 1–16). https://doi.org/10.1145/3663548.3675634
- Leung, M. W., Yen, I. H., & Minkler, M. (2004). Community based participatory research: A promising approach for increasing epidemiology's relevance in the 21st century. *International Journal of Epidemiology*, 33(3), 499–506. https://doi.org/10/d3nwv5
- Levin, J. (2011). The emergence of the research-development professional. *The Chronicle of Higher Education*, 57, 30.
- Li, D., & Koedel, C. (2017). Representation and salary gaps by race-ethnicity and gender at selective public universities. *Educational Researcher*, 46(7), 343–354. https://doi.org/10.3102/0013189X17726535
- Lindsay, S., & Fuentes, K. (2022). It is time to address ableism in academia: A systematic review of the experiences and impact of ableism among faculty and staff. *Disabilities*, 2(2), 178–203. https://doi.org/10.3390/disabilities2020014
- Lyall, C. (2019). Being an interdisciplinary academic: How institutions shape university careers. Palgrave Pivot Cham. https://doi.org/10.1007/978-3-030-18659-3

- Mäkinen, E. I., Evans, E. D., & McFarland, D. A. (2024). Interdisciplinary research, tenure review, and guardians of the disciplinary order. *The Journal of Higher Education*, 1–28.
- Mamboleo, G., Dong, S., Anderson, S., & Molder, A. (2020). Accommodation experience: Challenges and facilitators of requesting and implementing accommodations among college students with disabilities. *Journal of Vocational Rehabilitation*, 53(1), 43–54.m https://doi.org/10.3233/JVR-201084
- Marchetti, C. E., Foster, S. B., Schley, S., Bailey, M. B., & Kavin, D. (2024). Pathways of women who are deaf and hard-of-hearing to faculty careers in higher education. *ADVANCE Journal*, 4(2). https://doi.org/10.5399/osu/advjrnl.4.2.4
- Mason, E., & Learned, L. (2006). The role of "development" in a research administration office. Journal of Research Administration, 37(1), 23–34.
- Massachusetts Institute of Technology. (n.d.). General guidelines for ADA compatible laboratories. https://chemistry.mit.edu/about/quality-of-life-committee/disability-accommodations/general-guidelines-for-ada-compatible-laboratories/#:~:text=The%20recommended%20 range%20is%2028,for%20seated%20and%20standing%20individuals
- Massett, H. A., Hampp, S. L., Goldberg, J. L., Mooney, M., Parreco, L. K., Minasian, L., Montello, M., Mishkin, G. E., Davis, C., & Abrams, J. S. (2018). Meeting the challenge: The National Cancer Institute's central institutional review board for multi-site research. *Journal of Clinical Oncology*, 36(8), 819–824. https://doi.org/10.1200/JCO.2017.76.9836
- Mazumdar, M., Messinger, S., Finkelstein, D. M., Goldberg, J. D., Lindsell, C. J., Morton, S. C., Pollock, B. H., Rahbar, M. H., Welty, L. J., Parker, R. A., & Biostatistics, Epidemiology, and Research Design Key Function Committee of the Clinical and Translational Science Awards Consortium. (2015). Evaluating academic scientists collaborating in team-based research: A proposed framework. Academic Medicine, 90(10), 1302–1308.
- McDaniels, M., & Asiedu, E. (2023). The role of mentorship in disrupting ableism and advancing STEM. National Academies of Sciences, Engineering, and Medicine Committee on Beyond Compliance: Promoting the Success of People with Disabilities in the STEM Workforce.
- McDermott, R. J. (2022). Taking back the narrative in science and public health-the challenge is enormous (and necessary). *Health Behavior and Policy Review*, 9(5), 1009–1016. https://doi.org/10.14485/HBPR.9.5.1
- McWilliams, R., Hoover-Fong, J., Hamosh, A., Beck, S., Beaty, T., & Cutting, G. (2003). Problematic variation in local institutional review of a multicenter genetic epidemiology study. *JAMA*, 290(3), 360-366. https://doi.org/10.1001/jama.290.3.360
- Meister, D. (1999). The history of human factors and ergonomics. CRC Press. https://doi.org/10.1201/9781315276069
- Mello, M. M., Triantis, G., Stanton, R., Blumenkranz, E., & Studdert, D. M. (2020). Waiting for data: Barriers to executing data use agreements. *Science*, 367(6474), 150–152. https://doi.org/10.1126/science.aaz7028.
- Meurer, J. R., Fertig, J., Garrison, O., & Shaker, R. (2023). Team science criteria and processes for promotion and tenure of health science university faculty. *Journal of Clinical and Translational Science*, 7(1), e27. https://doi.org/10.1017/cts.2022.523
- Microsoft Teams. (2025). Plan for government clouds. https://learn.microsoft.com/en-us/microsoftteams/platform/concepts/cloud-overview
- Milkman, K. L., Akinola, M., & Chugh, D. (2015). What happens before? A field experiment exploring how pay and representation differentially shape bias on the pathway into organizations. *Journal of Applied Psychology*, 100(6), 1678–1712. https://doi.org/10.1037/apl0000022
- Mingers, J., & Leydesdorff, L. (2015). A review of theory and practice in scientometrics. European Journal of Operational Research, 246(1), 1–19. https://doi.org/10.1016/j.ejor.2015.04.002

- Mirel, B., Luo, A., & Harris, M. (2015). Research infrastructure for collaborative team science: Challenges in technology-supported workflows in and across laboratories, institutions, and geographies. *Seminars in Nephrology*, 35(3), 291–302. https://doi.org/10.1016/j.semnephrol.2015.04.009
- Misra, J., Lundquist, J., Holmes, E., & Agiomavritis, S. (2011). The ivory ceiling of service work. *Academe*, 97, 22–26.
- Moher, D., Naudet, F., Cristea, I. A., Miedema, F., Ioannidis, J. P., & Goodman, S. N. (2018). Assessing scientists for hiring, promotion, and tenure. *PLoS Biology*, 16(3), e2004089. https://doi.org/10.1371/journal.pbio.2004089
- Mulfinger, L. M., Dressler, K. A., James, L. E., Page, N., Serrano, E., & Vázquez, J. D. (2016).
  Trends in large proposal development at major research institutions. The Journal of Research Administration, 47, 40–57.
- National Academies of Sciences, Engineering, and Medicine (National Academies). (2018). Open science by design: Realizing a vision for 21st century research. The National Academies Press. https://doi.org/10.17226/25116
- \_\_\_. (2020). Re-envisioning promotion and advancement for STEM faculty: Proceedings of a workshop in brief. The National Academies Press. https://doi.org/10.17226/25742
- \_\_\_\_. (2023). Advancing antiracism, diversity, equity, and inclusion in STEMM organizations: Beyond broadening participation. The National Academies Press. https://doi.org/10.17226/26803
- National Cancer Institute. (2024). NCI research specialist award (R 50). https://www.cancer.gov/grants-training/grants-funding/funding-opportunities/r50
- National Council of University Research Administrators. (n.d.). About us. https://www.ncura.edu/aboutus.aspx
- National Institute of General Medical Sciences. (2025). Collaborative program grant for multidisciplinary teams (RM1). https://www.nigms.nih.gov/grants/RM1/Pages/Collaborative-Program-Grant-for-Multidisciplinary-Teams-%28RM1%29
- National Institute on Deafness and Other Communication Disorders. (2019). Assistive devices for people with hearing, voice, speech, or language disorders. https://www.nidcd.nih.gov/health/assistive-devices-people-hearing-voice-speech-or-language-disorders#4
- National Institute on Minority Health and Health Disparities. (2024, January 12). Transdisciplinary Collaborative Centers for Health Disparities Research Program (TCC). https://www.nimhd.nih.gov/programs/extramural/research-centers/tcc/
- National Institutes of Health. (2023, September 15). NIH final updated policy guidance for subaward/consortium written agreements (Notice No. NOT-OD-23-182). U.S. Department of Health & Human Services. https://grants.nih.gov/grants/guide/notice-files/NOT-OD-23-182.html
- National Organization of Research Development Professionals. (n.d.). What is research development? https://nordp.org/page/about
- National Research Council. (2015). Enhancing the effectiveness of team science. The National Academies Press. https://doi.org/10.17226/19007
- $National\ Science\ Foundation.\ (2023).\ STEM\ access for\ persons\ with\ disabilities\ (STEM-APWD)\ https://www.nsf.gov/funding/opportunities/dcl-stem-access-persons-disabilities-stem-apwd$
- \_\_\_\_. (2024). Science and technology centers: Integrative partnerships. https://www.nsf.gov/od/oia/ia/stc
- Nelson, A. (2022, August 25). Ensuring free, immediate, and equitable access to federally funded research (OSTP Memo). White House Office of Science & Technology Policy. https://www.whitehouse.gov/wp-content/uploads/2022/08/08-2022-OSTP-Public-Access-Memo.pdf

- Nguyen, M., Chaudhry, S. I., Desai, M. M., Dzirasa, K., Cavazos, J. E., & Boatright, D. (2023). Gender, racial, and ethnic inequities in receipt of multiple National Institutes of Health research project grants. *JAMA Network Open*, 6(2), e230855–e230855. https://doi.org/10.1001/jamanetworkopen.2023.0855
- Nogrady, B. (2023). Hyperauthorship: The publishing challenges for 'big team' science. *Nature*, 615(7950), 175–177. https://doi.org/10.1038/d41586-023-00575-3
- Nosek, B. A., Alter, G., Banks, G. C., Borsboom, D., Bowman, S. D., Breckler, S. J., Buck, S., Chambers, C. D., Chin, G., Christensen, G., Contestabile, M., Dafoe, A., Eich, E., Freese, J., Glennerster, R., Goroff, D., Green, D. P., Hesse, B., Humphreys, M., . . . Yarkoni, T. (2015). Promoting an open research culture. *Science*, 348(6242), 1422–1425. https://doi.org/10.1126/science.aab2374
- Oakes, J. M. (2002). Risks and wrongs in social science research. An evaluator's guide to the IRB. *Evaluation Review*, 26(5), 443–479. https://doi.org/10.1177/019384102236520
- Osinski, A. (2021). Towards a critical sustainability science? Participation of disadvantaged actors and power relations in transdisciplinary research. *Sustainability*, 13(3), Article 3. https://doi.org/10.3390/su13031266
- Parthasarathy, S., & Katzman, J. (2024). Bringing communities in, achieving AI for all. *Issues in Science and Technology*, 40(4), 41–44. https://doi.org/10.58875/SLRG2529
- Passmore, S. R., Kisicki, A., Gilmore-Bykovskyi, A., Green-Harris, G., & Edwards, D. F. (2022). "There's not much we can do..." researcher-level barriers to the inclusion of underrepresented participants in translational research. *Journal of Clinical and Translational Science*, 6(1), e4. https://doi.org/10/g6dnbw
- Pearson, Y., & Alexander, Q. (2020). The inclusion of persons with disabilities in engineering education and careers. In *Implementation Strategies for Improving Diversity in Organizations* (pp. 249–270).
- Peek, L., Tobin, J., Van De Lindt, J. W., & Andrews, A. (2021). Getting interdisciplinary teams into the field: Institutional Review Board preapproval and multi-institution authorization agreements for rapid response disaster research. *Risk Analysis*, 41(7), 1204–1212. https://doi.org/10.1111/risa.13740
- Perez-Riverol, Y., Gatto, L., Wang, R., Sachsenberg, T., Uszkoreit, J., Leprevost, F. D. V., Fufezan, C., Ternent, T., Eglen, S. J., Katz, D. S., Pollard, T. J., Konovalov, A., Flight, R. M., Blin, K., & Vizcaíno, J. A. (2016). Ten simple rules for taking advantage of Git and GitHub. *PLoS Computational Biology*, 12(7), e1004947. https://doi.org/10.1371/journal.pcbi.1004947
- Persson, H., Åhman, H., Yngling, A. A., & Gulliksen, J. (2015). Universal design, inclusive design, accessible design, design for all: Different concepts—one goal? On the concept of accessibility—historical, methodological and philosophical aspects. *Universal Access* in the Information Society, 14(4), 505–526. https://doi.org/10.1007/s10209-014-0358-z
- Piwowar, H. A., & Vision, T. J. (2013). Data reuse and the open data citation advantage. *PeerJ*, 1, e175. https://doi.org/10.7717/peerj.175
- Platt, A., Truong, T., Boulos, M., Carlson, N. E., Desai, M., Elam, M. M., Slade, E., Hanlon, A. L., Hurst, J. H., Olsen, M. K., Poisson, L. M., Rende, L., & Pomann, G. M. (2024). A guide to successful management of collaborative partnerships in quantitative research: An illustration of the science of team science. Stat, 13(2), e674. https://doi.org/10.1002/sta4.674
- Potter, M. B., Mansur, S., Rutman, S. P., & Brindis, C. D. (2024). Is team science valued in the academic promotions process? A mixed-methods case study. *Journal of Clinical and Translational Science*, 8(1), e28. https://doi.org/10.1017/cts.2024.7
- Preuss, M., Eck, K., Fechner, M., & Walker, L. (2018). Describing research development: A first step. Research Management Review, 23(1). https://eric.ed.gov/?id=EJ1215447

- Puertas-Martín, S., Banegas-Luna, A. J., Paredes-Ramos, M., Redondo, J. L., Ortigosa, P. M., Brovarets', O. O., & Pérez-Sánchez, H. (2020). Is high performance computing a requirement for novel drug discovery and how will this impact academic efforts? Expert Opinion on Drug Discovery, 15(9), 981–985. https://doi.org/10.1080/17460441.2020.1758664
- Resnik, D. B., Smith, E. M., & Shi, M. (2018). How U.S. research institutions are responding to the single Institutional Review Board mandate. *Accountability in Research*, 25(6), 340–349. https://doi.org/10.1080/08989621.2018.1506337
- Riley, E. A. U., & Schekman, R. (2021). Point of view: Open science takes on Parkinson's disease. *eLife*, 10, e66546. https://doi.org/10.7554/eLife.66546
- Rizzo, J.-R., Rosenblum, P., Samuel, C., Wittich, W., Martiniello, N., Beheshti, M., Johnson, A., Wu, Y.-H., Sukhai, M., & Bonnielin, S. (2024). Accessible scientific conferences for blind and low vision professionals and researchers: A necessary step for achieving STEMM equity. Disability & Society, 1–7. https://doi.org/10.1080/09687599.2024.2412269
- Rolland, B., Lee, C. P., & Potter, J. D. (2017). Greater than the sum of its parts: A qualitative study of the role of the coordinating center in facilitating coordinated collaborative science. *Journal of Research Administration*, 48(1), 65–85.
- Ross, R., Reeves, J., Scarpinato, K., & Pelham, M. (2019). Success factors for university research development offices and activities. *Journal of Research Administration*, 50(3), 107–124.
- Sabatello, M., Martschenko, D. O., Cho, M. K., & Brothers, K. B. (2022). Data sharing and community-engaged research. *Science*, 378(6616), 141–143. https://doi.org/10.1126/science.abq6851
- Schwartz, J., Roy, I., Hauptmann, M., Durme, Y. V., & Denny, B. (2019). Organizational performance: It's a team sport. https://www2.deloitte.com/us/en/insights/focus/human-capital-trends/2019/team-based-organization.html
- Sege, R., Nykiel-Bub, L., & Selk, S. (2015). Sex differences in institutional support for junior biomedical researchers. *JAMA*, 314(11), 1175–1177.
- Seidel, C. (2022). Research administration practices for proposal development and post-award management of stakeholders and community participants. In D. Lerner, M. E. Palm, & T. W. Concannon (Eds.), *Broadly engaged team science in clinical and translational research* (pp. 205–213). Spring Cham.
- Shaw, L. C., Errington, T. M., & Mellor, D. T. (2022). Toward open science: Contributing to research culture change. *Science Editor*, 45(1), 14–17. https://doi.org/10.36591/se-d-4501-14
- Singh Chawla, D. (2024). Accessibility worsens for blind and low-vision readers of academic PDFs. *Nature*. https://www.nature.com/articles/d41586-024-03953-7
- Smikowski, J., Dewane, S., Johnson, M. E., Brems, C., Bruss, C., & Roberts, L. W. (2009). Community-based participatory research for improved mental health. *Ethics & Behavior*, 19(6), 461–478. https://doi.org/10.1080/10508420903274971
- Smith, N., & Dropkin, D. (2018). Access and inclusion. In P. Buxton (Ed.), *Metric handbook* (6th ed., pp. 41–51). Routledge.
- Sotto-Santiago, S., Mac, J., & Genao, I. 2023. "Value my culture, value me": A case for culturally relevant mentoring in medical education and academic medicine. *BMC Medical Education*, 23(1), 229. https://doi.org/10.1186/s12909-023-04148-w
- Specht, A., & Crowston, K. (2022). Interdisciplinary collaboration from diverse science teams can produce significant outcomes. *PLoS ONE*, 17(11), e0278043. https://doi. org/10.1371/journal.pone.0278043
- Spellman, B., Gilbert, E., & Corker, K. S. (2018). Open science. In J. T. Wixted (Ed.), Stevens' handbook of experimental psychology and cognitive neuroscience (Vol. 5). https://doi.org/10.1002/9781119170174.epcn519

- Stanford Research Development Office. (2024). Who we are. https://doresearch.stanford.edu/who-we-are/stanford-research-development-office
- Steiner, J. S., Blum-Barnett, E., Rolland, B., Kraus, C. R., Wainwright, J. V., Bedoy, R., Martinez, Y. T., Alleman, E. R., Eibergen, R., Pieper, L. E., Carroll, N. M., Hixon, B., Sterrett, A., Rendle, K. A., Saia, C., Vachani, A., Ritzwoller, D. P., & Burnett-Hartman, A. (2023). Application of team science best practices to the project management of a large, multi-site lung cancer screening research consortium. *Journal of Clinical and Translational Science*, 7(1), 1–22. https://doi.org/10.1017/cts.2023.566
- Stephens, B., Downer, J. B., & Cummings, J. N. (2024). Teamwork coaching in the research development process. *Small Group Research*, 10464964241240721. https://doi.org/10.1177/10464964241240721
- Stewart, A. J., & Valian, V. (2018). An inclusive academy: Achieving diversity and excellence. The MIT Press.
- Stuart, T. E., & Ding, W. W. (2006). When do scientists become entrepreneurs? The social structural antecedents of commercial activity in the academic life sciences. *American Journal of Sociology*, 112(1), 97–144. https://doi.org/10.1086/502691
- Sutton, L., Berdan, L. G., Bolte, J., Califf, R. M., Ginsburg, G. S., Li, J. S., McCall, J., Moen, R., Myers, B. S., Rodriquez, V., Veldman, T., & Boulware, L. E. (2019). Facilitating translational team science: The project leader model. *Journal of Clinical and Translational Science*, 3(4), 140–146. https://doi.org/10.1017/cts.2019.398
- Swartz, T. H., Palermo, A.-G. S., Masur, S. K., & Aberg, J. A. (2019). The science and value of diversity: Closing the gaps in our understanding of inclusion and diversity. *The Journal* of *Infectious Diseases*, 220(Suppl 2), S33–S41. https://doi.org/10/ghgvxv
- University of California. (n.d.) Research policy analysis and coordination: Challenges to full-cost recovery. Office of the President. https://www.ucop.edu/research-policy-analysis-coordination/policies-guidance/indirect-cost-recovery/challenge-of-under-recovery. html#:~:text=The%20federal%20government%20placed%20a,to%20have%20 such%20a%20cap
- University of Southern California. (2022). University committee on appointments, promotions, and tenure: UCAPT manual. https://policy.usc.edu/appointments-promotions-tenure/
- U.S. Department of Agriculture. (2018). NIFA-19-010 2018 Farm Bill indirect cost provision. U.S. Department of Food and Agriculture. https://www.nifa.usda.gov/nifa-19-010-2018-farm-bill-indirect-cost-provision
- U.S. Department of Health and Human Services. (2018). Federal policy for the protection of human subjects ('Common Rule'). 45 C.F.R. § 46. https://www.hhs.gov/ohrp/regulations-and-policy/regulations/common-rule/index.html
- Vecchiola, C., Pandey, S., & Buyya, R. (2009, December 14–16). High-performance cloud computing: A view of scientific applications [Conference session]. I-SPAN 2009 The 10th International Symposium on Pervasive Systems, Algorithms, and Networks. https://doi.org/10.1109/I-SPAN.2009.150
- Venkatraman, V. (2010, April 16). Conventions of scientific authorship. *Science*. https://www.science.org/content/article/conventions-scientific-authorship-0
- Vidal-Verdú, F., & Hafez, M. (2007). Graphical tactile displays for visually-impaired people. *IEEE Transactions on neural systems and rehabilitation engineering*, 15(1), 119–130. https://ieeexplore.ieee.org/abstract/document/4126548
- Vogel, A. L., Knebel, A. R., Faupel-Badger, J. M., Portilla, L. M., & Simeonov, A. (2021). A systems approach to enable effective team science from the internal research program of the National Center for Advancing Translational Sciences. *Journal of Clinical and Translational Science*, 5(1), e163. https://doi.org/10.1017/cts.2021.811

- Wegner, D., Da Silveira, A. B., Marconatto, D., & Mitrega, M. (2024). A systematic review of collaborative digital platforms: Structuring the domain and research agenda. *Review of Managerial Science*, 18(9), 2663–2695. https://doi.org/10.1007/s11846-023-00695-0
- Widdowson, L., Rochester, L., Barbour, P. J., & Hullinger, A. M. (2020). Bridging the team coaching competency gap: A review of the literature. *International Journal of Evidence Based Coaching & Mentoring*, 18(2), 35–50.
- Wilkinson, M., Dumontier, M., Aalbersberg, I., Appleton, G., Axton, M., Baak, A., Blomberg, N., Boiten, J.-W., da Silva Santos, L. B., Bourne, P. E., Bouwman, J., Brookes, A. J., Clark, T., Crosas, M., Dillo, I., Dumon, O., Edmunds, S., Evelo, C. T., Finkers, R. Gonzalez-Beltran, A., Gray, A. J. G., Groth, P., Goble, C., Grethe, J. S., . . . Mons. B. (2016). The FAIR Guiding Principles for scientific data management and stewardship. *Scientific Data*, 3, Article 160018. https://doi.org/10.1038/sdata.2016.18
- Wilson, G. (2013). Software carpentry: Lessons learned. arXiv. https://doi.org/10.48550/arXiv.1307.5448
- World Health Organization. (2023, February 6). Artificial intelligence in mental health research: New WHO study on applications and challenges. https://www.who.int/europe/news/item/06-02-2023-artificial-intelligence-in-mental-health-research--new-who-study-on-applications-and-challenges
- Wu, Y. H., Martiniello, N., & Swenor, B. K. (2022). Building a more accessible conference for researchers with vision impairment. *JAMA Ophthalmology*, 140(2), 113–114. https://doi. org/10.1001/jamaophthalmol.2021.5613
- Yale University. (n.d.) Guidance on authorship in scholarly or scientific publications. Office of the Provost. https://provost.yale.edu/policies/academic-integrity/guidance-authorship-scholarly-or-scientific-publications
- Yang, L., Albats, E., & Etzkowitz, H. (2020). Interdisciplinary organization as a basic academic unit? *Industry and Higher Education*, 35(3), 173–187. https://doi.org/10.1177/0950422220956951



5

# **Evaluating Team Science**

This chapter identifies key outcomes, methods, and measures for evaluating team science training and performance. The chapter uses the term *team science evaluation* broadly to refer to the systematic assessment of team-based scientific research efforts (Stokols et al., 2008). This process involves measuring and analyzing various aspects of how science teams function and collaborate, the outcomes they produce, and/or the broader impacts they achieve. Team science evaluation can help reveal the dynamics within a science team; assess the effectiveness of the team's collaboration processes; and determine how the team's work contributes to advancing scientific knowledge, achieving institutional objectives, and generating societal benefits.

Team science evaluation can refer to both traditional evaluation efforts, dictated by funders, and the measurement of team processes and outcomes as part of research investigating scientific collaboration. Based on the committee's literature searches, this chapter provides guidance for research evaluators in understanding which phenomena to assess when determining whether grant recipients are effectively meeting their funding objectives. It also guides scholars studying scientific collaboration with a broad understanding of the different constructs involved in team functioning—from inputs to processes to outcomes.

#### IMPORTANCE OF TEAM SCIENCE EVALUATION

The effectiveness of a science team can have far-reaching implications for a wide array of individuals and groups, all of whom have a vested interest in the team's success (Falk-Krzesinski et al., 2011). For example,

the individual members of a science team—such as graduate students, postdoctoral researchers, faculty members, professional scientists, and administrators—are impacted directly by the team's functioning and effectiveness. There could be potential setbacks for an early career researcher if a dysfunctional team limits opportunities for publishing or networking. On the other hand, their professional growth, recognition, professional networks, and intellectual contributions could be enhanced through involvement in a productive science team. Additionally, the team can benefit as a whole from each member's contributions and collaborative processes. Successful team science collaboration can lead to groundbreaking research outputs and shared achievements (Thayer et al., 2018; Xu et al., 2022). Beyond the team, scientific institutions, such as universities and research centers, rely on the success of science teams to build reputations, secure funding, and fulfill their missions of advancing knowledge (e.g., Jones et al., 2008; National Science Foundation, n.d.). The impact of team science extends to an institution's ability to attract top-tier talent, secure multimillion-dollar grants (i.e., center grants), and maintain a leadership role in the advancement of knowledge. Entire scientific fields also stand to benefit from the efforts of science teams, driving innovation, expanding knowledge, and setting new standards within the discipline (Stokols et al., 2008). Society at large is another critical party interested in science teamwork, as the discoveries made by science teams could lead to technological advancements, informed public policies, solutions to pressing global issues, and ultimately improving quality of life and addressing societal needs (Stokols et al., 2008). The stakes can be high, such that, without effective team science, lifesaving discoveries or technological breakthroughs may remain out of reach.

Evaluating how science teams collaborate is vital for improving research partnerships across all science fields (Klein, 2008). Evaluation can help leaders and teams understand team dynamics, improve processes, adapt to challenges, manage risks, and foster inclusion and continuous improvement. Evaluating research outputs, such as publications, patents, and outreach efforts, can be vital for determining overall team effectiveness and highlighting the innovation and creativity fostered by team science. Evaluation can also help clarify the broader impacts of team science research, including societal, educational, and policy contributions, ensuring that wider goals are being met. This, in turn, can inform future resource allocation and funding decisions, ensuring the effective use of resources and justifying future investments. Moreover, evaluating teamwork processes, psychological states, and practices can help identify and refine best practices and successful strategies that could be shared and replicated across other team science initiatives (Klein, 2008; Stokols et al., 2008). Insights gained from team science evaluation may help lead to improvements in research

methodologies, processes, and administrative support, thereby optimizing the overall effectiveness and efficiency of collaborative work (Belcher et al., 2015; Love et al., 2022; Roelofs et al., 2019). In short, systematic evaluation of team science can enable researchers, institutions, and funders to better support and enhance the contributions of collaborative research to scientific advancement and societal benefit.

#### CHALLENGES OF TEAM SCIENCE EVALUATION

Evaluating team science effectively is complex because of the need to address impacts on many types of groups, team dynamics, research types, research and organizational contexts, project time frames, and more.

The effectiveness of team science can impact different groups at multiple levels (e.g., individual, team, institutional, scientific, societal), so evaluation may need to consider the impact of team science across these different levels. For example, a thorough understanding of team functioning goes beyond team-level factors (e.g., team output and collaboration) to consider individual (e.g., attributes, contributions, individual outcomes) and contextual factors (e.g., institutional characteristics and outcomes) that both shape and are shaped by team dynamics. Thus, a key challenge for evaluators is identifying the most relevant factors at each level—individual, team, and context—to be included in the evaluation process.

The dynamics of science teams are complex, as teams may be composed of individuals with different goals, cultural contexts, and disciplinary areas. For example, team members may be dispersed geographically or have status differences within the team (e.g., junior faculty vs. senior faculty, medical researcher vs. community member). Some teams have fluid membership, in that members may join or leave the research over time. Thus, tailored evaluation methods are needed for capturing team interactions and outputs effectively.

Evaluations need to adapt to the type of work teams are pursuing. For instance, a team focused on biomedical research may have different collaborative dynamics, communication styles, and success metrics than a team working in environmental science. Teams conducting biomedical research may involve clinical (e.g., doctors, nurses) and laboratory researchers with different working schedules. Environmental science teams may conduct field studies and plan for long-term data collection efforts.

The cultural context, including the norms and values that guide behavior within the team, can also vary widely depending on the team's geographic location, institutional setting, and the professional or personal backgrounds of its members. Similarly, the disciplinary focus of a team influences the methodologies it uses; the nature of its research questions; and the types of outputs it generates, such as publications, patents, or policy

recommendations. These complexities often necessitate flexible, context-specific approaches to team science evaluation that can accommodate the variable and evolving nature of team science endeavors (Hall et al., 2018; Stokols et al., 2003).

Team science evaluation can consider the timing of measurement as well as the evolution of team goals and dynamics over time (Mâsse et al., 2008; Stokols et al., 2008). Longitudinal data collection can be resource intensive, and teams often operate in different contexts, making it difficult to apply a standardized evaluation approach (Cummings et al., 2013; Hall et al., 2018; O'Connor et al., 2003). Identifying and consistently measuring relevant metrics that accurately reflect performance is another hurdle, as performance encompasses both outcomes and processes that can shift over time. Additionally, attributing changes in performance to specific factors is complicated by the influence of various internal and external variables (Ilgen et al., 2005). Temporal patterns and lag effects further complicate the assessment, as the impact of actions may not be immediately apparent (Ilgen et al., 2005). Furthermore, as team goals and success criteria often evolve throughout a project, evaluations may need to adapt to these changes.

The organizational context represents another key consideration when evaluating team science (Lee & Jabloner, 2017). For example, organizations outside of academia may have established training programs, best practices, and accountability and evaluation procedures (e.g., Barry et al., 2024; Savannah River Site, 2020) that can support team science. These practices often derive from hierarchical, command-and-control structures, yet productivity, cohesion, and retention within teams ultimately depend on the skills and behaviors of the direct leader and team members. Consequently, many industry organizations invest heavily in training employees in nontechnical areas, such as project management, conflict resolution, positive leadership, and effective communication (Carucci, 2018; Day et al., 2021), all of which enhance collaboration and team science effectiveness (Delise et al., 2010).

Industry organizations frequently use anonymous employee surveys to gather feedback on what is working well and where improvements are needed, subsequently creating action plans to address identified gaps. Additionally, mechanisms for confidential peer feedback can contribute to continuous improvement by informing development plans. Regular performance discussions between employees and leaders provide ongoing real-time feedback that reinforces positive behaviors and allows leaders to address issues before they escalate. When more complex issues arise that team members cannot resolve themselves, experts from human resources and/or compliance and ethics departments can assist in investigating and mediating solutions. This emphasis on accountability can cultivate a culture

in industry that supports team science. Although these systems are not flawless, they can establish a solid foundation for collaboration (Schneider et al., 1996). This feedback loop helps maintain productivity and ensures a positive return on investment.

However, since companies rarely share data on the outcomes linked to these practices, it is challenging to scientifically evaluate the strengths and weaknesses of various industry approaches to improving team science. In one exception, Google LLC has publicly shared the results of its internal research to understand what variables make some teams more successful than others, with a summary reported by Duhigg (2016) in *The New York Times*. Under Project Aristotle, researchers reviewed external literature and data from over a hundred teams throughout Google and found that psychological safety was the most important indicator for how a team would perform. High team psychological safety tended to track with equal speaking time and conversational turn-taking and high average social sensitivity (Duhigg, 2016). Although it is rare for companies to release such information externally, the results from Project Aristotle reinforce and are consistent with best practices for effective team science captured in the academic literature (e.g., Edmondson & Roloff, 2008).

In summary, the multifaceted nature of scientific ecosystems, the varied goals and strategies pursued by different science teams, the complexities of team dynamics over time, and the need to account for organizational practices and industry norms make it challenging to establish universal metrics or approaches for team science evaluation. The variable and evolving nature of science teams renders a one-size-fits-all approach to team evaluation inadequate. Instead, effective evaluation requires tailored methods for capturing the specific interactions, processes, and outcomes relevant to each team. This may involve adapting existing evaluation frameworks to align with a team's unique goals or developing entirely new metrics to accurately assess performance in context. As teams evolve—adapting to new challenges, incorporating new members, or shifting their research focus—the evaluation approach must also be flexible and adaptive to remain relevant and insightful, and to provide valuable feedback that guides the team's ongoing development and success.

#### A FRAMEWORK FOR GUIDING TEAM SCIENCE EVALUATION

Three overarching types of criteria could be considered when assessing the effectiveness of a work team (Hackman, 1987; Kozlowski & Ilgen, 2006). The first criterion, referred to in this chapter as *team dynamics*, assesses whether the team's social processes and emergent psychological states are effective and efficient in achieving the desired performance standards, while also fostering the team members' ability to learn and collaborate on

future tasks. The second criterion assesses whether the teams' performance output meets or exceeds the performance standards expected by those who receive and/or review it. The third criterion considers whether, on balance, the group experience satisfies, rather than frustrates, the personal needs of individual members. Building on this framework and considering the multilayered nature of scientific ecosystems, the following sections identify methods and measures for evaluating: (a) team dynamics (e.g., team composition; teamwork processes; emergent states), (b) team performance outputs (e.g., intellectual merit, broader impact), and (c) the team's impact on individual members (e.g., career success, opportunities for learning and development).

From a national laboratory perspective, one notable gap is the lack of formal evaluation of team dynamics by laboratory leaders, which the committee views as a limitation. Traditionally, teams in national labs have been assessed based on project outcomes, not on the quality of their collaboration. Adopting a more rigorous approach to evaluating teamwork from a team science perspective would be valuable. An essential question to consider would be: How effectively did the team function?

## **Evaluating Science Team Dynamics**

Some teams "burn themselves out" in the process of completing the team task, ultimately compromising their effectiveness. Other teams can sustain their performance, learn from one another, and improve over time. Likewise, team performance may be high, but members may be so dissatisfied with the team's dynamics that they do not want to continue working with the same members (this phenomenon is measured by what is known as team viability [Tekleab et al., 2009]).

The scientific literature on team functioning has identified attitudes, behaviors, and cognitive states (ABCs) that are "signatures" of highly effective teams (Salas et al., 2008).

Attitudes include psychological states such as *cohesion*, where members feel a strong bond and commitment to the team, and *trust*, which enables members to rely on one another; these attitudes play a pivotal role in sustaining high performance (Beal et al., 2003; De Jong et al., 2016). Additionally, *psychological safety*, an environment where team members feel safe to speak up, ask questions, and challenge ideas without fear of negative consequences, also enhances team performance (see also Chapter 4; Frazier et al., 2017).

Behaviors include core team processes that highlight interaction among members (Marks et al., 2001). For example, effective communication can ensure that information flows freely—reducing misunderstandings, aligning efforts, and improving performance (Marlow et al., 2018).

Coordination enables members to align their actions so that tasks are integrated seamlessly (Rico et al., 2008). Dysfunctional types of conflict, such as relational conflict (tension between team members) and process-related conflict (disagreements about logistics) reduce productivity, whereas task conflict (disagreements about the content of work) may increase performance under certain circumstances (de Wit et al., 2012).

Emergent *cognitive states*, such as shared mental models and transactive memory systems, allow teams to operate smoothly and efficiently by creating a common understanding of tasks, roles, and distributed knowledge (Mohammed et al., 2021). Information-sharing allows teams to capitalize on each member's unique expertise, whereas team-learning enables members to collectively acquire and apply knowledge to achieve common goals, both of which enhance team performance (Mesmer-Magnus & DeChurch, 2009; Wiese et al., 2021).

The broader team literature demonstrates that teams that develop and nurture the ABCs of teamwork are well positioned to improve team well-being and performance. But research on teamwork specific to the science team context rarely assesses team emergent states and processes, focusing instead on performance outcomes captured via archival measures such as publications, patents, or grants. However, teams that excel do not focus on task completion alone—they also invest in sustaining these critical processes and psychological states over time (Salas et al., 2008).

Researchers can assess the quality of a team's ABCs by collecting self-reported data via surveys. Table D-1 in Appendix D provides a sampling of survey-based assessments used in the team literature, including a construct/scale description, measurement specifics (number of items, dimensions, rating scale), and references indicating scale development and validation evidence. Following the popular input-mediator-output-input team framework (Ilgen et al., 2005), scales in Table D-1 are divided into seven main categories:

- 1. compositional/individual difference surveys (e.g., team roles, collaboration readiness);
- emergent states (affective states such as trust and cohesion, cognitive states such as team-learning and transactive memory systems, and behavioral states such as workload-sharing and task interdependence);
- 3. team behavioral processes (e.g., conflict, coordination);
- 4. team climate and context (e.g., work group inclusion, team perceived virtuality);
- 5. team leadership;
- 6. team outcomes (e.g., affective outcomes such as team viability and collaboration, productivity outcomes such as performance); and

7. composite team surveys, which measure a variety of dimensions, including team readiness, functioning, climate, and outcomes (e.g., TeamSTEPPS [Agency for Healthcare Research and Quality, n.d.]).

To facilitate an understanding of when these measures may be helpful, these categories are represented in the form of questions (e.g., Are team members positioned for effective teamwork? How happy are team members with how they worked together?).

Although most scales in Appendix D are self-reported, some are designed to be completed by peers (e.g., the comprehensive assessment of team member effectiveness by Ohland et al. [2012] for self- and peer evaluation). For some constructs—such as conflict (e.g., Jehn & Mannix, 2001), transactive memory systems (e.g., Lewis, 2003), and psychological safety (e.g., Edmondson, 1999)—there is consensus on measurement, with most research utilizing the most popular scale. In contrast, constructs such as cohesion, team climate, team viability, and team performance currently have no agreed-upon measure in the team literature (see Appendix D).

Considerable variability exists for these measures regarding validation evidence, with more robust methods showing consistent factor structures over multiple samples and demonstrating favorable psychometric properties and multiple forms of validity (e.g., content, discriminant, convergent, predictive). For example, seeking to validate 50-, 30-, and 10-item versions of their team process scale, Mathieu et al. (2020) utilized data from 700 teams across laboratory and field contexts to demonstrate a consistent confirmatory factor structure over 10 samples, along with high content and discriminant validity.

Context (e.g., level of virtuality, nature and difficulty of team tasks, resources and support, team culture) needs to be considered carefully when determining what scales to adopt. Because most of the scales in Appendix D were developed in the broader team literature, adaptations in wording may be needed to fit science teams. Although scale adaptations (e.g., changing the item context or referent, shortening scales, adding new items) are widespread in the team literature because of varying types of teams and tasks, adaptations may introduce threats to validity and psychometric properties (e.g., Heggestad et al., 2019). Justifying scale adaptations and providing evidence to support the validity of scales altered to fit a science team context would be possible but would require funding.

<sup>&</sup>lt;sup>1</sup> The items for many of the scales provided in Appendix D are freely available at https://ctsi.psu.edu/research-support/team-science-toolbox/assessment/

## **Evaluating Science Team Performance Outputs**

The outputs that science teams and multiteam systems produce—such as publications, invention reports, patents, funding, and developmental opportunities for junior scholars—can be valuable indicators of their effectiveness. These metrics can reflect a team's ability to generate novel ideas and solutions and advance scientific knowledge. A consistent flow of quality scientific output suggests that the team is actively engaged in research with practical applications (Hall et al., 2018; Trochim et al., 2008) and may indicate that the team is successfully integrating different expertise and perspectives and has strong collaboration and synergy. Thus, performance outputs can serve as key measures of team productivity and the tangible impact of research efforts (Xu et al., 2022; Yang et al., 2022).

#### **Publications**

Publishing scientific articles is a fundamental outcome of science teamwork, as publications can disseminate research findings to the broader community, contribute to the body of knowledge, and establish the credibility of the team's work. The collaborative process of writing and submitting manuscripts involves synthesizing different perspectives, adhering to rigorous methodologies, and presenting coherent and impactful narratives. The quality and quantity of publications can be evaluated by examining a variety of metrics such as the number of articles published, the impact factors of the journals in which they appear, and the citation rates they receive. Furthermore, evaluating the collaborative process leading to publication—such as the division of labor, the integration of different disciplinary insights, and the ability to meet deadlines—provides valuable information on the team's operational efficiency and cohesion (Hall et al., 2018; Nancarrow et al., 2013; Shuffler & Carter, 2018). Bibliometric analysis is a research evaluation methodology rooted in information science (Lyu et al., 2023). It uses statistical methods to analyze patterns in publications and citations, and it is rooted in the principle that the impact and influence of research can be quantified by examining how it is cited and used by other researchers.

There are different approaches to using publication output as a performance metric for science teams. Beyond simply counting the number of publications produced by a science team, researchers studying scientific collaboration have also assessed the quality of publications by considering their citation counts (Uzzi et al., 2013). Some scholars have used impact factors of the journals in which the publications appear (Llewellyn et al., 2020, 2024). Using impact factors and/or citation counts to evaluate science team performance has both advantages and disadvantages. On the positive side, these metrics can provide quantifiable measures of the potential

influence and reach of a team's research. High citation counts suggest that the team's work is being recognized and used by other researchers, indicating its relevance and impact within the scientific community. Similarly, publishing in high-impact journals can signal that the team's research meets high standards of quality and significance, as these journals often have rigorous peer-review processes and are widely read. Demonstrating team performance through these metrics may be beneficial for securing funding, establishing collaborations, and enhancing the reputation of both the team and their embedding institution(s) (Carpenter et al., 2014). However, there are also multiple drawbacks to relying solely on citation counts and impact factors to evaluate science team performance. These metrics can be misleading, as they do not necessarily reflect the quality or innovation of the research (Donaldson & Cooke, 2014; Michalska-Smith & Allesina, 2017). For instance, citation counts for a team could be inflated by a few highly cited papers, which may not represent the overall contribution of the team. Additionally, a journal's impact factor reflects the average citation rate of all articles within that journal, not the specific impact of any single article, which can make it a poor indicator of individual team performance in many cases (Waltman & Traag, 2021).

For teams with interdisciplinary and/or transdisciplinary research goals, evaluating the degree to which publications integrate knowledge, methods, and perspectives from multiple disciplines to address complex research questions can be relevant (Laursen et al., 2022; Tremblay et al., 2011). Evaluation of interdisciplinarity can be approached through several key indicators and methods. For example, a qualitative evaluation of interdisciplinarity could involve peer review and expert evaluation, where feedback is solicited from experts in different disciplines to assess the perceived interdisciplinarity of the team's work (Mansilla, 2006). These peer reviews and expert panels can provide valuable qualitative insights into how effectively the research integrates and advances knowledge across multiple fields. Additionally, self-reported interdisciplinarity can be included, where the research team conducts a self-assessment, reflecting on their interdisciplinary practices, the integration of knowledge from various disciplines, and the challenges and benefits they encountered while working across these different fields (e.g., Palmer et al., 2016).

Quantitative analysis of disciplinary differences can involve examining the disciplinary backgrounds of team members listed as authors on scientific publications, categorized by their departmental or institutional affiliations and areas of expertise. Additionally, analyzing the variety of citations and references in the team's publications can reveal the extent to which the team draws on and contributes to multiple disciplines. Assessing the variety of publication venues, particularly the range of journals in which the team's work is published, also provides insight into the research's relevance across

different academic fields (Carr et al., 2018). This can be measured by categorizing journals by disciplinary focus and calculating the proportion of publications in each category. Considering impact factors and target audiences of journals, as well as citation counts, is important in this context, as inclusion in high-impact, interdisciplinary journals indicates that the research is accessible and valuable to a broad academic audience beyond a single discipline (e.g., Carr et al., 2017; Leahey et al., 2017; Okamura, 2019).

### Invention Reports and Patents

Invention reports and patents can be valuable metrics for evaluating the success and impact of team science, particularly in fields where research has the potential to lead to practical applications and technological innovations. These indicators generally offer tangible evidence of a team's ability to translate scientific discoveries into real-world solutions, reflecting the practical significance and economic value of their research (Fortunato et al., 2018; Tigges et al., 2019; Wuchty et al., 2007).

Invention reports serve as preliminary documentation of novel ideas or technologies developed by a team, and they provide a gate check before applying for a patent (e.g., Harvard University, n.d.; National Institute of Health SEED, n.d.). These reports may be useful for evaluating the creativity and innovative capacity of a science team, as they capture the early stages of ideation and problem-solving. The number and quality of invention reports may indicate a team's effectiveness in generating new ideas and their potential for future technological breakthroughs (University of Michigan Innovation Partnerships, 2024; Tigges et al., 2019).

Patents, on the other hand, represent a more advanced stage of innovation, providing legal protection for inventions and enabling commercialization. The number of patents filed and granted may serve as a key indicator of a team's success, not only in developing novel technologies but also in advancing these ideas to a stage where they have clear utility and market potential (Allen et al., 2016; Vestal & Mesmer-Magnus, 2020). Patents also contribute to the broader impact of team science by facilitating the transfer of knowledge and technology from academic settings to industry and society, potentially leading to new products, processes, and services that address societal needs (National Center for Science and Engineering Statistics, 2024).

However, although invention reports and patents are important indicators of innovation, they have potential limitations in terms of evaluating team science across all fields. For example, newer fields of research may have higher invention potential compared with more established fields, by virtue of how much "blank space" exists. As a result, the number of

invention reports and patents could potentially be skewed by the type of scientific research being conducted, irrespective of a science team's quality. Therefore, although publications, invention reports, and patents can be useful for assessing certain aspects of team science performance, these approaches can be complemented by broader evaluative approaches that include narrative accounts of success. A holistic perspective to team science evaluation can enable a more comprehensive evaluation of team contributions, acknowledging the diverse forms of impact that team science can achieve across different disciplines (e.g., Wooten et al., 2014).

Invention reports and patents are also used as key output metrics in industry to help appraise, as part of a broader evaluation matrix, the return on investment at the project, department, and organizational levels. Also trackable are the conversion of patented and unpatented technologies into new commercial products (Nerkar & Shane, 2007) that yield monetary gains and the percentage of sales that come from newly innovated products. However, these output metrics need to be put into context with the macroeconomic environment, as changes in the economy will likely result in changes to inputs such as funding for research and development and outputs such as revenue (Mezzanotti & Simcoe, 2023). In addition, the quality of innovation management systems between and within companies and industries can vary widely, which can strongly influence output metrics and be disconnected from the quality of team science taking place on a particular project. The quality of innovation management systems is important enough that the International Organization for Standardization (2019) has issued a comprehensive set of guidelines: Innovation management—Innovation management system. As a result, from a team science perspective, control variables can be confounded by external and internal factors that hinder the ability to adequately isolate how an organization's structural changes directly impact the quality of team science and associated outcomes (Memon et al., 2024). Temporal impacts also need to be considered, as any change that is made to a procedure or process will usually take months or years to impact scientific and/or innovation output metrics (Miller et al., 2021). As a result, it can be difficult to parse the correlational and causational factors when using invention reports, patents, and other innovation metrics as a proxy for the effectiveness of team science.

#### Research Awards

Securing research funding is sometimes used as an output metric to evaluate a science team's success. Effective team science can lead to the development of new fields or lines of research that attract funding and inspire new sponsored programs or institutional strategic initiatives. This

influx of resources enables teams to explore innovative ideas, support interdisciplinary collaborations, and build research capacity that would otherwise be unattainable. For example, the National Academies Keck Futures Initiative (NAKFI) was a 15-year program supported by a \$40 million grant from the W. M. Keck Foundation. Its goal was to advance the future of science by supporting innovative interdisciplinary research ideas generated during think tank–style conferences. Each of these conferences focused on different real-world challenges. Attendees were able to compete for seed fund grants, allowing them to pursue what the NAKFI program viewed as bold and new research ideas (for more information, see National Research Council, 2018).

Funding serves as a catalyst for scholarly growth, providing the necessary support for conducting high-impact research, building infrastructure, and training the next generation of scientists. However, the true measure of success can lie in how well these financial resources are leveraged to generate meaningful scientific contributions, address complex societal challenges, and foster sustained academic and community engagement (Shrivastava et al., 2020; Tebes & Thai, 2018). Therefore, although the ability to secure research dollars is a critical component of evaluating team science, it needs to be considered in the context of how these resources are used to achieve broader intellectual and societal impacts.

## Other Performance Output Considerations

As mentioned throughout this discussion, those researching and evaluating science teams ought to take a broad perspective when considering scientific outputs. This can include intellectual output beyond publications and patents. For example, science teams can sometimes produce spin-off projects based on new ideas generated, which sometimes lead to new grants. They might create new methodologies from combining approaches developed in separate fields or from addressing a need in their innovation process. Science teams can also be evaluated on outcomes such as development of new scientific software or unique datasets that came out of their project, either out of need or from discovery; datasets are particularly important given that they require intensive interdisciplinary collaboration.

A related form of evaluation pertains to how the team members are changing. Science teams can create an informal learning culture where knowledge and skills are transferred as the research progresses. This could include learning new methods for research or creating a shared vocabulary that transcends disciplinary language barriers (Dietl et al., 2023; Tannenbaum et al., 2009). In this process, science team members are developing collaboration competencies. While these competencies are important, they might not always be tracked consistently (Strimel et al., 2014). This

informal learning, however, is crucial to the scientific ecosystem, as it is the main way new knowledge is acquired after graduate school.

Although some funders already account for educational outcomes (e.g., students graduated, theses supported), evaluation of team science needs to also consider interdisciplinary integration in student research (e.g., Laursen et al., 2023). Team science projects commonly lead to the development of new courses and even new degree programs based upon the need for interdisciplinary graduate training. What is more, the enhanced professional networks developed through collaboration cannot be captured via coauthor networks. These social–professional ties are important to track because they can lead to collaborations that are not normally traced with traditional methods.

New subgroups in professional societies can be produced when a research topic has broader implications and may lead to new interdisciplinary societies (e.g., funding for smart cities research led to the development of technical groups and dedicated meetings in computer science societies). In this vein, researcher output can also lead to articles reflecting on the implication of findings for science policy and to informal science articles, such as online magazines or blogs, or popular press articles in science magazines. These kinds of outreach activities become increasingly important to demonstrate the broader impact science teams can have on the community at large and to educate the public about new discoveries.

# Impact on Financial and Human Resources for Communities and Institutions

When evaluating team science output, it is also crucial to measure and document the impact these efforts have on the institutions and communities in which the science teams operate. For example, science team output may directly impact training and development opportunities for junior scholars. Indicators such as the creation of new academic programs and classes at universities, the integration of cutting-edge research findings into curricula, and the inclusion of team science outcomes in textbooks and other education resources are key metrics to consider (Steer et al., 2017; Wallen et al., 2019). These indicators could help assess whether the team is producing new knowledge that is being disseminated to relevant communities and whether the next generation of scientists is gaining a comprehensive understanding of interdisciplinary approaches and collaborative research methods.

Furthermore, it is important to track and document the competitive advantages that team science can provide for institutions, particularly in terms of financial profitability and resource acquisition. Many of the metrics typically used to assess the performance of science teams, such

as publications, patents, and grant funding (noted previously), can also be applied at the center, institution, or community levels. For example, metrics such as increased grant funding, which is often a result of successful interdisciplinary collaborations, can serve as valuable indicators of the initiative's financial impact. This approach provides valuable insights into whether science teams are achieving their desired outcomes and whether institutional-level interventions are effectively driving these results.

Additionally, documenting the recruitment and retention of talented personnel highlights the role of team science in making institutions more attractive to scholars and professionals. These factors are crucial for understanding how well the institution is leveraging its commitment to innovative, collaborative research to enhance its overall standing and resource base, as well as its ability to positively impact local and global communities for the benefit of their workforce and for scientific reach (e.g., Parilla & Haskins, 2023; Valero & Van Reenen, 2019).

At the macro level, the impact of team science on connections among groups and disciplines could be measured and documented to evaluate whether large-scale initiatives are having the desired impact on the communities that have invested in them and on the communities where these initiatives are invested. Metrics such as the development of more interconnected publication networks of scientists from different disciplines, the strengthening of bonds within and across research groups, and the establishment of connections with relevant agencies and other invested groups are all potential indicators of team science success. As an example, the National Science Foundation (NSF) had funded the Research Coordination Networks program to advance a field or create new directions in research by supporting investigators across international, geographic, and organizational boundaries (National Science Foundation, n.d.). Metrics that assess these connections might provide insight into how effectively team science initiatives are breaking down silos, facilitating knowledge, promoting the translation of scientific findings into practical applications, and improving the communities that they draw from and are embedded in. Therefore, by systematically tracking the development and evolution of connectivity among relevant parties, and their impact on science and on various communities, evaluators can better understand the role of team science in advancing scientific progress.

## **Evaluating Broader Impacts**

Defining societal impact in research and academia goes beyond the traditional metrics of success, such as impact factor, publication counts, invited talks, and patents. *Societal impact* is the tangible influence that research has on society, encompassing the ways in which knowledge creation improves public well-being, shapes policies, and drives societal advancements. In 2020,

the United Nations (UN) published the report Shaping the Trends of Our Time, which lists five interconnected megatrends—the greatest challenges facing humanity—as (a) climate change and environmental degradation, (b) demographic trends and population aging, (c) sustainable urbanization, (d) digital technologies, and (e) inequalities. The UN prioritized these megatrends, which will shape the world for the next several decades, because they are human challenges that can be modified by human choices (United Nations, 2020). Effective team science can help address each of these societal challenges by accelerating advances in science and technology and developing practical and scalable solutions. While traditional metrics highlight the academic reach of research, they often fail to capture how that research contributes to solving real-world problems such as those identified by the UN. To comprehensively assess societal impact, one must consider alternative measures, such as community engagement, public policy influence, and the practical application of research findings in nonacademic sectors (D'Este & Robinson-García, 2023).

One of the key challenges in measuring societal impact is the significant time lag between when research is conducted and when its broader societal effects become evident (Siar, 2023). For instance, a study on public health might not show its true societal benefits until years after the findings have been applied in policy reforms or health care interventions. Similarly, environmental research could take decades before its recommendations lead to significant ecological or regulatory changes. As a result, tracking societal impact requires a long-term view and an understanding that traditional, short-term metrics may not capture the full scope of influence.

Another difficulty lies in determining who is responsible for tracking societal impact. Unlike academic citations, which are relatively easy to quantify through established databases, societal impact is diffuse, involving multiple entities such as government agencies, nonprofits, industry partners, and community organizations. Researchers may lack the resources or time to systematically track how their work is being utilized outside academia (e.g., Oliver et al., 2014). As such, institutions may need to play a larger role in this effort, possibly through dedicated offices or staff that monitor the application of research in broader societal contexts. Collaboration with external partners who are involved in the implementation of research could also provide valuable insights.

Supporting such tracking efforts is another significant hurdle, as these efforts require a focused effort (Celeste et al., 2014), particularly to go beyond anecdotal evidence and develop robust methods for documenting influence. Universities and research institutions need to allocate resources to track the diffusion and implementation of research across various societal domains. This could involve establishing partnerships with policymakers, industries, or public organizations that can monitor and report on the

outcomes of research projects. Furthermore, platforms where researchers can share the broader impacts of their work, such as policy briefs or community impact reports, could help document societal contributions more systematically.

Ultimately, the challenge of defining and tracking societal impact invites a broader conversation about what one values in research. Expanding the understanding of impact to include societal contributions requires a shift in both institutional priorities and reward systems. Instead of focusing solely on traditional academic outputs, institutions can recognize and incentivize efforts that engage with communities, influence public policy, and address pressing societal challenges (Ozer et al., 2023). Broadening the criteria for success fosters a research ecosystem that not only advances knowledge but also drives meaningful change in the world.

The Translational Science Benefits Model<sup>2</sup> is a framework that helps determine the range of societal impacts research can have on society. The model takes into consideration a number of potential domains that can be influenced, such as policy, economic, health or clinical, as well as the community. One valuable aspect of this model is its ability to help researchers think in the planning phase about how their research will have impact and how they will measure the benefits of its impact. For example, in health care, researchers can examine whether scientific papers are referenced in policy documents and whether these policies lead to improved health care or societal outcomes.

Societal trust in science is another critical component of understanding science's broader impact. If people in society have a high degree of trust in scientific findings, research-backed policies and clinical practices may be more likely to be accepted and implemented effectively. Conversely, if public trust in science is low, even the most robust policies and practices may struggle to gain traction (Anderson et al., 2024; Goldenberg, 2023).

## Evaluating a Science Team's Impact on Individual Members

## Psychological Impact

Evaluating the impact of being part of a science team on an individual team member can involve several important dimensions (Tay et al., 2023). One key aspect is individual well-being, which encompasses job satisfaction, engagement, and a sense of meaning and professional identity (Gibson et al., 2023). Being part of a collaborative team can enhance these elements by providing a supportive environment where scientists feel valued and

<sup>&</sup>lt;sup>2</sup> More information about the Translational Science Benefits Model is available at https://translationalsciencebenefits.wustl.edu/

connected to a larger purpose. When team members experience high levels of well-being, they are more likely to be motivated, productive, and committed to their work (Gibson et al., 2023). Another important factor is the individual's experience of inclusion, or even fusion within the team (Swann et al., 2009). When scientists perceive a fair and inclusive environment, they are more likely to contribute effectively and feel a sense of belonging, which is crucial for maintaining a healthy and positive team dynamic (Gibson et al., 2023; Salas et al., 2015).

Learning and development are also critical outcomes of team participation. Scientists not only advance their own skills and knowledge through collaboration but also contribute to the development of those around them (Bennett & Gladlin, 2012). This reciprocal process of learning fosters a team culture of continuous improvement and innovation, which can have long-lasting benefits for both the individual and the group (Thayer et al., 2018).

### Professional Networks

Being part of a science team could significantly impact an individual scientist by broadening and diversifying their network of collaborators. This expanded network can be tracked over time, providing insights into the development of the scientist's professional relationships (Fortunato et al., 2018; Okamoto & Centers for Population Health and Health Disparities Evaluation Working Group, 2015). As a scientist works with new colleagues across different disciplines, their network might grow not only in size, but also in perspective diversity, which can lead to richer, more innovative collaborations (van Knippenberg et al., 2020). This diversification of connections may enhance the scientist's ability to tackle complex research problems by bringing in fresh perspectives and expertise from a wide array of fields.

Furthermore, the social capital gained from being part of a diverse science team or group may facilitate the recruitment and retention of even more diverse team members over time (e.g., Harris et al., 2025). As scientists work together with new colleagues from different networks, they not only produce collaborative outputs such as publications, but also can establish lasting professional relationships. This is particularly evident in academia, where the freedom to pursue various research topics and the support provided by universities or agencies—such as through multi-disciplinary workshops and collaborative programs—make it easier to connect with new people (Ertas et al., 2003; Hannon et al., 2018). Additionally, the transient nature of student populations in academic settings provides a continuous influx of new talent, further enriching the team's diversity and collaborative potential.

#### Career Success

Strengthening a network of collaborators, including participating in team science, can also have meaningful outcomes for a scientist's career. As collaborative relationships depend and become more productive, they can lead to more significant achievements, such as coauthored publications, joint grant applications, and shared researched projects, that are reflected on the scientist's curriculum vitae. The value of these connections is often seen in the sustained collaborations that result in regular outputs over many years, which can be particularly beneficial in an academic setting, where long-term partnerships may be more feasible (Bu et al., 2018). For example, it is generally a baseline requirement in industry that employees will develop a proven track record of being able to collaborate effectively; this ability can be considered in the recruitment and promotion processes (Klein & Falk-Krzesinski, 2017). A key differentiator in hiring can be an individual's experience with complex collaborations in team science, such as inter- or transdisciplinary work, especially with teams that are geographically dispersed. This may be true for global industrial firms, where team science can stretch across disciplines and geographies (Hung et al., 2021; Jones et al., 2008; Mazzucchelli et al., 2021).

#### Potential Limitations

It is important to recognize certain limitations when evaluating the impact of team science at the individual level. While many aspects discussed are inherently positive, such as the expansion of professional networks and the promotion of learning and development outcomes, some scholars have highlighted significant trade-offs and potential negative consequences associated with team participation (Benson et al., 2016; Conn et al., 2019). For instance, Forscher et al. (2023) underscored the risk of unaccountable leadership within large-scale team science initiatives, and Mäkinen et al. (2025) identified challenges faced by untenured faculty in gaining recognition for their contributions to interdisciplinary research. Additionally, research has shown that for junior employees, membership in multiple teams is associated with greater role ambiguity, lower job performance, and higher absenteeism (van de Brake et al., 2020). Berkes et al. (2024) demonstrated that researchers who engage in interdisciplinary work early in their careers tend to have less career success and stop publishing sooner than those who initially stay within their discipline. Ensuring that early career researchers receive guidance, recognition, and resources can help mitigate these risks and promote more sustainable career paths in interdisciplinary research. Taken together, these findings suggest the potential trade-offs inherent in team science participation warrant consideration.

#### CONCLUSION AND RECOMMENDATION

Conclusion 5-1: Data collection and evaluation, supported by both institutions and science team leaders, are critical for answering questions about key features of science teams:

- How social processes on the team are unfolding (e.g., team members' perception of experiences of participation on the team, team member satisfaction, psychological safety, trust).
- What the team is producing (e.g., successfully completed team objectives, publications, patents and invention reports, research grant applications and awards, educational outcomes).
- What impact the team is having on individual members (e.g., well-being, skill and knowledge advancement, growth of professional networks, career success).

Recommendation 5-1: Funding agencies, including the National Science Foundation, the National Institutes of Health, and the many other agencies and foundations that support research, should require that the science teams they support develop an evaluation plan to assess their effectiveness and impact. The plans should incorporate team dynamics (e.g., social processes), team performance (e.g., bibliometric metrics), and impact on members (e.g., learning and development outcomes). Regular review periods should be established with the team to monitor and track progress and team effectiveness.

#### REFERENCES

- Agency for Healthcare Research and Quality. (n.d.). TeamSTEPPS Teamwork Perceptions Questionnaire.
- Allen, T. J., Gloor, P., Fronzetti Colladon, A., Woerner, S. L., & Raz, O. (2016). The power of reciprocal knowledge sharing relationships for startup success. *Journal of Small Business* and Enterprise Development, 23, 636–651. https://doi.org/10.1108/JSBED-08-2015-0110
- Anderson, J. D., Malone, T., & Akridge, J. T. (2024). Strategies for land-grant universities to foster public trust. *Journal of Agricultural and Applied Economics*, 1–16.
- Barry, E. S., Varpio, L., Teunissen, P., Vietor, R., & Kiger, M. (2024). Preparing military interprofessional health care teams for effective collaboration. *Military Medicine*, 190(3-4), e804–e810. https://doi.org/10.1093/milmed/usae515
- Beal, D. J., Cohen, R. R., Burke, M. J., & McLendon, C. L. (2003). Cohesion and performance in groups: A meta-analytic clarification of construct relations. *Journal of Applied Psychology*, 88, 989–1004.
- Belcher, B. M., Rasmussen, K. E., Kemshaw, M. R., & Zornes, D. A. (2015). Defining and assessing research quality in a transdisciplinary context. *Research Evaluation*, 25(1), 1–17.
- Bennett, L. M., & Gadlin, H. (2012). Collaboration and team science: From theory to practice. Journal of Investigative Medicine: The Official Publication of the American Federation for Clinical Research, 60(5), 768–775. https://doi.org/10.2310/JIM.0b013e318250871d

- Benson, M. H., Lippitt, C. D., Morrison, R., Cosens, B., Boll, J., Chaffin, B. C., Fremier, A. K., Heinse, R., Kauneckis, D., Link, T. E., Scruggs, C. E., Stone, M., & Valentin, V. (2016). Five ways to support interdisciplinary work before tenure. *Journal of Environmental Studies and Sciences*, 6, 260–267. https://doi.org/10.1007/s13412-015-0326-9
- Berkes, E., Marion, M., Milojević, S., & Weinberg, B. A. (2024). Slow convergence: Career impediments to interdisciplinary biomedical research. *Proceedings of the National Academy of Sciences*, 121(32), e2402646121. https://doi.org/10.1073/pnas.2402646121
- Bu, Y., Ding, Y., Liang, X., & Murray, D. S. (2018). Understanding persistent scientific collaboration. Journal of the Association for Information Science and Technology, 69(3), 438–448.
- Carpenter, C. R., Cone, D. C., & Sarli, C. C. (2014). Using publication metrics to highlight academic productivity and research impact. *Academic Emergency Medicine*, 21(10), 1160–1172. https://doi.org/10.1111/acem.12482
- Carr, G., Blanch, A. R., Blaschke, A. P., Brouwer, R., Bucher, C., Farnleitner, A. H., Fürnkranz-Prskawetz, A., Loucks, D. P., Morgenroth, E., Parajka, J., Pfeifer, N., Rechberger, H., Wagner, W., Zessner, M., & Blöschl, G. (2017). Emerging outcomes from a cross-disciplinary doctoral programme on water resource systems. Water Policy, 19(3), 463–478. https://doi.org/10.2166/wp.2017.054
- Carr, G., Loucks, D. P., & Blöschl, G. (2018). Gaining insight into interdisciplinary research and education programmes: A framework for evaluation. *Research Policy*, 47(1), 35–48. https://doi.org/10.1016/j.respol.2017.09.010
- Carucci, R. (2018). When companies should invest in training their employees and when they shouldn't. *Harvard Business Review*. https://hbr.org/2018/10/when-companies-should-invest-in-training-their-employees-and-when-they-shouldnt
- Celeste, R. F., Griswold, A., & Straf, M. L. (2014). Committee on Assessing the Value of Research in Advancing National Goals. Measuring Research Impacts and Quality. https://www.ncbi.nlm.nih.gov/books/NBK253890/
- Conn, V. S., McCarthy, A. M., Cohen, M. Z., Anderson, C. M., Killion, C., DeVon, H. A., Topp, R., Fahrenwald, N. L., Herrick, L. M., Benefield, L. E., Smith, C. E., Jefferson, U. T., & Anderson, E. A. (2019). Pearls and pitfalls of team science. Western Journal of Nursing Research, 41(6), 920–940.
- Cummings, J. N., Kiesler, S., Bosagh Zadeh, R., & Balakrishnan, A. D. (2013). Group heterogeneity increases the risks of large group size: A longitudinal study of productivity in research groups. *Psychological Science*, 24(6), 880–890. https://doi.org/10.1177/0956797612463082
- Day, D., Bastardoz, N., Bisbey, T., Reyes, D., & Salas, E. (2021). Unlocking human potential through leadership training & development initiatives. *Behavioral Science & Policy*, 7(1), 41–54. https://doi.org/10.1177/237946152100700105
- D'Este, P., & Robinson-García, N. (2023). Interdisciplinary research and the societal visibility of science: The advantages of spanning multiple and distant scientific fields. *Research Policy*, 52(2), 104609.
- De Jong, B. A., Dirks, K. T., & Gillespie, N. (2016). Trust and team performance: A metaanalysis of main effects, moderators, and covariates. *Journal of Applied Psychology*, 101(8), 1134.
- de Wit, F. R., C., Greer, L. L., & Jehn, K. A. (2012). The paradox of intragroup conflict: A meta-analysis. *Journal of Applied Psychology*, 97(2), 360–390.
- Delise, L., Gorman, C. A., Brooks, A., Rentsch, J., & Steele-Johnson, D. (2010). The effects of team training on team outcomes: A meta-analysis. *Performance Improvement Quarterly*, 22(4), 53–80. https://doi.org/10.1002/piq.20068

- Dietl, J. E., Derksen, C., Keller, F. M., & Lippke, S. (2023). Interdisciplinary and interprofessional communication intervention: How psychological safety fosters communication and increases patient safety. *Frontiers in Psychology*, 14, 1164288.
- Donaldson, M. R., & Cooke, S. J. (2014). Scientific publications: Moving beyond quality and quantity toward influence. *BioScience*, 64(1), 12–13. https://doi.org/10.1093/biosci/bit007
- Duhigg, C. (2016, February 25). What Google learned from its quest to build the perfect team. The New York Times. https://www.nytimes.com/2016/02/28/magazine/what-google-learned-from-its-quest-to-build-the-perfect-team.html
- Edmondson, A. (1999). Psychological safety and learning behavior in work teams. *Administrative Science Quarterly*, 44(2), 350–383. https://doi.org/10.2307/2666999
- Edmondson, A. C., & Roloff, K. S. (2008). Overcoming barriers to collaboration: Psychological safety and learning in diverse teams. In E. Salas, G. F. Goodwin, & C. S. Burke (Eds.), *Team effectiveness in complex organizations* (pp. 217–242). Routledge.
- Ertas, A., Maxwell, T., Rainey, V. P., & Tanik, M. M. (2003). Transformation of higher education: The transdisciplinary approach in engineering. *IEEE Transactions on Education*, 46(2), 289–295.
- Falk-Krzesinski, H. J., Contractor, N., Fiore, S. M., Hall, K. L., Kane, C., Keyton, J., Thompson Klein, J., Spring, B., Stokols, D., & Trochim, W. (2011). Mapping a research agenda for the science of team science. *Research Evaluation*, 20(2), 145–158. https://doi.org/10.3152/095820211X12941371876580
- Forscher, P. S., Wagenmakers, E. J., Coles, N. A., Silan, M. A., Dutra, N., Basnight-Brown, D., & IJzerman, H. (2023). The benefits, barriers, and risks of big-team science. *Perspectives on Psychological Science*, 18(3), 607–623.
- Fortunato, S., Bergstrom, C. T., Börner, K., Evans, J. A., Helbing, D., Milojević, S., Petersen, A. M., Radicchi, F., Sinatra, R., Uzzi, B., Vespignani, A., Waltman, L., Wang, D., & Barabási, A.-L. (2018). Science of science. *Science*, 359(6379). https://doi.org/10.1126/science.aao0185
- Frazier, M. L., Fainshmidt, S., Klinger, R. L., Pezeshkan, A., & Vracheva, V. (2017). Psychological safety: A meta-analytic review and extension. *Personnel Psychology*, 70, 113–165.
- Gibson, C., Thomason, B., Margolis, J., Groves, K., Gibson, S., & Franczak, J. (2023). Dignity inherent and earned: The experience of dignity at work. *Academy of Management Annals*, 17(1), 218–267.
- Goldenberg, M. J. (2023). Public trust in science. *Interdisciplinary Science Reviews*, 48(2), 366–378.
- Hackman, J. R. (1987). The design of work teams. In J. W. Lorsch (Ed.), *Handbook of organizational behavior* (pp. 315–342). Prentice-Hall.
- Hall, K. L., Vogel, A. L., Huang, G. C., Serrano, K. J., Rice, E. L., Tsakraklides, S. P., & Fiore, S. M. (2018). The science of team science: A review of the empirical evidence and research gaps on collaboration in science. *American Psychologist*, 73(4), 532. https://doi.org/10.1037/amp0000319
- Hannon, C. (2018). How to plan a design workshop. Chris Hannon Creative. https://www.chrishannoncreative.com/blog/2018/8/9/how-to-plan-a-design-research-workshop
- Harris, H., Tan, I. M. C., Qiu, Y., Brouwer, J., Sosa, J. A., & Yeo, H. (2025). Faculty characteristics and surgery trainee attrition. JAMA Surgery. https://doi.org/10.1001/jamasurg.2025.0274
- Harvard University. (n.d.). *Inventions and invention reporting*. https://osp.finance.harvard.edu/inventions-and-invention-reporting
- Heggestad, E. D., Scheaf, D. J., Banks, G. C., Monroe Hausfeld, M., Tonidandel, S., & Williams, E. B. (2019). Scale adaptation in organizational science research: A review and best-practice recommendations. *Journal of Management*, 45(6), 2596–2627. https://doi. org/10.1177/0149206319850280

- Hung, S. W., Cheng, M. J., Hou, C. E., & Chen, N. R. (2021). Inclusion in global virtual teams: Exploring non-spatial proximity and knowledge sharing on innovation. *Journal of Business Research*, 128, 599–610.
- Ilgen, D. R., Hollenbeck, J. R., Johnson, M., & Jundt, D. (2005). Teams in organizations: From input-process-output models to IMOI models. *Annual Review of Psychology*, 56, 517–543. https://doi.org/10.1146/annurev.psych.56.091103.070250
- International Organization for Standardization. (2019). Innovation management—Innovation management system (ISO No. 56002:2019). https://www.iso.org/standard/68221.html
- Jehn, K. A., & Mannix, E. A. (2001). The dynamic nature of conflict: A longitudinal study of intragroup conflict and group performance. *Academy of Management Journal*, 44(2), 238–251. https://doi.org/10.5465/3069453
- Jones, B. F., Wuchty, S., & Uzzi, B. (2008). Multi-university research teams: Shifting impact, geography, and stratification in science. Science, 322(5905), 1259–1262.
- Klein, J. T. (2008). Evaluation of interdisciplinary and transdisciplinary research: A literature review. *American Journal of Preventive Medicine*, 35(Suppl 2), S116–S123.
- Klein, J. T., & Falk-Krzesinski, H. J. (2017). Interdisciplinary and collaborative work: Framing promotion and tenure practices and policies. *Research Policy*, 46(6), 1055–1061.
- Kozlowski, S. W., & Ilgen, D. R. (2006). Enhancing the effectiveness of work groups and teams. *Psychological science in the public interest*, 7(3), 77–124.
- Laursen, B. K., Motzer, N., & Anderson, K. J. (2022). Pathways for assessing interdisciplinarity: A systematic review. Research Evaluation, 31(3), 326–343. http://dx.doi.org/10.1093/reseval/rvac013
- \_\_\_\_. (2023). Pathway profiles: Learning from five main approaches to assessing interdisciplinarity. Research Evaluation, 32(2), 213–227. https://doi.org/10.1093/reseval/rvac036
- Leahey, E., Beckman, C. M., & Stanko, T. L. (2017). Prominent but less productive: The impact of interdisciplinarity on scientists' research. *Administrative Science Quarterly*, 62(1), 105–139. https://doi.org/10.1177/0001839216665364
- Lee, S. S.-J., & A. Jabloner (2017). Institutional culture is the key to team science. *Nature Biotechnology*, 35(12), 1212–1214.
- Lewis, K. (2003). Measuring transactive memory systems in the field: Scale development and validation. *Journal of Applied Psychology*, 88, 587–604. https://doi.org/10.1037/0021-9010.88.4.587
- Llewellyn, N., Carter, D. R., DiazGranados, D., Pelfrey, C., Rollins, L., & Nehl, E. J. (2020). Scope, influence, and interdisciplinary collaboration: The publication portfolio of the NIH Clinical and Translational Science Awards (CTSA) program from 2006 through 2017. Evaluation & the Health Professions, 43(3), 169–179. https://doi.org/10.1177/0163278719839435
- Llewellyn, N., Nehl, E. J., Dave, G., DiazGranados, D., Flynn, D., Fournier, D., Hoyo, V., Pelfrey, C., & Casey, S. (2024). Translation in action: Influence, collaboration, and evolution of COVID-19 research with Clinical and Translational Science Awards consortium support. Clinical and Translational Science, 17(1), e13700. https://doi.org/10.1111/ cts.13700
- Love, H. B., Fosdick, B. K., Cross, J. E., Suter, M., Egan, D., Tofany, E., & Fisher, E. R. (2022). Towards understanding the characteristics of successful and unsuccessful collaborations: A case-based team science study. *Humanities and Social Sciences Communications*, 9(1), 371.
- Lyu, P., Liu, X., & Yao, T. (2023). A bibliometric analysis of literature on bibliometrics in recent half-century. *Journal of Information Science*. https://doi.org/10.1177/01655515231191233
- Mäkinen, E. I., Evans, E. D., & McFarland, D. A. (2025). Interdisciplinary research, tenure review, and guardians of the disciplinary order. *The Journal of Higher Education*, 96(1), 54–81.

- Mansilla, V. B. (2006). Assessing expert interdisciplinary work at the frontier: An empirical exploration. *Research Evaluation*, 15(1), 17–29. http://dx.doi.org/10.3152/147154406781776075
- Marks, M. A., Mathieu, J. E., & Zaccaro, S. J. (2001). A temporally based framework and taxonomy of team processes. *The Academy of Management Review*, 26(3), 356–376.
- Marlow, S. L., Lacerenza, C. N., Paoletti, J., Burke, C. S., & Salas, E. (2018). Does team communication represent a one-size-fits-all approach?: A meta-analysis of team communication and performance. Organizational Behavior and Human Decision Processes, 144, 145–170.
- Mâsse, L. C., Moser, R. P., Stokols, D., Taylor, B. K., Marcus, S. E., Morgan, G. D., Hall, K. L., Croyle, R. T., & Trochim, W. M. (2008). Measuring collaboration and transdisciplinary integration in team science. *American Journal of Preventive Medicine*, 35(Suppl 2), S151–160.
- Mathieu, J. E., Luciano, M. M., D'Innocenzo, L., Klock, E. A., & LePine, J. A. (2020). The development and construct validity of a team processes survey measure. *Organizational Research Methods*, 23(3), 399–431. https://doi.org/10.1177/1094428119840801
- Mazzucchelli, A., Chierici, R., Tortora, D., & Fontana, S. (2021). Innovation capability in geographically dispersed R&D teams: The role of social capital and IT support. *Journal of Business Research*, 128, 742–751.
- Memon, M. A., Thurasamy, R., Ting, H., Cheah, J. H., & Chuah, F. (2024). Control variables: a review and proposed guidelines. *Journal of Applied Structural Equation Modeling*, 8(2), 1–18.
- Mesmer-Magnus, J. R., & DeChurch, L. A. (2009). Information sharing and team performance: A meta-analysis. *Journal of Applied Psychology*, 94(2), 535–546.
- Mezzanotti, F., & Simcoe, T. S. (2023, August). Research and/or development? Financial frictions and innovation investment (NBER Working Paper No. w31521). National Bureau of Economic Research. https://ssrn.com/abstract=4533235
- Michalska-Smith, M. J., & Allesina, S. (2017). And, not or: Quality, quantity in scientific publishing. *PloS One*, 12(6). https://doi.org/10.1371/journal.pone.0178074
- Miller, B. M., Metz, D., Schmid, J., Rudin, P. M., Blumenthal, M. S. (2021). Measuring the value of invention: The impact of Lemelson-MIT prize winners' inventions. *RAND*. https://www.rand.org/pubs/research\_reports/RRA838-1.html
- Mohammed, S., Rico, R., & Alipour, K. K. (2021). Team cognition at a crossroad: Toward conceptual integration and network configurations. *Academy of Management Annals*, 15(2), 455–501. https://doi.org/10.5465/annals.2018.0159
- Nancarrow, S. A., Booth, A., Ariss, S., Smith, T., Enderby, P., & Roots, A. (2013). Ten principles of good interdisciplinary team work. *Human Resources for Health*, 11, 1–11. https://doi.org/10.1186/1478-4491-11-19
- National Center for Science and Engineering Statistics. (2024, February 29). *Knowledge transfer indicators: Putting information to use.* https://ncses.nsf.gov/pubs/nsb20241/knowledge-transfer-indicators-putting-information-to-use
- National Institute of Health SEED. (n.d.). *Intellectual property and iEdison invention report requirements*. https://seed.nih.gov/small-business-funding/small-business-program-basics/grant-policy/ip#:~:text=Invention%20Reporting%20though%20iEdison,of%20the%20participating%20federal%20agencies
- National Research Council (2018). Collaborations of consequence: NAFKI's 15 years igniting innovation at the intersections of disciplines. The National Academies Press. https://doi.org/10.17226/25239
- National Science Foundation. (n.d.) Research coordination networks program. https://www.nsf.gov/funding/opportunities/research-coordination-networks/11691/nsf23-529

- Nerkar, A., & Shane, S. (2007). Determinants of invention commercialization: An empirical examination of academically sourced inventions. *Strategic Management Journal*, 28(11), 1155–1166.
- O'Connor, G. C., Rice, M. P., Peters, L., & Veryzer, R. W. (2003). Managing interdisciplinary, longitudinal research teams: Extending grounded theory-building methodologies. *Organization Science*, 14(4), 353–373. https://www.jstor.org/stable/4135115
- Ohland, M. W., Loughry, M. L., Woehr, D. J., Bullard, L. G., Felder, R. M., Finelli, C. J., Layton, R. A., Pomeranz, H. R., & Schmucker, D. G. (2012). The comprehensive assessment of team member effectiveness: Development of a behaviorally anchored rating scale for self and peer evaluation. *Academy of Management Learning & Education*, 11, 609–630. https://doi.org/10.5465/amle.2010.0177
- Okamoto, J., & Centers for Population Health and Health Disparities Evaluation Working Group. (2015). Scientific collaboration and team science: A social network analysis of the centers for population health and health disparities. *Translational behavioral medicine*, 5(1), 12–23.
- Okamura, K. (2019). Interdisciplinarity revisited: Evidence for research impact and dynamism. *Palgrave Communications*, *5*(1), 141. https://doi.org/10.1057/s41599-019-0352-4
- Oliver, K., Innvar, S., Lorenc, T., Woodman, J., & Thomas, J. (2014). A systematic review of barriers to and facilitators of the use of evidence by policymakers. *BMC Health Services Research*, 14, 1–12.
- Ozer, E. J., Renick, J., Jentleson, B., & Maharramli, B. (2023). Scan of promising efforts to broaden faculty reward systems to support societally-impactful research. The Pew Charitable Trusts.
- Palmer, M. A., Kramer, J. G., Boyd, J., & Hawthorne, D. (2016). Practices for facilitating interdisciplinary synthetic research: The National Socio-Environmental Synthesis Center (SESYNC). Current Opinion in Environmental Sustainability, 19, 111–122. https://doi. org/10.1016/j.cosust.2016.01.002
- Parilla, J., & Haskins, G. (2023, February 9). How research universities are evolving to strengthen regional economies. *Brookings*. https://www.brookings.edu/articles/how-research-universities-are-evolving-to-strengthen-regional-economies/
- Rico, R., Sánchez-Manzanares, M., Gil, F., & Gibson, C. (2008). Team implicit coordination processes: A team knowledge-based approach. *The Academy of Management Review*, 33, 163–184.
- Roelofs, S., Edwards, N., Viehbeck, S., & Anderson, C. (2019). Formative, embedded evaluation to strengthen interdisciplinary team science: Results of a 4-year, mixed methods, multi-country case study. *Research Evaluation*, 28(1), 37–50.
- Salas, E., Cooke, N. J., & Rosen, M. A. (2008). On teams, teamwork, and team performance: Discoveries and developments. *Human Factors*, 50(3), 540–547. https://doi.org/10.1518/001872008x288457
- Salas, E., Grossman, R., Hughes, A. M., & Coultas, C. W. (2015). Measuring team cohesion: Observations from the science. *Human factors*, 57(3), 365–374. https://doi.org/10.1177/0018720815578267
- Savannah River Site. (2020, February 10). General employee training. https://www.srs.gov/general/training/TREGGETALPLN.pdf
- Schneider, B., Ashworth, S. D., Higgs, A. C., & Carr, L. (1996). Design, validity, and use of strategically focused employee attitude surveys. *Personnel Psychology*, 49(3), 695–705. https://doi.org/10.1111/j.1744-6570.1996.tb01591.x
- Shrivastava, P., Smith, M. S., O'Brien, K., & Zsolnai, L. (2020). Transforming sustainability science to generate positive social and environmental change globally. *One Earth*, 2(4), 329–340.

- Shuffler, M. L., & Carter, D. R. (2018). Teamwork situated in multiteam systems: Key lessons learned and future opportunities. *American Psychologist*, 73(4), 390. https://psycnet.apa.org/doi/10.1037/amp0000322
- Siar, S. (2023). The challenges and approaches of measuring research impact and influence on public policy making. *Public Administration and Policy*, 26(2), 169–183.
- Steer, C. J., Jackson, P. R., Hornbeak, H., McKay, C. K., Sriramarao, P., & Murtaugh, M. P. (2017). Team science and the physician–scientist in the age of grand health challenges. *Annals of the New York Academy of Sciences*, 1404(1), 3–16.
- Stokols, D., Fuqua, J., Gress, J., Harvey, R., Phillips, K., Baezconde-Garbanati, L., Unger, J., Palmer, P., Clark, M. A., Colby, S. M., Morgan, G., & Trochim, W. (2003). Evaluating transdisciplinary science. Nicotine & Tobacco Research: Official Journal of the Society for Research on Nicotine and Tobacco, 5(Suppl 1), S21–S39. https://doi.org/10.1080/14622200310001625555
- Stokols, D., Hall, K. L., Taylor, B. K., & Moser, R. P. (2008). The science of team science: Overview of the field and introduction to the supplement. *American Journal of Preventive Medicine*, 35(Suppl 2), S77–S89. https://doi.org/10.1016/j.amepre.2008.05.002
- Strimel, G., Reed, P., Dooley, G., Bolling, J., Phillips, M., & Cantu, D. V. (2014). Integrating and monitoring informal learning in education and training. *Techniques: Connecting Education & Careers*, 89(3), 48–54.
- Swann Jr, W. B., Gómez, A., Seyle, D. C., Morales, J., & Huici, C. (2009). Identity fusion: The interplay of personal and social identities in extreme group behavior. *Journal of Personality and Social Psychology*, 96(5), 995.
- Tannenbaum, S. I., Beard, R. L., McNall, L. A., & Salas, E. (2009). Informal learning and development in organizations. In *Learning, training, and development in organizations* (pp. 303–331). Routledge.
- Tay, L., Batz-Barbarich, C., Yang, L. Q., & Wiese, C. W. (2023). Well-being: The ultimate criterion for organizational sciences. *Journal of Business and Psychology*, 38(6), 1141–1157.
- Tebes, J. K., & Thai, N. D. (2018). Interdisciplinary team science and the public: Steps toward a participatory team science. *American Psychologist*, 73(4), 549.
- Tekleab, A. G., Quigley, N. R., & Tesluk, P. E. (2009). A longitudinal study of team conflict, conflict management, cohesion, and team effectiveness. *Group & Organization Management*, 34(2), 170–205.
- Thayer, A. L., Petruzzelli, A., & McClurg, C. E. (2018). Addressing the paradox of the team innovation process: A review and practical considerations. *American Psychologist*, 73(4), 363.
- Tigges, B. B., Miller, D., Dudding, K. M., Balls-Berry, J. E., Borawski, E. A., Dave, G., Hafer, N. S., Kimminau, K. S., Kost, R. G., Littlefield, K., Shannon, J., Menon, U., & Measures of Collaboration Workgroup of the Collaboration and Engagement Domain Task Force, National Center for Advancing Translational Sciences, National Institutes of Health (2019). Measuring quality and outcomes of research collaborations: An integrative review. *Journal of Clinical and Translational Science*, 3(5), 261–289. https://doi.org/10.1017/cts.2019.402
- Tremblay, D., Roberge, D., Cazale, L., Touati, N., Maunsell, E., Latreille, J., & Lemaire, J. (2011). Evaluation of the impact of interdisciplinarity in cancer care. *BMC Health Services Research*, 11, 1–10. https://doi.org/10.1186/1472-6963-11-144
- Trochim, W. M., Marcus, S. E., Mâsse, L. C., Moser, R. P., & Weld, P. C. (2008). The evaluation of large research initiatives: A participatory integrative mixed-methods approach. *American Journal of Evaluation*, 29(1), 8–28. https://doi.org/10.1177/1098214007309280
- United Nations. (2020, September). Report of the UN Economist Network for the UN 75th anniversary: Shaping the trends of our time. United Nations. https://www.un.org/en/desa/unen/report

- University of Michigan Innovation Partnerships. (2024). *Impact report*. https://innovationpartnerships. umich.edu/wp-content/uploads/2024/09/91224\_V3\_Digital\_Impact\_Report\_2024.pdf
- Uzzi, B., Mukherjee, S., Stringer, M., & Jones, B. (2013). Atypical combinations and scientific impact. *Science*, 342(6157), 468–472. https://doi.org/10.1126/science.1240474
- Valero, A., & Van Reenen, J. (2019). The economic impact of universities: Evidence from across the globe. *Economics of Education Review*, 68, 53–67. https://doi.org/10.1016/j.econedurev.2018.09.001
- van de Brake, H. J., Walter, F., Rink, F. A., Essens, P. J., & van der Vegt, G. S. (2020). Benefits and disadvantages of individuals' multiple team membership: The moderating role of organizational tenure. *Journal of Management Studies*, 57(8), 1502–1530.
- van Knippenberg, D., Nishii, L. H., & Dwertmann, D. J. (2020). Synergy from diversity: Managing team diversity to enhance performance. *Behavioral Science & Policy*, 6(1), 75–92
- Vestal, A., & Mesmer-Magnus, J. (2020). Interdisciplinarity and team innovation: The role of team experiential and relational resources. *Small Group Research*, 51(6), 738–775. https://doi.org/10.1177/1046496420928405
- Wallen, K. E., Filbee-Dexter, K., Pittman, J. B., Posner, S. M., Alexander, S. M., Romulo, C. L., Bennett, D. E., Clark, E. C., Cousins, S. J. M., Dubik, B. A., Garcia, M., Haig, H. A., Koebele, E. A., Qiu, J., Richards, R. C., Symons, C. C., & Zipper, S. C. (2019). Integrating team science into interdisciplinary graduate education: An exploration of the SESYNC Graduate Pursuit. *Journal of Environmental Studies and Sciences*, 9, 218–233. https://doi.org/10.1007/s13412-019-00543-2
- Waltman, L., & Traag, V. A. (2021). Use of the journal impact factor for assessing individual articles: Statistically flawed or not? *F1000Research*, 9, 366. https://doi.org/10.12688/f1000research.23418.2
- Wiese, C. W., Burke, C. S., Tang, Y., Hernandez, C., & Howell, R. (2021). Team learning behaviors and performance: A meta-analysis of direct effects and moderators. *Group & Organization Management*, 47(3). https://doi.org/10.1177/10596011211016928
- Wooten, K. C., Rose, R. M., Ostir, G. V., Calhoun, W. J., Ameredes, B. T., & Brasier, A. R. (2014). Assessing and evaluating multidisciplinary translational teams: A mixed methods approach. *Evaluation & the Health Professions*, 37(1), 33–49. https://doi.org/10.1177/0163278713504433
- Wuchty, S., Jones, B. F., & Uzzi, B. (2007). The increasing dominance of teams in production of knowledge. *Science*, 316(5827), 1036–1039. https://doi.org/10.1126/science.1136099
- Xu, F., Wu, L., & Evans, J. (2022). Flat teams drive scientific innovation. Proceedings of the National Academy of Sciences, 119(23), e2200927119. https://doi.org/10.1073/ pnas.2200927119
- Yang, Y., Tian, T. Y., Woodruff, T. K., Jones, B. F., & Uzzi, B. (2022). Gender-diverse teams produce more novel and higher-impact scientific ideas. *Proceedings of the National Academy of Sciences*, 119(36), e2200841119. https://doi.org/10.1073/pnas.2200841119



6

## Forward-Looking Research Recommendations and Infrastructure Needs

This report has discussed the existing research on science teams, as well as research on teams more generally, with particular attention to best practices, external support, and evaluation. Team science has already demonstrated benefits in innovation and problem-solving, and there is reason to believe that strengthening both the way team science is performed and the way it is researched and supported would strengthen these benefits and lead to more effective teams. This final chapter reviews the subjects discussed in the previous chapters and assesses needs for additional research specific to the science of team science.

#### **BEST PRACTICES**

The current literature on science teams indicates a pressing need for systematic empirical research to evaluate interventions in real-world team settings. While numerous best practices and interventions have been identified, much of the supporting evidence remains anecdotal or correlational. Chapter 3 highlights how most studies are primarily case based or correlation driven, leaving a gap in experimentally validated best practices that can be applied widely across various team science contexts. By funding real-time studies, research sponsors can facilitate rigorous experimental research that could lead to more generalizable and reliable data, fostering the development of evidence-based guidelines that directly improve team science processes and outcomes. Table 3-1 in Chapter 3 details possible future research questions to address that would strengthen the evidence for best practices.

Science teams often encounter a variety of challenges, ranging from interdisciplinary differences to unique project requirements and team compositions. Thus, interventions need to be tailored to each team's specific circumstances to enhance effectiveness. Chapter 3 notes the diversity of team needs and configurations, which affect the efficacy of interventions labeled as "best practices." For example, research shows that establishing norms around tool use needs to account for individual team characteristics, such as team composition and the cultural or linguistic backgrounds of team members, to promote efficient collaboration. Consequently, a reflexive approach that adapts practices based on each team's requirements and experiences can lead to more effective team functioning. This tailored approach not only respects the unique nature of each science team but also encourages team members to critically evaluate and refine their workflows, fostering adaptability and resilience.

Virtual collaboration tools, if thoughtfully selected and integrated, can address geographic and temporal challenges that often arise in hybrid or distributed teams. Chapter 3 emphasizes that these tools need to be evaluated based on their alignment with team configuration, accessibility needs, and the skills of team members. For example, ensuring that tools include features such as screen sharing, closed captioning, and language translation can help accommodate team needs, increasing participation and improving communication. However, improper or inconsistent use of these tools can hinder trust and create conflict, particularly if team members feel that tools lack transparency or do not provide a secure environment for open communication.

Moreover, periodic reassessment of tool efficacy is recommended as team compositions and technological options evolve. Regularly revisiting tool usage norms can help avoid potential misalignments between team needs and available functionalities, ensuring that the technology remains a facilitator rather than a barrier to effective collaboration. The evolving nature of team science necessitates a flexible approach to tool integration, underscoring the importance of establishing clear norms and ensuring that all members are adequately trained to use these tools effectively.

In summary, supporting science teams through tailored approaches, well-integrated virtual tools, and robust research funding are essential for enabling effective and adaptable team science practices. These recommendations reflect the heterogenous configurations and complex dynamics of science teams, underscoring the importance of real-time research, a flexible methodology, and carefully chosen virtual collaboration tools to foster productive and resilient scientific collaboration.

#### INSTITUTIONAL AND EXTERNAL SUPPORT

As discussed in Chapter 4, a comprehensive institutional approach is critical to support successful team science outcomes. This includes considering team science in decisions regarding personnel, funding, incentives, technology, and policy. However, many institutions lack clear practices or infrastructure for facilitating such a holistic approach, and limited research exists to identify which specific supports are most effective in fostering successful team science environments. Institutional challenges, such as inadequate incentives for teamwork in promotion and tenure criteria, limited access to collaborative technologies, and lack of administrative support, can all hinder team performance. To optimize team science, more research is needed to understand how institutional practices influence collaboration outcomes, allowing for the development of informed strategies for supporting effective teamwork.

To advance team science, funders and institutions need to encourage best practices by recognizing team science contributions in hiring, tenure, promotion, and awards. Traditional evaluation methods, such as publication count and grant funding, often overlook collaborative efforts and are disadvantageous to scientists involved in cross-disciplinary projects. Revising these criteria to value team science roles—such as leadership in large projects and contributions to shared research outcomes, as well as additional contributions that may include mentoring, community-based research, and other work—would incentivize engagement and address current disincentives. Additionally, simplifying data sharing, open-science practices, and ethical review processes can alleviate administrative burdens on science teams, thereby enhancing collaboration. For instance, open-science frameworks that support sharing data and research products streamline workflows and promote transparency, benefiting both individual scientists and their teams.

Active support from funders for research on team science effectiveness is also crucial. Investing in studies on team science competencies, the impact of professional development (e.g., workshops, courses), and metrics to evaluate development of these competencies will provide insights into optimizing team dynamics. This includes research on how incentives affect team science prevalence and success across institutions, helping funders understand and refine policies that foster collaboration. Furthermore, studies examining institutional support structures—like research development professionals and team facilitators—can reveal their contributions to project efficiency and success. For a sustained impact, funders and institutions need to ensure accountability in adapting policies to support team-based science effectively. Addressing these areas comprehensively will not only support team science but also contribute to high-quality, impactful scientific outcomes across disciplines.

#### **EVALUATION**

To effectively evaluate science team outcomes, funders, institutions, and researchers need to prioritize three core questions that capture the essential aspects of team effectiveness: the social processes within the team, the tangible outputs produced, and the positive impact on individual members. Evaluating team dynamics, including social processes such as communication, cohesion, and psychological safety, is fundamental, as these factors directly influence a team's ability to collaborate and achieve high performance. Research suggests that well-functioning social processes foster trust, reduce conflict, and enable teams to address complex tasks more efficiently.

Evaluating outputs—such as publications, patents, and other research deliverables—enables a direct assessment of the team's productivity and contributions to scientific knowledge. By measuring both quantity and quality of these outputs, evaluators can gain insights into the team's innovative capacity and impact within the broader scientific community. That said, evaluation needs to move beyond that which is easily counted. This can include the study of how scientific collaborations adapt scientific concepts, how they change professional networks and societies, as well as the nature and form of material artifacts produced. Additionally, understanding the broader impact of these outputs, including societal or policy contributions, provides a comprehensive view of the team's success.

Finally, assessing the team's impact on individual members is crucial, as participation in team science can significantly affect career development, professional satisfaction, and future collaborative opportunities. Positive experiences within science teams can lead to personal growth, expanded professional networks, and improved career trajectories. Emphasizing individual well-being and professional development aligns with the evolving goals of modern scientific research, which increasingly prioritizes inclusivity and career sustainability for team members. For a comprehensive evaluation, funding agencies and institutions can consider multiple evaluation criteria tailored to specific team goals, resources, and evaluation purposes. A multidimensional evaluation approach that captures short- and long-term outcomes enables a nuanced understanding of team performance, guiding future investments and fostering a supportive environment for team science initiatives to thrive.

## **IDENTIFICATION OF RESEARCH GAPS**

While there has been substantial application of team science principles derived from the social and organizational sciences (e.g., social and organizational psychology, organizational behavior, cognitive science), the

empirical basis for applying these insights to science teams is still limited. This gap is crucial to address, as science teams are distinct from the types of teams typically studied outside of science, raising concerns about the generalizability of these findings. The committee suggests that a better understanding of these distinctions requires context-specific research focused on the unique structures and dynamics of science teams.

Funders need to provide more funding mechanisms for the scientific study of science teams. Indeed, major funding bodies such as the National Science Foundation and National Institutes of Health emphasize the critical importance of team science in solving complex global problems, but the support needed to fund the study of science teams is lacking. This disparity is problematic given the complexity of studying science teams, which requires both significant financial resources and specialized methodological approaches tailored to scientific collaboration contexts.

#### CONCLUSION AND RECOMMENDATION

Conclusion 6-1: The research translating findings on the functioning of teams general to the functioning of science teams is incomplete for reasons that include insufficient funding and a lack of professional recognition and reward for the study of science teams. Specifically needed are:

- Studies focused on the science team context that explore the application of existing theories to the unique processes and dynamics that distinguish science teams from traditional organizational teams.
- Studies of team science competencies and interventions that allow for robust statistical analyses or pre- and post-testing to build the empirical evidence base for team science learning, training, and professional development in real time.
- Data-driven research, including longitudinal studies, to better understand team science outcomes.
- Research to identify and investigate institutional policies and practices that reinforce or serve as barriers to team science, such as the support structures universities can provide (e.g., research development professionals, team facilitators), how these affect team science processes and outcomes, and other incentives that influence the conduct of team science.
- Research into the effectiveness of virtual collaboration tools and how these relate to the specific configuration of hybrid and virtual teams, including factors such as geographic distribution, temporal dynamics, and communication needs.

Recommendation 6-1: Funders interested in supporting the conduct of science should prioritize research on, and provide sufficient funding for, the application of findings from the broader study of teams to the science context. Areas of prioritization may include but are not limited to studies that incorporate both qualitative and quantitative data to build empirical evidence about the science context and research evaluating institutional policies and supports for science teams.

## Appendix A

## Committee Biosketches

DIANA L. BURLEY (Co-chair) is a global cybersecurity expert with more than 30 years of experience driving digital transformation, implementing cybersecurity workforce initiatives, and promoting an equitable global technology community. She is currently vice provost for research and innovation at American University where she also leads the Khan Cyber & Economic Security Institute and serves as a member of the faculty. As both the university's chief research officer and chief innovation officer, Burley oversees the university-wide research and development portfolio, research partnerships, and strategic initiatives to catalyze discovery. She also directs the university's Translating Research into Action Center (TRAC), a \$6 million National Science Foundation-funded initiative launched to strengthen the university's capacity to advance positive societal impact through evidencedriven policy and practice. Burley regularly advises government officials and offers thought leadership at executive forums. Her board service includes the Cyber Future Foundation and the Global Cyber Security Advisory Group, and she has been honored by the Executive Women's Forum, SC Magazine, Association for Computing Machinery, and others for her leadership in building the global cybersecurity workforce. Burley earned her PhD from Carnegie Mellon University.

MO WANG (Co-chair) is the University Distinguished Professor, Lanzillotti-McKethan Eminent Scholar Chair and the associate dean for research and strategic initiatives at the Warrington College of Business at the University of Florida. He is also the chair of the Management Department and the director of the Human Resource Research

Center at the University of Florida. Wang specializes in research areas of organizational behavior and human resource management, especially on team science. He received numerous research awards for his research in these areas, including the Academy of Management (AOM) HR Division Scholarly Achievement Award; Careers Division Best Paper Award; European Union's Erasmus Mundus Scholarship Award for Work, Organizational, and Personnel Psychology; Emerald Group's Outstanding Author Contribution Awards; Society for Industrial-Organizational Psychology's (SIOP) William A. Owens Scholarly Achievement Award; and the Journal of Management Scholarly Impact Award. Wang also received a number of early/mid-career contribution awards from AOM, the American Psychological Association (APA), the Society for Industrial and Organizational Psychology (SIOP), and the Society for Occupational Health Psychology (SOHP). He is an elected Foreign Member of Academia Europaea and a fellow of AOM, APA, the Association for Psychological Science, and SIOP. Previously, Wang was the president of SOHP and SIOP. He received his PhD from Bowling Green State University.

DOROTHY CARTER is an associate professor of management in the Eli Broad College of Business at Michigan State University. Her research focuses on how leaders, teams, and larger multiteam systems can collaborate effectively to achieve shared goals in a variety of contexts including the military, space exploration, and team science. Carter's research on leadership and teamwork has appeared in numerous top journals within the organizational sciences, as well as in interdisciplinary outlets. She has received several research awards including the Rising Star in Leadership Research Award from the Academy of Management's Network of Leadership Scholars, the Charles B. Knapp Early Career Scholar in the Social Sciences Award from the University of Georgia, and the Early Career Award from the Interdisciplinary Network for Groups Research. Carter's research program has been supported by funding from the National Aeronautics and Space Administration, the National Science Foundation, the Army Research Institute, and the National Institutes of Health. She received her MS and PhD in industrial/organizational psychology from the Georgia Institute of Technology.

ERIN KAI-LING CHIOU is an associate professor of human systems engineering at The Polytechnic School, part of the Ira A. Fulton Schools of Engineering at Arizona State University. Her research focuses on human-automation interaction, job design with information and communication technologies, and trust in sociotechnical systems. Chiou's multidisciplinary and team-based research addresses complex work domains like education, health care, and defense. Recent research projects have been supported

APPENDIX A 195

by the Department of Homeland Security, the National Science Foundation, the National Institutes of Health, and the Chief Digital and Artificial Intelligence Office. Chiou has served as a member of the Diversity and Inclusion Committee for the Human Factors and Ergonomics Society, including as chair for two years, and was a recipient of the society's inaugural Inclusion Award in 2024. She is a co-editor of Advancing Diversity, Inclusion, and Social Justice Through Human Systems Engineering (2020), a winner of the Outstanding Academic Title recognition by Choice for the 2020 Open and Affordable Textbook Award from Rutgers University Libraries. Chiou is also an associate editor for the Journal of Cognitive Engineering and Decision-Making. She received her MS and PhD in industrial and systems engineering (human factors and ergonomics) from the University of Wisconsin-Madison, where she was a National Science Foundation Graduate Research Fellow. Chiou has served as a reviewer and committee member for several National Academies of Sciences, Engineering, and Medicine efforts, including as a member for the consensus study on Human-AI Teaming: State-of-the-Art and Research Needs.

**DEBORAH DIAZGRANADOS** is an associate professor in the Department of Psychiatry at Virginia Commonwealth University's School of Medicine and the Director of Evaluation and Team Science for the Kenneth and Dianne Wright Center for Clinical and Translational Research. A leading expert in leadership and collaboration, her work emphasizes the dynamic interplay of context and individual connections in fostering effective teamwork. DiazGranados views collaboration as a dynamic process profoundly shaped by context and the strategic connections individuals make. Her research on shared leadership and collaboration has been particularly influential in demonstrating how to integrate knowledge across disciplines for innovation. DiazGrandos' contributions are evident in numerous publications within highly respected academic journals like the Journal of Applied Psychology, Academic Medicine, Academy of Management Annals, American Psychologist, and Current Directions in Psychological Science. Furthermore, the practical implications of her research extend beyond scholarly circles, with specific articles cited in key policy documents for governmental agencies and health care organizations, underscoring its tangible impact on practice and policy development. With a robust track record, DiazGranados has been a principal investigator (PI) or Co-PI on grants and contracts totaling over \$10 million. She holds an MS and PhD in industrialorganizational psychology from the University of Central Florida.

STEPHEN M. FIORE is the director of the Cognitive Sciences Laboratory and Pegasus Professor with the University of Central Florida's (UCF's) Cognitive Sciences Program in the Department of Philosophy and the

Institute for Simulation & Training. He maintains a multidisciplinary research interest that incorporates aspects of the cognitive, social, organizational, and computational sciences in the investigation of learning and performance in individuals and teams. Fiore's primary area of research is the interdisciplinary study of complex collaborative cognition and the understanding of how humans interact socially and with technology. He is the past president of the International Network for the Science of Team Science and past president of the Interdisciplinary Network for Group Research. Fiore was a member of the expert panel for the Organisation for Economic Co-operation and Development's Programme for International Student Assessment, which focused on collaborative problem-solving skills. He was previously a member of the National Assessment of Educational Progress report Collaborative Problem Solving. Fiore was inducted into UCF's Scroll & Quill Society as recognition of his scholarship having an international impact. He also received UCF's Luminary Award as recognition for his work having a significant impact on the world, and UCF's Reach for the Stars Award, as recognition for bringing international prominence to the university. Fiore was given the title of Pegasus Professor, UCF's highest academic rank. He has a PhD in cognitive psychology from the University of Pittsburgh. Fiore has contributed to working groups for the National Academies of Sciences, Engineering, and Medicine in understanding and measuring 21st century skills and was a committee member of their Science of Team Science consensus study, which produced the 2015 report, Team Science Effectiveness.

MICHAEL C. FRANK is Benjamin Scott Crocker Professor of Human Biology in the Department of Psychology at Stanford University and director of the Symbolic Systems Program. He studies children's language learning and development, with a focus on the use of large-scale datasets to understand the variability and consistency of learning across cultures. Frank is a founder of the ManyBabies Consortium and has led opendata projects including Wordbank and the ongoing Learning Variability Network Exchange (LEVANTE) project. He served as president of the Cognitive Science Society, has edited for journals including Cognition and Child Development, and is the current co-editor in chief of the Open Encyclopedia of Cognitive Science. Frank has received awards including the Troland Award from the National Academy of Sciences and the FABBS Early Career Impact Award. He received his PhD in brain and cognitive sciences from the Massachusetts Institute of Technology

NICOLE B. MOTZER is the director of the Office of Research Development at Montana State University. She is actively engaged in the National Organization of Research Development Professionals and at the forefront

APPENDIX A 197

of cutting-edge approaches to growing university research enterprises. Prior to that, she served as assistant director for interdisciplinary science for the National Socio-Environmental Synthesis Center, Motzer's expertise thus lies at the intersection of research development, interdisciplinarity, and team science. She has mentored hundreds of scholars and dozens of research teams through team science best practices, led several first-of-their-kind team science training and funding opportunities, and regularly coached research groups across the country and even the globe. In addition to a team science facilitator and trainer, Motzer is an interdisciplinary collaborator in topics ranging from interdisciplinary evaluation to marine spatial planning to grassland health. She is co-primary investigator (co-PI) on a National Aeronautics and Space Administration's Land Cover Land Use Change Award and on a National Science Foundation (NSF) GRANTED Award, and she served as PI of a global effort funded by the Gordon and Betty Moore Foundation. Motzer was named a visiting fellow for Michigan State University's Center for Interdisciplinarity for her practical contributions to interdisciplinary science as well as her work on interdisciplinary and research evaluation. She completed her PhD in geographical sciences at the University of Maryland, College Park as an NSF graduate research fellow. Motzer participated in and reviewed the proceedings for the National Academies of Sciences, Engineering, and Medicine's "Measuring Convergence in Science and Engineering" workshop.

MARK T. PETERS is the president and chief executive officer of MITRE, the global not-for-profit technology company that operates research and development centers worldwide. Previously, he was the executive vice president for National Laboratory Management and Operations at Battelle Memorial Institute with responsibilities for governance and oversight of the U.S. Department of Energy (DOE) and U.S. Department of Homeland Security national laboratories for which Battelle has a significant lab management role. Peters was the director of Idaho National Laboratory and president of Battelle Energy Alliance, LLC. He was responsible for the management and integration of a large, multipurpose laboratory whose mission focuses on nuclear energy, national and homeland security, and energy and environmental science and technology. Peters served 2 years as chairman of the National Laboratory Directors' Council, an independent body that coordinates initiatives and advises the DOE and other national laboratory affiliates. Prior to joining Battelle, he served as the associate laboratory director for Energy and Global Security at Argonne National Laboratory. Peters serves as a senior adviser on nuclear energy technologies, research and development programs, and nuclear waste policy. As an expert in nuclear fuel cycle technologies and nuclear waste management, he is often called upon to provide expert testimony to Congress and to advise in formulation of policies for nuclear fuel cycles, nonproliferation, and nuclear waste disposal. Peters was honored as a fellow of the American Nuclear Society (ANS) for outstanding accomplishments in nuclear science and technology. He served on the ANS Public Policy Committee, the executive committee of the ANS Fuel Cycle and Waste Management Division. Peters serves on several boards and advisory committees, including the Idaho Power Board. He has received extensive management and leadership education and training, including completion of the Strategic Laboratory Leadership Program at the University of Chicago Booth School of Business. Peters received a BS in geology from Auburn University and a PhD in geophysical sciences from the University of Chicago. He is an elected member of the National Academy of Engineering.

SUSAN J. SIMKINS (formerly Susan Mohammed) is a professor of industrial-organizational psychology at The Pennsylvania State University, where she leads the Teams, Cognition, and Time lab. As the director of team science for the Penn State Clinical and Translational Science Institute, she merges science and practice by educating and consulting with team leaders and members on improving their team processes and outcomes. For the past three decades, Simkins has investigated the drivers of effective teamwork and performance. Her research focuses on team cognition, team composition/diversity, and the role of time in team and leadership research, and her most recent work is applying these research topics to humanrobot teaming. The National Science Foundation, the National Institutes of Health, and the Office for Naval Research fund Simkins' research, and she is a fellow of the Association of Psychological Science and the Society for Industrial and Organizational Psychology. She serves as an associate editor for the Academy of Management Annals and the Journal of Business and Psychology. Simkins received her PhD from The Ohio State University.

KELLY D. TAYLOR is an assistant professor of medicine in the Division of Prevention Science and director of the Center for Pandemic Preparedness and Response in the Institute for Global Health Sciences at the University of California San Francisco. Her research aims to both train leaders in health equity science and build transformative public health solutions with community partners to advance equity and saves lives by centering the perspective of those individuals and communities who are experiencing risks for poor health outcomes and empowering both them and their care teams to optimally mitigate that risk. As a part of that work, Taylor leads Combating Unequal Treatment in Healthcare Through Virtual Awareness and Training in Empathy, which is a team science developed study that explores decreasing bias in health care experienced by Black, Indigenous, and People of Color using patient-provider virtual reality simulations. She was invited to speak at the 2023 Congressional Black Caucus's Legislative Conference

APPENDIX A 199

on Unmasking Bias in Medicine. Taylor was awarded National Institutes of Health pre-doctoral and postdoctoral traineeships to study health services research and AIDS prevention studies respectively. She holds degrees from the University of Michigan, Vanderbilt University, and the University of California, Berkeley.

CORY VALENTE is the senior human resources director of talent acquisition and the corporate director for the office of inclusion at the Dow Chemical Company. He is responsible for developing and implementing strategies and best practices to advance an inclusive workplace culture and to position Dow to attract, recruit, and retain a skilled and engaged workforce to deliver against the company's strategic priorities. Previously, Valente served on a business leadership team as a research and development (R&D)/technical service and development director responsible for developing and implementing an innovation strategy to fuel business growth through more sustainable processes and products. Since joining Dow, his career has been at the intersection of innovation and workplace culture, serving in a diverse range of roles of increasing responsibility in R&D, human resources, and the office of inclusion. Valente is the author of more than 25 peer-reviewed publications in the chemical sciences. He holds a PhD in organic chemistry from York University and completed a postdoctoral fellowship with 2016 Nobel Laureate in Chemistry Sir Fraser Stoddart at Northwestern University.

CHRISTOPHER W. WIESE serves as an assistant professor in the School of Psychology and the area director of the Industrial-Organizational Psychology Program. He is also a prominent board member of the International Network for the Science of Team Science and the Team Science Lead for the Georgia Clinical and Translational Science Awards Program. Previously, Wiese held various roles that have honed his expertise in organizational psychology and team dynamics. An expert in team science, he specializes in studying team resilience, learning, and the temporal dynamics of team phenomena. Wiese has conducted numerous workshops and consulting sessions on team composition, conflict management, team effectiveness, and leadership. His innovative Team Science Boot Camp is a testament to his commitment to advancing team science methodologies, and his distinguished work has been sponsored by prestigious institutions such as the National Science Foundation, National Institutes of Health, National Aeronautics and Space Administration, and the Air Force Research Laboratory, highlighting his significant contributions to the field. Wiese received his PhD in industrialorganizational psychology from the University of Central Florida.



## Appendix B

# Team Science Background

# TEAM SCIENCE AND CROSS-DISCIPLINARY COLLABORATIONS: DEFINITION AND EVOLUTION

The scholarly study of science has a long tradition with a set of foundational fields going back decades. In this section, the committee briefly describes how science has been studied as a scholarly area of inquiry. The committee then transitions to one of the newer fields of study science, the science of team science. In this context, the committee provides important definitions to help scaffold our discussion and briefly describe the differing forms of cross-disciplinary collaborations that can occur in team science.

## Studying Science and Scientists

One of the earliest fields to study science is "the history and philosophy of science" (Pinnick & Gale, 2000). This field has a long tradition of scholarly work examining science and medicine through a historical lens. Taking a broad perspective, it examines how the understanding of the natural world has changed over the centuries and how scientific innovation affects society culturally, economically, and politically. Similar, but more narrowly focused, is the field of "social studies of science" (e.g., Nola, 2013; Woolgar, 1991) and the closely aligned field, "science and technology studies" (Hackett et al., 2008). Both examine how science affects society with an emphasis on the social dimensions of science and the role and ethical implications of science and technology.

Taking a more quantitative approach, the field of "scientometrics" was born out of a need to better measure and analyze science, technology, and innovation. From its origins, it relied heavily on bibliometrics and citation analyses to understand scientific impact and map scientific fields (de Solla Price, 1981; Garfield, 2009; Lane, 2009; Leydesdorff et al., 2018). Out of this came a more practice-oriented study of science, called "science of science policy" (Fealing, 2011). Through a blend of quantitative data and qualitative information, the science of science policy seeks to develop a more quantitatively informed basis for science policy and works to develop models that can help guide investments in science. These have evolved into a related area of study known as the "science of science." This field is a blend of scientometrics, network science, and big data analytics. The goal is both to improve the understanding of how knowledge is produced and disseminated and to make recommendations for improving the scientific ecosystem (Fortunato et al., 2018).

The aforementioned areas of study all examine different concepts, processes, and outcomes associated with science. Some fields study science to add to our understanding while others also seek out ways to improve the scientific ecosystem. To varying degrees, each field has also sometimes studied collaboration in science but rarely make it a focal issue of inquiry. Because of this, the science of team science (Hall et al., 2008) was specifically developed as a scholarly examination of teamwork in science. The goal of the science of team science is to improve the understanding of how scientists interact as members of a team, and how their collaboration helps build and integrate knowledge across disciplinary, professional, and institutional boundaries (Stokols et al., 2008). From this improved understanding, the science of team science aims to help science teams make full use of their intellectual capacity (Salazar et al., 2012). The science of team science fundamentally reframes science collaboration as a process of teamwork to be mastered (Fiore, 2008). In contrast to the other studies of science, teams are the focal area of study in the science of team science to improve our fundamental understanding of the collaborative production of knowledge and to develop new methods and models to improve science teamwork (National Academy of Sciences, 2005).

Some recent studies of collaboration in team science focus on large-scale projects across multiple teams and multiple universities and countries. One of these is "big team science" (Baumgartner et al., 2023; Coles et al., 2023), which is a recent community of researchers conducting complex collaborations where teams of scientists representing dozens of labs around the world are collaborating on related projects. These initiatives were undertaken to increase the robustness of research in what has traditionally been small sample studies, and big team science has made important progress in bringing teams together to study a variety of scientific issues. To accomplish

APPENDIX B 203

this, big team science recruits dozens of laboratories from around the world, asking each to run identical studies to increase samples to an order of magnitude larger than is typically found (e.g., social science experiments might have several dozen participants, but big team science studies can have several thousand; Coles et al., 2022). However, in doing so, they face both longstanding and novel challenges when it comes to coordinating collaboration (see Coles et al., 2023). Primarily out of Australia, the scholarly study of a particular form of translational science has emerged, referred to as integration and implementation sciences at Integration and Implementation Insights (i2Insights). 1 i2Insights, or I2S, is both a community of practice as well as a clearinghouse for material developed to improve collaboration in complex scientific areas. Focusing on methods, frameworks, and concepts for addressing complex societal and environmental issues, i2Insights seeks to publish and share a variety of research developments in areas such as climate change while attending to varied interdisciplinary approaches (e.g., systems thinking, action research, sustainability science). i2S is a form of convergence research that brings together differing approaches for studying and improving significant societal and environmental problems (NRC, 2014).

Developed primarily by the Swiss Academies of Arts and Sciences, the Network for Transdisciplinary Research (td-net<sup>2,3</sup>) is a repository of resources that are designed to support collaborative problem solving when team members come from different fields.

#### Definitional Distinctions in Team Science

To discuss these complex forms of collaboration, the committee finds it necessary to define several key terms regarding teamwork in science. First, the committee defines *team science* as collaborative, interdependent research conducted by more than one individual. Most shared publications were published by science teams of approximately two to ten members (Wuchty et al., 2007). Second, the committee also recognizes that there are often larger groups of more than ten scientists collaborating on complex problems (e.g., astrophysics), and publications produced by hundreds to even thousands of co-authors are becoming more prevalent (Nogrady, 2023). Nonetheless, these larger groups often experience the same challenges when interacting. Last, when trying to address what counts as

<sup>&</sup>lt;sup>1</sup> More information about Integration and Implementation Insights can be found at https://i2insights.org/about

<sup>&</sup>lt;sup>2</sup> More information about the Network for Transdisciplinary Research is available at https://transdisciplinarity.ch/

<sup>&</sup>lt;sup>3</sup> The td-net toolbox is available at https://naturalsciences.ch/co-producing-knowledge-explained/methods/td-net toolbox

success for these teams, the 2015 National Academies committee simply considered effectiveness to mean achieving team goals and objectives (National Research Council, 2015). However, the 2015 committee also acknowledged that there are multiple collaborative processes that can be assessed to understand what leads to effectiveness (National Research Council, 2015).

Team science is an increasingly critical area of inquiry given that modern science, more and more, embraces research ideas that cross disciplines. Therefore, team science varies in form, since disciplinary differences and integration can vary. Specifically, team science can vary in the degree in which members are integrating concepts, techniques, and theories from different fields. To provide additional definitional grounding, the committee chose to briefly describe how these collaborations can vary. While there are differing views in the literature and varying approaches internationally, our aim was to offer definitions that encompass the key elements found in the approaches that cross disciplines. For further exploration, refer to the works of Klein (2010), Wagner et al. (2011), Hall et al. (2012), National Academies (National Academy of Sciences et al., 2005), National Research Council (NRC; 2014, 2015), and Stokols et al. (2008).

The first type of cross-disciplinary research is multidisciplinarity. Multidisciplinarity involves collaboration among multiple disciplines toward a shared objective. These types of studies aim to facilitate a comprehensive analysis of the research issue. In multidisciplinary research, scientists may work on their own, in parallel, or sequentially. In doing so, they may convene periodically to exchange what they have accomplished and discovered. While each discipline's contributions complement each other, their methods, concepts, and theories are typically not integrated. Additionally, individual scientists remain ensconced in their own forms of disciplinary perspectives.

The next form of cross-disciplinary research is interdisciplinarity. Unlike the mere complementarity found in multidisciplinary research, this approach requires scientists to blend or juxtapose concepts and methods from various disciplines. The main goal is the systematic integration of information, data, techniques, tools, perspectives, concepts, and/or theories from multiple disciplines or bodies of specialized knowledge (National Academy of Sciences et al., 2005). The objective of interdisciplinarity is to provide a more integrated and fundamental understanding of the subject matter by addressing problems that exceed the boundaries of any one discipline.

A more deeply integrated research approach is transdisciplinarity. What distinguishes this approach is its advancement of discipline-specific theories, concepts, and methods (Hall et al., 2008). Moreover, it often requires considering problems across various levels of analysis (e.g., individual, group, community). In doing so, transdisciplinarity often involves "translational" partners from diverse sectors of society (e.g., nongovernmental

APPENDIX B 205

organizations, community, industry) in the research process. These translational partners typically include individuals outside science, such as community members, to provide real-world insights while also improving the likelihood of translation. Overall, transdisciplinary research aims to cultivate a holistic understanding of the examined problem (Hadorn et al., 2010). Because of its deeply integrated nature, transdisciplinary research primarily concentrates on societal issues and the generation of actionable knowledge (Brandt et al., 2013). When defining their research process, transdisciplinary teams focus on problem identification, structuring, and analysis (Pohl & Hadorn, 2008). More specifically, transdisciplinary research becomes essential when there is uncertainty surrounding scientific knowledge about a significant problem domain, when the specific characteristics of problems are unclear or open to interpretation, and when there are significant stakes for those affected by these problems. In these types of situations, transdisciplinary research is able to address problem domains in a manner that enables better comprehension of the intricacies of the problem and deeper consideration of different perspectives on problems. Transdisciplinary teams are able to consider social and scientific perspectives while working to establish connections between theoretical and contextspecific knowledge to build new knowledge and practices.

### KEY TAKEAWAYS FROM THE REPORT: ENHANCING THE EFFECTIVENESS OF TEAM SCIENCE

From 2013 to 2015, the National Academies' Committee on the Science of Team Science worked on a report to offer research-based guidance aimed at enhancing the processes and outcomes of collaboration in science (NRC, 2015). This project was supported by the National Science Foundation (NSF) and by Elsevier. The overarching objective was to improve the effectiveness of collaboration within science teams, research centers, and institutes. The audience for the report ranged from the NSF and other public and private sources of research funding to the broader scientific community, research centers, institutes, and universities.

This committee's charge was to conduct a consensus study to recommend opportunities for enhancing the effectiveness of collaborative research and explore what factors impact team dynamics, effectiveness, and productivity. They were to investigate these factors at the team, center, or institute level to understand how they influence science team effectiveness. The committee was tasked with exploring different management approaches and leadership styles that influence effectiveness. They were also tasked to examine how tenure and promotion policies help or hinder academic researchers who participate in research teams. Finally, the committee was charged with considering the organizational factors (e.g., human resource

policies, cyberinfrastructure) that might influence the effectiveness of science teams along with organizational structures, policies, and practices aimed at promoting effective teams.

At the outset, the 2015 National Academies committee identified a set of key features that pose challenges for science teams (NRC, 2015). The committee found that a high variety of membership within a team occasionally led to varying perspectives and approaches that could complicate collaboration. Relatedly, the committee discovered that there was a need for deep knowledge integration arising from the goal of merging disparate expertise and disciplinary backgrounds. Large science teams often face difficulties due to their size, making coordination and communication more complex. Misalignment of goals within and across science teams can hinder collaboration and cohesion within the broader research environment. Some science teams experience permeable boundaries that can result in ambiguity regarding roles and responsibilities or obscure the extent to which individuals should be contributing to the differing projects. Geographic dispersion was found to add another layer of complexity, requiring effective virtual communication and coordination mechanisms. Furthermore, high task interdependence necessitates close coordination and cooperation among team members to achieve shared objectives.

To address these challenges, the 2015 National Academies committee considered the broad body of literature and consulted experts from the social sciences to understand findings on teams and organizations. They concluded that there was a robust body of research going back decades showing how team processes related to team effectiveness. Furthermore, the committee identified a set of interventions that support teamwork and offer the most promising route to enhance team effectiveness. These included team composition, development, and leadership.

Regarding team composition, the committee found that research conducted in nonscience contexts indicated that the makeup of a team significantly impacts its effectiveness (NRC, 2015). This relationship depended on several factors, including task complexity, level of interdependence among team members, and duration the team was together. Task-relevant variety was an important factor affecting team effectiveness as it influenced membership. The committee noted that leveraging analytic methods developed in nonscience contexts, along with researcher networking tools established within scientific domains, could enable practitioners to more systematically attend to team composition. The committee suggested that scientists could use these tools to optimize team compositions to enhance overall effectiveness, while also considering task complexities, interdependencies, and the value of diversity (NRC, 2015).

Out of the above conclusions, the report recommended the use of task analytic methods and tools. These methods and tools assist in identifying APPENDIX B 207

requisite knowledge, skills, and attitudes, and they can help ensure that task-related differences align with team member experience and project requirements (NRC, 2015). Additionally, the committee suggested considering the application of tools like research networking systems, which were specifically designed to help with the assembly of science teams. Finally, they suggested future collaborations to help assess these methods to ensure their effectiveness and relevance in team science as well as provided guidance on how to improve them (NRC, 2015).

In the context of professional development for science teams, the committee noted that research outside of science has demonstrated that several types of interventions can improve team processes and outcomes. Recognizing this, the committee recommended that "team-training researchers, universities, and science team leaders should partner to translate, extend, and evaluate the promising training strategies shown to improve the effectiveness of teams in other contexts" (NRC, 2015, p. 8). Related, when considering educating scientists to work on teams, the committee found that "colleges and universities are developing cross-disciplinary programs designed to prepare students for team science" (NRC, 2015, p. 9). However, at that time, there had been scant empirical research on the extent to which students learn the targeted competencies and on whether their acquisition contributes to team science effectiveness.

Similarly, when considering leadership and team science, the committee found that decades of "research on team and organizational leadership in contexts other than science provide a robust foundation of evidence to guide professional development for leaders of science teams" (NRC, 2015, p. 9). From this, the committee recommended that "researchers, universities, and leaders of team science projects should partner to translate and extend the leadership literature to create and evaluate science leadership development opportunities" (NRC, 2015, p. 9).

Regarding research universities, the 2015 committee noted efforts to initiate and foster interdisciplinary team science, as seen by the establishment of research centers and institutes. However, the impact of these endeavors on the quantity and quality of team science remains largely unassessed. Relatedly, when considering reward structures within research universities, the committee concluded that promotion and tenure review policies typically lack "comprehensive, clearly articulated criteria for evaluating individual contributions to team-based research" (NRC, 2015, p. 11). Additionally, if criteria do exist, the recognition and rewards for team-based research vary significantly both within and across universities, potentially leading to disparities in incentives for participation (NRC, 2015). These inconsistencies may deter young faculty from pursuing team science in environments where such collaborations are undervalued. To address these issues, the committee recommended that universities and scholarly associations work to "develop"

and evaluate broad principles and more specific criteria for allocating credit to team-based work" (NRC, 2015, p. 11), so that promotion and tenure committees more accurately assess candidates and foster an environment conducive to team science.

The 2015 committee also noted that public and private funding agencies can play a pivotal role in cultivating a culture within the scientific community that promotes and facilitates team science. In addition to providing financial support, these agencies have the capacity to influence practices through the development of reports espousing the value and importance of team science, as well as providing training workshops. To further advance this culture of collaboration, the committee recommended that funders collaborate with the scientific community on several fronts (NRC, 2015). First, they "should encourage the development and implementation of new collaborative models, such as research networks and consortia" to facilitate interdisciplinary research efforts (NRC, 2015, p. 11). Second, funders can incentivize team science by helping to develop promotion and tenure policies that recognize and reward collaborative contributions. Third, they can allocate resources to support team science initiatives, such as establishing information repositories and offering training modules to enhance collaboration competencies. By actively engaging with the scientific community and implementing these recommendations, funding agencies can effectively promote and support the growth of team science.

The 2015 committee also considered additional ways funders can support the science of team science. They concluded that funding agencies generally lack consistency in how they evaluate scientific merit with collaborative merit, particularly concerning how teams execute the work. Often, agency announcements seeking team science proposals lack clarity regarding the expected level of collaboration and knowledge integration. To address these issues, the committee recommended funders mandate proposals for team-based research to include detailed collaboration plans (NRC, 2015). Additionally, funders can provide guidance to scientists on incorporating these plans into their proposals, along with criteria for reviewers to evaluate them effectively. Furthermore, authors of proposals for interdisciplinary or transdisciplinary research projects can be required to articulate how they will integrate differing disciplinary perspectives, concepts, theories, and methods. By implementing these approaches, the committee noted that funding agencies can help steward the development and evaluation of team science (NRC, 2015).

The 2015 committee concluded that for progress to be made in the science of team science, more efforts are necessary to assess and enhance the tools, interventions, and policies proposed in the report (NRC, 2015). However, there is a notable absence of funding programs dedicated to investigating the effectiveness of science teams. Considering this, the committee

APPENDIX B 209

concluded that there needs to be support for fundamental research on team science to inform continuous enhancements in its effectiveness (NRC, 2015). Furthermore, the committee pointed to a set of inter-related challenges that complicate research on science teams. These include the multifaceted and sometimes competing goals of team science projects and the multilevel perspective required to study science teams and processes at individual, team, and organizational levels. To address these challenges and promote advancements in team science, the committee recommended that public and private organizations allocate funding to support research on team science effectiveness. Additionally, the committee called for support for ongoing evaluation and refinement of their recommended interventions and policies. The committee recommended future research be conducted not just on science teams but also on the role of scientific organizations, such as research centers, networks, and consortia in bolstering science teams and larger groups (NRC, 2015). Finally, the committee noted that collaboration with universities and the scientific community is crucial to facilitate researchers' access to team science personnel.

#### REFERENCES

- Baumgartner, H. A., Alessandroni, N., Byers-Heinlein, K., Frank, M. C., Hamlin, J. K., Soderstrom, M., Voelkel, J. G., Willer, R., Yuen, F., & Coles, N. A. (2023). How to build up big team science: A practical guide for large-scale collaborations. *Royal Society Open Science*, 10(6), 230235. https://doi.org/10.1098/rsos.230235
- Brandt, P., Ernst, A., Gralla, F., Luederitz, C., Lang, D. J., Newig, J., Reinert, F., Abson, D. J., & Von Wehrden, H. (2013). A review of transdisciplinary research in sustainability science. *Ecological Economics*, 92, 1–15. https://doi.org/10.1016/j.ecolecon.2013.04.008
- Coles, N. A., DeBruine, L. M., Azevedo, F., Baumgartner, H. A., & Frank, M. C. (2023). 'Big team' science challenges us to reconsider authorship. *Nature Human Behaviour*, 7(5), 665–667. https://doi.org/10.1038/s41562-023-01572-2
- Coles, N. A., March, D. S., Marmolejo-Ramos, F., Larsen, J. T., Arinze, N. C., Ndukaihe, I. L., Willis, M. L., Foroni, F., Reggev, N., Mokady, A., Forscher, P. S., Hunter, J. F., Kaminski, G., Yüvrük, E., Kapucu, A., Nagy, T., Hajdu, N., Tejada, J., Freitag, R. M. K., Zambrano, D., Som, B., Aczel, B., Barzykowski, K., Adamus, S., . . . Liuzza, M. T. (2023). A multi-lab test of the facial feedback hypothesis by the Many Smiles Collaboration. *Nature Human Behaviour*, 6(12), 1731–1742. https://doi.org/10.1038/s41562-022-01458-9
- de Solla Price, D. (1981). The analysis of scientometric matrices for policy implications. *Scientometrics*, 3, 47–53. https://doi.org/10.1007/BF02021863
- Fealing, K. (2011). The science of science policy: A handbook. Stanford University Press.
- Fiore, S. M. (2008). Interdisciplinarity as teamwork: How the science of teams can inform team science. Small Group Research, 39(3), 251–277. https://doi.org/10.1177/1046496408317797
- Fortunato, S., Bergstrom, C. T., Börner, K., Evans, J. A., Helbing, D., Milojević, S., Petersen, A. M., Radicchi, F., Sinatra, R., Uzzi, B., Vespignani, A., Waltman, D., & Barabási, A. L. (2018). Science of science. *Science*, 359(6379), eaao0185. https://doi.org/10.1126/science. aao0185

- Garfield, E. (2009). From the science of science to Scientometrics visualizing the history of science with *HistCite* software. *Journal of Informetrics*, 3(3), 173–179. https://doi.org/10.1016/j.joi.2009.03.009
- Hackett, E. J., Amsterdamska, O., Lynch, M., & Wajcman, J. (Eds.). (2008). The handbook of science and technology studies. MIT Press.
- Hadorn, G. H., Pohl, C., & Bammer, G. (2010). Solving problems through transdisciplinary research. In R. Frodeman, J. T. Klein, & C. Mitcham (Eds.), The Oxford Handbook of Interdisciplinarity (pp. 431–452). Oxford University Press.
- Hall, K. L., Feng, A. X., Moser, R. P., Stokols, D., & Taylor, B. K. (2008). Moving the science of team science forward: Collaboration and creativity. *American Journal of Preventive Medicine*, 35(2), S243–S249. https://doi.org/10.1016/j.amepre.2008.05.007
- Hall, K. L., Vogel, A. L., Stipelman, B. A., Stokols, D., Morgan, G., & Gehlert, S. (2012). A four-phase model of transdisciplinary team-based research: Goals, team processes, and strategies. *Translational Behavioral Medicine*, 2(4), 415–430. https://doi.org/10.1007/s13142-012-0167-y
- Klein, J. T. (2010). A taxonomy of interdisciplinarity. In R. Frodeman, J. T. Klein, & C. Mitcham (Eds.), *The Oxford Handbook of Interdisciplinarity* (pp. 15–30). Oxford University Press.
- Lane, J. (2009). Assessing the impact of science funding. *Science*, 324(5932), 1273–1275. https://doi.org/10.1126/science.1175335
- Leydesdorff, L., Wagner, C. S., & Bornmann, L. (2018). Betweenness and diversity in journal citation networks as measures of interdisciplinarity—A tribute to Eugene Garfield. *Scientometrics*, 114, 567–592. https://doi.org/10.1007/s11192-017-2528-2
- National Academy of Sciences, National Academy of Engineering, & Institute of Medicine. (2005). *Facilitating interdisciplinary research*. The National Academies Press. https://doi.org/10.17226/11153
- National Research Council (NRC). (2014). Convergence: Facilitating transdisciplinary integration of life sciences, physical sciences, engineering, and beyond. The National Academies Press. https://doi.org/10.17226/18722
- \_\_\_. (2015). Enhancing the effectiveness of team science. The National Academies Press. https://doi.org/10.17226/19007
- Nogrady, B. (2023). Hyperauthorship: The publishing challenges for 'big team' science. *Nature*, 615(7950), 175–177. https://doi.org/10.1038/d41586-023-00575-3
- Nola, R. (2013). Social studies of science. In The Routledge companion to philosophy of science (pp. 291–300). Routledge.
- Pinnick, C., & Gale, G. (2000). Philosophy of science and history of science: A troubling interaction. *Journal for General Philosophy of Science*, 31(1), 109–125. https://doi.org/10.1023/A:1008353021407
- Pohl, C., & Hadorn, G. H. (2008). Core terms in transdisciplinary research. In G. H. Hadorn, H. Hoffmann-Riem, S. Biber-Klemm, W. Grossenbacher-Mansuy, D. Joye, C. Pohl, U. Wiesmann, & E. Zemp (Eds.), *Handbook of transdisciplinary research* (pp. 427–432). Springer, Dordrecht. https://doi.org/10.1007/978-1-4020-6699-3\_28
- Salazar, M. R., Lant, T. K., Fiore, S. M., & Salas, E. (2012). Facilitating innovation in diverse science teams through integrative capacity. *Small Group Research*, 43(5), 527–558. https://doi.org/10.1177/1046496412453622
- Stokols, D., Hall, K. L., Taylor, B. K., & Moser, R. P. (2008). The science of team science: Overview of the field and introduction to the supplement. *American Journal of Preventive Medicine*, 35(2), S77–S89. https://doi.org/10.1016/j.amepre.2008.05.002

APPENDIX B 211

Wagner, C. S., Roessner, J. D., Bobb, K., Klein, J. T., Boyack, K. W., Keyton, J., Rafols, I., & Börner, K. (2011). Approaches to understanding and measuring interdisciplinary scientific research (IDR): A review of the literature. *Journal of Informetrics*, 5(1), 14–26. https://doi.org/10.1016/j.joi.2010.06.004

- Woolgar, S. (1991). The turn to technology in social studies of science. Science, Technology, & Human Values, 16(1), 20–50. https://doi.org/10.1177/016224399101600102
- Wuchty, S., Jones, B. F., & Uzzi, B. (2007). The increasing dominance of teams in production of knowledge. *Science*, 316(5827), 1036–1039. https://doi.org/10.1126/science.1136099



## Appendix C

## **High-Functioning Science Teams**

What does a high-functioning, effective science team look like? According to Hackman (1987), team effectiveness is not just the product of collaboration; it is also a function of the social processes within a group and the personal satisfaction of its members. While the success of science teams is often measured through metrics such as research impact, successful grant applications, accepted journal publications, filed patents, and community connections, these outcomes alone do not reveal the processes that teams undergo to achieve these goals. In other words, they do not capture how the team functions on a day-to-day basis, such as team member behavior or their psychological states. It is crucial to understand both how teams function and what indicates that they are functioning well. For instance, if one attempts to improve a team based solely on outcome-based metrics, the only guidance they can provide is to enhance these metrics (e.g., secure more grants, file more patents).

Building on the foundation laid by the 2015 report *Enhancing the Effectiveness of Team Science* (National Research Council [NRC], 2015), this appendix aims to equip readers with an understanding of the mechanisms that underpin successful team science by illuminating the processes that contribute to the evolution of high-functioning science teams. First, the text explores how teams operate by offering two perspectives: a systems perspective and a temporal perspective. These perspectives will give the reader an understanding that teams neither operate in a vacuum, nor do they remain static over time. Subsequently, this appendix will highlight key indicators of successful scientific teamwork, providing a practical guide for identifying the components of effective scientific collaboration.

Before presenting these perspectives and indicators, it is necessary to first establish a shared understanding of terminology. Formally, a *science team* consists of two or more scientists working collaboratively toward a shared goal (NRC, 2015). In practice, science teams often include a diverse array of members, such as principal and co-investigators, research associates, postdoctoral fellows, graduate and undergraduate students, financial and administrative staff, and community members. The roles within science teams are often functionally complementary, contributing both directly (e.g., idea generation, experiment execution) and indirectly (e.g., processing annual reports, hiring postdoctoral fellows, purchasing research equipment) to the generation of new scientific knowledge.

Larger science teams may constitute a scientific "multiteam system"—in other words, interdependent networks of multiple component teams working toward broader objectives while maintaining their respective goals and responsibilities (Carter et al., 2019; Zaccaro et al., 2020). As an organizational form, the multiteam system structure is becoming increasingly prevalent, particularly in scientific endeavors, as it allows for exploring and investigating more complex challenges. Thus, science teams can be viewed as existing within a system composed of individuals, teams, multiteam systems, and institutes. This recognition of teams existing in a system is the first perspective to be discussed.

Importantly, teams are not static entities; they change and evolve over time. Although a team is technically formed once individuals are assigned to it, teams need time to develop and integrate psychologically before members fully consider themselves part of a cohesive unit. Therefore, it is crucial to understand that time is a key factor in how teams function (Cronin et al., 2011; Mohammed et al., 2008). Time not only influences the way teams operate but also provides valuable insights that can inform team science practices. For example, measuring trust immediately after a team is formed would be premature, as trust between team members develops over time (Grossman & Feitosa, 2018). Assessing a team too early would not yield an accurate understanding of trust within the team. Additionally, there are specific stages in a team's evolution where certain behaviors become particularly important. For instance, while communication is critical for success throughout the team's life cycle, clarifying roles and responsibilities is especially vital during the initial formation of the team (Kozlowski & Bell, 2007); see Chapter 3 in this report for a discussion of relevant best practices. Understanding these temporal dynamics is essential for fostering effective teamwork.

Much like the systems perspective, there are many ways in which time can be incorporated into understanding team functioning. One could take a team evolution or team development approach, which examines the stages or phases that teams go through as they get to know each other

and work toward achieving their goals. For instance, Tuckman's five-stage model (Tuckman, 1965) outlines the stages of forming, storming, norming, performing, and adjourning, where teams move from initial formation and conflict to effective collaboration and eventual disbandment. This model highlights how team dynamics evolve over time, with each stage building on the previous one. Another classic model is the punctuated equilibrium model (Gersick, 1988, 1989), which suggests that teams experience periods of stability punctuated by significant shifts in behavior, often occurring midway through a project. These shifts lead to a more focused and productive second phase of work, emphasizing the impact of critical moments in team development. A last example is Kozlowski & Bell's (2007) process model of team compilation, which not only describes the stages teams go through (team formation, team compilation, role compilation, team compilation) but also where these processes occur (individual level, dyadic level, team level).

Performance episodes offer another way to apply a temporal lens to team dynamics. These episodes refer to distinct periods during which a team works toward specific objectives, often engaging in multiple episodes simultaneously, each with a different focus and duration (Marks et al., 2001; Weingart, 1997; Zaheer et al., 1999). These episodes can range from short-term cycles lasting a few hours or days to long-term cycles extending over months or even years. Short-term episodes might involve daily operations, problem-solving sessions, or completing specific sections of a project, while long-term episodes could encompass strategic planning, the development of complex initiatives, or the overall progression of a major project. The input-mediator-output-input (IMOI) model (Ilgen et al., 2005) is often applied to understanding performance episodes. This model describes team development as a cyclical process where inputs (e.g., team composition, resources) influence mediators (e.g., team processes, emergent states) that, in turn, affect outputs (e.g., team performance, member satisfaction). These outputs then become inputs for the next cycle, influencing subsequent team interactions and development. The IMOI model highlights the dynamic and iterative nature of team functioning, emphasizing that teams continually evolve through feedback loops and ongoing interactions.

In summary, understanding what makes a science team high functioning requires more than just measuring their outcomes. By exploring both systems and temporal perspectives, we can gain insights into what makes these teams effective and how effective teams operate. Whether building, developing, or evaluating science teams, understanding the context in which they exist (i.e., a systems approach) and how they evolve over time (i.e., a temporal approach) is crucial for achieving successful outcomes. In the following section, the systems approach will serve as the organizing framework, focusing on the key constructs and competencies at the individual,

team, and multiteam system levels. Critically, as noted in Chapter 2, mid-to large-scale research on science teams has been limited in recent years. Therefore, this section draws on both small-scale research from the science of team science to more extensive studies from organizational sciences to highlight the most important constructs and competencies of high-functioning science teams.

#### FACTORS OF SUCCESSFUL SCIENCE TEAMS

It is crucial to acknowledge that a wide array of factors contribute to team success across various contexts. Extensive reviews and meta-analyses within the organizational sciences have identified numerous individual, team-, and multiteam-level factors that predict team effectiveness. The discussion that follows will concentrate on the team- and multiteam-level factors most pertinent to the success of science teams. Discussion of individual-level factors can be found in Chapter 3.

#### Science Team-Level Factors

The effectiveness of a team is not solely determined by the individual characteristics of its members but also by the dynamics that occur at the team level. Two critical categories of team-level factors that significantly influence team success are team processes and team emergent states. Team processes refer to the specific behaviors and interactions that teams engage in as they work together, such as communication, coordination, and conflict (Mathieu et al., 2008). These processes are essential for ensuring that team members collaborate effectively and efficiently toward achieving their goals. On the other hand, team emergent states are conditions that develop over time within the team, such as trust, psychological safety, cohesion, and shared mental models (Mathieu et al., 2008). These emergent states are crucial for fostering an environment where team members feel safe to share ideas, take risks, and support one another. The following discussion will focus on the team processes and emergent states that are most vital to the success of science teams.

## Team Learning Behaviors

Another important team behavioral process is team learning behaviors. Team learning behaviors refer to collaborative actions that contribute to either altering or reinforcing the team's shared understanding (Wiese & Burke, 2019) and are critically important to the success of science teams. This is particularly true in scientific contexts, where teams constantly integrate new information, adapt to evolving research findings, and synthesize

different expertise. Science teams often operate in highly complex and dynamic environments, requiring them to collectively learn from both successes and failures to remain innovative and effective. These behaviors foster the team's ability to process and apply new knowledge, ensuring that the team is continually advancing its understanding and improving its approach to solving problems.

Much like communication, there are various forms of team learning behaviors. These are often categorized by where the learning occurs—either internal to the team, such as within discussions or task processes, or external to the team, such as gathering insights from outside sources. Additionally, team learning behaviors can be classified as exploitative, where the focus is on integrating existing knowledge and refining processes, or explorative, where the emphasis is on generating new ideas or gaining novel insights (Harvey et al., 2022). Regardless of the type, team learning behaviors are significantly related not only to team performance but also to the development of critical emergent states, such as psychological safety. This dynamic creates a positive feedback loop, as learning behaviors strengthen trust and cohesion, which, in turn, enhance the team's ability to continue learning and innovating.

## Leadership

Leadership literature has evolved over the past few decades, indicating a shift from viewing leadership as an inherent trait of a single individual to understanding it as a set of behaviors or styles that can be exhibited by anyone on the team (Pearce & Conger, 2003; Pearce et al., 2008). This shift is especially relevant for science teams, where different expertise and perspectives are essential. In such teams, members with specialized knowledge may need to step up and demonstrate leadership at different points in time, depending on the demands of the project and the challenges faced. The dynamic nature of scientific work requires a flexible leadership approach that leverages the unique strengths of each team member.

While there are many different theories on leadership, the concept of shared leadership best captures the view of leadership as something that can be demonstrated by any team member. Shared leadership happens over time, and leadership roles are spread throughout the team. This concept can be understood through five key dimensions (D'Innocenzo et al., 2016). The first is the locus of leadership, which distinguishes between leadership that originates from outside the team (external) and leadership that comes from within the team (internal). The second dimension is the formality of leadership, which differentiates between leadership exercised by someone in a formal leadership role (e.g., a designated team leader) and informal leadership, where influence is exerted by individuals not explicitly tasked with leading.

A third dimension involves the distribution of leadership, which refers to how equally leadership roles are shared within the team. Some individuals view leadership as a collective process, where leadership responsibilities are shared equally by all members, while others see leadership as being distributed among members in varying degrees, with different individuals influencing the team in unique ways. Another important aspect of shared leadership is its temporal nature. Leadership within teams is not static; it shifts over time, with different members assuming leadership roles depending on the team's evolving needs and challenges. Finally, it is essential to recognize that leadership is not confined to a single set of behaviors. Leadership can range from traditional behaviors, such as decision-making and direction setting, to more task-oriented actions, such as organizing resources, facilitating collaboration, or providing support that helps the team achieve its goals. As mentioned earlier, the concept of shared leadership is particularly well suited for science teams, which typically consist of a group of experts with different expertise who contribute to the team's goals over an extended period of time. This variety creates opportunities for individuals to demonstrate their expertise and take on leadership roles at different stages of the team's life cycle. Shared leadership also has a demonstrated relationship with team performance.

#### **Faultlines**

One potential issue that can arise within science teams is the development of faultlines. Faultlines are divisions within a team that arise from differences between members, such as differences in expertise, discipline, or background (Jehn & Bezrukova, 2010; Thatcher & Patel, 2012; Thatcher et al., 2024). In science teams, faultlines are particularly important to consider due to the high likelihood of variety in expertise and functional roles. Science teams often consist of members from various disciplines, such as biologists, engineers, and social scientists, working together on complex problems. This variety, while essential for innovation and problem-solving, also increases the likelihood that faultlines will emerge, as differences in perspectives, knowledge, and approaches can create perceptual divisions within the team.

The impact of faultlines on team dynamics depends on whether they remain dormant or become activated. Dormant faultlines refer to the potential for division based on team composition, where members may recognize differences but do not yet act on them. In contrast, activated faultlines occur when these divisions become operational and influence team interactions, leading to subgroup formation and reduced cohesion. Faultlines typically become activated when the team faces challenges or stressors, such as conflicting goals, communication breakdowns, or perceived inequality in

contributions, which exacerbate underlying differences and cause these divisions to surface.

Research has shown that activated faultlines are more detrimental to team functioning than dormant faultlines. While dormant faultlines may influence team conflict by creating an underlying tension, activated faultlines have far more significant consequences. They lead to increased conflict, reduced information sharing, lower team satisfaction, and ultimately, diminished team performance (Thatcher et al., 2024). In science teams, where collaboration and integration of different knowledge is critical, activated faultlines can severely hinder the team's ability to achieve its goals. Therefore, managing faultlines effectively is essential for maintaining team cohesion and ensuring the success of scientific collaborations.

## Team Cognition

An essential indicator of a high-functioning science team is team cognition. Team cognition refers to the shared knowledge structures and patterns among team members that enable them to anticipate each other's needs and coordinate their actions efficiently (Mohammed et al., 2021). As team cognition involves the alignment and integration of individual cognitive states, it is classified as an emergent state, meaning that it evolves and strengthens over time through interaction and collaboration. There are two primary types of team cognition: shared mental models and transactive memory systems. Shared mental models refer to the common understandings that team members have about key aspects of their tasks, roles, and the environment in which they are working. Shared mental models enable team members to develop similar expectations, allowing for smooth coordination without the need for explicit communication at every turn (Van den Bossche et al., 2011). Transactive memory systems, on the other hand, refer to the division of knowledge within the team, where individuals not only know their specific areas of expertise but also understand who within the team holds particular knowledge (Mohammed et al., 2021). This system enables members to effectively locate and retrieve information by relying on the specialized knowledge of others.

For science teams, team cognition is particularly important, as these teams often consist of members with different disciplinary backgrounds and specialized expertise. The complexity of scientific work means that team members not only bring their knowledge to the table but also integrate and apply the knowledge of others. Shared mental models can help science teams align on project goals, methodologies, and expected outcomes, ensuring that all members are working toward a unified vision. Meanwhile, transactive memory systems allow team members to efficiently leverage the expertise distributed across the team, enhancing problem-solving and innovation.

However, achieving strong team cognition in science teams can be especially challenging due to the different perspectives and specialized knowledge each member brings. The process of building shared understanding and trust across different disciplines can be time-consuming, but it is critical for the team's overall success. Without effective team cognition, science teams may struggle with miscommunication, fragmented efforts, and missed opportunities for interdisciplinary collaboration. Therefore, fostering the development of shared mental models and transactive memory systems is essential to ensure that science teams can function cohesively and achieve their complex objectives.

### Psychological Safety

Science teams are more likely to thrive when they foster a strong sense of psychological safety. Psychological safety is the collective perception within a team that reflects an interpersonal environment where members feel safe to take risks, share ideas, and engage in open dialogue without fear of negative repercussions (Edmondson & Lei, 2014; Frazier et al., 2017). In teams where psychological safety is high, individuals are more willing to voice concerns, offer unique perspectives, and admit mistakes, knowing that these behaviors will be met with support rather than judgment. This climate of openness and trust is particularly critical for science teams, where innovation, complex problem-solving, and continuous learning are essential.

In teams with high psychological safety, members engage in candid communication and are unafraid to challenge ideas or offer constructive criticism. They are more likely to embrace feedback and engage in productive conflict, which fosters greater learning and collaboration. For science teams where interdisciplinary work often requires the integration of different knowledge sets and expertise, psychological safety allows team members to propose new approaches and solutions without the fear of embarrassment, a condition which is necessary for bringing together people from different thought worlds. In this environment, team members feel empowered to push boundaries and provide their unique perspectives, knowing that their contributions will be valued even if they challenge the status quo.

Research has consistently demonstrated the positive impact of psychological safety on team dynamics and outcomes. Studies have shown that psychological safety enhances team learning behaviors, promotes knowledge sharing, and encourages creative problem-solving (Edmondson & Bransby, 2023). Additionally, psychological safety mitigates potential negative effects of team differences by fostering greater cohesion and collaboration among members with differing perspectives. In the context of science teams, psychological safety serves as the foundation for effective

collaboration, ensuring that teams can fully leverage their collective knowledge to achieve their scientific goals.

The effectiveness of science teams is not determined solely by individual characteristics but also by vital team-level factors. The behaviors these teams engage in (i.e., team processes) and the dynamics that manifest along the way (i.e., emergent states) are indicative of high-functioning science teams. Team processes, such as communication, coordination, and conflict management, represent the behaviors and interactions that enable teams to work together efficiently. These go together with the emergent states, including trust, psychological safety, and shared mental models that evolve over time, creating an environment where team members feel safe to share ideas and collaborate effectively. The processes and emergent states mentioned here are particularly important for science teams given their interdisciplinary nature and the complexity of their goals. There are many other processes and emergent states that contribute to successful teamwork across various domains. Ultimately, fostering both strong team processes and positive emergent states is crucial for ensuring that science teams can collaborate effectively and achieve their collective goals.

### Multiteam System Factors

Much of what has been discussed so far has been well established since the publication of the previous report in 2015. Concepts such as team cognition, psychological safety, and the importance of shared leadership have long been recognized as critical components of high-functioning science teams. These factors contribute to the effective coordination of diverse expertise, the integration of knowledge, and the overall success of teams working toward complex goals. However, one area of research that has advanced significantly since the last report is the study of multiteam systems. Multiteam systems, which involve multiple interdependent teams working together to achieve a shared superordinate goal, have become increasingly relevant in the realm of scientific research (Zaccaro et al., 2020). A complementary development is the exponential growth in the use of multiteam systems within the scientific community, as evidenced by the rise of large-scale initiatives such as Science and Technology Centers (STCs) and Engineering Research Centers (ERCs). These centers are built on the principle of crossdisciplinary collaboration, bringing teams together from various fields and institutions to tackle complex scientific and technological challenges. Given this shift, it is crucial to examine the specific attributes of multiteam systems that contribute to the success of their scientific endeavors. Understanding

<sup>&</sup>lt;sup>1</sup> For more information on STCs and ERCs, see https://www.nsf.gov/od/oia/ia/stc and https://www.nsf.gov/eng/engineering-research-centers

these factors—such as boundary status, component team distance, and the type of superordinate goals—will allow better assessment of what drives high performance in science multiteam systems and how these teams can be supported and structured to maximize their impact on research and innovation.

### Multiteam System Attributes

Much like the previous section, which focused on individual- and team-level factors associated with high-functioning teams, the first aspect of a multiteam system that will be discussed are the key characteristics of the multiteam system itself. These characteristics are critical because they shape how teams within the system interact, coordinate, and ultimately achieve their shared objectives. In the context of science multiteam systems—multiteam systems composed of science teams—understanding these characteristics is important for ascertaining the most critical processes to focus on when developing and evaluating a science multiteam system. The three primary characteristics of multiteam systems that will be discussed are boundary status, component team distance, and superordinate goal type. Each attribute plays a distinct role in influencing the effectiveness of the scientific multiteam system.

Boundary status refers to whether the teams within the multiteam system come from within a single organization (internal boundary status) or across multiple organizations (external boundary status). Internal multiteam systems tend to operate under a shared set of norms, communication protocols, and authority structures, which can facilitate coordination and collaboration. On the other hand, external multiteam systems, composed of teams from different organizations or institutions, often face more challenges in aligning their goals, managing resources, and navigating different organizational cultures. For example, Kotarba et al. (2023) found that in a multi-institutional cross-disciplinary translational team focused on long COVID-19 research, teams from different universities faced challenges in coordinating efforts due to their external boundary status. Each institution had different operational norms and patient populations, which complicated collaboration and goal alignment across the team. For science multiteam systems, which often involve teams from multiple institutions (e.g., universities, research centers), understanding the boundary status is crucial, as it helps to anticipate challenges related to communication, trust-building, and the harmonization of different institutional priorities, all of which are vital for the success of large-scale interdisciplinary projects (Zaccaro et al., 2020).

The second attribute, component team distance (CTD), refers to the geographical, cultural, functional, or disciplinary distance between the teams that make up the multiteam system. In high CTD systems, teams are more

likely to face challenges in coordinating efforts due to physical separation, differences in disciplinary languages, or contrasting cultural norms. For scientific multiteam systems, CTD can be particularly relevant because science teams often work across institutions, regions, and even countries, introducing significant logistical and communication challenges. High CTD can also exacerbate misunderstandings or delays, slowing research progress or hindering collaboration. For example, Ingersoll et al. (2024) demonstrated the complexities of conducting a multisite clinical trial using virtual multiteam systems, where geographically dispersed teams had to navigate communication barriers, time zone differences, and coordination issues to maintain fidelity and progress in data collection. However, understanding CTD can also help teams develop strategies for bridging these divides, such as paying special attention to developing communication norms, creating cross-team coordination roles, or holding regular in-person meetings to strengthen relationships and align goals.

The final attribute to consider is the superordinate goal type, which refers to the overarching goal all teams within the multiteam system are working toward. Superordinate goals in multiteam systems can generally be classified as either intellectual or physical. Intellectual goals involve generating knowledge, such as developing new theories or conducting scientific research, while physical goals may include building a tangible product or implementing a solution. In multiteam systems, it is crucial that teams not only focus on their individual or team-specific goals but also recognize how their contributions align with and support the broader system-level objective. As Carter et al. (2019) emphasize, one of the key features of multiteam systems is the hierarchical structure of goals, where component teams have both local (subordinate) goals and shared (superordinate) goals. Aligning these goals is essential for the system's success, as misalignment can lead to internal tensions, competition, or even conflict between teams, which can undermine the overall performance of the multiteam system. Ensuring that teams remain focused on the broader objective while pursuing their local goals is a balancing act that requires careful management (Shuffler et al., 2015).

## Crucial Multiteam System Indicators

Beyond the structural components of multiteam systems, behavioral processes and emergent states play a critical role in determining their overall effectiveness. Many of these factors have been discussed in the previous section, such as communication, team cognition, and psychological safety, which are essential for team success at the multiteam level (Shuffler & Carter, 2018; Zaccaro et al., 2020). However, these elements take on added complexity at the multiteam system level, where coordination and

collaboration occur not just within teams but also across multiple teams. While many of the core team-level components remain relevant, there are unique considerations specific to multiteam systems, especially within the scientific context. In the following section, three critical factors that are indicative of science multiteam system effectiveness are highlighted: boundary spanners, who facilitate cross-team communication and bridge organizational divides; inter-team coordination, which ensures the alignment and integration of outputs across teams; and balancing countervailing forces, which helps maintain system cohesion while allowing individual teams to achieve their specific objectives.

A crucial component of high-functioning, effective science multiteam systems is boundary spanners. Boundary spanners are individuals who facilitate communication and coordination across the boundaries between different teams, especially when those teams come from different disciplines, organizations, or geographic locations. They are particularly important in complex multiteam system structures, such as those with high CTD or external boundary status, where teams face greater challenges in aligning goals, managing cultural differences, and coordinating tasks. This can be seen in the challenges and recommendations provided by Kotarba et al. (2023). Boundary spanners help bridge gaps between teams by facilitating information exchange, trust-building, and alignment of objectives, which are crucial for maintaining system-wide cohesion (Carter et al., 2019; Zaccaro et al., 2020). Importantly, boundary spanners operate at both the component team level and the multiteam system level—coming not only from leadership but also from team members who help manage interdependencies across the system.

The boundary spanner role is critical for the success of science multiteam systems because it ensures that teams can collaborate effectively despite differences in expertise, institutional/organizational norms, and goals. Boundary spanners help manage the complex dynamics inherent in interdisciplinary and multi-institutional projects, ensuring that communication and information flow freely and that teams are aligned toward the multiteam system's superordinate goals. For instance, boundary spanners likely play important roles in large-scale, federally funded projects such as STCs and ERCs, where teams from various institutions collaborate to solve complex, multidisciplinary challenges. In these projects, boundary spanners facilitate collaboration between laboratories working on different aspects of the center's overarching goal, such as combining expertise in artificial intelligence with domain-specific knowledge like climate science or engineering. They help ensure that research findings from one team are effectively integrated into the work of other teams and that all teams remain aligned with the overarching center objectives. In science multiteam systems, where research teams from different fields or institutions collaborate to address

complex problems, boundary spanners ensure that these collaborations are productive and not hindered by miscommunication or misalignment.

Relatedly, another component of high-functioning science multiteam systems is their ability to coordinate each component team within their system. Inter-team coordination refers to the processes by which teams manage inter-dependencies and synchronize their activities to achieve both their own goals and the system's superordinate goals (Ziegert et al., 2022). High-functioning multiteam systems require coordination both within and between teams, which may involve shifting focuses over time as projects evolve (Zaccaro et al., 2020). Early on, multiteam systems may prioritize task allocation and communication protocols, but as time progresses, coordination shifts toward aligning outputs and integrating efforts across teams. Ensuring that there is alignment between superordinate and subordinate goals facilitates effective inter-team coordination, as teams can see how their efforts contribute to the broader objectives.

Inter-team coordination is critical for the success of science multiteam systems because it enables teams to work in parallel while ensuring that their outputs are integrated into a cohesive whole. Returning to the example of STCs and ERCs, in these large, federally funded initiatives, teams with specialized expertise often address different facets of a complex problem. For instance, one team might focus on developing computational models that simulate neural circuits responsible for visual processing, while another team conducts experimental studies on how the brain processes social intelligence. Without effective inter-team coordination, these efforts could remain siloed, preventing the full integration of computational insights with biological data. However, when coordination mechanisms are in place, the contributions of each team—whether focused on machine learning, neuroscience, cognitive science, or any other scientific field—are aligned and complement one another, leading to scientific progress aligned with the STC's or ERC's superordinate goals.

Lastly, a unique element of effective science multiteam systems is balancing countervailing forces. High-functioning multiteam systems pay special attention to the unique needs of both the component teams and the overall multiteam system. Something that may be good for a single team, such as increasing cohesion or prioritizing its own goals, can sometimes detract from the performance of the entire multiteam system (Carter et al., 2019). For example, promoting strong cohesion within one team might reduce its willingness to collaborate with other teams, leading to silos and reduced system-level performance. Similarly, focusing too much on local team goals can result in a misalignment with the superordinate goals of the multiteam system, undermining the overall success.

Balancing these forces is critical for the success of science multiteam systems because it ensures that the system can leverage the strengths of each team while maintaining a focus on system-wide objectives. In the case of large, interdisciplinary initiatives like Clinical and Translational Science Awards, where teams often come from different fields with distinct goals and priorities, balancing these forces is essential for integrating different perspectives and achieving the broader research objectives.<sup>2</sup> For example, one team may focus on clinical trial design while another works on community engagement. Without careful management, these distinct priorities could lead to misalignment or competition for resources. However, by balancing the needs of individual teams with the overall goals of the multiteam system, the system can capitalize on the unique expertise of each team while ensuring progress toward the shared mission.

High-functioning science multiteam systems can take many forms, shaped by factors such as boundary status, component team distance, and the nature of the superordinate goal. These elements play a critical role in determining how teams interact, navigate logistical and disciplinary differences, and align their goals to drive the collective success of the system. Equally important are the processes and emergent states that support these interactions, such as boundary spanning, inter-team coordination, and balancing competing priorities, which together define a high-functioning science multiteam system. Despite the importance of these attributes and processes, empirical research specifically focused on scientific multiteam systems remains relatively limited, with a few notable exceptions (e.g., Kotarba et al., 2023). This gap is unfortunate given the increasing prevalence of large-scale scientific initiatives, such as STCs and ERCs, which have immense potential to advance scientific discovery and innovation.

#### REFERENCES

- Carter, D. R., Asencio, R., Trainer, H. M., DeChurch, L. A., Kanfer, R., & Zaccaro, S. J. (2019). Best practices for researchers working in multiteam systems. In K. L. Hall, A. L. Vogel, & R. T. Croyle (Eds.), *Strategies for team science success* (pp. 391–400). Springer International Publishing. https://doi.org/10.1007/978-3-030-20992-6\_29
- Cronin, M. A., Weingart, L. R., & Todorova, G. (2011). Dynamics in groups: Are we there yet? *Academy of Management Annals*, 5(1), 571–612. https://doi.org/10.5465/19416520.2011.590297
- D'Innocenzo, L., Mathieu, J. E., & Kukenberger, M. R. (2016). A meta-analysis of different forms of shared leadership-team performance relations. *Journal of Management*, 42(7), 1964–1991. https://doi.org/10.1177/0149206314525205
- Edmondson, A. C., & Bransby, D. P. (2023). Psychological safety comes of age: Observed themes in an established literature. *Annual Review of Organizational Psychology and Organizational Behavior*, 10, 55–78. https://doi.org/10.1146/annurev-orgpsych-120920-055217

<sup>&</sup>lt;sup>2</sup> For more information about Clinical and Translational Science Awards, see https://ncats.nih.gov/research/research-activities/ctsa.

Edmondson, A. C., & Lei, Z. (2014). Psychological safety: The history, renaissance, and future of an interpersonal construct. *Annual Review of Organizational Psychology and Organizational Behavior*, 1(1), 23–43. https://doi.org/10.1146/annurev-orgpsych-031413-091305

- Frazier, M. L., Fainshmidt, S., Klinger, R. L., Pezeshkan, A., & Vracheva, V. (2017). Psychological safety: A meta-analytic review and extension. *Personnel Psychology*, 70(1), 113–165. https://doi.org/10.1111/peps.12183
- Gersick, C. J. G. (1988). Time and transition in work teams: Toward a new model of group development. Academy of Management Journal, 31(1), 9–41. https://doi.org/10.5465/256496
  \_\_\_\_\_. (1989). Marking time: Predictable transitions in task groups. Academy of Management Journal, 32(2), 274–309. https://doi.org/10.5465/256363
- Grossman, R., & Feitosa, J. (2018). Team trust over time: Modeling reciprocal and contextual influences in action teams. *Human Resource Management Review*, 28(4), 395–410. https://doi.org/10.1016/j.hrmr.2017.03.006
- Hackman, J. R. (1987). The design of work teams. In J. W. Lorsch (Ed.), *Handbook of organizational behavior* (pp. 315–342). Prentice-Hall.
- Harvey, P. J.-F., Bresman, D. H., Edmondson, P. A. C., & Pisano, P. G. P. (2022, January 10). A strategic view of team learning in organizations. *Academy of Management Annals*, 16(2). https://doi.org/10.5465/annals.2020.0352
- Ilgen, D. R., Hollenbeck, J. R., Johnson, M., & Jundt, D. (2005). Teams in organizations: From input-process-output models to IMOI models. *Annual Review of Psychology*, 56(1), 517–543. https://doi.org/10.1146/annurev.psych.56.091103.070250
- Ingersoll, B., Espinel, A., Nauman, J., Broder-Fingert, S., Carter, A. S., Sheldrick, R. C., Stone, W. L., & Wainer, A. L. (2024). Using virtual multiteam systems to conduct a multisite randomized clinical trial in the part C early intervention system: Benefits, challenges, and lessons learned. Contemporary Clinical Trials, 143, 107585. https://doi.org/10.1016/j.cct.2024.107585
- Jehn, K. A., & Bezrukova, K. (2010). The faultline activation process and the effects of activated faultlines on coalition formation, conflict, and group outcomes. Organizational Behavior and Human Decision Processes, 112(1), 24–42. https://doi.org/10.1016/j.obhdp.2009.11.008
- Kotarba, J. A., Molldrem, S., Smith, E., Spratt, H., Bhavnani, S. K., Farroni, J. S., & Wooten, K. (2023). Exploring team dynamics during the development of a multi-institutional cross-disciplinary translational team: Implications for potential best practices. *Journal of Clinical and Translational Science*, 7(1), e220. https://doi.org/10.1017/cts.2023.640
- Kozlowski, S. W. J., & Bell, B. S. (2007). Team learning, development, and adaptation. In V. I. Sessa & M. London (Eds.), Work group learning: Understanding, improving and assessing how groups learn in organizations (pp. 15–44). Taylor & Francis Group/Lawrence Erlbaum Associates.
- Marks, M. A., Mathieu, J. E., & Zaccaro, S. J. (2001). A temporally based framework and taxonomy of team processes. *The Academy of Management Review*, 26(3), 356–376.. https://doi.org/10.5465/amr.2001.4845785
- Mathieu, J., Maynard, M. T., Rapp, T., & Gilson, L. (2008). Team effectiveness 1997-2007: A review of recent advancements and a glimpse into the future. *Journal of Management*, 34(3), 410–476. https://doi.org/10.1177/0149206308316061
- Mohammed, S., Hamilton, K., & Lim, A. (2008). The incorporation of time in team research: Past, current, and future. In E. Salas, G. F. Goodwin, & C. S. Burke (Eds.), *Team effectiveness in complex organizations: Cross-disciplinary perspectives and approaches* (pp. 321–348). Routledge/Taylor & Francis Group.
- Mohammed, S., Rico, R., & Alipour, K. K. (2021). Team cognition at a crossroad: Toward conceptual integration and network configurations. *Academy of Management Annals*, 15(2), 455–501. https://doi.org/10.5465/annals.2018.0159

- National Research Council. (2015). Enhancing the effectiveness of team science. National Academies Press. https://doi.org/10.17226/19007
- Pearce, C. L., & Conger, J. A. (Eds.). (2003). Shared leadership: Reframing the hows and whys of leadership. Sage Publishing. https://doi.org/10.4135/9781452229539
- Pearce, C. L., Conger, J. A., & Locke, E. A. (2008). Shared leadership theory. *The Leadership Quarterly*, 19(5), 622–628. https://doi.org/10.1016/j.leaqua.2008.07.005
- Shuffler, M. L., & Carter, D. R. (2018). Teamwork situated in multiteam systems: Key lessons learned and future opportunities. *American Psychologist*, 73(4), 390–406. https://doi.org/10.1037/amp0000322
- Shuffler, M. L., Jiménez-Rodríguez, M., & Kramer, W. S. (2015). The science of multiteam systems: A review and future research agenda. *Small Group Research*, 46(6), 659–699. https://doi.org/10.1177/1046496415603455
- Thatcher, S. M. B., Meyer, B., Kim, Y., & Patel, P. C. (2024). A meta-analytic integration of the faultlines literature. *Organizational Psychology Review*, 14(2), 238–281. https://doi.org/10.1177/20413866231225064
- Thatcher, S. M. B., & Patel, P. C. (2012). Group faultlines: A review, integration, and guide to future research. *Journal of Management*, 38(4), 969–1009. https://doi.org/10.1177/0149206311426187
- Tuckman, B. W. (1965). Developmental sequence in small groups. *Psychological Bulletin*, 63(6), 384–399. https://doi.org/10.1037/h0022100
- Van den Bossche, P., Gijselaers, W., Segers, M., Woltjer, G., & Kirschner, P. (2011). Team learning: Building shared mental models. *Instructional Science*, 39(3), 283–301. https://doi.org/10.1007/s11251-010-9128-3
- Weingart, L. R. (1997). How did they do that? The ways and means of studying group process. Research in Organizational Behavior, 19, 189–239.
- Wiese, C. W., & Burke, C. S. (2019). Understanding team learning dynamics over time. *Frontiers in Psychology*, 10, 1417.
- Zaccaro, S. J., Dubrow, S., Torres, E. M., & Campbell, L. N. P. (2020). Multiteam systems: An integrated review and comparison of different forms. *Annual Review of Organizational Psychology and Organizational Behavior*, 7(1), 479–503. https://doi.org/10.1146/annurev-orgpsych-012119-045418
- Zaheer, S., Albert, S., & Zaheer, A. (1999). Time scales and organizational theory. *The Academy of Management Review*, 24(4), 725–741. https://doi.org/10.5465/amr.1999.2553250
- Ziegert, J. C., Knight, A. P., Resick, C. J., & Graham, K. A. (2022). Addressing performance tensions in multiteam systems: Balancing informal mechanisms of coordination within and between teams. *Academy of Management Journal*, 65(1), 158–185. https://doi.org/10.5465/amj.2019.1043

# Appendix D

## Sampling of Survey-Based Team Assessments

Table D-1 provides a sampling of survey-based assessments used in the team literature, including a construct/scale description, measurement specifics (number of items, dimensions, rating scale), and references indicating scale development and validation evidence.

230

#### THE SCIENCE AND PRACTICE OF TEAM SCIENCE

## TABLE D-1 Sampling of Survey-Based Team Assessments

Construct/Scale Name	Construct Definition/Scale Description	
What Individual Charact	eristics Describe Your Team?	
To what extent are team members positioned for effective teamwork?		
Beliefs About Groups	Beliefs about whether groups are desirable and effective	

(BAG)

Beliefs about whether groups are desirable and effective

Team Role Experience and Orientation (TREO)

Predisposition to occupy different team roles

Multidimensional Perceived Person-Group Fit (MPPGF)

Fit between individuals and the group is measured on several dimensions including:

- Needs-supplies match
- Shared interests
- Perceived demographic similarity
- Complementary attributes
- · Values congruence
- · Goals similarity
- Common workstyle

#### Are team members ready for team and cross-disciplinary collaboration?

Research-orientation scale

Indicator of collaboration readiness that assesses team members' values and attitudes toward cross-disciplinary research on a continuum including unidisciplinary, multidisciplinary, and interdisciplinary/transdisciplinary research orientations

APPENDIX D 231

Measurement Specifics	Scale Development Validation Evidence
16-item BAG scale assessing 4 dimensions: preference for group versus individual work, positive group performance beliefs, negative group performance beliefs, and beliefs that others will work hard on group tasks. Items were rated on a 5-point scale ranging from strongly disagree (1) to strongly agree (5)	Karau et al. (2009)
strongly disagree (1) to strongly agree (3)	
48-item scale measuring members' propensities to occupy six different team roles, including:  Organizer  Doer  Challenger  Innovator  Team builder  Connector	Mathieu et al. (2015b)
Rated on a 5-point scale (1=not at all; 5=to a very great extent)	
28-item scale with 4 items each of the following 7 subscales: needs—supplies match, shared interests, perceived demographic similarity, complementary attributes, values congruence, goals similarity, and common workstyle.  The extent to which the person performing the self-assessment agrees with the fit-related items is rated on a 5-point Likert scale (1=strongly disagree, 5=strongly agree)	Li et al. (2018)
10-item scale assessing team members' proclivity toward	Hall et al. (2008)

10-item scale assessing team members' proclivity toward unidisciplinary (3 items, "researchers use theories and methods from a single discipline"), multidisciplinary (2 items, "researchers work in parallel or sequentially from disciplinary-specific base to address a common problem"), and interdisciplinary/transdisciplinary research orientations (5 items, "researchers work jointly but still from disciplinary-specific basis to address common problem" and "researchers work jointly using shared conceptual framework drawing together disciplinary-specific theories, concepts, and approaches to address common problems"; Rosenfield, 1992, p. 1351). Items rated on a 5-point scale ranging from strongly disagree to strongly agree

Hall et al. (2008) Rosenfield (1992)

continued

Construct/Scale Name	Construct Definition/Scale Description
Interdisciplinary Perspectives Index (IPI)	Attitudes toward working with team members from different fields and beliefs about the outcomes of cross-disciplinary work
Transdisciplinary Orientation Scale	"Values, attitudes, beliefs, conceptual skills and knowledge, and behavioral repertoires that predispose an individual to collaborating effectively in cross-disciplinary science teams" (Misra et al., 2015, p. 1)
What Are the Shared Affe	ctive, Cognitive, and Behavioral States in Your Team?
How strongly do team me emotionally and attituding	embers connect with each other and the team's purpose
Team cohesion	Interpersonal/social cohesion: emotional bonding among team members, reflecting trust, mutual respect, and camaraderie.  Task cohesion: shared commitment to the team's mission that reflects dedication and responsibility to complete tasks and solve problems
Team trust	Trust is defined as a latent variable resulting from 2 distinct but related formative indicators (propensity to trust and perceived trustworthiness), which lead to 2 reflective and behavioral indicators (cooperation and monitoring) between team members
Perceived collective efficacy	Team members' perceptions of the group's ability to succeed
Collective team identification	Members' emotional commitment to the team and its goals
To what extent is team kn	owledge shared and differentiated in your team?
Referee Shared Mental Models Measure (RSMMM)	The extent to which team members believe that they share a similar understanding of taskwork (what the team is doing) and teamwork (how the work gets done and who accomplishes what)

	Scale Development/
Measurement Specifics	Validation Evidence
6-item scale assessing attitudes about cross-disciplinary work, including the extent to which they value interdisciplinary work, are optimistic about the scientific outcome of such work, have tolerance of and open-mindedness toward research perspectives other than their own, use multiple research methods from many disciplines, believe that a high degree of goodwill exists among their research collaborators, and believe that the benefits of interdisciplinary research outweigh the inconveniences	Misra et al. (2009)
12-item scale to measure team members' transdisciplinary orientation on two dimensions: (a) values, attitudes, and beliefs (6 items) and (b) conceptual skills and behaviors (6 items) on a 5-item Likert scale ranging from strongly disagree to strongly agree	Misra et al. (2015)
6-item scale measuring interpersonal (3 items) and task-oriented (3 items) team cohesion on a 1–7 Likert-type scale	Mathieu et al. (2015a)
21-item multidimensional measure with 4 dimensions, including (a) propensity to trust (6 items), (b) perceived trustworthiness (6 items), (c) cooperation (6 items), and (d) monitoring between team members (3 items) rated on a 7-point scale ranging from 1=completely disagree to 7=completely agree	Costa and Anderson (2010)
4-item scale rated on a 5-point scale (1=never, 5=most of the time)	Salanova et al. (2003)
4-item scale rated on a 7-point scale (1=completely disagree, 7=completely agree)	Van Der Vegt and Bunderson (2005)
13-item measure assessing whether team members believe they are on the same page about relevant team knowledge, including decision-making, technology, procedures, interactions, and priorities. Items are rated on a 7-point Likert scale (1=totally disagree, 7=totally agree)	Sinval et al. (2020)

#### THE SCIENCE AND PRACTICE OF TEAM SCIENCE

Construct/Scale Name	Construct Definition/Scale Description
Transactive memory systems scale	Unique member expertise plus a shared awareness of who knows what in the team
Team learning	The process through which team members collectively acquire, share, and apply knowledge to achieve common goals
How strongly do team n	nembers depend on each other and share the workload?
Task interdependence	The degree to which team members rely on one another to complete their tasks and achieve the team's goals
Workload sharing	Equitable distribution of tasks and responsibilities among team members to ensure that the overall workload is managed efficiently and fairly
What Is the Quality of T	eam Processes and Interactions in Your Team?
Conflict	Task conflict involves differences among team members about the content and outcomes of the tasks being performed Relationship conflict refers to personal incompatibilities among team members and is characterized by tension and animosity between team members  Process conflict centers on issues about logistics, the procedures used to accomplish tasks, and who will perform what roles
Conflict management	Strategies and processes used to resolve disputes among team members in a constructive manner
Team processes	Team processes capture interaction between team members. Transition processes: teams engage in evaluation and planning activities Action processes: teams perform activities that directly contribute to goal attainment Interpersonal processes: teams foster motivation, manage emotions, and resolve conflict
Implicit coordination	The process by which team members align their actions and synchronize their efforts without explicit communication

Measurement Specifics	Scale Development/ Validation Evidence
15-item scale divided into 3 dimensions (specialization, credibility, and coordination) of 5 items each and rated on a 5-point Likert scale (1=strongly disagree, 5=strongly agree)	Lewis (2003)
7-item scale rated on a 5-point Likert scale (1=never, 7=always)	Edmondson (1999)
5-item scale rated on a 5-point Likert scale (1=strongly agree,	Van Der Vegt et al.
5=strongly disagree)	(2000)
3-item scale rated on a 5-point scale (1=strongly disagree, 5=strongly agree)	Campion et al. (1993)
9-item scale with 3 sub-scales of 3 items each (task conflict, relationship conflict, and process conflict) rated on a 7-point Likert scale (1=not at all, 7=a lot)	Jehn and Mannix (2001)
4-item scale rated on a 7-point Likert-type scale (1=strongly disagree, 7=strongly agree)	Tekleab et al. (2009)
50-item full scale (15-item transition processes scale with 3 subscales (mission analysis, goal specification, strategy formulation); 20-item action processes scale with 4 sub-scales (monitoring progress toward goals, systems monitoring, team monitoring and backup, coordination); 15-item interpersonal process scale with 3 sub-scales (conflict management, motivating and confidence-building, affect management)	Mathieu et al. (2020)
30-item shorter form 10-item shorter form	
4-items rated on a 5-point scale (1=extremely inaccurate, 5=extremely accurate)	Fisher et al. (2012)

Construct/Scale Name	Construct Definition/Scale Description
Is the Climate in Your Te	eam Supportive of Scientific Collaboration?
Team climate	Collective perceptions and shared attitudes of team members regarding their team environment
Work group inclusion	The extent to which group members feel included in the team includes 2 dimensions: belonging (supportive and caring relationships among members) and uniqueness (differences between members are valued and respected)
Psychological safety	Shared belief among team members that the environment is safe for interpersonal risk-taking such that members can ask questions, admit mistakes, and raise concerns without fear of embarrassment or ridicule
Team Perceived Virtuality	"Shared affective-cognitive emergent state that is characterized by team members' co-constructed and collectively experienced 1) distance and 2) information deficits, thereby capturing the unrealized nature of the team as a collective system" (Handke et al., 2021, p. 626). Socio-constructivist perspective on team virtuality emphasizing how team members perceive virtuality
How Much Do Team Mo	embers Take on Different Leadership Roles in the Team?
Team leadership	Influence across 2 main phases in the team life cycle:  1. Transition phase: planning the team's work and ensuring that taskwork and teamwork goals will be reached  2. Action phase: activities that contribute directly to goal accomplishment
Temporal leadership	The extent to which leaders schedule deadlines, synchronize team member behaviors, and allocate temporal resources

Measurement Specifics	Scale Development/ Validation Evidence
The Team Climate Inventory is a 14-item scale (short version) that measures team climate in the areas of vision, participative safety, task orientation, and support for innovation rated on a 5-point Likert scale (1=strongly disagree, 5=strongly agree)	Kivimaki & Elovainio (1999) Strating & Nieboer (2009)
The Work Group Inclusion scale is a 10-item scale rated on a 5-point scale (1=strongly disagree, 5=strongly agree)	Chung et al. (2020)
7-item scale rated on a 7-point scale (1=very inaccurate, 5=very accurate)	Edmondson (1999)
10-item scale with 2 dimensions: (a) collectively experienced distance (5 items) and (b) collectively experienced information deficits (5 items) rated on a 7-point scale ranging from strongly disagree to strongly agree	Handke et al. (2021, 2024)
The Team Leadership Questionnaire (TLQ) is an integrative measurement tool assessing 15 team leader function categories divided into 2 primary phases:	Morgeson et al. (2010)
<ol> <li>Transition: compose team, define mission, establish expectations and goals, structure and plan, train and develop team, sensemaking, provide feedback</li> <li>Action: monitor team, manage team boundaries, challenge team, perform team task, solve problems, provide resources, encourage team self-management, support social climate</li> </ol>	
7-item scale rated on a 5-point scale (1=not at all, 5=a great deal)	Mohammed & Nadkarni (2011)

Construct/Scale Name	Construct Definition/Scale Description
What Are the Affective a	and Performance Outcomes of Team Interactions?
How happy are team m	embers with how they worked together?
Team viability	Willingness for team members to continue working together in the future
Individual-level team satisfaction	Overall contentment and positive feelings team members have about their experiences in a team, including relationships with colleagues and task fulfillment
Collaboration	Team members working together to achieve common goals by sharing knowledge and communicating clearly to integrate different perspectives and ideas

How productive is your team?	
Team performance	The extent to which a team collectively achieves its goals and objectives
Comprehensive assessment of team member effectiveness	<ol> <li>5 broad areas of team effectiveness:</li> <li>1. Contributing to the team's work</li> <li>2. Interacting with teammates</li> <li>3. Keeping the team on track</li> <li>4. Expecting quality</li> <li>5. Having relevant knowledge, skills, and abilities</li> </ol>

Measurement Specifics	Scale Development/ Validation Evidence
4-item scale rated on a 7-point Likert-type scale (1=strongly disagree, 7=strongly agree)	Tekleab et al. (2009)
5-item scale rated on a 7-point Likert-type scale (1=very dissatisfied, 7=very satisfied)	Tekleab et al. (2009)
18-item scale measuring satisfaction with collaboration (8 items: acceptance of ideas, communication, researchers' strengths, organization, resolution of conflict, working styles, outside involvement, and discipline involvement), collaboration impact (6 items: meeting productivity, products productivity, overall productivity, research productivity, quality of research, and time burden), and trust and respect (4 items: being comfortable in showing limits, trusting colleagues, being open to criticism, and respect) in research teams. Satisfaction with collaboration and 3 of the collaboration impact items are rated on a 5-point scale ranging from inadequate to excellent. 3 collaboration impact items and trust and respect items are rated on a 5-point Likert-type scale ranging from strongly disagree to strongly agree	Mâsse et al. (2008)
5-item scale (team learning behavior, context support, team leader coaching, team psychological safety, team efficacy) rated on a 5-point Likert scale (1=never, 5=always)	Edmondson (1999)
33-item short scale designed for self- and peer-evaluations of team members in 5 areas: contributing to the team's work, interacting with teammates, keeping the team on track, expecting quality, and having relevant knowledge, skills, and abilities rated on a 5-point Likert-type scale (1=strongly disagree, 5=strongly agree)	Ohland et al. (2012)
87-item full scale designed for self-and peer-evaluations in the 5 areas listed above rated on a Behaviorally Anchored Rating Scale in which raters read through behavioral examples describing a 1, 3, and 5 rating and select the option that best matches ratees' behavior	

Construct/Scale Name	Construct Definition/Scale Description
What Is the Overall State	of Readiness, Functioning, Climate, and Outcomes in Your Team?
TeamSTEPPS Teamwork Perceptions Questionnaire (T-TPQ)	A tool developed by the Agency for Healthcare Research and Quality that helps to identify the strengths and areas for improvement in team dynamics in health care settings
Team Diagnostic Survey (TDS)	An extensive instrument based on a conceptual model of the five enabling conditions that increase the likelihood that a team will perform effectively. It also assesses individual affective reactions

Note: The items to many of the scales provided in Appendix D are freely available at https://ctsi.psu.edu/research-support/team-science-toolbox/assessment/

#### REFERENCES

- Agency for Healthcare Research and Quality. (n.d.). TeamSTEPPS Teamwork Perceptions Questionnaire. https://www.ahrq.gov/sites/default/files/wysiwyg/teamstepps-program/tools/ts-tpq-questionnaire.pdf
- \_\_\_\_. (2023). TeamSTEPPS: Research/evidence base. https://www.ahrq.gov/teamstepps-program/evidence-base/index.html
- Campion, M. A., Medsker, G. J., & Higgs, A. C. (1993). Relations between work group characteristics and effectiveness: Implications for designing effective work groups. *Personnel Psychology*, 46(4), 823–847.
- Chung, B., Ehrhart, K., Shore, L. M., Randel, A., Dean, M., & Kedharnath, U. (2020). Work group inclusion: Test of a scale and model. *Group and Organization Management*, 45(1), 75–102. https://doi.org/10.1177/1059601119839858
- Costa, A. C., & Anderson, N. (2010). Measuring trust in teams: Development and validation of a multifaceted measure of formative and reflective indicators of team trust. *European Journal of Work and Organizational Psychology*, 20(1), 119–154.
- Edmondson, A. (1999). Psychological safety and learning behavior in work teams. *Administrative Science Quarterly*, 44(2), 350–383. https://doi.org/10.2307/2666999
- Fisher, D. M., Bell, S. T., Dierdorff, E. C., & Belohlav, J. A. (2012). Facet personality and surface-level diversity as team mental model antecedents: Implications for implicit coordination. *Journal of Applied Psychology*, 97(4), 825.

Measurement Specifics	Scale Development/ Validation Evidence
35-item instrument assessing 5 dimensions (7 items each) of team functioning in a health care context: team structure, team leadership, situation monitoring, mutual support, and communication. Items are rated on a 5-point Likert scale ranging from strongly agree to strongly disagree	Agency for Healthcare Research and Quality (n.d., 2023)
The TDS measures 5 enabling conditions underlying team success, including whether the task requires a real team rather than in name only (8 items), whether the team has a compelling direction (10 items), a well-designed enabling team structure (20 items), a supportive organizational context (11 items), and available expert coaching (27 items). The survey also measures process criteria of team effectiveness (9 items), the quality of team interpersonal processes (7 items), and individual learning and well-being (10 items).  Most items are rated on a 5-point scale ranging from highly inaccurate to highly accurate	Wageman et al. (2005)

- Hall, K. L., Stokols, D., Moser, R. P., Taylor, B. K., Thornquist, M. D., Nebeling, L. C., Ehret, C. C., Barnett, M. J., McTiernan, A., Berger, N. A., Goran, M. I., & Jeffery, R. W. (2008). The collaboration readiness of transdisciplinary research teams and centers findings from the National Cancer Institute's TREC Year-One evaluation study. American Journal of Preventive Medicine, 35(2 Suppl), S161–S172. https://doi.org/10.1016/j.amepre.2008.03.035
- Handke, L., Costa, P., & Feitosa, J. (2024). How virtual are we? Introducing the team perceived virtuality scale. *Journal of Business and Psychology*, 1–24.
- Handke, L., Costa, P. L., Klonek, F. E., O'Neill, T. A., & Parker, S. K. (2021). Team perceived virtuality: An emergent state perspective. *European Journal of Work and Organizational Psychology*, 30(5), 624–638.
- Jehn, K. A., & Mannix, E. A. (2001). The dynamic nature of conflict: A longitudinal study of intragroup conflict and group performance. Academy of Management Journal, 44(2), 238–251. https://doi.org/10.5465/3069453
- Karau, S. J., & Elsaid, A. M. M. K. (2009). Individual differences in beliefs about groups. *Group Dynamics: Theory*, *Research*, *and Practice*, *13*(1), 1–13. https://dx.doi.org.ezaccess.libraries. psu.edu/10.1037/a0013366
- Kivimaki, M., & Elovainio, M. (1999). A short version of the Team Climate Inventory: Development and psychometric properties. *Journal of Occupational and Organizational Psychology*, 72(2), 241–246.
- Lewis, K. (2003). Measuring transactive memory systems in the field: Scale development and validation. *Journal of Applied Psychology*, 88(4), 587–604. https://doi.org/10.1037/0021-9010.88.4.587

- Li, C. S., Kristof-Brown, A. L., & Nielsen, J. D. (2018). Fitting in a group: Theoretical development and validation of the Multidimensional Perceived Person–Group Fit scale. *Personnel Psychology*, 72, 139–171. https://doi.org/10.1111/peps.12295
- Mâsse, L. C., Moser, R. P., Stokols, D., Taylor, B. K., Marcus, S. E., Morgan, G. D., Hall, K. L., Croyle, R. T., & Trochim, W. M. (2008). Measuring collaboration and transdisciplinary integration in team science. *American Journal of Preventive Medicine*, 35(2), S151–S160. https://doi.org/10.1016/j.amepre.2008.05.020
- Mathieu, J. E., Kukenberger, M. R., D'Innocenzo, L., & Reilly, G. (2015a). Modeling reciprocal team cohesion–performance relationships, as impacted by shared leadership and members' competence. *Journal of Applied Psychology*, 100(3), 713–734.
- Mathieu, J. E., Luciano, M. M., D'Innocenzo, L., Klock, E. A., & LePine, J. A. (2020). The development and construct validity of a team processes survey measure. *Organizational Research Methods*, 23(3), 399–431. https://doi.org/10.1177/1094428119840801
- Mathieu, J. E., Tannenbaum, S. I., Kukenberger, M. R., Donsbach, J. S., & Alliger, G. M. (2015b). Team role experience and orientation: A measure and tests of construct validity. *Group & Organization Management*, 40(1), 6–34.
- Misra, S., Stokols, D., & Cheng, L. (2015). The transdisciplinary orientation scale: Factor structure and relation to the integrative quality and scope of scientific publications. *Journal of Translational Medicine and Epidemiology*, 3(2), 1042.
- Mohammed, S., & Nadkarni, S. (2011). Temporal diversity and team performance: The moderating role of temporal leadership. *Academy of Management Journal*, 54(3), 489–508.
- Morgeson, F. P., DeRue, D. S., & Karam, E. P. (2010). Leadership in teams: A functional approach to understanding leadership structures and processes. *Journal of Management*, 36(1), 5–39.
- Ohland, M. W., Loughry, M. L., Woehr, D. J., Bullard, L. G., Felder, R. M., Finelli, C. J., Layton, R. A., Pomeranz, H. R., & Schmucker, D. G. (2012). The comprehensive assessment of team member effectiveness: Development of a behaviorally anchored rating scale for self and peer evaluation. *Academy of Management Learning & Education*, 11, 609–630. https://doi.org/10.5465/amle.2010.0177
- Rosenfield, P. L. (1992). The potential of transdisciplinary research for sustaining and extending linkages between the health and social sciences. *Social Science & Medicine*, 35(11), 1343–1357.
- Salanova, M., Llorens, S., Cifre, E., Martínez, I. M., & Schaufeli, W. B. (2003). Perceived collective efficacy, subjective well-being and task performance among electronic work groups: An experimental study. Small Group Research, 34(1), 43–73. https://doi.org/10.1177/1046496402239577
- Sinval, J., Aragão e Pina, J., Sinval, J., Marôco, J., Santos, C. M., Uitdewilligen, S., Maynard, M. T., & Passos, A. M. (2020). Development of the referee shared mental models measure (RSMMM). Frontiers in Psychology, 11, Article 550271. https://doi.org/10.3389/fpsyg.2020.550271
- Strating, M. M., & Nieboer, A. P. (2009). Psychometric test of the Team Climate Inventoryshort version investigated in Dutch quality improvement teams. *BMC Health Services Research*, 9(1), 1–7.
- Tekleab, A. G., Quigley, N. R., & Tesluk, P. E. (2009). A longitudinal study of team conflict, conflict management, cohesion, and team effectiveness. *Group & Organization Management*, 34(2), 170–205.
- Van Der Vegt, G., Emans, B., & Van De Vliert, E. (2000). Team members' affective responses to patterns of intragroup interdependence and job complexity. *Journal of Management*, 26(4), 633–655. https://doi.org/10.1177/014920630002600403

Van Der Vegt, G. S., & Bunderson, J. S. (2005). Learning and performance in multidisciplinary teams: The importance of collective team identification. *Academy of Management Journal*, 48(3), 532–547.

Wageman, R., Hackman, J. R., & Lehman, E. (2005). Team diagnostic survey: Development of an instrument. *The Journal of Applied Behavioral Science*, 41(4), 373–398.



# Appendix E

# Glossary

accessibility Highlights the importance of making environ-

ments, products, and services usable by people with disabilities. This ensures that all individuals, regardless of their abilities, can participate fully in

society.

best practice An activity or strategy that can enable a science

team to collaborate effectively to achieve its goals.

competencies The knowledge, skills, and abilities required for

team success.

context-driven

competencies

Competencies required for a specific team working

on a specific scientific task.

cross-disciplinary Collaboration among multiple disciplines toward

a shared objective.

faultlines Ways in which teams may split into sub-groups

based on one or more attributes.

group escalation The tendency of teams to commit poor decisions

resulting from pressures for conformity.

246	THE SCIENCE AND PRACTICE OF TEAM SCIENCE
hyperauthorship	Refers to when scientific papers have hundreds to thousands of authors.
interdisciplinary	Blending or juxtaposing concepts and methods from various disciplines.
multidisciplinary	Each discipline makes separate contributions in an additive way.
multiteam systems	Interdependent sets of two or more component teams pursuing shared superordinate goals.
project management	The systematic and deliberate application of knowledge, tools, and expertise to ensure the successful completion of complex projects in a timely and efficient manner.
psychological safety	The belief that a group is safe to engage in interpersonal risk.
researcher	Any member of a science team conducting and disseminating scientific research.
task-contingent competencies	Competencies specific to a type of task but can be applied across different teams.
taskwork	The activities associated with achieving a team's goals.
team charter	A document written by team members at the beginning of a team's life cycle that defines acceptable
	team behaviors.
team-contingent competencies	team behaviors.  Competencies specific to a particular team that can be applied generically across scientific tasks.
_	Competencies specific to a particular team that can

APPENDIX E 247

team task analysis A process to identify work behaviors and associated

knowledge, skills, and abilities for successful job or

task performance.

teamwork The interactions among team members that are

essential for effective collaboration.

transdisciplinary Advancement and integration of discipline-specific

theories, concepts, and methods.

transdisciplinary orientation

A characteristic that reflects the values, attitudes, beliefs, and behaviors for effective cross-disciplinary

collaboration.

transportable competencies

Competencies that have the ability to be applied

across a range of tasks and teams.

workshops Organized, structured interventions designed to

bring together individuals to engage in collaborative learning, skill development, and reflection over

a few hours to a few days.

